The Alignment of Strategic Objectives within the context of Temporary Multi-Organisations

Amos Peter Haniff

Submitted for the degree of Doctor of Philosophy

Heriot-Watt University

School of Energy, Geoscience, Infrastructure and Society

January 2017

The copyright in this thesis is owned by the author. Any quotation from the thesis or use of any of the information contained in it must acknowledge this thesis as the source of the quotation or information.

ABSTRACT

Current research investigating the alignment of projects with organisational strategy has predominantly focused on formation and implementation of strategic objectives within the boundary of a single, permanent organisation. Within the construction industry, the temporary organisation, created by the client organisation to deliver the project, is formed from multiple organisations that are brought together, under contractual conditions, to engage in a single endeavour. This creates a situation whereby multiple strategic objectives, and hence, multiple perceptions of project success, are pursued at anyone time. The research, forming this thesis, investigates how the varied organisations that comprise a *Temporary Multi-Organisation* (TMO) seek to align multiple strategic objectives within the context of a single construction project, and realise project success. In achieving this aim, the study exposes a number of ambiguities and difficulties organisations face when seeking to realise strategic objectives through a TMO.

Four cases of recently complete construction projects within the public sector were selected for empirical study. Qualitative data collected from interviews with actors across organisations participating in each TMO, was supported by secondary data comprising of project documentation from each case. The study finds that the strategy of the TMO, which evolves to realise project success, is guided by the varied strategic objectives of organisations participating in the project, and is influenced by the environmental conditions, procurement strategies, client complexity and leadership style of the client project manager, which together, influence the strategic behaviour of the TMO.

This research contributes to theories of strategic fit and the theory of temporary organisations through a model of strategic alignment within the context of a TMO that explains the complex interactions, which occur when multiple organisations engage within a single construction project. The research also contributes to the understanding of project success through a model of identifying the varied and competing success criteria within a TMO. Finally, the research contributes to leadership theory, through analysis of leadership styles within the context of a TMO.

DEDICATION

For Lisa, Qasim and Safia... Daddy's home. xxx

ACKNOWLEDGEMENTS

I wish to thank my supervisors Professor Stephen Ogunlana, Dr. Graeme Bowles and Dr. John Sanders for their guidance, constructive feedback, encouragement and patience, throughout the dissertation process. I would also like to thank the numerous organisations and participants for their support and contribution towards the research in this thesis. Without their agreement, time, and providing and allowing access to data, this study would not have been possible.

There are also a significant number of other people to whom I owe my gratitude. I am unable to acknowledge them all personally, but I am eternally grateful to my colleagues, family and friends for their support, valuable advice and words of encouragement over the years, in particular, at times of frustration. I also wish to offer a special thanks to all my fellow writers at Professor Rowena Murray's writing retreat. Your help and advice was invaluable.

DECLARATION STATEMENT

TABLE OF CONTENTS

ABS	STRACT	ii
DEI	DICATION	iii
ACI	KNOWLEDGEMENTS	iv
DeC	LARATION STATEMENT	v
TAF	BLE OF CONTENTS	vi
LIS	TS OF FIGURES	xi
T IS'	TS OF TABLES	vii
	TO DELEVEANT DIDLICATIONS DV THE CANDIDATE	
	T OF RELEVEANT PUBLICATIONS BY THE CANDIDATE	XIII
LIS	T OF ABBREVIATIONS	xiv
CHA	APTER 1 INTRODUCTION	1
1.1	Introduction to thesis	1
1.2	Background to study	1
	1.2.1 Temporary organisations	3
	1.2.2 Temporary multi-organisations	3
	1.2.3 Strategic alignment of projects	4
1.3	Research problem	5
	1.3.1 Current studies of strategic alignment of projects	5
	1.3.2 Strategic alignment of projects within the construction industry	
	133 Project success	8
	13.4 Problem statement	8
14	Research significance	
1.7	1 4 1 Contribution to temporary organisational theory	8
	1.4.2 Contribution to theories of strategic fit	0
	1.4.2 Contribution to understanding of project success	9
15	1.4.5 Contribution to understanding of project success	9
1.5	Nethedalass	10
1.6	Methodology	10
1.7	Structure of thesis	
1.8	Conclusion to chapter	14
CHA	APTER 2 THE STRATEGIC CONTEXT OF PROJECTS	15
2.1	Introduction to chapter	15
2.2	The nature of strategy	15
	2.2.1 Chaffee's three models of strategy	15
	2.2.2 Mintzberg's five P's for strategy	17
2.3	Strategic fit and alignment	18
2.4	The strategic hierarchy	20
2.5	Streams of project management research	23
	2.5.1 Quantitative stream of project management research	23
	2.5.2 Human resource stream of project management research	24
	2.5.3 Strategic stream of project management research	24
2.6	Project success criteria	25
2.7	The strategic alignment of projects	29
,	2.7.1 Hierarchy of project objectives	30
	2.7.2 Project strategy	34
28	Conclusion to chanter	
2.0	Conclusion to enapter	

CH	APTER 3 TEMPORARY ORGANSITIONAL FORMS	
3.1	Introduction to chapter	
3.2	Temporary organisational forms	
3.3	Theory of temporary organisations	
	3.3.1 Time	
	3.3.2 Task	41
	3.3.3 Team	42
	3.3.4 Transition	44
3.4	Leadership of temporary organisations	45
3.5	Types of temporary organisations	
3.6	Temporary multi-organisations	53
	3.6.1 Client complexity with the TMO	56
	3.6.2 TMO procurement strategy in the construction sector	57
3.7	Project governance	62
3.8	Research gaps and research questions	65
3.9	Conceptual Model	67
3.10	Conclusion to chapter	69
CH	APTER 4 RESEARCH METHODOLOGY	
4.1	Introduction to chapter	70
4.2	Research philosophy	70
	4.2.1 Ontological assumption	72
	4.2.2 Epistemological assumption	73
	4.2.3 Assumptions about human nature	76
	4.2.4 Methodological assumptions	77
	4.2.4.1 Action research	78
	4.2.4.2 Ethnography	79
	4.2.4.3 Grounded Theory	79
	4.2.4.4 In-depth surveys	80
	4.2.4.5 Case study	80
	4.2.5 Axiological assumptions	
4.3	Research Design	
	4.3.1 Overview of Research Design	83
	4.3.2 Establishing case study type	
	4.3.3 Selecting between single and multiple case studies	
	4.3.4 Unit of analysis	
	4.3.5 Validity and Reliability	91
4.4	Data Collection	92
	4.4.1 Number of case studies	92
	4.4.2 Case study selection	93
	4.4.3 Case study protocol	96
	4.4.4 Instrumentation Choice	97
	4.4.5 Semi-structured interviews	
	4.4.6 Interview design	
	4.4.7 Within-case sampling	
	4.4.8 Documents and archival records	
4.5	Data Analysis	
	4.5.1 Data reduction - Computer assisted data analysis	
	4.5.2 Data Reduction – Template analysis	
	4.5.3 Data display and conclusions	
4.6	Research Ethics	110
	4.6.1 Harm to participants	110
	4.6.2 Informed Consent	111

	4.1.1	Maintenance of privacy	112
	4.6.3	Confidentiality	112
4.7	Concl	usion to Chapter	112
СН	APTER	5 FINDINGS: WITHIN-CASE ANALYSIS	114
5.1	Introd	uction to chapter	114
5.2	Case A	A – New Student Residence	116
	5.2.1	Background to the case	116
	5.2.2	Client organisation's strategic objectives	118
	5.2.3	Business level strategies	118
	5.2.3	3.1 Department of Student Services strategy	119
	5.2.3	3.2 Department of Hospitality Services strategy	119
	5.2.3	3.3 Department of Estates strategy	119
	5.2.4	TMO strategic objectives	120
	5.2.5	Procurement process	121
	5.2.6	Governance structure	122
	5.2.7	Client behaviour	123
	5.2.8	Supply chain	124
	5.2.9	TMO strategy	125
	5.2.9	9.1 Time	125
	5.2.9	9.2 Team	126
	5.2.9	9.3 Task	127
	5.2.9	9.4 Transition	128
	5.2.10	Project success	128
	5.2.11	General findings from Case A	131
	5.2.1	11.1 Organisational strategy	131
	5.2.1	11.2 Mechanisms to maintain alignment	131
	5.2.1	11.3 TMO strategic behaviour	132
	5.2.1	11.4 Project success	132
5.3	Case I	B – New Four-lane Carriageway	134
	5.3.1	Background to the case	134
	5.3.2	Client organisation's strategic objectives	136
	5.3.3	TMO strategic objectives	137
	5.3.4	Procurement processes	138
	5.3.5	Governance structure	139
	5.3.6	Client behaviour	139
	5.3.7	Supply Chain	140
	5.3.8	TMO Strategy	141
	5.3.8	8.1 Time	141
	5.3.8	8.2 Team	142
	5.3.8	3.3 Task	143
	5.3.8	3.4 Transition	143
	5.3.9	Project Success	144
	5.3.10	General findings from Case B	147
	5.3.	10.1 Organisational strategy	147
	5.3.11	Mechanisms to maintain alignment	147
	5.3.1	11.1 TMO strategic behaviour	148
. .	5.3.1	11.2 Project Success	148
5.4	Case (C – New Laboratory and Office Facility	150
	5.4.1	Background to the case	150
	5.4.2	Client strategic objectives	152
	5.4.3	TMO strategic objectives	153
	5.4.4	Procurement processes	154

	5.4.5	Governance Structure	154
	5.4.6	Client behaviour	155
	5.4.7	Resistance to change	
	5.4.8	TMO Behaviour	
	5.4.	8.1 Time	
	5.4.	8.2 Team	
	5.4	83 Task	158
	5.1. 5.4	8.4 Transition	150
	549	Project Success	159
	5 / 10	General Findings from Case C	157
	5.4.10	10.1 Organisational Stratagy	102
	5.4.	10.2 Machanisms to maintain alignment	102
	5.4.	10.2 TMO starts sis habasi say	102
	5.4.	10.4 D	
	5.4.	10.4 Project success	
5.5	Case	D – New Tourist Visitor Centre	
	5.5.1	Background to case	
	5.5.2	Client Strategic Objectives	167
	5.5.3	TMO Strategic Objectives	169
	5.5.4	Procurement processes	170
	5.5.5	Governance processes	171
	5.5.6	Client Behaviour	172
	5.5.7	TMO Strategy	173
	5.5.	7.1 Time	173
	5.5.	7.2 Team	173
	5.5.	7.3 Task	
	5.5.	7.4 Transition	
	5.5.8	Project Success	
	5.5.9	General Findings from Case D	
	5.5	9.1 Organisational Strategy	179
	5 5	9.2 Mechanisms to maintain alignment	179
	5.5	9.3 TMO Strategic Behaviour	180
	5.5.	9.4 Project Success	
	5 5 10	Conclusion to Chapter	101
	5.5.10		101
CH	APTEF	R 6 CROSS-CASE ANALYSIS AND DISCUSSION	
6.1	Introd	uction to chapter	
6.2	Case	Study Characteristics	185
6.3	Hiera	rchy of Strategic Objectives	187
	6.3.1	Hierarchy of Strategic Objectives within each case study	187
	6.3.2	Cross-case synthesis of strategic objectives	190
	6.3.3	Alignment of strategic objectives	
6.4	Gover	mance of temporary organisations	
	6.4.1	Cross-case synthesis of procurement strategies	
	6.4.2	Project Governance Structure	
	6.4.3	Client Complexity	198
65	Leade	ership of Temporary Multi-Organisations	199
6.6	TMO	Strategy	
5.0	661	Time	201
	667	Tack	201
	663	Team	
	664	Transition	204 205
67	0.0.4 Droia	TTAIISIUUII	203 200
0.7		Success anitaria within a TMO	
	0./.1	Success criteria within a TMO	

	6.7.1.1 Tension 1: Between project objectives	209
	6.7.1.2 Tension 2: Between completion of tasks	211
	6.7.1.3 Tension 3: Between TMO member organisations	212
	6.7.1.4 Tension 4: Between organisations and TMO actors	213
	6.7.1.5 Tension 5: Between stakeholders within the client organisation	213
	6.7.2 Dimensions of project success	214
6.8	Model of strategic alignment within a TMO	217
	6.8.1 Model development	218
	6.8.2 Discussion of model	218
	6.8.2.1 Strategic drivers	219
	6.8.2.2 Project success criteria	220
	6.8.2.3 Procurement strategies	220
	6.8.2.4 Governance mechanisms	221
	6.8.2.5 Environmental influences	221
	6.8.2.6 TMO Strategic objectives	222
	6.8.2.7 TMO Behaviours	223
	6.8.2.8 Project success	223
	6.8.3 Benefits of the model	224
6.9	Summary and conclusion to chapter	225
СП		227
	APTER / CUNCLUSION	441
7.1	Deview of size and chiestives	227
1.2	7.2.1 Objective 1: Alignment of strategic abjectives	227
	7.2.1 Objective 1: Alignment of strategic objectives	227
	7.2.2 Objective 2: Effectiveness of governance mechanisms	230
	7.2.3 Objective 3: TMO Strategy	231
7 2	7.2.4 Objective 4: Perceptions of project success	232
1.3	7.2.1 Contribution	233
	7.3.1 Contribution to theories of strategic fit	234
	7.3.2 Contribution to perception of project success	234
	7.3.3 Contribution to the theory of temporary organisations	235
7 4	7.3.4 Contribution to project leadership	237
1.4	Contribution to practice	237
1.5	Limitation	238
7.6	Further research	239
7.7	Final conclusions	240
LIS	T OF REFERENCES	243
API	PENDIX A – INTERVIEW GUIDE	
	DENDIV D CASE STUDY DECISTED	262
AL	$\mathbf{E} \mathbf{N} \mathbf{D} \mathbf{A} \mathbf{D} = \mathbf{C} \mathbf{A} \mathbf{S} \mathbf{E} \mathbf{S} \mathbf{I} \mathbf{U} \mathbf{D} \mathbf{I} \mathbf{K} \mathbf{E} \mathbf{G} \mathbf{I} \mathbf{S} \mathbf{I} \mathbf{E} \mathbf{K} \dots$	
API	PENDIX C – DATA SOURCE LOGS	
C.1	Case Study A - Data Source Log	265
C.2	Case Study B - Data Source Log	266
C.3	Case Study C - Data Source Log	267
C.4	Case Study C - Data Source Log	268
API	PENDIX D – INITIAL TEMPLATE	269
API	PENDIX E – FINAL TEMPLATE	270
API	PENDIX F – THEMATIC MATRICES	272
F.1	Case Study A – Thematic Matrix	273
F.2	Case Study B – Thematic Matrix	275
F.3	Case Study C – Thematic Matrix	277
E1	Case Study D – Thematic Matrix	279
1.4		

LISTS OF FIGURES

Figure 1.1: Structure of thesis	.14
Figure 2.1: Planning hierarchy. Source: Mintzberg (1994)	.22
Figure 2.2: Relative importance of success dimensions. Source: Shenhar et al. (2001))
	.28
Figure 2.3: The Hierarchy of Objectives, Strategies and Projects. <i>Source:</i> Archibald	
(1987)	.31
Figure 2.4: The Hierarchy of Objectives, Strategies and Projects. Source: Morris and	L
Jamieson (2005). Adapted from Turner (1999)	.32
Figure 2.5: Hierarchy of Strategic Objectives. Source: Haniff & Fernie (2008)	.33
Figure 3.1: Contrasting organizational forms. Source: Modig (2007)	.48
Figure 3.2: Four types of inter-organizational projects. Source: Adapted from Jones a	and
Lichtenstein (2008)	.49
Figure 3.3: Framework of project business: Four distinctive management areas. Sour	ce:
Artto and Kujala (2008)	.51
Figure 3.4: Types of Project Based Firms (PBFs). Source: (Whitley, 2006)	.52
Figure 3.5: TMO configuration in traditional, general contracting. <i>Source</i> : Adapted	
from Murdoch and Hughes (1997)	.58
Figure 3.6: The vicious circle of construction procurement. Source: (Curtis et al., 199	91)
	.59
Figure 3.7: TMO configuration in Design and Build. Source: Adapted from Murdoch	1
and Hughes (1997)	.60
Figure 3.8: TMO configuration in construction management. <i>Source</i> : Adapted from	
Murdoch and Hughes (1997)	.60
Figure 3.9: TMO configuration in management contracting. <i>Source</i> : Adapted from	
Murdoch and Hughes (1997)	.61
Figure 3.10: Conceptual model of strategic alignment of TMOs	.68
Figure 4.1: The subjective – objective dimensions: <i>Source</i> : Burrell and Morgan (197	9)
	.71
Figure 4.2: Overview of the research design	.85
Figure 4.3: Basic Types of Design for Case Studies. Source: Adapted from Yin (2014	4)
	.90
Figure 5.1: Within-case analyses chapter structure	114
Figure 5.2: Case study A - I MO Structure	120
Figure 5.3: Causal network diagram Case Study	130
Figure 5.4: Case Study B - TMO Structure	130
Figure 5.5: Case Study B, Causal Network Diagram	140 151
Figure 5.0: Case study C - TMO Structure	121
Figure 5.7: Causal Network Diagram - Case Study C	101
Figure 5.0: Case D TWO Structure	10/ 179
Figure 5.7: Causal Indiwork Diagram Case D	197
Figure 6.1. Transformational process from temporary to permanent organisation	104)07
Figure 0.2. Transformational process from temporary to permanent organisation2 Figure 6.3. Model of project success within a TMO	≌07 200
Figure 0.3. Would of project success over time	≌U≯)15
Figure 6.5. Model of strategic alignment within a TMO	213)10
	217

LISTS OF TABLES

Table 3.1: Characteristics of transformational and transactional leaders. <i>Source</i> :	
Adapted from Bass (1990)	46
Table 3.2: Points for research into the client role in construction management. Sourc	e:
Adapted from Cherns and Bryant (1984)	55
Table 4.1: Network of Basic Assumptions Characterizing The Subjective - Objective	e
Debate within Social Science. Source: Morgan and Smircich (1980)	72
Table 4.2: Research Methodology. Source: Adapted from Remenyi et al. (1998)	78
Table 4.3: Components of Case study design. Source: Yin (2014)	83
Table 4.4: Process of theory building from case study research. Source: Eisenhardt	
(1989b)	84
Table 4.5: Definitions and Examples of Case Study Types. Source: Adapted from	
Baxter and Jack (2008)	87
Table 4.6: Cases, units of analysis and embedded units of analysis.	90
Table 4.7: Validation and reliability tactics applied within the research	92
Table 4.8: Summary of cases studies forming the research	95
Table 4.9: Summary of case study protocol	96
Table 4.10: Triangulation methods pursued in data collection stage	98
Table 4.11: Participant groups for data collection	102
Table 4.12: Number of interviews within each case data	102
Table 4.13: Secondary data types used in study	103
Table 4.14: Comparison of data models of data analysis. Source: Creswell (2007)	104
Table 4.15: Summary of initial coding set	108
Table 5.1: Case studies informing the research	115
Table 5.2: Structure of each case study report	115
Table 5.3: Case A: time-ordered display of key events	116
Table 5.4: Case B: Summary of key events	135
Table 5.5: Case C: Summary of key events	150
Table 5.6: Case D: Summary of key events	166
Table 6.1: Summary of Case Study Characteristics	185
Table 6.2: Hierarchy of strategic objectives within a TMO	189
Table 7.1: Distinctions between temporary organisations and TMOs	236

LIST OF RELEVEANT PUBLICATIONS BY THE CANDIDATE

Haniff A.P. & Fernie, S (2008) *Projects: Where strategies collide*. In Proceedings of the CIB 2008 Conference "Transformation through Construction" 15-18 November 2008, Dubai

Haniff, A.P & Ogunlana, S (2015) *Strategic alignment within a TMO: Perceptions of project success.* In Proceedings of ARCOM 31st Annual Conference. University of Lincoln 7th-9th September 2015

Haniff, A.P & Ogunlana, S (2015) 'Towards a Theory of temporary multiorganizations: Implications for project success' in Proceedings Paper for the Engineering Project Organization Society. University of Edinburgh 24th-26th June 2015

LIST OF ABBREVIATIONS

- APM Association for Project Management
- CDM Construction Design and Management coordination
- CGI Computer Generated Imagery
- HMO House in Multiple Occupation
- ID Interpretation Design
- JMC Joint Main Contractor
- JVC Joint Venture Contractor
- MC Management Consultant
- M&E Mechanical Engineering
- OGC Office of Government and Commerce
- PBF Project Based Firm
- PM Project Management
- PMC Project Management Consultant
- PMI Project Management Institute
- QS Quantity Surveyor
- SBU Strategic Business Unit
- TRPM Technical Services Project Management
- TMO Temporary Multi-Organisation

CHAPTER 1 INTRODUCTION

1.1 Introduction to thesis

Current research into the alignment of projects with organisational strategy has predominantly focused on the formation and implementation of strategic objectives within the boundary of a single organisation. Within the construction industry, the temporary organisation, created by a client organisation to implement the construction project, is formed from multiple organisations that are brought together to engage in a single endeavour. This creates a situation whereby multiple strategic objectives, and hence, multiple perceptions of project success, are pursued at any one time. The research within this thesis sets out to investigate how varied organisations that comprise a Temporary Multi-Organisation (TMO) seek to align multiple strategic objectives within the context of a single construction project, and realise project success. In pursuit of this aim, the research explores the complex interactions, processes, influences and constraints that occur when multiple organisations engage within a construction project.

This introductory chapter provides the background to the field of study, summarising the theoretical perspectives considered within the research and the purpose of the study. This is followed by an explanation and statement of the research problem addressed within the thesis. The significance of the research, including contribution to theory, is discussed. A statement of the research aims and objectives is presented, together with a brief overview of the research methodology. The chapter concludes with an outline of the structure of the thesis, demonstrating how the aims and objectives of the research are achieved.

1.2 Background to study

The significant growth in project management across different industries has seen projects become an integral component of an organisation's strategic operations. This wider adoption of project organising has not only generated new business practices, but also challenges the common perceptions of project success. Projects are no longer perceived as endeavours to provide tangible products or results. Rather, projects are now viewed as vehicles for business transformation, continuous improvement, organisational change, value creation and implementation of strategic objectives (Winter *et al.*, 2006a; Winter *et al.*, 2006b; Maylor, 2001). Traditionally, a project was

deemed successful if it achieved the project scope within the predetermined schedule, budget and quality constraints (Oisen, 1971). Within the strategic management domain, a project is successful when it enables the realisation of an organisations, long-term, strategic aspirations (Shenhar *et al.*, 2001).

Central to the discussion of projects to implement organisational strategy, is the concept of strategic fit. This considers the degree of alignment that exists between the competitive environment and organisational conditions (Ginsberg and Venkatraman, 1985). Within the project environment this involves either, ensuring that projects, selected for implementation, fit with the strategic objectives of the organisation (Cooper *et al.*, 1999; Aalto, 2000; Meskendahl, 2010), or that the project management processes are appropriately aligned to realise the organisations business strategy (Milosevic and Srivannaboon, 2006; Joshi *et al.*, 2003).

Following Hofer and Schendel's (1978) suggestion that strategy be constrained by the upper level within a strategic hierarchy, the proposition is made that alignment of a project with an organisation's strategic objectives is achieved by setting strategy at the corporate level and cascading down through strategic and operational levels, to be implemented as projects (Archibald, 1988; Youker and Brown, 1998). Kerzner (2001) presents a hierarchy illustrating how corporate strategic plans flow horizontally across strategic business units to support plans and budgets. Whereas, Morris and Jamieson (2004) show how organisations position their programmes and projects to achieve strategic objectives.

However, it is recognised that strategy occurs at differing levels in an organisation, and that each level will have its own, distinctive and often competing, strategic intentions (Turner, 1999; Haniff and Fernie, 2008; Morris and Jamieson, 2005). Subsequently, stakeholders at each level of a strategic hierarchy are likely to focus on the unique strategic objectives developed in response to the specific environmental conditions in which it competes. As the filtering of strategic objectives from the corporate to the project level involves a number of complex interactions, processes, customers and varied strategic constraints, it is proposed that organisational strategies formulated at the corporate level will be subject to incremental, internal and external influences before, and during, implementation as a project.

1.2.1 Temporary organisations

The strategic stream in project management research has also renewed interest into project organisational structures and how they are coordinated to achieve project success. Whereas during the 1970's, research focused on organisational behaviour and matrix organisations (Knight, 1976), more recent research has concentrated on the examination of projects as temporary organisational forms (Lundin and Soderholm, 1995; Jacobsson *et al.*, 2015; Bakker, 2010; Turner and Muller, 2003). Termed to reflect their transitory nature, the dominant discourse makes the assumption that temporary organisations are formed within the organisational boundary of a permanent organisation to manage complex organisational tasks (Turner and Muller, 2003; Packendorff, 1995; Goodman and Goodman, 1976).

A consistent theme within the literature has been on defining the characteristics of a temporary organisation and justifying their existence (Miles, 1964; Bennis and Slater, 1968; Goodman and Goodman, 1976; Miles, 1977; Soderlund, 2004a; Jacobsson *et al.*, 2015). Lundin and Soderholm (1995) present a theory of temporary organisations demarcating between permanent and temporary organisational behaviour. Developed as an opposing model to Cyert and March's (1963) behavioural theory of the firm, Lundin and Soderholm (1995) identify four concepts of limited time of the project, the task that must be accomplished, the team that is formed around that task, and the transition as a result of the task, which determines the behaviour of temporary organisational actors.

1.2.2 Temporary multi-organisations

Following Lundin and Soderholm's (1995) much cited framework, there has been a growth of studies exploring intra-organisational temporary systems (Bakker, 2010; Turner and Muller, 2003; Jacobsson *et al.*, 2015). Conversely, discourse on temporary systems involving inter-organisational relationships has not received the same attention. Instead, the assumption is made that many of the propositions underpinning Lundin and Soderholm's (1995) basic concepts apply to all temporary organizational types. However, within the construction industry, the temporary organisation is not formed within the boundaries of a single parent organisation. Rather, it operates within an environment of overlapping organisational boundaries where multiple organisations simultaneously seek to ensure their own organisational strategies are realised through a single project. Within a construction project, the composition of the temporary system typically consists of multiple firms, each making representation on a single endeavour.

Defined by Cherns and Bryant (1984: 181) as a "*temporary multi-organisation*", typical membership includes consultants, contractors, sub-contractors and suppliers, with a client system that is itself organisationally complex. Not only do these actors participate in the project, but as key stakeholder, also have significant influence over project related decisions (Cleland, 1998).

It is the fundamental differences between inter and intra-organisational types that have implications for the pursuit of strategic alignment and the measurement of project success. Firstly, actors within TMOs are brought together under contractual conditions to provide specific elements of management, services or resources to deliver a facility on behalf of a client body. Hence, much of the research into TMO relationships focuses on procurement strategies (Lizarralde *et al.*, 2011). Secondly, the inter-organizational nature of TMOs often results in actors having different levels of expertise, overlapping areas of responsibilities and disparate strategic objectives (Jones and Lichtenstein, 2008). Thirdly, engagement in a construction project is over different points in time and changes throughout the project lifecycle as specific organisational services or resources are required (Cherns and Bryant, 1984). Consequently, actors within a TMO are never fully integrated, either within or between organisations involved in a construction project (de Blois and Lizarralde, 2010).

1.2.3 Strategic alignment of projects

The most important distinction with regards to the strategic alignment of projects, concerns the degree of autonomy a project has in relation to the client organisation (Lampel and Jha, 2004). The dominant discourse within the literature makes the assumption that temporary organizations are subordinate to a single parent organisation and will serve as an *'obedient servant'* to the parent organisation, as its most important stakeholder (Artto *et al.*, 2008). The notion is that strategic objectives formulated at the upper levels of the strategic hierarchy are fixed for the subordinate, temporary organisation to implement as directed. Alignment within this context is measured on how well the project management process supports the parent organisation's strategy (Srivannaboon, 2006). In contrast, within any given construction project, there exists a number of organisations that are guided by their own strategic objectives, organisational structure, individual set of stakeholders and own rationale for participating in the TMO, from which varied perceptions of project success will be measured. Consequently, it is unlikely that the activities of the TMO will be constrained to the strategic directives of a

single parent organisation. Instead, it is proposed that the TMO will be in a state of compromise and negotiation as it seeks to align the project with multiple organisational strategies.

In considering the argument above, this thesis sets out to address an important gap in construction management research, by investigating how individual organisations, that comprise a TMO, seek to align multiple strategic objectives through a single construction project. In considering the complex interactions that occur between organisations within a TMO and the complexity of strategy itself (Hambrick, 1983; Mintzberg *et al.*, 1998), it is proposed that alignment of an organisation's corporate strategy with the implementation of a construction project will not be as prescriptive as current literature assumes. Instead, project success will be subjective and will follow a complexity of strategic influences.

1.3 Research problem

This thesis identifies three related problems that are addressed through the research, leading to a problem statement in Section 1.3.4. The first considers the limitations of current studies into the strategic alignment of projects. The second considers the nature of the construction industry, in respect to the alignment of strategic objectives. The third problem considers the difficulty in reaching agreement on project success, when multiple organisations engage in a single construction project.

1.3.1 Current studies of strategic alignment of projects

As previously discussed, the dominant discourse explores the strategic alignment of projects within the intra-organisational context (Milosevic and Srivannaboon, 2006; Skulmoski and Hartman, 2000; Ritson *et al.*, 2012). Of the models that have been developed explaining how strategic objectives are implemented through a project, focus has been on the inherent constraints within a single organisational hierarchy (Morris and Jamieson, 2004; Turner, 1999). More often, implementation of strategy is seen as a deliberate process where strategy is articulated at a higher level of an organisation and realised at the project level, through operational detail and tactical adjustment (Archibald, 1988; Youker and Brown, 1998). However, within a project environment the strategic plan is rarely acted upon in a prescriptive manner as models suggest and most strategic actions manifest themselves in a more, haphazard, process (Hauc and Kovac, 2000).

The underlying assumption made within a strategic hierarchy is that strategy involves clear communication of strategic intentions and objectives, against which line managers devise their own operating targets and plans (Snow and Hambrick, 1980). Consequently, the strategy implementation process is entrusted to the organisations internal systems and procedures (Mintzberg and Waters, 1985; Hrebiniak and Joyce, 1985), and in the form of projects becomes the responsibility of the project manager, who typically has little involvement in the strategy formation process (Pellegrinelli and Bowman, 1994). It is this gap between formulation and implementation that is considered to be the main deficiency in most strategic planning processes (Grundy, 1998). Also, because such a separation hinders the rapid implementation of business strategies and ignores emergent influences (Crawford, 2005).

Mintzberg (1994) proposes that strategies emerge over time, rather than being a deliberate course of progression. Although, most projects follow some form of strategic planning process, strategy may not be realized in a ridged or formal manner as the higher levels of the hierarchy assumes (Mintzberg and Waters, 1985; Morris and Jamieson, 2004). Typically, the realisation of strategy is subject to external forces (Porter, 1985) and internal influences (Rumelt, 1974). As such, strategic objectives developed at one point in time, at a specific hierarchical level, will be subject to incremental deviations in response to external and internal pressures at each stage of the strategic hierarchy. Despite this, current models demonstrating how organisational strategy is translated through a project, do not take into consideration external or emergent strategic influences.

1.3.2 Strategic alignment of projects within the construction industry

In comparison to temporary organisations within an intra-organisational context, there has been relatively little research considering the strategic alignment of projects within the construction industry. Previous studies have investigated the competitive strategies of construction organisations through business level and project management processes (Tan *et al.*, 2012; Meredith and Chinowsky, 2000; Budayan *et al.*, 2015), but studies investing the pursuit of the strategic aspirations of multiple organisations participating in a single construction project remain sparse. Unlike projects operating within single organisational boundary, the TMO operates in an environment of overlapping organisational boundaries, where multiple strategic hierarchies converge. Within the construction industry, a single construction project represents a strategic investment for

a number of organisations. This not only includes the investment decision of the client to develop a facility as part of a wider organisational strategy, but also the strategic decisions of the consultants and contractors to commit recourses to a specific project in order to achieve specific competitive advantages.

Current research within TMOs tend to focus on formal procurement strategies and contractual relationships, as mechanisms to maintain alignment with the client's strategic objectives (Lizarralde *et al.*, 2011). But, even in themselves, procurement systems provide strong evidence of the predominance of the single project perspective (Blismas, 2001). Whereas, procurement strategies provide formal conditions of responsibility of each organisation participating in the project, they do not determine the strategic behaviours of individual TMO actors. As a result of the uncertainty, high risk and problems associated within a project environment (Lau and Rowlinson, 2011), coupled with the partial engagement of TMO actors over time (Cherns and Bryant, 1984), behaviours within the TMO are likely to evolve and change over the duration of the project.

This creates difficulties for the notion of strategic alignment within the context of a TMO. At the operational level of any given construction project there exist a significant number of stakeholders in the form of the project team, who differentiate in terms of skills, professional body and loyalty to ones own firm (Walker, 2007). Moreover, the consequences of fragmentation between parties has been discussed by Latham (1994) and confirmed in Egan's (1998) report, which views the construction industry as:

"...typically dealing with the project process as a series of sequential and largely separate operations undertaken by individual designers, constructors, and suppliers who have no stake in the long-term success of the product and no commitment to it" (Egan, 1998: 13)

TMO actors develop personal conflict of allegiances, not only to the client, but also to their employer, professional body and bounded rationality (Cyert and March, 1963). As each team member is commonly employed by different organisations there does not exist one organisational strategy, rather there exists a number of strategies, all being pursued at any one time.

1.3.3 Project success

A further difficulty deriving from procurement processes is that success of the project is constrained to the efficiency of the TMO in delivering the project within time, cost and quality objectives as stated in the conditions of contract. Whereas, efficiency in managing projects may contribute to project success, it does not necessarily result in the project being successful (De Wit, 1988). This is due to the subjective nature of success (Pinto and Slevin, 1988), and the notion that varied stakeholders within a construction project have different opinions on what constitutes an effective project and make assessment on varied, often subjective, criteria (Freeman and Beale, 1992; Belassi and Tukel, 1996).

In considering the multiple stakeholders that have an interest in a construction project, there is a high likelihood that there will be a difference in the perception of project success within and between organisations participating in the project. Whereas, members of the temporary organisation are likely to measure success on the efficient implementation of project processes, members of the parent organisation are likely to measure success on the contribution the project makes to the individual levels within the client hierarchy, and to the long-term strategic objectives of the business (Shenhar *et al.*, 2001; Baccarini, 1996).

1.3.4 Problem statement

Therefore, the research problem addressed in this thesis is stated as follows:

The construction industry presents a situation where multiple organisations seek to align multiple strategic objectives through a single a project. Consequently, the realisation of organisational strategy, and project success, will be influenced by factors internal and external to the TMO.

The literature review, methodology, data collection and data analysis within this thesis, are designed to address the research problem, as stated.

1.4 Research significance

Through investigation into the strategic alignment of a TMO, this thesis contributes to theory and research in the following areas:

1.4.1 Contribution to temporary organisational theory

Since publication of Lundin and Soderholm's (1995) seminal paper introducing a theory of temporary organisations, there has been a number of studies exploring the temporary

organisational form within the intra-organisational context (Bakker, 2010; Janowicz-Panjaitan *et al.*, 2009; Bechky, 2006). In contrast, research into the inter-organisational context of temporary organisations has been relatively sparse. This is despite Cherns and Bryant's (1984) publication defining the basic characteristics of a TMO and providing guidance for research into the temporary organisational form, being introduced into the literature eleven years before Lundin and Soderholm's (1995) theory of temporary organisations. Through empirical study, this research examines the unique strategic behaviours of TMO actors within the themes of *time, task, team* and *transition,* thereby extending knowledge of temporary organisations to greater consideration of inter-organisational implications. At the same time, TMO characteristics, as defined in Cherns and Bryant's (1984) pilot study, are empirically examined.

1.4.2 Contribution to theories of strategic fit

Current models of alignment explain how strategy is set at the corporate level of a parent organisation and implemented by a temporary organisation created within the organisational boundaries of a single organisation. In contrast, within any given construction project there exist a number organisations that are influenced by their own strategic objectives, organisational structure, individual set of stakeholders and strategic rationale for joining the TMO. To date, a definitive model explaining how multiple organisations align strategic objectives through a single construction project has not been presented within the literature. It is argued that within the context of a construction project, strategic alignment will be significantly more complex than within an intraorganisational context. This is not only as a result of the increased number of stakeholders, but also the intra-inter originations relations (Lizarralde *et al.*, 2011), the complexity of the client system (Cherns and Bryant, 1984), environment conditions (Porter, 1980), the varied perceptions of project success (De Wit, 1988), and the strategic behaviours of actors within a temporary organisation (Lundin and Soderholm, 1995).

1.4.3 Contribution to understanding of project success

The third contribution to theory concerns the perceptions of project success. Despite the abundance of studies investigating the success of projects, there are actually limited studies considering how success of a construction project supports the strategic objectives of a client organisation. Most studies considering project success tend to focus on success factors that need to be in place for success to be realized, with less

research into the criteria on which success should be measured (Ika, 2009). Based on the proposition that strategy occurs at different levels within an organisation (Hofer and Schendel, 1978), and that varied stakeholders will have different perspectives of project success (De Wit, 1988), this thesis makes contribution to theory, by identifying the varied perceptions of project success within the context of a construction project, and establishing how these link to the realisation of strategic objectives.

1.5 Research aims and objectives

In considering the research problems established in Section 1.3, and the contribution to theory affirmed in Section 1.4, the aim of this thesis is to investigate how varied organisations within a temporary multi-organisation seek to align multiple strategic objectives through a single construction project, and realise project success. This will be explained through a theoretical model of strategic alignment within the context of a TMO, as an alternative to models explaining strategic alignment within the context of a single organisation. To address this aim, the following research objectives of the study are pursued:

- **Objectives 1:** Explore how multiple organisations within a TMO align varied strategic objectives through a single construction project.
- **Objective 2:** Examine the effectiveness of mechanisms implemented to maintain alignment of strategic objectives.
- **Objective 3:** Explore the strategy of the TMO in the pursuit of the varied organisational strategies inherent within a TMO
- **Objective 4:** Explore the linkages between project success and the intended strategic objectives of TMO actors.

1.6 Methodology

The research within this thesis can be characterised as exploratory within the context of construction management and strategy. The aims and objectives of the research are presented in Section 1.5 and an outline and structure of the thesis is presented in Section 1.7. The research questions informing the study were developed from gaps identified within the current literature, presented in Chapter 3. The methodology and research

approach designed to address the research questions and research objectives are detailed in Chapter 4.

A multiple case study design, using replication as the underlying logic (Yin, 2014; Eisenhardt, 1989b) was developed to provide the empirical data to achieve the aims and objectives of the thesis. Cases studies were selected through purposeful sampling using pre-defined criteria. Four case studies, comprising of recently completed construction projects within the public sector and used to provide data for the research. Data within each case study was formed from a combination of collection methods, which included semi-structured interviews, archives and documentation. Triangulation was achieved through multiple sources of evidence. Interviews were conducted at the micro level between organisations participating within the TMO, including key stakeholders within the client organisations. To ensure reliability, all interviews were professionally transcribed and managed using NVivo computer assisted qualitative data analysis software, for coding and analysis.

1.7 Structure of thesis

The structure of the thesis, identifying the purpose of each chapter is illustrated in Figure 1.1. An outline of each chapter is as follows:

- **Chapter 1** provides the background to the field of study, summarising the theoretical perspectives considered within the research and the purpose of study. Research aims and objectives are justified through a discussion of the research problems addressed within the thesis and the contribution the research makes to theory.
- Chapter 2 provides the first of two chapters forming the critical analysis of literature and themes explored within the research. This, first chapter positions projects within the context of organisational strategy and provides a review of the extant literature contributing to the strategic alignment of projects. The review includes an overview of the main theoretical models of strategy and examination of the different perspectives to strategic fit from within the strategic management literature, including the concept of the hierarchy of strategic objectives. To position the study within the contemporary literature, a short history of project management research is

presented, introducing the strategic stream of project management research. This is followed by a critique of current models explaining the strategic alignment of projects and an introduction to the concept of project strategy.

- Chapter 3 provides a review of the extant literature on temporary organisational forms that are created to implement organisational strategy through projects. In particular, the review focuses on the behaviour of temporary multi-organisations within the construction industry. The chapter presents a review of the early discourse on temporary systems from within the organisational behaviour literature, before providing a critique of Lundin and Soderholm's (1995) basic concepts that contribute to the theory of temporary organisations. Variations of temporary organisational forms are explored before defining a TMO. Following Cherns and Bryant's (1984) study, the issues of client complexity within the context of the construction industry are discussed. The examination of TMOs concludes by investigating the role of procurement strategies and governance mechanisms in the strategic alignment of projects. Finally, a discussion of research gaps, research questions and a conceptual model, to guide the methodology of the study, is presented.
- Chapter 4 describes the rationale for the research method pursued within this study. This includes a discussion of the philosophical assumptions that influence the methodological choice. Justification for the case study research approach is provided, along with a detailed description of the research design guiding the data collection and analysis. The methods of data collection utilised within the research are discussed with an explanation of the instruments used to analyse the collected data, and from which conclusions are drawn from the findings. Finally, the principle of ethical research relevant to the thesis are considered and discussed.
- Chapter 5 is the first of two chapters reporting the empirical findings and analysis from the collection of data. Through a process of within-case analysis, this chapter offers an in-depth investigation into each of the four cases. For the purpose of replication logic, each case is segmented in specific themes, which are used to address the research questions of the study. These include the structure of the TMO, an examination of the strategic objectives, mechanisms implemented to maintain alignment of strategic objectives, internal and external influencing factors, an

examination of the project strategy, and an enquiry into the varied stakeholder perception of project success.

- Chapter 6 discusses the findings through the process of cross-case analysis, based on the *a priori* themes identified within literature review and *a posteriori* themes emerging from the findings. Themes are explored and examined in relation to the research objectives of the study. Two important models, developed from the analysis, are presented to address the aims of the thesis. The first is a model of project success identifying the varied perceptions of project success within a TMO, and the implications for strategy alignment. The second is a model of strategic alignment within the context of a TMO, which explains the complex interactions, internal and external influences, and the strategic behaviours of TMO actors, that determine the strategy of the TMO in pursuit of project success and realisation of strategic objectives.
- **Chapter 7** presents the final conclusions of the research. This chapter begins with a review of the aims the study and an explanation of how each of the research objectives were achieved. A discussion of the theoretical and practical contributions is also presented. In reflecting upon the limitations of the study, recommendations for future research are proposed.



Figure 1.1: Structure of thesis

1.8 Conclusion to chapter

This introductory chapter provides the general context of the thesis, summarising the theoretical perspectives considered within the research and justification for the study. The research aims and objectives are stated, with an outline of the structure providing guidance as to how the aims and objectives of this thesis will be achieved. The following chapters develop the theoretical context through a review of the literature and the development of conceptual diagram to guide the methodology of the study.

CHAPTER 2

THE STRATEGIC CONTEXT OF PROJECTS

2.1 Introduction to chapter

This chapter is the first of two forming the critical analysis of literature and themes explored within the study. The chapter positions projects within the context of organisational strategy and provides a review of the extant literature contributing to the strategic alignment of projects. The review commences with an overview the main theoretical models of strategy. This is followed by examination of the different perspectives to strategic fit from within the strategic management literature. The concept of the hierarchy of strategic objectives is then introduced. The second part of the chapter provides a summary of the evolution of project management and the move towards implementing strategy through projects, before presenting a critique of current models of strategic alignment of projects. Finally, the chapter concludes by introducing the concept of project strategy, exposing the challenges of aligning projects with organisational strategy.

2.2 The nature of strategy

A significant number of scholars find a lack of consensus for the definition of 'strategy' within the literature (Hofer and Schendel, 1978; Chaffee, 1985; Bourgeois, 1980; Shirley, 1982). Hambrick (1983) proposes that this lack of consistency is due to two factors; firstly, that strategy is multidimensional in character, and secondly, strategy is situational and varies by industry. Chaffee (1985) also suggests that the lack of consensus regarding a definition is due to the pluralistic nature of strategy and the fact that various models exist to explain the concept. To explore the varied perspectives, the review draws a comparison between Chaffee's (1985) three models of strategy and Mintzberg's (1987) classification.

2.2.1 Chaffee's three models of strategy

Chaffee (1985) suggests that the term 'strategy' refers to three mental models of 'linear', 'adaptive' and 'interpretive'. The focus of the 'linear' model is on long term strategic planning. This focus on planning is also inherent in Chandler's (1962) definition as "the determination of the basic long-term goals of enterprise and the adoption of courses of action and the allocation of resources necessary to carry out these goals" (Chandler, 1962: 13). Within this perspective, strategy consists of

decisions, actions and plans delivered in a sequential and hierarchical process. Chaffee (1985) advises that it is this model that forms the foundations of strategic planning where conceptually, managers at every level of the hierarchy must agree on a detailed, integrated plan of action for the coming year (Lorange and Vancil, 1976). This is despite Mintzberg's (1994) observation that the commitment required at each level of the hierarchy, together with the lack of flexibility required by strategic thinkers, resulted in a shift away from strategic planning by the mid 1970's.

The 'adaptive' model is characterised by Hofer (1973) as being "concerned with the development of a viable match between the opportunities and risk present in the external environment and the organisations capabilities and resources for exploiting these opportunities" (Hofer, 1973: 3). Within this perspective the organisation is expected to define its relationship to its environment in the pursuit of objectives (Bourgeois, 1980). This requires constant monitoring of the external environment and making simultaneous internal changes. Therefore, the key premise of the 'adaptive' model is that strategy should favourably 'match' the business with its environment (Andrews, 1971; Porter, 1980; Hatten and Schendel, 1977). However, whereas the 'adaptive' model assumes that the organisation must change with the environment, Miles and Snow (1978) argue that strategy is relatively fixed and therefore constrains the organisation in its response to environmental changes. Despite this, the 'adaptive' model goes someway towards explaining the complexity of the changing environment and is consistent in current models of strategy formation (Lynch, 2006; Pearce and Robinson, 2003; Grant, 2005).

The third model in Chaffee's (1985) definition is the '*interpretative*' strategy. This view is based on the social contract that portrays the organisation as a collection of cooperative agreements entered into by individuals with free will. Within the '*interpretative*' model, strategy becomes a mental process where the organisation's existence relies on its ability to attract enough individuals to cooperate in mutually beneficial exchange. In contrast to the adaptive model where the environment is defined, the '*interpretative*' model manages the environment through symbolic actions and communication. Therefore, within this context, strategy becomes an organisational wide activity and is dependent on the motivation of stakeholders to believe and act in ways that are expected, in order to produce favourable results for the organisation. This is in contrast to the 'linear' model where strategy is predetermined by senior management.

2.2.2 Mintzberg's five P's for strategy

Mintzberg (1987) presents five dimensions of strategy, which he refers to as the "Five P's". These are defined as *'plan'*, *'ploy'*, *'pattern'*, *'position'* and *'perspective'*. Strategy as a *'plan'*, aligns with Chaffee's (1985) *'linear'* model, in that strategy setting involves a deliberate conscious set of guidelines that determine decisions for the future (Mintzberg, 1978). As a *'plan'*, Mintzberg (1987) proposes that strategy can also be a *'ploy'* defined as a specific manoeuvre intended to outwit an opponent or competitor.

The third dimension in Mintzberg's (1987: 12) framework is strategy as a "*pattern in a stream of decisions*", which aligns with Chaffee's (1985) '*interpretive*' model. This dimension considers the resulting behaviour of strategy over time. For Mintzberg and Waters (1985), most manifestations of strategy are implicit, fragmented and fluid. A distinction is drawn between deliberate strategies, as those realised as intended, and emergent strategies, which are realised despite, or in the absence of, intertion. Mintzberg and Waters (1985) argue that for a strategy to be realised as intended, at least three conditions need to be satisfied. First, there must exist precise intentions of the organisation articulated in a specific level of detail. Second, there must be effective control mechanisms in place to ensure that the intentions are indeed organisational. Third, strategic intentions must be realised exactly as intended, and that no external force could have interfered with them. Therefore, in accepting that organisational strategies emerge unintentionally, it is proposed that the only way of defining strategy is to analyse the strategic actions and results after the event (Mintzberg, 1978).

Mintzberg's (1987) 'position' dimension supports Chaffee's (1985) 'adaptive' model, in that strategy involves locating the organisation within the external environment and seeking a match between the internal and external context (Hofer and Schendel, 1978). Mintzberg (1987) affirms that a 'position' can be predetermined through a 'plan' or 'ploy', but realised through a pattern of behaviour and deployment of resources. Whereas, the 'position' dimension considers the external environment, the 'perspective' dimension considers the internal environment of the organisation. This includes the internal structures, culture and processes. Successful implementation of strategy within Chafee's (1985) '*adaptive*' model and Mitzberg's (1987) '*position*' and '*perspective*' dimensions are dependent on alignment either between the internal and external environment or within the organisation. This is discussed further in the following section.

2.3 Strategic fit and alignment

The concept of strategic fit has become a central theme within the strategic management literature. Examination of the literature reveals different terms used to define the principle. These include 'congruence', 'match', 'consistency' and 'alignment'. As a core construct of contingency theory, the basic proposition of strategic fit is that the degree of alignment between organisational and environment conditions has a significant effect on performance (Ginsberg and Venkatraman, 1985). Fry and Smith (1987) advise that, if conditions or organisational components 'fit well' then the organisations functions effectively. In contrast, if components 'fit poorly' the converse will apply.

Despite the general agreement within the literature regarding the notion of fit, there remains criticism regarding clarification of the concept (Venkatraman and Camillus, 1984; Van de Ven, 1979). This can partially be attributed to the fact that differing views exist of how fit should be achieved and which organisational components or conditions need to be aligned. A review of the literature finds at least three different perspectives to explain the domain of fit. The first perspective focuses on strategy formulation and sees fit as alignment between strategy and the external environment. Fundamental to strategic management, the concept of strategic fit involves aligning organisational resources with environmental opportunities and threats (Andrews, 1971; Chandler, 1962). Drawing on industrial organisation economics (Bain, 1950), Porter (1980) identifies five key industry forces that need to be recognised by the organisation when devising strategy. Whereas, Hatten and Schendel (1977) determined the need to formulate differential strategies through strategic groups within specific industries. Also, in considering the product life-cycle growth rate, Hambrick *et al.* (1982) studied the relationship between strategy and relative market share.

The second perspective focuses on strategy implementation, and sees fit as alignment between strategy and internal elements. Strategic alignment is achieved by tailoring internal administrative and organisational mechanisms in line with organisational strategy, with almost no direct reference to external influences (Venkatraman and Camillus, 1984). The dominant theme in this perspective is the strategy-structure relationship. This follows Chandler's (1962) proposition that a change in strategy requires change in all organisational activities so that the structure of the organisation fits with the new strategic objectives. The salient factor within the strategy-structure perspective is that the strategy is articulated at the outset and alignment is achieved by adapting mechanisms and structure to the articulated strategy (Ansoff, 1965). In considering diversification strategy, Rumelt (1974) related the fit between strategy and structure on performance. Lenz (1980) also found combinations of strategy and structure on high performance firms differed from combinations associated with low performance firms.

However, Venkatraman and Camillus (1984) caution that strategy implementation is more than a fit between strategy and structure. Norburn and Miller (1981) propose a fit between strategy and reward systems. Schwartz and Davis (1981) argue for the need to ensure alignment between strategy, organisation and culture. Whereas, Waterman Jr. (1982) argued for alignment among the seven internal elements of strategy, structure, systems, style staff, skills and subordinate goals, as presented within the McKinsey 7-S framework (Waterman Jr. *et al.*, 1980).

The third perspective takes an integrated approach, where strategic management involves both strategy formulation and implementation, covering both organisational and environmental decisions. Central to this theme is the proposition that there must be alignment between the context, strategy and structure (Galbraith *et al.*, 1986; Venkatraman and Camillus, 1984). Models such as Miles and Snow's (1978) attempt to specify a relationship between structure, strategy and process by proposing that firms should develop relatively stable patterns of behaviour in order to accomplish alignment with environmental conditions. Nadler and Tushman (1980) take a total systems approach and suggests that fit occurs when each organisational components needs, demands goals and objectives are aligned with other components needs, demands goals and objectives. In considering the ability to realise the value of technologies, Henderson and Venkatraman (1993) present a model of alignment between the market, structure and IT processes. From a management perspective, Chorn (1991) proposes that strategic fit rests on the alignment between the four '*logics*' of the competitive situation, strategy, culture and leadership. More recently, Carmeli *et al.* (2010) explores how leadership

enables a firm to change and adapt to its external environment and, by implication enhance performance. They argue that while internal fit is of importance, it is of little value by itself unless the organisational system as a whole is aligned with its external competitive environment.

Despite, wide ranging discussion regarding the notion of fit, a number of authors recognise inherent problems with the concept. Zajac *et al.* (2000) observe that the static orientation of fit implies alignment only at a single point in time and does not appreciate the extent of strategic change. Venkatraman (1989) also concludes that existing studies focus on cross-sectional approaches and calls for the development of mechanisms to test fit within a longitudinal perspective. Moreover, the static orientation of fit does not adequately take into the consideration the emergent strategies discussed in Mintzberg's (1987) framework. As affirmed by Zajac *et al.* (2000) organisations face multiple environmental and organisational contingencies and, therefore, the concept of fit and alignment need to be conceptualised in more dynamic terms to reflect uncertainty and the changing environment.

2.4 The strategic hierarchy

A further complexity in the concept of alignment is the recognition that strategy occurs at differing levels within an organisation. In proposing that each level be constrained by the upper level, Hofer and Schendel (1978) categorises the levels of a strategic hierarchy as *'corporate'*, *'business'* and *'functional'*. The proposition is made that strategic priorities between levels of the organisation need to be aligned for organisational performance to be enhanced (Joshi *et al.*, 2003).

Corporate-level strategy defines the scope of the organisation in terms of the industries and markets in which it competes and the range of business it pursues (Andrews, 1971; Ansoff, 1965). Hofer and Schendel (1978) provides the following definition:

"Corporate-level strategy is concerned primarily with answering the question of what set of businesses should we be in. Consequently, scope and recourse deployments among business are the primary components of corporate strategy" (Hofer and Schendel, 1978: 27).

Referred to as the "*primary strategy*" of the organisation (Bourgeois, 1980: 27), an organisations corporate strategy can be operationalized in terms of the distribution of a firm's assets, employment, capital-budget or other indexes of a firm's resources among the range of industries and markets it operates in (Beard and Dess, 1981). Thus, at this

level the primary role of an executive board is to ensure the growth, risk reduction and profitability of the organisation through the synergy of the diverse business units within the firm (Grant, 2010; Rumelt, 1982).

Business level strategy focuses on the competitive actions of a firm in a particular industry or product/market segment (Hofer and Schendel, 1978). Bourgeois (1980: 27) refers to this as the "secondary strategy" of the organisation. Beard and Dess (1981) defines business level strategy in terms of variation in a firm's characteristics, relevant to competitive success or failure within a given industry. As such, a firm has a unique business-level strategy for each industry within which it competes, and "the relevant characteristics of the firm's business level strategy would be measured relative to the range and norms on each characteristic in each of the industries" (Beard and Dess, 1981: 667).

From a resource-based perspective, the type of strategy the firm chooses to pursue will be greatly influenced by its core competencies (Prahalad and Hamel, 1990). As each strategic business unit (SBU) is fundamentally constrained by the resources and capabilities at the upper level of the hierarchy, competitive advantage is sustained by redeploying the unique valuable resources of the SBU to enable the creation and enhancement of dynamic capabilities in response to the changing environment (Teece *et al.*, 1997). For Hofer and Schendel (1978) achieving synergy at this level becomes more significant, as it needs to focus on the integration of different functional area activities within each, single SBU.

The focus of the functional-level strategy is on the maximisation of resource productivity within each function of a single SBU (Ginsberg and Venkatraman, 1985). At this level, synergy and the development of distinctive competencies becomes the key strategic component, where synergy involves the coordination and integration of activities within a single function (Hofer and Schendel, 1978). It is the functional strategies that define the operations and approaches within each functional department and, in turn, are translated into a sub-hierarchy of actions for implementation (Mintzberg, 1994).



Figure 2.1: Planning hierarchy. Source: Mintzberg (1994)

Mintzberg (1994) presents a model of strategic planning, illustrated in Figure 2.1. The model demonstrates how strategic objectives formulated at the corporate level cascade down a hierarchy to be implemented as actions at the operational level. In explaining the model Mintzberg (1994) acknowledges the general assumption in strategic planning that objectives determined by senior management are implemented as articulated. However, research finds that strategies derive from a two way process, where objectives at the lower level impact on strategic objectives at the upper levels. Slack *et al.* (2006) presents a model illustrating how corporate objectives impact on business objectives, which influence operations strategy. The salient issue in this model is the recognition that circumstances, experiences and capabilities at the operational level reveal an emergent sense of what the strategy should be. Therefore, strategic actions of the
organisation are not only implemented from a top-down hierarchy, but also from bottom-up hierarchal influences.

2.5 Streams of project management research

Prior to discussing the varied models of strategic alignment of projects and project success, it is helpful to understand the difficulties that exist when seeking to implement organisational strategy through projects. This is explained through an overview of project management research, which according to Söderlund (2002) has evolved in two streams. With reference to the evolution of the discipline, these streams of research have been more or less influential in shaping modern project management practice and behaviour.

2.5.1 Quantitative stream of project management research

The first stream focuses on the mathematical approach of planning and control techniques developed in parallel by, both, the US Department of Defence (DoD) and the chemical industry between 1958 and 1959 (Morris, 1997). These efforts resulted in two of the most important early contributions to project management research, most notably the development of Programme Evaluation and Review Technique (PERT) by US Navy Special Projects and the Critical Path Method (CPM) by E.I du Pont de Nemours Company (Siemens, 1971; Archibald, 1987; Fondahl, 1987). On the apparent successful application of these techniques the US DoD developed further systems based tools, including publication of the DoD/NASA PERT/cost guide (1962) to be used by contractors on US defence projects.

However, as observed by Thomas (2006), many of the techniques used at this time were developed on an *ad hoc*, trial and error basis, rather than being a deliberate management activity. It was only due to the efforts of the US DoD themselves, that the concept of project management became well publicised, with CPM and PERT being the fundamental models of the concept. Despite these efforts, by the mid-60's, the preoccupation with quantitative planning techniques began to receive criticism (Packendorff, 1995). The reaction from contractors to adopt the new techniques was decisively negative and from a financial perspective, customers realised that the deterministic systems resulted in expenditure of considerable cost and effort by the contractor (Kerzner, 2003; Archibald, 1987). Moreover, researchers were also

questioning the underlying contribution of rational techniques to project success (Avots, 1969).

2.5.2 Human resource stream of project management research

In a shift from quantitative techniques, the second stream of project management research was concerned with human resource management, leadership and organizational theory. Cleland and Ireland (2006) identify two seminal papers that were influential to the development of the second stream. Firstly, the much cited paper published in the Harvard Business Review by Gaddis (1959), was the first to introduce the concept of project management as a recognised job description, outlining the leadership and responsibility role of the project manager. The second significant contribution, also published in the Harvard Business Review functional teamwork approaches in organisational design (Fish, 1961). The contingency approach to organisational structures was also being developed during this period (Lawrence and Lorch, 1967), which was followed by Galbraith (1971), who described a series of organisational forms ranging from function, product/project, and the benefits of creating project teams from varied departments to form matrix structures.

The most influential movement in terms of establishing a project management discipline came from the formation of the Project Management Institute (PMI) who sought to homogenize the practice by developing generic sets of standards in the form of the Project Management Body of Knowledge (PMI, 1987). The aim of these guides was two-fold. Firstly, to formally standardise the growing discipline of project management by presenting a set of guidelines that are deemed to be best practice. Secondly to accommodate the widening discipline of project management by presenting a generic set of tools and management methods. Although criticised for its mechanistic process and instrumental rationality (Hodgson and Cicmil, 2006), it is this body of knowledge with the emphasis on a standardised rational approach that underpins project management practice.

2.5.3 Strategic stream of project management research

The review of the literature finds that a third stream of research evolving during the late 1980's, as project management practices transferred from construction and engineering into mainstream management. This shift saw a widening of the discipline where project

management became the dominant model in many organisations for strategy implementation (Pellegrinelli and Bowman, 1994; Van Der Merwe, 2002; Hauc and Kovac, 2000). The extension of project management practices also saw Management By Projects (MBP) emerge in response to the growing demand to provide organisations with the flexibility and response to compete under turbulent conditions (Gareis, 1989). This approach gives the advantage of enabling organisations to employ project management principles, used on large external projects, for smaller internal projects (Sharad, 1986; Gareis, 1991). One of the salient benefits presumed in the literature is that the project management approach enables organisational strategy to be implemented more efficiently and effectively, thus shorting the time from strategy formulation to strategy implementation (Hauc and Kovac, 2000; Partington, 1996). By employing a project management approach to realising organisational strategies, businesses are able to partially eradicate the traditional bureaucratic, mechanistic structures, which according to Burns & Stalker (1961) are inherently resistant to strategic change.

Despite these developments, the current conceptual base of project management continues to attract criticism for its lack relevance to current business practice (Packendorff, 1995; Maylor, 2001), the dominant allegiance to its '*hard paradigm*' (Pollack, 2007: 226) and the assumption in main stream project management literature that a single theoretical base exists to adequately explain the actual management of projects (Winter *et al.*, 2006b). This is in contrast to the field of strategy, where there exist varied opinions and schools of thought to explain the concept (Chaffee, 1985; Mintzberg, 1987; Mintzberg *et al.*, 1998). Indeed, the homogenization of project management practice through the varied influential bodies of knowledge has resulted in project management becoming a mechanistic process driven by instrumental rationality (Hodgson and Cicmil, 2006). Moreover, the criticism in contemporary literature of the widespread assumption that all projects are fundamentally similar and can be managed by a universal set of project management activities (Evaristo and van Fenema, 1999; Winter *et al.*, 2006b; Blismas *et al.*, 2004; Dvir *et al.*, 1998) has implications for the strategic alignment of projects and project success.

2.6 Project success criteria

Although project success has been discussed extensively within the project management literature, the concept of success remains, both, subjective and ambiguous, with little

agreement concerning the criteria by which success of a project should be measured (Pinto and Slevin, 1988; Freeman and Beale, 1992). The review of literature finds that the majority of discourse on project success has tended to focus on factors that need to be in place for success to be realised (Ika, 2009). However, as cautioned by De Wit (1988), efficiency in managing projects may contribute to project success, but does not necessarily result in the project being successful. This is because varied stakeholders involved in a project will have different opinions on what constitutes success and, therefore, make assessment on varied success criteria (Davis, 2014).

Traditionally, the dominant discourse within the project literature has long been on achieving the specified project objectives within the *'iron triangle'* of time, cost and quality (Oisen, 1971). Project management standards have since extended this criteria to include project performance (BSI, 1996; APM, 1995). In taking a converse perspective, Avots (1969) proposed that if a project exceeds its completion time, exceeds it budget or does not meet the quality and performance criteria determined in the specification, the project is deemed a failure.

More recently, the concept of project success has been perceived to involve broader objectives from the viewpoint of stakeholders throughout the project life cycle (De Wit, 1988). Cleland (1998) defines the 'project stakeholder' as

"...people or groups that have, or believe they have, legitimate claims against the substantive aspects of the project. A stake is an interest or share or claim in a project; it can range from informal interest in the undertaking; at one extreme, to a legal claim of ownership at the other" (Cleland, 1998: 55).

Internal stakeholders are those stakeholders within the project organisation who play an active role in the development and implementation of the project, and external stakeholders are the groups and individuals who may not actively participate in the project, but may influence the project outcome in either a negative or positive way (Pinto, 2013; Winch, 2007).

In their study of 94 projects to determine patterns of causes of project failure, Pinto and Mantel (1990) identify three success criteria, which includes: 1) implementation of the project; the perceived value of the project; and client satisfaction of the delivered project. Whereas, Westerveld's (2003) Project Excellence Model considers five

subjective measures of success from varied stakeholder groups that include the client, project personnel, end users, contracting partners, and external stakeholders.

In accepting that success of a project will be defined by the perception of varied stakeholders, a number of authors began to draw distinction between the efficiency in the implementation of the project and the perceived value to the organisation as a result of implementing the project. In Baccarini's (1999: 25) theoretical framework, two components of project success are identified:

- **Project Management Success:** Focuses upon the project process and, in particular, the successful accomplishment of cost, time, and quality objectives. It also considers the manner in which the project management process was conducted.
- **Product Success:** Considers the effects of the project's final outcome.

Despite the distinction, Collins and Baccarini's (2004) empirical study of 150 Australian project managers found that the traditional project management objectives of time, cost and quality was still dominant in the Australian project management community, with only a small percentage perceiving the meeting of owners needs as a success criterion.

In their empirical study of success factors across 71 large multi-national organisations, Cooke-Davies (2002) also draw on De Wit's (1988) differentiation between 'project management success', as being measured against the traditional gauges of project performance, and 'project success' as being measured against the overall objectives of the project. Whereas, Lim and Mohamed (1999) took a micro and macro perspective in their study of project professionals in Kuala Lumpur. Within their framework, they suggest that the micro viewpoint considers project achievements at the smaller component level. This included the measurable project management objectives of time, cost, quality and performance. In contrast, the macro perspective of project success addresses the question of whether the original project concept has been achieved or not.

In taking a strategic perspective Shenhar *et al* (2001) presents a model illustrating four dimensions of project success from the findings of an empirical study into project managers within the technology sector. The model shown in Figure 2.2 demonstrates the time dependence of each dimension, with the relative importance of each dimension varying during and after completion of the project.



Figure 2.2: Relative importance of success dimensions. Source: Shenhar et al. (2001)

- The first dimension considers the efficiency in delivering the project within the time, budget and requirement goal. Shenhar *et al.* (2001) advises that this dimension can be assessed only in the very short-term during the execution of the project and immediately after its completion, and was considered critical to all project managers in expressing the competence in which the project was managed.
- The second dimension addresses the importance placed on client requirements and meeting the client needs. This includes the meeting of performance measures, functional requirements, and technical specifications. This dimension can only be assessed a short time after the project has been delivered to the client and the customer is using the final output.
- The third dimension addresses the immediate impact the project has on the business, in terms of sales, income and profit, as a result of investment into the project. This dimension can only be assessed after a reasonable time to achieve sales, usually one or two years after handover of the project.
- The fourth dimension addresses the issue of preparing the organization and technological infrastructure for the future, and questions if the organisation has explored new opportunities for further markets, ideas innovations and future products as a result of the project. According to Shenhar *et al.* (2001), this dimension can only be realistically assessed after a period of, at least, two to five years.

The significance of Shenhar's *et al's* (2001) framework is that it highlights the strategic value of projects to an organisation. In revisiting De Wit (1988), Cooke-Davies (2007) argues that a project is successful only if it delivers the benefits that were envisaged by the stakeholders that agreed to undertake the project in the first instance. Similarly, Hartman (2000) suggests that a project is successful if it aligns with organisational strategy. Jugdev and Muller (2005) also argue that the narrow focus on achieving the time, cost and quality criteria results in project management only providing operational value to the organisation and calls for need for further investigation into owners attitude towards project success. The salient factor is that it is the owner who is responsible for the project delivering the organisational strategy. Therefore, it is the owner that affects the view of a project within an organisation and therefore influences the perception of project success (Davis, 2014). This is despite efficient delivery of the project within the time, cost, quality and performance criterion.

2.7 The strategic alignment of projects

As discussed in Section 2.5.3, a key theme evolving from the third stream of project management research is the use of projects as a vehicle to implement organisational strategy (Pellegrinelli and Bowman, 1994). This theme recognises that projects are initiated to achieve business results (Dye and Pennypacker, 2002; Patanakul and Shenhar, 2012). Cleland and Lewis (2006) emphasise that projects do not stand alone in the enterprise, rather they are 'building blocks' of the organisational initiatives.

Consequently, a number of scholars have explored methods to ensure that projects deliver the organisational strategic intentions. The majority of studies investigating the strategic alignment of projects have focused their attention on project portfolio management (Archer and Ghasemzadeh, 1999; Aalto, 2000). Within this stream alignment is maintained through attention to evaluation, prioritisation and selection of projects that fit with the strategic objectives of the organisation (Meskendahl, 2010; Aalto, 2000; Cooper *et al.*, 1999). More recently, in their longitudinal study of senior management, Unger *et al.* (2012), defined the project portfolio through termination of projects that no longer conformed to the corporate strategy.

A second stream considers how project processes are aligned. Using case study methodology, Milosevic and Srivannaboon (2006) develop an empirically-based framework of aligning the competitive elements of the business strategy, according to

Porter's (1980) generic strategies, to the elements of project management adapted from Shenhar's (2004) 'Strategic Project Leadership' framework. According to Milosevic and Srivannaboon's (2006) framework, a firms business strategy drives the project management elements of strategy, organization, process, tools, metrics and culture. Similarly, in their empirical study of four organisations, Cooke-Davies *et al* (2009) proposed that in order to maximise value, a fit between the strategic drivers of the business strategy and the project management systems, grouped under policy, people, structure and process, must exist. Furthermore, Budayan *et al* (2015) identifies a relationship between differentiation strategies in the construction industry and project management processes, in particular, time management, cost management and quality management.

2.7.1 Hierarchy of project objectives

Following Hofer & Schenel's (1978) hierarchy of strategy and Mintzberg's (1994) hierarchy of strategic planning, a number scholars have developed hierarchical models of project objectives, demonstrating how organisational strategy is translated into projects.

Archibald (1988) presents a hierarchy, shown in Figure 2.3, illustrating how strategic objectives set at the policy level cascade down through strategic and operational levels to be implemented as projects. Within the model, policy objectives refer to the overall goals of the organisation, established by senior management. The objectives at this level corresponds to the corporate management objectives in Mintzberg's (1994) model, shown in Figure 2.1. The second level of the hierarchy, corresponds to the business management objectives in Mintzberg's (1994) model. These are the sub-objectives that support the corporate level objectives. Youker and Brown (1998) refer these as the strategic objectives of the project. Finally, the operational level relates to the project objectives. Within Archibald's (1988) model these are defined in terms of the deliverables of the project, as described in the detailed project strategies and plans.



Figure 2.3: The Hierarchy of Objectives, Strategies and Projects. *Source:* Archibald (1987)

In his discussion of strategic planning, Kerzner (2004) presents a model showing how corporate strategic plans flow horizontally across a number of SBU's, and vertically to support plans and budgets, thus recognising the individual objectives of each SBU within the organisation. In adapting Turner's (1999) model, Morris and Jamieson (2005) also present a model depicting the sequencing from the business strategy to the individual objectives of the project team members. As shown in Figure 2.4, the model demonstrates how the business strategy consists of a portfolio of programmes and projects that need to be aligned to realise the business objectives.



Figure 2.4: The Hierarchy of Objectives, Strategies and Projects. *Source*: Morris and Jamieson (2005). Adapted from Turner (1999)

Within their empirical study across four organisations into the way corporate strategy is created and moved into projects and programmes, Morris and Jamieson (2004) found that goals and strategies created at the corporate level cascade across SBU's, consistent with Turner's (1999) and Kerzner's (2004) models. In turn, each SBU, in conjunction with corporate strategy planners, develops their own business goals and strategies. In their empirical study, Morris and Jamieson (2004) also found, that in all four cases the programme and project team, of each subsequent SBU, developed project strategies that aligned with the business and corporate strategy.

This is in contrast to Young *et al's* (2012) single case study into the effectiveness of project management within the investment section of public sector offices. They found that programme and project management processes were not focused on the realisation of strategic goals. Instead, the focus of the business units was on individual asset investment, with little emphasis of alignment with higher-level strategies. However, in a follow up study, Young and Grant (2015) found that, over a period of time, some projects did make a contribution to strategic goals, consistent with Shenhar's (2001) model in Figure 2.2.



Figure 2.5: Hierarchy of Strategic Objectives. Source: Haniff & Fernie (2008)

In further adapting Morris and Jamieson's (2005) model, Haniff & Fernie (2008) present a hierarchy of strategic objectives demonstrating how the context of the corporate strategy is communicated down through the strategic hierarchy to be translated into project management activities. The model shown in Figure 2.5 recognises the varied levels of strategy that occur within an organisation. Besides the varied SBU's and their individual portfolios, Haniff and Fernie (2008) show how each programme consists of number of interrelated projects. Alignment at the programme level is maintained through management and coordination of the related projects in response to changes in the business environment (Lycett *et al.*, 2004; Pellegrinelli *et al.*, 2007).

Consistent with Archibald's (1988) hierarchy of objectives, detailed project strategies and plans are developed at the operational level to achieve the project and business objectives. However, as shown in Haniff and Fernie's (2008) model, implementation of the project becomes the responsibility of a project manger that has had little, or no, involvement in the strategy formation process (Crawford, 2005). As such, project decisions are based on senior management direction, external influences and the bounded rationality of the project manager (Cyert and March, 1963). Although prescriptive, the hierarchy in Figure 2.5 demonstrates that the strategic alignment of projects is not a straightforward process of communicating policy objectives from the corporate to the operational level. Within the hierarchy, strategic intentions filter down through a number of management levels, complex interactions and processes. Not only are strategies communicated and translated into projects from the top down, but the impact and affect of the lower level strategic objectives influence the strategies at the upper levels, consistent with Slack *et al's* proposition (2006).

2.7.2 Project strategy

Morris and Jamieson (2005) caution that hierarchies of strategic objectives only depicts the intended, deliberate sequencing of corporate strategy to projects, and does not take into consideration the emergent nature strategy. The significant limitation of the models discussed in Section 2.7.1, is the notion of a single, unified project strategy developed to realise the strategic intentions of the organisation. Presented within the hierarchy of strategies (Morris and Jamieson, 2005; Haniff and Fernie, 2008), project strategy is defined as "*a direction in a project that contributes to success of the project and its environment*" (Artto *et al.*, 2008: 8).

The dominant discourse within the project strategy literature makes the assumption that the project is under the control of a single parent organisation (Shenhar *et al.*, 2007; Turner, 1999; Morris and Jamieson, 2005; Cleland, 2007; Milosevic and Srivannaboon, 2006). As such, projects lack the autonomy to develop strategies or mechanisms independently (Lampel and Jha, 2004). Instead the project serves as an *'obedient servant'* to the parent organisation and conforms to the parent organisations directions, in the implementation of the business strategy (Artto *et al.*, 2008: 49).

In their review of project strategies, Artto *et al.* (2008) conclude that it is not always appropriate for a single organisation to determine the strategy or dictate the operations of a project. Individual projects operate in varied competitive environments and establish their own project strategy (Yang, 2012). Loch's (2000) study of 90 new product development projects from multiple business units illustrate how autonomous projects contribute to the strategic positioning of a parent organization. Artto *et al.* (2008) also recognises the body of literature that views project strategy as a project execution strategy, whereby the authority of the project is limited to designing and implementing strategy for the management of the project, but not in the actual overall

business context that the project is designed to implement. Lam *et al* (2004) explains this through design and build projects in the construction industry, where the client articulates their requirements and the project strategy is developed to execute the directives in accordance with the client specified success criteria.

Artto *et al.* (2008) also acknowledge the existence of projects that are positioned in a complex organisational environment with several powerful stakeholders, rather than one strong parent organisation. Such an arrangement is typical within large-scale construction projects that are executed in a consortium of several firms (Davies *et al.*, 2009). Within these complex environments, the project strategy becomes self-established and is related to the projects own governance structure, rather than an individual organisation (Floricel and Miller, 2001; Miller *et al.*, 2014). However, as suggested by Haniff and Fernie (2008) the notion of self-originated project strategy is not confined to large-scale construction projects. Indeed, any construction project where there exist multiple participating organisations will develop its unique project strategy in order to execute the project. The difficulty occurs when the interests of stakeholders are, inevitably, varied (Toor and Ogunlana, 2010), which influences the perceptions of project success and impacts on the alignment of strategic objectives.

2.8 Conclusion to chapter

A review of the extant literature into the strategic context of projects finds a number of limitations when seeking to align projects with organisational strategy. Firstly, from a theoretical perspective, there appears to be a gap between strategy and project management concepts. The review has identified varied perceptions and schools of strategy that exist to explain the concept (Mintzberg *et al.*, 1998; Chaffee, 1985). Section 2.3 of the literature review also finds that the focus of strategic management literature tends to be on the formation of organisational strategy. In contrast, as discussed in Section 2.5, the dominant strand in the project management literature is the rationale, deterministic model that governs the single project paradigm (PMI, 1987). This model focuses on the implementation of projects bounded by the success criteria of time, cost and quality, and makes little reference, or has any involvement, in the strategy formation process.

Secondly, Mintzberg and Waters (1985) propose that strategies emerge over time, rather than being a deliberate course of progression. Despite this, models depicting a strategic

hierarchy of objectives assume that strategy is formulated at the higher level, cascades down and is translated into projects for implementation (Archibald, 1988; Youker and Brown, 1998). Typically, strategic plans are subject to environmental influences at each level of the hierarchy. However, the ridged processes inherent within the project management bodies of knowledge, do not consider the emergent nature of strategy. Instead, within a hierarchy of strategic objectives, projects are 'given' to a project manager for implementation, regardless of the emergent strategies or environmental forces that potentially impacts on the perceived success criteria.

Finally, models of strategic fit tend to only consider the alignment of strategic objectives within the boundaries of a single parent organisation, where the project strategy is formulated to implement the strategic intentions of one organisation. However, Section 2.7.3 identify contexts where multiple organisations collaborate on a single project, each seeking to realise individual strategic intentions. Not only does this have implications for the concept of project strategy, but also for the concept of project success. As discussed in Section 2.6, multiple stakeholders involved in a project invariably have different success criteria. This includes the success criteria of different organisations involved in a project and the varied success criteria within each different hierarchical level of an organisation. This complexity is discussed further in the following chapter and explored within the study.

CHAPTER 3 TEMPORARY ORGANSITIONAL FORMS

3.1 Introduction to chapter

The previous chapter provided a review of the theory underpinning research in the strategic alignment of projects. This chapter provides a review of the extant literature on temporary organisational forms created to implement organisational strategy through projects. In particular, the review focuses on the behaviour of temporary multi-organisations within the construction industry.

The chapter begins by reviewing the early discourse on temporary systems from within the organisational behaviour literature. This is followed by a review of Lundin and Soderholm's (1995) basic concepts that contribute to the theory of temporary organisations. Extending Lundin and Soderholm's (1995) concept of teams, theories of leadership applied to temporary organisations is then examined. Variations of temporary organisational forms from within the literature are considered, before introducing the temporary multi-organisation. Within this, the issue of client complexity within the context of the construction industry is discussed, followed by critical examination of procurement strategies and governance mechanisms in the strategic alignment of projects. The chapter concludes with identification of the research gaps, leading to the research questions of the study and development of a conceptual model to guide the methodology in Chapter 4.

3.2 Temporary organisational forms

The earliest published research on the topic of temporary organisational forms is accredited to that of Miles (1964; 1977), who recognised the difficulties 'permanent' systems face when implementing change within the context of educational innovations. Concerned with rapid societal changes within the United States at the height of the Cold War, Bennis (1965) foretold of an increase in the use of '*temporary systems*' to implement change within wider society. Drawing on the distinction between organisational types proposed by Burns and Stalker (1961), Bennis and Slater (1968) argues that temporary systems are more likely to be organised within an organic structure, where the arrangements are more fluid, rather than the mechanistic model, in which the structure is more ridged. In predicting a growth in project-based

organizational forms, Bennis and Slater (1968) define "*adaptive structures*" as consisting of:

"Adaptive, problem-solving, temporary systems of diverse specialists, linked together by coordinating and task-evaluating executive specialists in an organic flux – this is the organization form that will gradually replace bureaucracy" (Bennis and Slater, 1968: 74)

Palisi (1970) also agues, that as a consequence of their relatively flat hierarchical structure and specific goal focus, *"transitory organizations"* are likely to be less bureaucratic than permanent organisations. Thus, there is less reliance on authoritarianism and hierarchical power in the decision making-process (Palisi, 1970; Miles, 1964). Bennis and Slater (1968) further propose that the structure of a temporary system leads to a reduction in intergroup conflict, as a result of greater collaboration and increased motivation, due to the conditions under which individuals can gain increased satisfaction with the task itself.

Early empirical studies into the application of temporary systems included Keith (1978), who investigated the role clarity of teachers in the development of "*task forces*" created to implement change in metropolitan schools. Whereas, Lanzara (1983) provided an account of the role "*ephemeral organisations*" played in disaster recovery, following an earthquake in Southern Italy. However, it was Goodman and Goodman (1976) who were the first to present the characteristics of the temporary system within the organisational context. In recognising the use of *ad hoc* groups formed to produce theatre projects, they define a temporary system (or organisation) as "*a diverse set of skilled people working together on a complex task over a limited period of time*" (Goodman and Goodman, 1976: 494).

Goodman and Goodman (1976) suggest that temporary systems are created in response to four concurrent problems organisations face in the accomplishment of specific tasks. Firstly, when the task is of such complexity that it requires an integrated effort of organisational members to complete it. The secondly problem concerns unique tasks that do not fit with regular processes and procedures of the permanent organisation. Thirdly, when the tasks are of critical or significant importance to the organisation that a new structure needs to be created to manage them. Fourthly, when the task is defined in terms of specific goals, thus, setting a time limit to the task so the permanent organisation will know when it is complete. The emergence of temporary organisations within contemporary project management literature follows Packendorff's (1995) review of the main deficiencies within the theoretical field of project management. Published within a special edition of the Scandinavian Management Journal on 'Temporary Organisations and Project Management' (Lundin, 1995), Packendorff's (1995) main concern was that within the general systems perspective, projects are explained in terms of inputs and outcomes. Thus, neglecting the strategic rationale for implementation of the project in the first instance. In proposing a change in metaphor for the project from "a temporary endeavour" (PMI, 1987) to a "temporary organisation" (Packendorff, 1995: 321), Packendorff (1995) further argues that when projects are regarded as tools, the various motives of actors for participating in the project are overlooked. Turner and Muller (2003) also propose that viewing the project as a temporary organisation introduces many of the elements of project management, including dealing with the conflict of interest between the various stakeholders, realising the complex roles of the project manager, and the implementation of information, communication and monitoring systems.

3.3 Theory of temporary organisations

The most cited explanation of the temporary organisational form comes from Lundin and Soderholm (1995). Within their paper, also published within the Scandinavian Management Journal, Lundin and Soderholm (1995) outline a theory of temporary organisations through the examination of four basic interrelated concepts of *'time'*, *'task'*, *'team'* and *'transition'*, to demarcate between temporary organisational forms and the environment of the permanent organisation sponsoring the project. Each concept is explored, with regards to the extant literature, in the following sections.

3.3.1 Time

Time is considered to be the distinguishing feature of the temporary organisation (Lundin and Soderholm, 1995). However, within the literature there exist two perspectives regarding the concept of temporariness. One approach views temporariness as the short duration of the project and the requirement for immediate goals (Palisi, 1970). Whereas, the other viewpoint perceives temporariness as the short participation of the actors in the temporary organisation (Lanzara, 1983). Bakker *et al* (2013) observe that the nature of temporary organisations creates an awareness among actors that the

project is limited in scope by a deadline. Consequently, termination of the project is imminent, at which point the team will disband (Baker and Faulkner, 1991; Saunders and Ahuja, 2006; Morley and Silver, 1977).

A number of scholars have investigated the consequences of limited duration on both, the behaviour and the social integration of team members within a temporary organisation. For example, Lindkvist (2005) proposes that a short project duration does not allow adequate time for actors within temporary originations to develop communities of practice, share task-relevant knowledge or become *"tightly-knit"* (Lindkvist, 2005: 1189), in a social structured sense. Morley and Silver (1977) also found that the limited duration of projects within the film industry did not allow sufficient time to develop task familiarity and good working relationships among temporary organisations with short duration, focus attention on the immediate present and are less likely to be concerned about the impact of current team behaviour in future situations.

Other researchers have considered the impact of the limited lifespan of the temporary organisation on team members. Turner and Muller (2003) propose that the transient nature of the temporary organisation, where there exist a defined beginning and end, creates a pressure of urgency among team members to deliver the desired outcomes within the specific timescales. In recognising the stress caused by the pressure of limited time, Morley and Silver (1977) advocate increased collaboration and interpersonal support between colleagues to alleviate the anxiety. Conversely, Bryman et al. (1987b) suggest that, whereas, longer projects allow more time for fostering group relations, the greater urgency of shorter projects engenders an orientation to the task. Consequently, within Bryman et al's (1987b) empirical study of construction projects, it was found that leaders within temporary organisations were focused more on the task to be accomplished, rather than relationships between actors. Similarly, in Saunders and Ahuja's (2006) conceptual paper, they argue that temporary teams will be primarily concerned with the effective accomplishment of current tasks, rather than building longterm relationships. This is because temporary teams are highly goal driven and less concerned with developing long-term efficiencies in team processes. Significantly, Saunders and Ahuja (2006) also propose that, within the time concept, members of temporary teams tend to be rewarded for successful completion of task over social ties and operational efficiency.

3.3.2 Task

The second concept in the theory of temporary organisations concerns the tasks that are executed in order to realise goals. Goodman and Goodman (1976) propose that the case for the creation of a temporary organisation is motivated by the unique tasks that are required to be accomplished. According to Lundin and Soderholm (1995) the temporary organisation is dependent on one, or a very limited number of, defined tasks. It is therefore the task definitions that provide the "*raison d'être*" for its existence (Lundin and Soderholm, 1995: 438). They further determine that the task and devotion to goals are more important to participants of temporary organisations, in comparison to employees within permanent organisations.

The assumption is made that tasks performed by the temporary organisation must have a degree of complexity (Meyerson *et al.*, 1996). Some scholars measure this in terms of the difficulty of the work to be accomplished (Liu, 1999; Hanisch and Wald, 2011). Whereas, Lundin and Soderhulm (1995) measure the complexity of tasks in terms of uniqueness. Within their theory, Lundin and Soderhulm (1995) draw a distinction between temporary organisations that focus on performing tasks that will be repeated in the future, and temporary organisations that are created for one single and specific situation that will not occur again. It is this latter perspective of temporary organisations that is prominent within the literature.

Literature conveys the presumption that tasks performed by temporary organisations are so unique that they do not lend themselves to repetition (Goodman and Goodman, 1976; Gann and Salter, 2000; Prencipe and Tell, 2001). Moreover, it is suggested that the more unique the project task, the fewer actors, within the temporary organisation, will have the limited and immediate knowledge of how to execute them (Lundin and Soderholm, 1995; Linderoth, 2002). Despite benefits of employing temporary organisations to perform unique tasks (Miles, 1964), some authors raise concern regarding the impact of performing unique, non-repetitive activities on organisational learning. According to Ibert (2004) the unique conception of projects limits the possibility of generating lessons learnt. Brady and Davies (2004) also advise that the development of project capabilities. Moreover, Grabher (2004) warns that knowledge accumulated in the course of a project is at risk of being dispersed, as the temporary organisation disbands and members are assigned to different tasks.

However, in contrary to Liu's (1999) proposition, Hanisch & Wald (2011) logically argue that task complexity does not necessarily equate to the complexity of a project. Project complexity has its routes in general systems theory and the notion that an organisation can be likened to that of an open system of independent, interacting component parts (Boulding, 1956; Von Bertalanffy, 1968; Katz and Khan, 1966). At a basic level, complexity refers to the number of components in the system, or the number of combinations that need to be considered in making a decision (Sterman, 2000). Projects are defined as complex systems as they involve a number and of variety of components and interdependencies among them (Davies and Mackenzie, 2014; Shenhar and Dvir, 1996). This includes the size of the system, the arrangement of hierarchical structures, and the complex social systems within a project (Bresnen et al., 2005; Baccarini, 1996). In recognising the numerous interactions within a project, Bakhshi et al. (2016) advise that complex systems display a variety of behaviours, including selforganisation, emergent properties, and non-linear behaviour that is often counterintuitive. Therefore, taking into consideration the number of interactions within a project, it can be argued that task complexity is but one view of complexity within the context of a project (Hanisch and Wald, 2011).

3.3.3 Team

Closely related to the concept of task is the design of the temporary organisation. The unique characteristic of the temporary organisational form is that it typically consists of individuals with diverse skills that are temporarily grouped together to perform specific, complex tasks (Bakker *et al.*, 2013). General project management texts suggest that selection for membership of the team should be based on the necessary skill set required to complete the task (Larson and Gray, 2014), whereas other authors suggest that selection of team members is based on interpersonal skills and competences (Bennis and Slater, 1968; Bryman *et al.*, 1987a). It is also proposed, that in many cases, actors in temporary organisations might have never have worked together in the past, and most probably will not work together on future projects, or at least do not anticipate future collaboration beyond the life span of the temporary organisation (Bennis, 1965; Goodman and Goodman, 1976; Saunders and Ahuja, 2006).

However, Mohammed and Nadkarni (2011) warn that "*temporal diversity*" within temporary organisations is considered to be a "*double-edged sword*" (Mohammed and Nadkarni, 2011: 489). From one perspective, authors have contended that a multidisciplinary team enhances effectiveness by balancing multiple team performance and creating synergy (Pinto, 2013; Larson and Gray, 2014). Conversely, other authors have observed that differentiation can create ambiguity and conflict among team members (McGrath, 1991). Furthermore, according to Lindkvist (2005), when the temporary organisations consists of a mix of individuals with highly specialized competencies, there is difficulty in establishing shared understanding or a common knowledge base.

The main area of concern within the concept of teams is the inadequate time available for development and integration. Within temporary organisations, individual actors lack the time to engage in the usual forms of confidence building activities that contribute to the development and maintenance of trust found in the more traditional, enduring organisational forms (Meyerson *et al.*, 1996). Despite this, a number of scholars argue that teams are, in fact, able to overcome obstacles and develop levels of trust (Robert Jr *et al.*, 2009; Guangquan *et al.*, 2007).

Meyerson *et al.* (1996) explains this through the concept of '*swift trust*', which they define as a '*unique form of collective perception and relating that is capable of managing issues of vulnerability, uncertainty, risk and expectation*' (Meyerson *et al.*, 1996: 167). Saunders and Ahuja (2006) attribute swift trust, in a temporary organisation, to the fact that team members primary concern is the effective accomplishment of tasks. They argue that through the concept of swift trust, team members act as if trust were present from the start of the project, thus enabling them to take action with the uncertainty, ambiguity and vulnerability that arise while working on complex interdependent tasks with individuals they have never interacted with before. Researchers further propose that, in projects where there is not enough time to slowly build trust, team members will assume trustworthiness among fellow actors and begin working as if trust was present from the start (Jarvenpaa *et al.*, 1998; Jarvenpaa and Leidner, 1999; Powell *et al.*, 2004).

However, the defining characteristic of teams within temporary organisations is that members are only assigned temporarily to the project. Lundin and Soderholm (1995) advise that actors within temporary organisations have a functional department or organisation to which they are involved. As such, members are dependent on other organisational contexts besides that of the temporary organisation. In many cases this is not an issue. Knight (1976) suggest that the functional organisations exists as a '*fallback*' for project members, where they can engage with colleagues in their own specialism, and provide a base to return to when the project is complete. However, the difficulty arises when there exists competing teams or competing organisational structures (Lundin and Soderholm, 1995). This is particularly the case in complex organisational environments consisting of several powerful stakeholders, as defined by Artto *et al.* (2008) in their review of project strategy, discussed in Section 2.7.2.

3.3.4 Transition

The fourth theme within Lundin and Soderholm's (1995) framework concerns the concept of 'transition'. In their paper, they propose two distinct meanings for the term. The first definition refers to the actual transformation as a result of the task itself, in terms of distinctive organisational change between 'before' and 'after' the project. Drawing on the emergent nature of strategy, as discussed by Mintzberg and Waters (1985), Lundin and Söderholm (2013) advise that the "end-state" (Lundin and Söderholm, 2013: 590) of the project task is subject to change and adapted over time.

The second meaning of transition refers to the possible (or desirable) perceptions of the transformation or change among the project participants. According to Lundin and Soderholm (1995) this latter meaning of transition is more important to the inner functioning of the project, as it focuses on perceptions of casual relationships, ideas about how to proceed to the final outcome and the conclusion of the project. Focusing on the second meaning of transition, Burstrom and Jacobsson (2012) conducted an empirical study to identify and understand challenges related to *'transition processes'* that occur between temporary and permanent organisations. Drawing on three related projects they found that transition processes are characterised by operational complexities that demand project stakeholders perform multiple translational and transformative activities at the individual, functional, operational and strategic levels of the permanent organisation.

Jacobsson *et al.* (2013) further argue that, from an intra-organisational perspective, it is unlikely that temporary organisations can exist detached from the permanent

organisation. As it is a subset of personnel from the permanent structure that are drawn together to perform a specific task, the temporary organisation is viewed as a form of *'transitory unit'* in the permanent organisation (Jacobsson *et al.*, 2013: 259). However, Artto (2013) perceives the notion that the temporary organisation remains the extended counterpart of a single permanent organisation, as being problematic. Within this context, the temporary organisation emerges as an *'obedient servant'* to the permanent organisation and implements its project strategy to fulfil the mission of the parent (Artto *et al.*, 2008), as discussed in Section 2.7.2. This, in effect, challenges the emergent nature of temporary organisations and the notion that temporary organisations make autonomous decisions to self-manage itself in pursuit of self-originated goals (Artto, 2013). As the permanent organisation continues to determine the actions of the temporary organisation, the boundaries between them become blurred.

3.4 Leadership of temporary organisations

A further theme within Lundin and Soderholm's (1995) team concept is leadership of temporary organisations. Although there has been an extensive body of literature on the topic of leadership within permanent organisations, empirical research investigating leadership styles within the context of temporary organisations remains scarce (Söderlund, 2011; Janowicz-Panjaitan *et al.*, 2009). Turner and Muller's (2005) review also reveal that leadership styles are typically overlooked as a success factor in projects. This is despite the proposition in general management literature that an appropriate leadership style can lead to higher performance. Of the few exceptions, Aga *et al.* (2016) found a positive relationship between project success and leadership styles of project managers in the role of team-building in Ethiopian Non-Government projects. Kangis and Lee-Kelley (2000) also identified a relationship between the perception of difficulty involved in a project and the type of leadership style adopted by project managers in clinical research organisations. Furthermore, Bryman *et al.* (1987b) found that, compared to permanent organisations, leaders of temporary organisations on construction projects, tend to be more participatory in style.

Within the context of the permanent organisation, much of the research on leadership has drawn on the dichotomy between task-focused leadership and person-focused leadership styles, as presented in Fielder's (1967) contingency theory of leadership. Fielder (1967) proposed that the primary concern of task-orientated leaders is to ensure that subordinates perform to a high level. In contrast, person-focused leaders are

primarily concerned with developing good relationships with their subordinates. It is this style of leadership that is, generally, considered to be more effective.

The contemporary view proposes that no single leadership style is suitable for every situation (Kangis and Lee-Kelley, 2000). Effective leadership is widely understood to be contingent on *'the leader, his followers, the organization and the social milieu'* (McGregor, 1966: 76). Drawing on the contingency perspective, Bass an Avolio (1994) present their full range transactional-to-transformational range of leadership styles, summarised in Table 3.1. First introduced by Burns (1978) within his descriptive research on political leaders, transactional leadership refers to the exchange relationship between leader and follower to meet their own self interests. Bass (1990) proposes that this may take the form of contingent rewards in exchange for effort.

"...making, and fulfilling, promises of recognition, pay increases, and advancement for employees who perform well. By contrast employees who do not do good work are penalized" (Bass, 1990: 20).

It may also take the form of *management by exception*, in which the leaders, either, monitors their followers performance and takes corrective action for deviations, or waits until problems arise before taking corrections. In contrast, it may also take the form of *laissez-fair*, in which the leader avoids taking action altogether (Bass, 1999).

TRANSFORMATIONAL LEADER	TRANSACTIONAL LEADER				
Charisma: Provides vision and sense of mission, instils pride, gains respect and trust.	Contingent Reward: Contracts exchange of rewards of effort, promises rewards for good performance, recognises accomplishment.				
Inspiration: Communicates high expectations, uses symbols to focus efforts, expresses important purposes in simple ways.	Management by Exception (active): Watches and searches for deviations from rules and standards, takes corrective action.				
Intellectual Stimulation: Promotes intelligence, rationality, and careful problem solving.	Management by Exception (passive): Intervenes only if standards are not met				
Individual Consideration: Gives personal attention, treats each employee individually, coaches, advises.	Laissez-Faire: Abdicates responsibilities, avoids making decisions.				

 Table 3.1: Characteristics of transformational and transactional leaders. Source:

 Adapted from Bass (1990)

Conversely, transformational leadership refers to the leader moving the follower beyond immediate self-interests, energising them and promoting positive change in individuals and groups (Avolio *et al.*, 1991; Bass, 1999). Transformational leadership may be achieved either, through the charisma, or idealised influence of the leader to inspire

followers. It may also be achieved through inspiration by communication of a desirable vision of the future, or realized through intellectually stimulating followers to become more innovative and creative. Finally, it may be attained through individual consideration to the developmental needs and coaching of followers (Bass, 1990; Bass, 1999).

Within the context of the permanent organisation it is transformational leadership, in particular, with its principles of communication of vision, affordance of autonomy via the extension of trust, and the idealised influence of the leader, that has been found to be most effective in terms of fostering value-adding activity such as innovation, or greater engagement amongst followers (Avolio and Bass, 2002). However, despite studies investigating transformational leadership within the management of projects (Aga *et al.*, 2016), this review finds problems in the transactional-transformational perspective when applied to temporary organisations.

The first problem is the association of temporary organisational actors with the permanent organisation. As observed by Tyssen *et al* (2014), the significant challenge for leaders of projects is that they, actually, have little *de facto* authority. As a consequence of the limited duration of the temporary organisation, responsibility is over a defined period. Once the project is complete, actors within the temporary organisation will either return to their functional department or progress to other projects. Significantly, project members are aware from commencement that at some defined point in time, the temporary organisation will cease to exist (Miles, 1964; Keith, 1978). Studies find that it is the line manager within the permanent organisation that has the most influence over project team members, as leaders of temporary organisations have no influence over the transactional exchange of promotions, career prospects or rewards of employees (Kangis and Lee-Kelley, 2000).

Tyssen *et al* (2014) further argue that the short duration of a project also does not allow adequate time for the development of deeper social relations required for transformational leadership interactions to take place. This is significant, as according to Muller and Turner (2007), leaders emotional intelligence affects their perception of success, which can feed through to the project team. In their study of construction projects, Bryman *et al.* (1987a) suggests that the limited period of time in which to complete a task, results in a tendency for leaders being highly task-orientated.

Moreover, as a result of time pressures, members of temporary originations are more likely to focus their attention on task-related interactions over development of social relations (Kelly and Loving, 2004; Saunders and Ahuja, 2006). Nonetheless, these tensions do not negate the importance and utility of transformational leadership. Rather they support the contention that transactional and transformational leadership are not mutually exclusive and, in fact, represent ideal points in the contingency-based leadership paradigm (Burns, 1978; Bass, 1985; Avolio *et al.*, 1991).

3.5 Types of temporary organisations

In seeking to gain a greater understanding of the intricacy in managing temporary organisations, a number of researchers have focused on the variations in the temporary organisational forms and developed typologies to describe the differences. Modig (2007) presents a continuum that ranges from *'stationary'* to *'temporary'* organisations. The defining deference between each end of the continuum is the anticipated lifespan of the organisation. In conducting an empirical study of four cases across different industries, temporary organisations are classified along the continuum according to the forms of employment, work processes and resource network.



Figure 3.1: Contrasting organizational forms. Source: Modig (2007)

As shown in Figure 3.1, the '*pure*' temporary organisations, on far right of the continuum, represents projects that employ members for only the duration of the project, develop work routines internally and are dependant on member's networks to secure access to critical resources. At the other extreme, employees within '*pure*' stationary organisations have long-term employment contracts, pre-defined work processes and organisational wide resource networks. According to Modig (2007), the mid section of the continuum represents project orientated organisations with project offices or project programmes, such as consultancy firms or construction contractors. These type of organisational structures (Linderoth, 2002).



Figure 3.2: Four types of inter-organizational projects. *Source:* Adapted from Jones and Lichtenstein (2008)

Jones and Lichtenstein (2008) also take a multi-firm perspective in their discourse on inter-organisational projects. In considering the two dimensions of 'temporal embeddedness' and 'social embeddedness' Jones and Lichtenstein (2008) identify four types of inter-organisational projects, shown in Figure 3.2. Within the framework, 'temporal embeddedness' refers to the expected duration of a project and associated mechanisms for coordination of collaborative activities between member organisations (Clark, 1985). Drawing on Granovetter's (1992) definition of embeddedness, Jones and Lichtenstein (2008) propose that 'social embeddedness' comprises of 'rational embeddedness', which is the degree to which actors know and consider one another's needs and goals (Granovetter, 1992) and 'structural embeddedness', which is defined as "the impersonal configuration of linkages between or people units" (Nahapiet and Ghoshal, 1998a: 244). Using these dimensions, Jones and Lichtenstein (2008: 241) classify 'single project organising' as one-time projects that are relatively short in duration, enacted by organisations that have rarely interacted in the past. In contrast, they suggest that 'network alliances' and 'constellations' are enacted by organisations that interact repeatedly. In the case of *constellations*, Jones and Lichtenstein (2008) suggest that this type of project, typically involves a single client.

The concept of structural embeddedness does create difficulties for the notion of swift trust, discussed in Section 3.3.3 (Meyerson *et al.*, 1996; Saunders and Ahuja, 2006). Jones and Lichtenstein (2008) propose that the temporal nature of intra-organisational

projects, actually forces actors to move among collaborative partners. Temporary organisational actors not only have direct relations with other team members, but also are also indirectly linked by third parties who are likely to have further interactions with each other. They therefore assert that trust within this context derives from a shared institutionalised understanding that evolves out of prior interactions between and among organisations. As such, Jones and Lichtenstein (2008) argue that participants in the temporary system who do not know each other are not developing trust swiftly, rather they are operating under established, shared collaborative rules contained within an industry macro-culture that has been facilitated by a process of structural embeddedness.

Jones and Lichtenstein's (2008) framework also creates difficulties in the suggestion that members of temporary organisations do not anticipate future collaboration beyond the life span of the temporary organisation (Bennis, 1965; Goodman and Goodman, 1976; Saunders and Ahuja, 2006). It is noted that in some industries, members have high expectations of interacting with the same people on future projects. This is a common occurrence in the film industry (Bechky, 2006) and the construction industry where temporary organisational forms are structured as '*quasifirms*' by stable, and recurring relations between the main contractor and a small pool of subcontractors (Eccles, 1981). Findings within construction management research actually suggest that when temporary organisational actors have prior working relationships with other actors, project performance is enhanced (Rowlinson, 1988; Shoesmith, 1996). Hence, partnering arrangements have been actively encouraged within the construction industry over, relatively, recent years (Latham, 1994).

Drawing on Soderlund's (2004b) framework for analysis of project research, Artto and Kujala (2008) present a matrix using dimensions of single v multiple projects, and single v multiple to firms, to categorize four distinctive areas of project management research, as shown in Figure 3.3. Within the framework, management of a project addresses the management of a single project, within a single firm. As discussed within Sections 2.7.2 and 3.4.4, the contextual assumption within the literature is that the temporary organisation created to undertake the project is under the governance of a single parent organisation (Shenhar *et al.*, 2007; Turner, 1999; Morris and Jamieson, 2005). As such, the operations of the temporary organization are determined through the hierarchy of the permanent structure.

	One firm	Many firms
One project	1. Management of a project	2. Management of a project network
Many projects	3. Management of a project-based firm	4. Management of a business network

Figure 3.3: Framework of project business: Four distinctive management areas. *Source*: Artto and Kujala (2008)

The second quadrant in Artto and Kujala's (2008) framework is management of a project network. This area addresses the management of a single project across multiple participating firms. Other terms include 'project-based enterprise' (DeFillippi and Arthur, 1998), 'project coalition' (Winch, 2008), 'temporary project coalition' (Pryke, 'multi-organization enterprise' (Grun, 2004) and 2004), *'temporary multi*organisations' (Cherns and Bryant, 1984). In contrast to the management of a business network, in quadrant four, management of the project network takes a "short-term" perspective" (Hellgren and Stjernberg, 1995: 377). On completion of the project, the network disbands. The defining characteristic of this temporary organisational type is that the complex network of firms and other actors involved in the execution of a single project, have their own objectives, interests and expectations, based on the business objectives of the permanent organisation and the objectives of individual actors (Artto and Kujala, 2008). This view is also shared by Jones and Lichtenstein (2008) who caution intra-organisational projects involve multiple organisational actors with disparate goals, overlapping areas of responsibility and differing level of expertise.

The management of a project-based firm is an area which addresses the managerial issues of a firm that conducts a part of its activities as projects (Hobday, 2000). Whitley (2006) propose two dimensions to differentiate between different types of project-based firms. The first dimension, termed the *'singularity'* of goals, contrasts those project-

based firms producing a single or small number of outputs from those conducting a series of related projects producing similar kinds of outcomes. The notion is that the more singular the output, the more likely the temporary organization will have to deal with exceptions to routines and adjust to variations in the work environment (Perrow, 1967). The second dimension considers the extent to which the tasks and roles are predictable and stable over a number of projects. Drawing on these, Whitley (2006: 84) presents a matrix, shown in Figure 3.4, that defines four types of temporary organisations, ranging from *'organisational'*, where there exists multiple, variable outputs and the skills and roles are changeable, as found in consultancy organisations, to *'hollow'*, where single outputs are produced through standardized, separate roles and skills, and the expertise are fairly predictable, as found in construction projects.

	Singularity of goals and outputs				
	Low	High			
Low	Organisational PBFs producing multiple and varied outputs with different and changeable skills and roles	Precarious PBFs producing risky, unusual outputs with varied and changeable skills and roles.			
and stability of work roles High	Craft PBFs producing multiple, incrementally related outputs with distinct stable roles and skills.	Hollow PBFs producing single outputs and coordinating tasks through standardised, separate and stable roles and skills.			

Figure 3.4: Types of Project Based Firms (PBFs). *Source*: (Whitley, 2006)

The fourth quadrant in Artto and Kujala's (2008) framework, shown in Figure 3.3 is the management of a business network, which relates to the activities of several firms that occasionally engage in mutual projects. Firms within the business network create synergies through partnerships, collaborations, alliances and joint ventures (Walker and Johannes, 2003; Davies and Walker, 2008). Artto and Kajala (2008) recognize a rise in the development of business networks, which they attribute to increased outsourcing, de-regulation, new technologies, and the challenge of complexity that has seen distinct capabilities spread across several originations. The salient characteristic of the business network is that, firms may participate in a number of projects, in different roles

(Eloranta *et al.*, 2006). As a consequence, the role of the partnering firm may change from customer, to competitor, to supplier. This results in relations between firms in the business network being contradictory and conflicting (Artto and Kujala, 2008), which has implications for the permanent network and actors within temporary organizations created to implement the varied projects.

3.6 Temporary multi-organisations

Stringer (1967) uses the term "*multi-organisation*" to described the collective allocation of resources to construction projects within the public sector, in which the individuals are, commonly, members of separate organisations. In drawing a comparison with the permanent organisational form, he defines the multi-organisation as:

"...the union of parts of several organisations each being part of a subset of interests of its own organisation. It is defined by the performance of a particular task (which may be a continuing one) through the interaction of between individuals" (Stringer, 1967: 107).

A multi-organisation, therefore, constitutes a "socio-technical system...a system in which social and interpersonal relationships are partly conditioned by the task – and vice versa" (Stringer, 1967: 107). Winch (1988) argues that all organisations have a technical system comprising of work processes serviced by the social systems of personal and group interaction. However, in construction these systems operate, both, in a single organisation and across the boundaries of the firm (Shirazi *et al.*, 1996).

Within the context of the construction industry, Stringer (1967) asserts that the task of the multi-organisation requires collaboration from a number of organisations. Each actor will have other interests apart from the construction of the building, and once the building is complete, their collaboration ends. Consequently, the multi-organisation creates a situation where there exist conflicting and competing objectives being pursued through the diverse values that prevail among members. Stringer (1967) concludes that effective communication is, therefore, essential if the multi-organisation is to perform its tasks adequately.

Cherns and Bryant (1984) extend the definition of the multi-organisation form to a temporary multi-organisation. Drawing on Stringer (1967) and Bryant *et al.* (1978), Cherns and Bryant (1984: 181) characterise the concept of a construction project as an

engagement of parts of several, separate and diverse organisations for the limited and defined purpose of bringing a building into being. This includes the client, consultants and contractor. As determined by its temporal nature, once the building is complete, members of the TMO will eventually disperse, returning back to their own organisation, or to a new project (Cherns and Bryant, 1984; Tyssen *et al.*, 2014).

One of the conclusions from Cherns and Bryant's (1984) pilot study into the client's role in construction management was that very little research had been conducted into multi-organisational dynamics. This view was re-emphasised more than ten years later by Shirazi *et al.* (1996) and Murray *et al.* (1999). Cherns and Bryant (1984: 181-83), therefore, present '*twenty, points*', summarised in Table 3.2, as propositions for future research into the nature of TMOs and guidance for the collection of valid data from client systems, which they acknowledge as being tentative and subject to modification and qualification in light of further evidence. Despite this, Cherns and Bryant (1984) salient points concern the complexity of the client system and the structure of the TMO which, in contrast to permanent organisations, includes multiple organisations and actors.

e points in time of 11. The reasons for early decisions have their orights in the chems, organisation actors and suppliers, culture, procedures and structures.	10 12. Any attempt to understand the clients' roles in the organization and management of construction projects must consider technical and econon factors, and social and political forces acting within the client organisatio	Ire and mode will 13. The decision to build is a large scale innovation decision with consequent the course of the for existing patterns of resource sharing and risk taking in terms of power conflicts and political behaviour within the client organisation	s potential for 14. Conflicts and behaviours within the client organisation can critically affe- take place across formation, development and performance of the TMO	by the capabilities of [15. The performance of the TMO can be fully explained and understood only of the form of the client organisation is seen as a complex system with a past, present an future.	mplexity is managed 16. Effective research into the client's involvement in construction managem nent and performance dependant on gaining access to valid data about the client's background, actions prior to inception and during the course of the project.	nally than is 17. Gaining access to valid data about the client role in the TMO implies discarding notions of random selection. TMO as a focus of study implies intense penetration into a few carefully selected client systems.	1 a single client 18. In seeking access to data the researcher has to develop a professional ent system can be 2 nt system can be relationship with a potential client within whom the need for access has to justified and entry negotiated for detailed, sensitive and confidential information about the client system.	 and difficulties 19. Critical access to valid data is feasible only if 1) a basis exists between th resolved conflict researcher and client system that offers something to the client in exchan, access, and 2) access must provide for a deep and continued penetration i the client system. 	nfluence over the 20. The involvement of the client system and its influence within the TMO is in those taken later. In the initial phases. Thereafter, involvement tends to be remitted to lowe levels of authority within the client system which retreats to reactive moc	
A construction project is an engagement of over different several organisations – consultants, contractors, subcontr with a client system that is itself organizationally comple	The management of a construction project from inception function of a TMO comprising relevant parts of compone	The TMO is a device for handling uncertainty, the structu change over time as the focus of uncertainty shifts during project	The TMO is an important focus of research because of its minimizing risk, allocating risk and enabling learning to component organisations.	The actual performance of the TMO is determined more its component organisations and their coordination than t contract.	The complexity of the client system itself and the way co has the most critical influence of the formation, developn of the TMO.	Most client systems are much more complex organization commonly acknowledged by their advisers.	Many architects, and other advisors insist on dealing with representative within whom all internal politics of the cli contained.	Many of the problems concerning design changes, delays during the construction phase have their origins in the un within the client organisation.	The earliest decisions taken by client system have more i way the TMO is formed and subsequent performance that	
<u>.</u>	5.	3.	4.	5.	6.	7.	8.	9.	10	

Table 3.2: Points for research into the client role in construction management. Source: Adapted from Cherns and Bryant (1984)

3.6.1 Client complexity with the TMO

In considering the role of the client in construction, Cherns and Bryant (1984) argue that the client in not a unitary concept. Instead they maintain that the progress of a construction project involves various groups within the client organisation whose interests may differ and conflict. For Cherns and Bryant (1984: 181) the client represents a '*complex system*' consisting of both, congruent and competing, influential interest groups. These different influences within the client system are themselves part of the TMO and can critically affect its performance (Cherns and Bryant, 1984).

Newcombe (2003) also propose that the concept of the client as a single entity is obsolete and does not reflect the reality of stakeholder configurations for most projects, where the client is only one stakeholder out of many. In perceiving the project organisation as "*coalitions of powerful individuals or individuals or interest groups which pursue multiple goals*" Newcombe (1996: 76) describes the stakeholder configuration on a construction project to include clients, project managers, designers, sub-contractors, suppliers, funding bodies, users and the general community. In his study of stakeholder power within a major redevelopment of a railway yard, Newcombe (2003) concludes that the project should be managed to the benefit of all stakeholders and this is the responsibility of the project manager as their agent. This principle is also supported by Mazur *et al*'s (2014) hypothesis that the mediating mechanisms of internal and external relationships, implemented by the project manager, lead to the success of a project.

Bourne and Walker (2006) also use stakeholder analysis to identify conflicts evolving from power gaps and political groupings among client stakeholders. In bringing together elements of risk management and leadership theory, they suggest that the diverse range of the perceptions of project success can be reconciled through the project manager's attention to the needs and expectations of the multi-stakeholder client body. Bertelsen and Emmitt (2005) attempt to categorise construction project stakeholders by conceptualising the client stakeholder as three distinct client groups, consisting of owners, users and society, each having differing perspectives of the value of the building prior, during and after construction. Whilst, Sidwell (1990) finds that project owners and stakeholders have divergent value systems, which are often incompatible with those of other stakeholders. As a result, the perception of project success becomes diverse.

Thomson (2011) proposes that the pluralistic nature of the multi-stakeholder client body often leads to conflict between client stakeholders. In defining the pluralism of the construction client, Thomson (2011) draws on a definition outside of construction, that "...organisations are more or less heterogeneous assemblages of actors, interests and inclinations where conflicts of interests are inescapable" (Alvesson et al., 2009: 253). In his pilot study into the client complexity within an office refurbishment project for a public sector client, Thomson (2011) explores the consequences of ambiguous stakeholder perceptions of project success. His findings support the proposition that client requirements continue to emerge throughout project progression, and that a single project sponsor cannot reconcile the varied perceptions of success held by a multi-stakeholder client body.

Bana e Costa *et al.* (2001) also highlight the consequences of client complexity in their observation of stakeholder conflicts within a case study for a new railway link to the Port of Lisbon. Within their study, they found horizontal conflicts deriving from the different value systems between the public sector departments involved in the project. Similarly, Olander (2007) observed the competing value systems, differentiation, internal complexity, and changing relative importance of clients constituent stakeholders within their stakeholder analyse of three public sector construction projects. Although the notion of the client as a system (Arnaud, 1998) has been demonstrated through these studies, Points 8 of Cherns and Bryant's (1984) framework, in Table 3.1, propose that the internal project stakeholders, forming the TMO, are impatient of client complexity. Instead, there is preference for dealing with a single client representative, within whom all the politics of among the client body is contained.

3.6.2 TMO procurement strategy in the construction sector

In contrast to the structural configurations within the permanent organisation, Lizarralde *et al.* (2011) proposes that the formal structural configuration of the TMO commonly focuses on the analysis of procurement strategies and the contractual relationships that determine the formal governance mechanisms (Turner and Simister, 2001; Walker and Hampson, 2008; Kelly *et al.*, 2002). Within the context of the construction industry, the focus of procurement has seen a shift from low cost to best value for the client (Oyegoke *et al.*, 2009). Client value is emphasised through a number of UK government reports (Latham, 1994) and is inherent in the Office of Government Commerce's (2003) definition of procurement strategy.

"...procurement strategy identifies the best way of achieving the objectives of the project and value for money, taking account of the risks and constraints, leading to decisions about the funding mechanism and asset ownership for the project. The aim of a procurement strategy is to achieve the optimum balance of risk, control and funding for a particular project" (OGC, 2003: 2)

The procurement strategy is delivered through an appropriate procurement route. This includes the contract strategy that is assumed best to meet the needs of the client (OGC, 2003). Although, the Royal Institute of Chartered Surveyors (RICS) offers *"general principles"* for the selection of an appropriate procurement route (RICS, 2013: 11), it is generally accepted that no single approach is suitable for all contingencies (Rowlinson, 1999; Taylor *et al.*, 1999). As such, clients will select a procurement route that has either worked for them in the past or which they know to be suitable, taking into account their prioritised objectives and their attitude to risk (Morledge, 2002).

There are four predominant procurement routes within the UK construction industry: traditional contracting, design and build, construction management and management contracting (Oyegoke *et al.*, 2009; RICS, 2013). Each type determines the complex, structural configuration of the TMO.



Figure 3.5: TMO configuration in traditional, general contracting. *Source*: Adapted from Murdoch and Hughes (1997)

The traditional approach to procurement, also referred to as general contracting (GC Works), is characterised by the separation of design from the construction process (Murdoch and Hughes, 1997). As illustrated in Figure 3.5, the configuration of the TMO consists of the architect and other design professionals that form the design team
design team and the main contractor. Typically, there is a lack of integration across this boundary (Rowlinson, 1999). It is also common for the contractor to sub-contract the majority of site works through a network of builders and suppliers (Eccles, 1981).

RICS (2013: 12) suggest that this route is suitable for inexperienced or occasional construction clients due to its *"least risk"* approach and the level of certainty about design, cost and duration inherent in the strategy, if properly implemented. Under normal conditions of the traditional procurement route, design is completed before competitive tenders are invited and before the main construction contract is let. Consequently, the client will usually select and appoint the design team prior and before appointment of the main contractor, thus maintaining ownership of the design. Assuming no changes are introduced, cost certainty and project duration can be determined prior to the commencement of the construction phase.



Figure 3.6: The vicious circle of construction procurement. Source: (Curtis et al., 1991)

Murdoch and Hughes (1997) warn that problems with the traditional procurement route can occur when work does not concur with the design, or when the design information is incomplete. Indeed, the biggest drawback within the tradional procurement strategy is in the competitive bidding process that forces contractors to submit low, often unrealic, bids and look to alternative means to recoup lost profit, mainly through claims (Ioannou and Leu, 1993). Viewed as the *'vicious circle'* of the construction procurement process (Curtis *et al.*, 1991; Rowlinson, 1999), clients will be seeking to maximize value for money through the contractual agreement, whilst contractors and professionals will be seeking to maximize profit, as illustrated in Figure 3.6.



Figure 3.7: TMO configuration in Design and Build. *Source*: Adapted from Murdoch and Hughes (1997)

In contrast to the traditional procurement strategy, the design and build approach, shown in Figure 3.7, is characterised by the single point of responsibility offered to the client by the contractor and the opportunity of overlapping the design and construction phases (Rowlinson, 1999). RICS (2013: 10) advises that in the deign and build procurement route, the contractor assumes the risk and responsibility for designing and building the project. As with traditional approach, there are variations within design and build, as described by Akintoye (1994). But typically, consultants on the project are employed directly by the design and build contractor. The most significant feature of the design and build arrangement is, therefore, the lack of an independent certification role in the contract, instead the contractor is responsible for all elements of the project (Murdoch and Hughes, 1997).



Figure 3.8: TMO configuration in construction management. *Source*: Adapted from Murdoch and Hughes (1997)

Rowlinson (1999) clarifies the difference between construction management and management contracting. The essential distinction is between the organisational forms

is the relationship of the client to the contractors. Under a construction management route, illustrated in Figure 3.8, the client places a direct contract with each of the specialist design professionals and trade contractors. This ensures that the risk and responsibility is not allocated to a single contractor (RICS, 2013). In order to coordinate and facilitate collaboration between TMO members, the client employs the services of a construction manager, who acts as the role of consultant to advise the client (Murdoch and Hughes, 1997).



Figure 3.9: TMO configuration in management contracting. *Source*: Adapted from Murdoch and Hughes (1997)

In contrast, management contracting, illustrated in Figure 3.9, allows for the involvement of a management contractor, or consultant, to manage the entire building process (Oyegoke, 2001). In this route the client engages a management contractor to participate in the project at the early stage, in order to contribute construction expertise to the design and manage the construction operations on site (Murdoch and Hughes, 1997). Rowlinson (1999) explains that, in contrast to construction management, trades contractors have a direct contract with the management contractor, who also acts as client's representative and has authority to make decisions affecting the designers and contractors.

Although each procurement route is an idealisation of a complex temporary organisation (Rowlinson, 1999), Lizarralde *et al.* (2011) argue that procurement strategies oversimplify the relationship between actors in the TMO, through four key points. First, it is suggested that informal communication links between organisations and individuals does not necessarily correspond to the formal contractual links in the procurement strategy (Gluch and Raisanen, 2009). Second, TMOs are not static. They

evolve to adapt to the dynamic conditions of the project and its environment. This concurs with Point 1 of Cherns and Bryant's (1984) framework, in that engagement of organisations within the TMO are over different points in time. For this reason the structural configuration constantly changes as different actors join and leave the TMO (Murdoch and Hughes, 1997). Third, Lizarralde *et al.* (2011) find that legally bound contracts consider and represent the client as a single heterogeneous entity. This assumption underestimates the complexity of the project client, which includes various, diverse organisations with a complex internal structure, with heterogeneous interests, methods of work and expectations of the project, as advised in Lizarraldle *et al*'s (2011) fourth point.

3.7 Project governance

As discussed in Section 3.6.2, the structure of a TMO involves a complex arrangement of multiple clients and stakeholders. Indeed, general procurement strategies anticipate that the client establishes contracts with one or more organisations (Lizarralde *et al.*, 2011). Successful delivery of the project, therefore, involves several independent economic transactions, under conditions of uncertainty and conflicting objectives (Ahola *et al.*, 2014). This complexity is addressed through the governance of a project, which according to Ahola *et al.* (2014) originates from two streams of literature. Firstly, transaction action cost economics that represents a contractual approach to economic transactions in which efficiency of the transaction is the main objective (Williamson, 1975; 1985). Secondly, the corporate governance literature that focuses on agency theory and the exchange relationship between the agent and the principle (Eisenhardt, 1989a; Jensen and Meckling, 1976).

In conducting citation analysis of studies into project governance, Ahola *et al* (2014) identify two distinct perspectives within the project management literature. The first stream views project governance as a process external to any specific project. According to this perspective, individual projects are subjected to governance by their owner organisation (APM, 2004; PMI, 2013). Drawing on the definition for corporate governance presented by the Organisation for Economic Co-operation and Development (OECD), which proposes that *"corporate governance involves a set of relationships between a company's management, its board, its shareholders and other stakeholders"* (OECD, 2004: 11), the Association of Project Management (APM) present the following definition for project governance:

"The governance of project management concerns those areas of corporate governance that are specifically related to project activities. Effective governance of project management ensures that an organisation's project portfolio is aligned to the organisation's objectives, is delivered efficiently and is sustainable. Governance of project management also supports the means by which the board, and other major project stakeholders, are provided with timely, relevant and reliable information" (APM, 2004: 4)

The focus of governance within this perspective is on maintaining the alignment of various project objectives with the strategic objectives of a single organisation (Müller, 2009). This includes alignment of governance at the different project related levels of project, programme and portfolio management (Bekker and Steyn, 2009), as illustrated in Figure 2.5.

The second literature stream evolving from Ahola *et al's* (2014) analysis, treats project governance as a process internal to any specific project. According to this view, the primary purpose of project governance is to ensure that the project meets the goals and expectations subjected to it by the varied stakeholders within the temporary organisation (Turner, 2006). This perspective aligns with Artto *et al's* (2008) proposition that the project is positioned in a complex organisational environment of several powerful stakeholders, rather one strong parent organisation, as discussed in Section 2.7.2. Turner and Simister (2001) argue that the purpose of project governance, therefore, is to create a cooperative project organisation, in which all participants are motivated to achieve common objectives and align individual goals. In their investigation of Public Private Partnerships (PPP) in Indonesia, Abednego and Ogunlana (2006) emphasised the importance of risk allocation in the governance mechanism. Whereas, Ruuska *et al.* (2011: 650) present the following key elements of governance to be considered among projects involving multiple stakeholder organisations:

- Contracts between involved actors
- How procurement is organised and carried out
- How network of suppliers are managed by project actors
- How risks are managed and shared by project actors
- How work is monitored and coordinated during the project lifecycle
- How the project actors collaborate and develop practices, and
- How communication between project actors is organised.

Bekker (2014) attempts to categorise project governance schools of thought further, by drawing a distinction between multi-firm project governance and large capital project governance. He proposes that multi-firm governance concentrates on the contractual relationships among firms participating in a single, or multiple projects. The focus of the governance process is, therefore, on employment relations and supply-chain dynamics (Winch, 2001). Other authors have investigated governance mechanism for knowledge transfer between multiple organisations (Pemsel and Müller, 2012). In the case of the Dutch construction industry, Bosch-Sijtsema and Postma (2010) find that trust, as well as, formal contracts serves as the governance mechanism among participating firms in a single project. Furthermore, in examining relationships in the '*temporary project coalition*' through the application of social network analysis, Pryke (2004) proposes that governance involves a multi-layer of independent networks, which includes contractual relationships, performance incentives and information exchanges among participating firms.

As a result of the intrinsic complexity of large capital projects, Bekker (2014) proposes that the emphasis of governance is on creating an environment, in which the activities can be efficiently managed, shielded from the external environmental, political and strategic influences. This sees a shift in focus from contractual relationships and control, as discussed in Section 3.7.2, to the development of an appropriate governance framework in which project decisions can be made. Whereas the terms 'governance' and 'governing' both refer to the power relations between stakeholders and the conditions that frame them (Lizarralde *et al.*, 2013), Sanderson (2012) highlights the distinction, by clarifying that governance considers the "form of organisation designed *ex ante*" whilst, governing considers the "mirco-processes of organizing emerging ex post" (Sanderson, 2012: 433). However, following the lessons learned from a study of project governance within the construction of two major nuclear power plants, Ruuska *et al.* (2011) propose that governance mechanisms, actually, evolve during the construction phase of the project through a process of self-regulation.

Bekker (2014) makes the argument that the performance of large projects is affected by complex institutional environments, and by the underlying business network of organisations (Artto and Kujala, 2008). Ruuska *et al* (2011) also emphasises stakeholder complexity by arguing that large projects face challenges of governing a project's internal complex supply chain of multiple firms, and of simultaneously

governing the network of external stakeholders. Despite the focus of research being on large capital projects (Bekker and Steyn, 2009; Ruuska *et al.*, 2011; Abednego and Ogunlana, 2006), the complex integration of various internal and external stakeholders and the complexity of the composition of different contracting companies from different organisations working across organisational boundaries, also applies to TMOs, as defined by Cherns and Bryant (1984). As observed by Bekker (2014), the difference between approaches to governance, appear to be driven more by stakeholder complexity, rather than complexity of the project.

3.8 Research gaps and research questions

The review of the literature, conducted in Chapters 2 and 3, finds that the majority of research enquiring into the strategic alignment of projects, and project success, has predominantly focused on temporary organisations within the intra-organisational context. As demonstrated in Chapter 2, current models of strategic alignment focus on the realisation of corporate level strategic objectives through the implementation of projects contained within the boundaries of a single organisation. Within this arrangement the dominant discourse makes the assumption that the temporary organisation created to implement the project will arrange its operations and perform activities under the directives of a principal, permanent organisation.

Also, as discussed in Section 3.3, much of the discourse contributing to the theory of temporary organisations makes the assumption that many of the propositions underpinning Lundin and Soderholm's (1995) basic concepts, apply to all temporary organisational types. However, as discussed in Sections 3.5 and 3.6, there are fundamental differences between temporary intra and inter-organisations that have significant implications in the pursuit of strategic alignment and project success. The literature review finds that, in comparison to intra-organisational studies, research considering the inter-organisational context of temporary organisations is currently limited, particularly in the field of construction where the dominant body of literature focuses on formal procurement strategies and the contractual relationships between the client and organisations forming the TMO.

The literature suggests that the inter-organisational nature of the TMO results in the creation of an environment where multiple organisations seek to pursue disparate strategic objectives. Whereas, models have been developed to illustrate alignment

within the context of a single organisation (Archibald, 1988; Kerzner, 2004; Morris and Jamieson, 2005), there are no accepted models explaining how strategies are aligned within the context of a TMO. The review also finds a lack of empirical studies investigating strategic alignment within a project environment. As stated in Chapter 1, the first objective of this study is to explore how multiple organisations within a TMO align varied strategic objectives through a single project. In order to investigate the nature of alignment, the varied strategies being pursued by multiple organisations participating in a TMO need to be established. Hence, the first research question of this thesis is as follows:

Research Question 1: What are the varied organisational strategies being pursued within a temporary multi-organisation?

A second research gap concerns the dynamic environment in which the TMO is operating. The literature suggests that participating organisations will pursue disparate strategic objectives through a TMO (Jones and Lichtenstein, 2008) resulting in negative behaviours between parties (Curtis *et al.*, 1991). Despite this, project management literature does not appear to consider emergent strategies throughout the project. Instead, procurement strategies are formed at the start of the project and make the assumption that all external factors and contingencies have been identified and considered prior to the formation of the TMO. Yet, as discussed in section 3.7, governance processes are implemented in order to ensure the clients strategic objectives are maintained through the process. Therefore, in considering client complexity, the complexity of the TMO environment, and external influences, the following research questions are asked:

Research Question 2: What mechanisms are implemented to maintain the alignment of a construction project with the client strategic objectives? and

Research Question 3: *How effective are these mechanisms in maintaining the strategic alignment of a construction project?*

The third research gap considers the behaviour of TMO actors in pursuit of the, possible, disparate strategic objectives. The literature on project strategy (Artto *et al.*, 2008) and Lundin and Solderhom's (1995) theory of temporary organisations contribute to the understanding of the difficulties in pursuing strategic objectives through projects, but there is currently a lack empirical research to sufficiently explain the strategic

behaviours that evolve when temporary organisation consists of members from multiple organisations. Furthermore, the literature offers no guidance as to how project strategy is developed within this context (Artto *et al.*, 2008). Whereas, the client seeks to maintain alignment of strategic objectives through formal contracts, there is a gap in the literature to explain how representative actors of TMO member organisations seek to balance the pursuit of strategic objectives throughout a construction project. The fourth research question of this study therefore asks:

Research Question 4: *How do varied TMO actors pursue strategic objectives within a single construction project?*

Finally, the literature review finds a gap in the research into project success criteria. This is despite the plethora of discourse on the concept of project success, as discussed in Section 2.6. Literature proposes that the varied stakeholders involved in a project will have different opinions on what constitutes success, and will therefore, make assessment on varied success criteria (Davis, 2014). However, if the complexity of measuring success within the boundaries of a single organisation is accepted (Shenhar *et al.*, 2001), this complexity must be significantly magnified when considering the multiple organisational boundaries that overlap in a TMO. This is further compounded when considering the varied levels of each organisation's strategic hierarchy (Hofer and Schendel, 1978), and client complexity within a construction project (Cherns and Bryant, 1984). The review of the literature also finds a lack of empirical studies identifying the success criteria of varied stakeholders with the TMO, or an explanation as to how the varied success criteria links to the multiple strategic objectives. To address these gaps, the fifth research question of this study asks:

Research Question 5: *How do the varied actors engaged within a TMO measure the success of a construction project?*

3.9 Conceptual Model

In order to answer the research questions presented in Section 3.8, a conceptual model is developed from the initial themes evolving from the literature review. The conceptual model guides the development of data collection instruments, designed to achieve the aims and objectives of the study, stated in Section 1.5. A detailed descripting of the research design is addressed in the following chapter.

The conceptual model illustrated in Figure 3.10 integrates the theoretical concepts of strategic alignment, discussed in Chapter 2, with concepts of temporary multi-organisations explored in Chapter 3. It was established from the strategic hierarchy of objectives, discussed in Section 2.4, that strategy occurs at different levels within an organisation. As such, the client should not be considered as a unitary concept, instead the client comprises of various stakeholder groups. It is this client complexity, discussed in Section 3.7.1 that determines the varied project success criteria of the stakeholder groups.



Figure 3.10: Conceptual model of strategic alignment of TMOs

The conceptual model proposes that the strategy of the TMO, developed to pursue strategic objectives and realise client project success criteria, will be influenced by a number of factors. These include the formal procurement strategy that determines the

structural configuration of the TMO, as discussed in Section 3.7.2, and the informal governance arrangements, implemented to maintain alignment of the client's strategic objectives, that emerge throughout the life span of the TMO, as discussed in Section 3.8. The strategy of the TMO will also be a subject to internal and external environmental influences, which may impact on the decisions and actions the TMO takes in pursuit of strategic objectives. Within the literature review, internal influences include the leadership of the TMO, as discussed in Section 3.5, and the client structure, as discussed in Section 3.7.1. Finally, drawing on Lundin and Soderholm's (1995) framework, as discussed in Section 3.5, the model proposes that the behaviour of TMO actors, and hence the TMO strategy, will be influenced by the time, task, team and transition factors, specific to the construction project. On completion, the success of the construction project will be assessed against the client success criteria.

The aim of this research is to investigate how varied organisations in a temporary multiorganisation seek to align multiple strategic objectives through a single construction project, and realise project success. As stated in Section 1.5, this will be explained through a theoretical a model of strategic alignment within the context of a TMO. The conceptual model in Figure 3.10 forms the foundation of the final theoretical model, which is developed from the empirical investigation of this thesis, and presented in Chapter 6. The conceptual model, developed from the literature, does not take into consideration the individual objectives of each TMO member organisation, or how varied actors within the TMO measure success. These constructs are investigated within the study and included in the final model, presented in Figure 6.5.

3.10 Conclusion to chapter

This chapter concludes the theoretical underpinning of the study. Chapter 2 examined the differing perspectives of strategic alignment, project success and project strategy, whereas this chapter introduced the concept of temporary organisations, characterising the difference and complexity of temporary multi-organisations. Through the integration of these theoretical areas, limitations in the extant literature have been identified. The research questions of the study and conceptual model presented in Figure 3.10 have been developed in response to the limitations. Drawing on the theoretical foundations and the themes evolving from the literature review, the following chapter defines the method used to address the research questions and achieve the research aims and objectives of the thesis.

CHAPTER 4 RESEARCH METHODOLOGY

4.1 Introduction to chapter

The previous chapters provided a review of the underpinning theory informing the research in this thesis. This chapter describes the rationale for the research method pursued within this study. The chapter begins with a discussion of the philosophical assumptions taken in the thesis, which in turn, influenced the methodological choice. The following section of the chapter provides justification for the case study research approach and presents a detailed description of the research design guiding the data collection and analysis. The methods of data collection utilised within the research are then discussed, before explaining the instruments used to analyse the collected data from which conclusions are drawn from the findings. Finally, the principle of ethical research relevant to the study are considered and discussed.

4.2 **Research philosophy**

Whereas, philosophy is concerned to know what kind of things exist in the world and our warrant to know them, social research is concerned with their knowable properties (Williams and May, 1996). Research philosophy, therefore, considers the nature and development of knowledge and the search for truth (Menacere, 2016; Saunders *et al.*, 2009). However, as proposed by Williams and May (1996), the nature of *'truth'* is only partially explained through scientific investigation, in social research, the nature of truth also involves abstract reasoning of the researcher.

Central to Burrell and Morgan's (1979) thesis is the notion that "all theories of organisation are based upon a philosophy of science and theory of society" (Burrell and Morgan, 1979: 1). Burrell and Morgan (1979) present an argument that all social scientists approach their subject via explicit or implicit assumptions about the nature of the social world and the way it may be investigated. The first assumption concerns the ontological nature of the phenomena under investigation, and raises the question of whether reality is constructed externally from individual consciousness, or the product of individual consciousness. The second assumption is of an epistemological nature and is concerned with how knowledge can be obtained and how can the truth be discovered. The third set of assumptions concern human nature and, in particular, the relationship between human beings and their environment. This assumption seeks to establish if

humans are responding in a deterministic fashion to situations encountered in their world, or if human beings are more creative and act on 'free will' and voluntarism. Burrell and Morgan (1979) emphasise that all social science must be predicted upon assumptions of human nature, as it is humans who are essentially the subject and object of enquiry.

Burrell and Morgan (1979) propose that the three sets of assumptions have direct implications on the methodological nature of enquiry. Each assumption has important consequences for the way in which social scientist attempts to investigate and obtain knowledge about the social world. As such, different ontologies, epistemologies and assumptions about human nature are likely to influence social scientists towards different methodologies. Guba and Lincoln (1998) further propose that the basic beliefs of any paradigm can be summarized by the responses to the ontological, epistemological and methodological assumptions.

The subjective approach to social science		Assumption		The objectivist approach to social science
Nominalism	←	Ontology		Realism
Anti-positivism	←	Epistemology	\longrightarrow	Positivism
Voluntarism	←───	Human Nature		Determinism
Ideographic	←	Methodology		Nomothetic

Figure 4.1: The subjective – objective dimensions: *Source:* Burrell and Morgan (1979)

The model in 4.1 summarizes the ontological, epistemological and assumptions about human nature, that influence the choice of methodology. As shown, each assumption sits within a continuum between the two polarized philosophical dimensions of 'objectivism' and 'subjectivism'. In the model, the objectivist dimension views the social world from a hard, external reality, where the scientific endeavor focuses on the analysis of relationships and regularities between the various elements, which it comprises. The concern within this dimension is the definition of these elements and the discovery of ways in which the relationships can be expressed. In contrast, the subjective dimension stresses the importance of the experience of individuals in the creation of the social world. As such, the search for understanding focuses upon different issues and approaches them in different ways. Consequently, the principal concern within the subjectivist perspective is the comprehension in the way in which the individual creates, modifies and interprets their social world.

	The subjective approach to social science					The objectivist approach to social science
Core ontological assumptions	Reality as a projection of human imagination	Reality as a social construction	Reality as a realm of symbolic discourse	Reality as a contextual field of information	Reality as a concrete process	Reality as a concrete structure
Basic Epistemological Stance	To obtain phenomenolo gical insight, revelation	To understand how social reality is created	To understand patterns of symbolic discourse	To map contexts	To study systems, process, change	To construct a positivist science
Assumptions About Human Nature	Man as pure spirit, conscious being	Man as a social constructor the symbol creator	Man as an action the symbol user	Man as an information processor	Man as an adaptor	Man as a responder
	Exploration of pure subjectivity	Hermeneutics	Symbolic analysis	Contextual analysis of Gestalten	Historical analysis	Lab experiments surveys

Table 4.1: Network of Basic Assumptions Characterizing The Subjective - Objective	ve
Debate within Social Science. Source: Morgan and Smircich (1980)	

Although, objectivism and subjectivism are illustrated as extremes within Burrell and Morgan's (1979), in reality there are a number of positions that exist somewhere between these limits (Holden and Lynch, 2004). Based on the core assumptions of the nature of science, Morgan and Smircich (1980) identify several taxonomies that lie between the six extreme philosophical perspectives debated by Burrell and Morgan (1979). Table 4.1 shows how a researcher's core ontological assumptions influences the epistemological position and position taken on human nature. Each of these assumptions is further explored in shaping the methodological choice of this thesis.

4.2.1 Ontological assumption

The first level of debate towards a research methodology is the ontological assumption. This is concerned with the nature of reality. As illustrated in Figure 4.1, Burrell and Morgan (1979) present a dichotomy between the two extremes of nominalism and realism. The nominalist position evolves around the assumption that the social world, external to individual cognition, consists of nothing more than names, concepts and labels, which are used to structure reality. In contrast, the realist assumption postulates that the social world, external to individual cognition, consists of pre-existing hard, tangible and relatively immutable structures.

Bryman and Bell (2007) also draw a distinction between the two dimensions of 'objectivism' and 'constructivism'. Within this dichotomy 'objectivism' is defined as the ontological position that asserts "social phenomena and their meanings have an existence that is independent of social actors" (Bryman and Bell, 2007: 22). At the other extreme of their continuum, 'constructivism' is defined as the ontological position "...which asserts social phenomena and meaning are continually being accomplished by social actors (Bryman and Bell, 2007: 23).

In considering the ontological debate, this thesis holds with the subjective assumption that external realities and the social world exist as subjective constructions of the mind, and drawing on Bryman and Bell's (2007: 23) definition, are not only produced through social interaction, but that they are in a constant state of revision.

4.2.2 Epistemological assumption

The second level of debate towards a research methodology is the epistemological assumption. This is concerned with the science of the nature of knowledge and what is regarded as acceptable knowledge in the field of study. Whereas, Burrell and Morgan's (1979) continuum is polarized between '*positivism*' and '*anti-positivism*', other researchers have focused their discussion on '*interpretivism*' to denote an alternative to the positive position (Saunders *et al.*, 2009; Bryman and Bell, 2007; Collis and Hussey, 2009).

At the epistemological level, the positivist paradigm seeks to explain the social world by searching for regularities and causal relationships between constituent elements. As explained by Burrell and Morgan (1979), positive epistemology is, in essence, based upon the traditional approaches that dominate the natural sciences. At the epistemological level, the positivist position takes the view that knowledge is external, objective, observable, measurable and independent of social actors (Saunders *et al.*, 2009; Collis and Hussey, 2009). In defining the philosophical perspective further

Remenyi et al. (1998) states that:

"being a positivist, or perhaps a logical positivist, implies that the researcher is working with an observable social reality and that the end product of such research can be the derivation of laws or law-like generalizations similar to those produced by the physical and natural sciences" (Remenyi et al., 1998: 32). Within positivism the researcher is an objective analyst of a tangible social reality. As Remenyi *et al.* (1998) further explains "*underlying positivism is the assumption that the researcher is independent of and neither affects nor is affected by the subject of the research*" (Remenyi *et al.*, 1998: 33). The positive position, therefore, advocates the application of the methods of natural sciences to the study of social reality, which includes the formulation of a hypothesis that can be tested and, thereby, allow the explanation of laws of science to be assessed (Bryman and Bell, 2007).

An alterative inquiry is presented by Guba and Lincoln (1998) who analyze four competing paradigms of 'positivism', 'postpositivism', 'critical theory' and 'constructionism'. Within their framework, Guba and Lincoln (1998) suggest that the postpostivist paradigm has evolved as a response to the limitations and criticism of positivism. According to Guba and Lincoln (1998), postpositivsm takes the position of critical realism, where "reality is assumed to exist but to be only imperfectly apprehendable because of basically human intellectual mechanisms and the fundamentally untraceable nature of phenomena" (Guba and Lincoln, 1998: 205). In other words, "perception is not reality as critical realist might aver: instead a perception for realists is a window on to reality through which a picture of reality can be triangulated with other perceptions" (Perry et al., 1999: 18).

In contrast to the positive position, anti-positivist rejects the notion that science can generate objective knowledge (Burrell and Morgan, 1979). The anti-positivism position advocates that social reality and the world are relative, and can only be understood from the viewpoint of the individuals directly involved within the activities being studied. The anti-positivist researcher, therefore, rejects the standpoint of the observer. Instead, arguing that inferences about a social reality cannot be made without an internal perspective of the social reality (Burrell and Morgan, 1979).

Within the traditions of anti-posivitism, interpretivism views the investigator and the object of investigation as being intrinsically linked (Guba and Lincoln, 1998). Consequently there is a need to minimise the distance between the investigator and the investigated object (Collis and Hussey, 2009). In drawing a distinction with functionalism, that advocates a research process in which the researcher is independent of the subject matter, Burrell and Morgan (1979) advise that the interpretive paradigm advocates a research process where the social world is best understood from within. It

seeks explanation within the reality of individual consciousness and subjectivity, from perspective of the research participant. In its approach to social science interpretivsm tends to be nominalist, anti-positivist, voluntarist and ideographic. The concern, therefore, within the interpretive paradigm is to understand the fundamental nature of the social world at the level of subjective experience.

Symon and Cassell (2012) advise that interpretivism covers a variety of philosophical approaches. These include the German intellectual tradition of hermeneutics (Taylor, 1971) and the Verstehen tradition in German sociology (Weber, 1947). Prasad (2005) further describes how interpretive traditions emerge from the position that takes human interpretation as the starting point for developing knowledge about the social world. Consequently, within the intrepretivist epistemology there is a necessity for the researcher to understand differences between humans in their role as social actors (Saunders *et al.*, 2009).

As informed by Crotty (1998), intrepretivist traditions include phenomenology and symbolic interaction. Within these perspectives, phenomenology is defined as:

"a theoretical point of view that advocates the study of direct experience taken at face value: and one which sees behavior as determined by the phenomena of experience, rather than by external, objective and physically described reality" (Cohen et al., 2007: 22).

In contrast to the positivist researcher, the phenomenologist does not consider the world to consist of an objective reality, but instead considers subjective consciousness as prevailing. As such, each situation is considered to be unique and its meaning being a function of the circumstances and the individuals involved (Remenyi *et al.*, 1998). Therefore, phenomenologist researchers do not consider themselves independent of what is being researched, but is an intrinsic part of it.

Symbolic interactionism makes the argument that interaction takes place in such a way that the individual is continually interpreting the symbolic meaning of their environment and acts on the basis of the implicated meaning (Bryman and Bell, 2007). Blumer (1962) refers to this position as *"the peculiar and distinctive character of interactions as it takes place between human beings"* (Blumer, 1962: 79). According to Denzin (1978) symbolic interaction rests on three basic assumptions. Firstly, social reality as it is sensed, known and understood is a social production, and interacting individuals produce and define their own definitions of situations. Second, humans are

assumed to be capable of engaging and 'minded' in self-reflective behavior, in that, they are capable of shaping and guiding their own behavior and that of others. Third, in the course of shaping and guiding their own behavioral standpoint and fitting that standpoint to the behavior of others, humans interact with one another. Interaction is, therefore, symbolic because it involves the manipulation of symbols, words, meanings, and languages.

In considering the ontological and epistemological arguments presented in this chapter, this study holds with the interpretivist paradigm. The epistemological stance of the research sits between obtaining phenomenological insight and seeking to understand patterns of symbolic discourse, as illustrated within Morgan and Smircich's (1980) guiding framework, illustrated in Table 4.1. The research objectives discussed in Chapter 1 and the research questions developed in Chapters 2 and 3 seek to gain a greater understanding of the strategic interactions within temporary multi-organizations. Therefore, in following the ontological position that reality is socially constructed, the epistemological position of this study seeks to understand how social reality is created.

4.2.3 Assumptions about human nature

Within Burrell and Morgan's (1979) framework, the third set of assumptions that contribute towards consideration of a methodology, is the relationship between human beings and their environment. Burrell and Morgan's (1979) model presents the opposing dimensions of human nature 'determinism' and 'volunterism'. as Determinism portrays humans as responding mechanically and deterministically to their environment. Within the objective extreme of Morgan and Smircich's (1980) framework, human beings are perceived as products of the external forces in the environment to which they are exposed. In contrast, volunteerism assumes humans are initiators of their own actions with free will and creativity, producing their own environments (Cohen et al., 2007). At the subjective extreme of Morgan and Smircich's (1980) framework, it is assumed that humans are intentional beings, directing their psychic energy and experience in ways that constitute the world in a meaningful, intentional form. In considering that the subject of this research is, essentially, the strategic behaviour of human beings within temporary organisational environment, this study holds with the volunteerism position.

4.2.4 Methodological assumptions

The three sets of assumptions have direct implications on the methodological concerns of the researcher since the contrasting ontologies, epistemologies and assumptions about human nature, will in turn, require different research methods (Cohen *et al.*, 2007). Within Burrell and Morgan's (1979) model, the nomothetic dimension emphasizes the importance of basing research upon a systematic protocol and techniques employed in the natural sciences. Here, scientific investigation is focused upon the process of testing hypothesis in accordance with the principles of scientific rigor. Hence, methodology is predominantly concerned with the construction of scientific tests and the use of quantitative techniques for data analysis.

From the subjective perspective, the idiographic dimension is based on the view that the social world can only be understood by obtaining first hand knowledge of the subject under investigation. The principle concern is with the understanding of the ways in which the individual creates, modifies and interprets the world in which they exist. The idiographic approach, therefore, emphasizes the detailed analysis of subjective accounts and allows research subjects to unfold its nature and characteristics during the process of the investigation (Burrell and Morgan, 1979). As such, the idiographic approach embraces qualitative methods of investigation, but can also include quantitative. The norms of the natural science model and positivism are, therefore, rejected in preference for a view of social reality as a constantly shifting emergent property of individuals' creation (Bryman and Bell, 2007).

Following the interpretive paradigm that this thesis will be following, an inductive approach to the research is assumed. Although, as advised by Saunders *et al* (2009), it is accepted that the labelling of interpretive paradigms being associated with inductive approaches, and positive paradigms being associated with deductive approaches is not a fixed rule. Despite this, within an inductive approach to research, theory is developed from the observation of empirical reality. Thus, general inferences are induced from particular instances (Collis and Hussey, 2009). Contribution to theory is the outcome of the research, as it evolves from the observation and findings of the study. This is in contrast to the deductive process where the researcher develops a theory and hypothesis and then designs a research strategy to test hypothesis (Saunders *et al.*, 2009).

In describing some of the main methodologies used with in the social sciences, both Collis and Hussey (2009) and Remenyi *et al.* (1998) draw a distinction between those methodologies typically used within a positive paradigm and those used within interpretivism. However, they also advise that a number of research approaches are adaptable from either a positivist or an interpretivist philosophical position. In accepting that methodological choice is guided by philosophical positions, Saunders *et al* (2009) also propose that the choice of methodology should be guided by the suitability to answer the research question, the ability to meet the research objectives, and the amount of time required and resources available.

Methodology	Philosophical Position
Action Research	Strictly interpretivist
Ethnography	Strictly interpretivist
Grounded theory	Strictly interpretivist
In-depth surveys	Mostly interpretivist
Case Study	Have scope to be either positive or interpretivist

Table 4.2: Research Methodology. Source: Adapted from Remenyi et al. (1998)

Table 4.2 identifies the main methodologies within the interpretivist paradigm. In respecting Saunders *et al's* (2009) advice and the framework presented by Morgan and Smircich (1980), each methodology is considered for appropriateness within this study and discussed in the following sections.

4.2.4.1 Action research

Kurt Lewin (1946) defined action research as the process of organisational change having "spiral of steps, each of which is composed of a circle of planning, action, and fact-finding about the result of the action" (Lewin, 1946: 38). In contrast to typical models of social enquiry, action research acknowledges the researcher as an active participant in the process, rather than a passive observer (Connaughton and Weller, 2013). Typically, the researcher is part of the organisation in which the research and the change process is taking place, rather than being an object of study (Coghlan and Brannick, 2005; Saunders *et al.*, 2009). Action research also involves a degree of intervention on the part of researcher in the phenomenon being studied. This is emphasised by the iterative nature of the methodology where action planning is taken, following a diagnoses of the problem and evaluation of the recommendations (Saunders *et al.*, 2009).

Although it acknowledged that the application of action research in construction management has been growing (Connaughton and Weller, 2013), the methodology was not considered appropriate for this study. In the first instance, organisational engagement for research of a strategic nature would have proved difficult due to issues of commercial sensitivity. Secondly, subjective assessment of project success, as discussed in Section 2.6, could only be analyzed on completion of the project task undertaken by the TMO. Thirdly, it was determined that active participation in the study would only address the research questions from a single organizational perspective.

4.2.4.2 Ethnography

Emanating from the field of anthology, ethnography is firmly rooted within the inductive approach (Saunders *et al.*, 2009). The purpose is to interpret the social world in the same way as the research subjects would describe and explain it. The main method of data collection in ethnography is participant observation, where the researcher becomes a full member of the group being studied (Collis and Hussey, 2009). Due to the wide variation in human behaviours, ethnography involves unstructured fieldwork, observing people in their natural settings, and participating in daily activities, which necessities the research processes to be unstructured, flexible and responsive to change (Burgess, 1982; Saunders *et al.*, 2009).

Although a significant number of disciplines use ethnography, the methodology was not considered appropriate to address the aims of this study. Ethnography normally takes place within in a clearly defined natural setting. It is also, usually, necessary for the researcher to become involved within the group being studied for a substantial period of time (Remenyi *et al.*, 1998). Due to the limitations of resources and time, this would restrict the collection of data to either the project level, or the organisation level of the client organisation, and would, therefore, not address the overall research questions of the study.

4.2.4.3 Grounded Theory

Grounded theory is commonly associated with the inductive approach. The theorydiscovery methodology enables the researcher to develop a theoretical account of the general features of a topic whilst simultaneously grounding the account of empirical observations or evidence (Glaser and Strauss, 1967). In this way, theory from data is systematically gathered and analyzed throughout the research process, and involves an iterative process between data collection and data analysis to establish the conditions in which a theory will or will not hold (Strauss and Corbin, 1998). This is in contrast to positive studies that begin with a theoretical framework, before establishing hypothesis and collecting data to test the hypotheses. Remenyi *et al.* (1998) propose that the grounded methodology normally relies heavily on the use of in-depth interviews with experts in the field of study for the collection of evidence that will be used in an inductive way in the generation of theory.

However, as discussed in Chapter 3, research questions have evolved from the literature review, along with a conceptual model to be examined. Therefore, theory evolving from the research is not grounded under the definition of classic grounded theory (Glaser and Strauss, 1967), rather the study contributes to existing theory through an inductive process.

4.2.4.4 In-depth surveys

Remenyi *et al.* (1998) describes in-depth surveys as being used to generally attempt to obtain detailed in-depth evidence from a relatively small number of informants through a series of interviews. In contrast to questionnaire surveys, in-depth surveys tend to be unstructured and informal, designed to elicit in-depth responses and nuances. Although, able to be used from an interpretivist position, in-depth surveys were not considered appropriate to address the aims of the study as they do not readily allow for data collection within a bounded context, that enables focus on the specific research questions raised in Chapter 3.

4.2.4.5 Case study

The case study methodology is widely used in organisational studies and across social disciplines, including sociology, anthropology and industrial relations (Hartley, 2004). Eisenhardt (1989b) describes it as "*a research strategy which focuses on understanding the dynamics present within single settings*" (Eisenhardt, 1989b: 534). According to Yin (2014), the distinctive need for case study research arises out of the desire to understand complex social phenomena that allows investigators to focus on a '*case*' and to retain a holistic and real-world perspective. Hence, Yin (2014) defines a case study as an empirical enquiry that:

"...(1) investigates a contemporary phenomenon in depth and within its real world context, especially when (2) the boundaries between the phenomenon and context may not be clearly evident" (Yin, 2014: 16).

A distinctive feature of case studies is that they can employ an embedded design that consists of multiple levels of analysis within a single case (Eisenhardt, 1989b; Yin, 2014). Consequently, the 'monographic approach' (Hamel et al., 1993: 1) employs various methods (Hartley, 2004) and multiple sources of evidence to investigate a particular contemporary phenomenon (Robson, 2002). Yin (2014) also informs that within case study research there will be many more variables of interest than data points and, therefore, relies on triangulating data from multiples sources of evidence. Data collection methods include, but not limited to, archives, interviews, questionnaires and observation.

Despite the multiple methods used within cases study research, the approach has become a common strategy for conducting qualitative research (Stake, 1995). As explained by Meyer (2001), the key difference between the case study and other qualitative designs, such as grounded theory and ethnography, is that the case study is open to the use of theory or conceptual categories that guide the research and analysis of data. In contrast, grounded theory and ethnography prepossesses the theoretical perspectives are grounded in and emerge from first hand data (Glaser and Strauss, 1967). This is notwithstanding Eiesenhardt's (1989b) seminal paper describing how to build theories from case studies.

In considering the philosophical position of this thesis, and the research questions to be addressed, the case study methodology was deemed the most appropriate form of enquiry for this study. The potential to conduct intensive and detailed analysis through individual cases allowed for an in-depth understanding of the strategic nature of TMOs, which, as established in Chapter 3, has not been previously documented. Robson (2002) also perceives that case study design is useful when the aim of the study is to gain a rich understanding of the research perspective and the process being endorsed. Moreover, Sutrisna and Barrett (2007) propose that the case study approach is a reliable method for capturing rich information in complex situations such as construction projects.

4.2.5 Axiological assumptions

Finally, a number of scholars have introduced another level of questions considered necessary to fully define enquiry within a paradigm (Heron and Reason, 1997; Saunders *et al.*, 2009; Fitzgerald and Howcroft, 1998). The axiological level considers the role of values in research and the role the investigator's value plays in the all stages of the research process (Saunders *et al.*, 2009). Heron (1996) argues that an individual's values are the guiding reason for all human action. Whereas, the ontological, epistemological and methodological assumptions are about matters to do with truth, Heron and Reason (1997) propose that the axiological assumption is about values of being and about what humans state are to be valued by virtue of what they are.

From the positivist perspective the researcher's view of the role of values in research is that the research is undertaken in a value-free way (Saunders *et al.*, 2009). The researcher is independent of the data, and maintains an objective stance. In contrast, Saunders *et al.* (2009) defines the interpetivist, axiological perspective as the research being value bound, where the researcher is part of what is being researched and cannot be separated.

4.3 Research Design

Research design refers to the provision of a framework for the collection and analysis of data (Bryman and Bell, 2007). Hartley (2004) adds that research design should also make the argument for the logical steps that link the research questions to data collection, analysis and interpretation, in a coherent way. As discussed in Section 4.2.4, the case study methodology was judged to be the most appropriate strategy to address the aim, objectives and research questions addressed in this study. However, as noted by Meyer (2001), in comparison to other research strategies, there are virtually no specific requirements guiding case study research.

This section presents an outline of the research design, by drawing on the frameworks of Yin (2014) and Eisenhardt (1989b) who provide valuable guidance for the design of case-studies. The appropriate elements of both Yin (2014) and Eisenhardt's (1989b) frameworks are considered, in relation to the philosophical position of this study, as discussed in Section 4.2. Drawing on these, an overview of the research design for this study is presented.

4.3.1 Overview of Research Design

Yin (2003) presents a positivist/post-positive approach to research design in which he sets out a prescriptive process that includes five components of research design that are considered equally important. These are summarized in Table 4.3, with the application to the research design of this thesis identified at each stage.

Stage	Component	Explanation	Application to this thesis
1	A case study's questions	Clarification of the nature of study questions through literature and previous studies.	• Addressed in Section 3.8 of the literature review
2	Its proposition, if any	The propositions direct focus on what should be examined within the scope of the study. Some studies, for legitimate reasons, may not have study propositions, eg. Exploratory research.	• Conceptual framework presented in Section 3.9
3	The unit of analysis	Defining of the entity in the case study to be investigated and bounding the case. A unit of analysis can be single organisation, a single location, an individual or single event (Bryman and Bell, 2007)	• Addressed in Section 4.3.4 of the methodology chapter
4	The logic of linking the data to the propositions	Developing a strategy to matching case study data to concepts of interest. Techniques include pattern matching, explanation building, time-series analysis, logic models, and cross-case synthesis.	• Cross-case synthesis in Chapter 6
5	The criteria for interpreting the findings.	Determine how the data from the case will inform the findings. This involves identifying and addressing rival explanations.	• Addressed in Chapter 6

Table 4.3: Components of Case study design. Source: Yin (2014)

In contrast, Eisenhardt (1989b) takes an inductive approach to research design, in which she sets out a *'roadmap'* for the building of theories from case study research. The framework, illustrated in Table 4.4, synthesizes the previous work on qualitative methods with the design of case study research by building upon Glaser and Struass' (1967) research into grounded theory.

Step	Activity	Application to this thesis
Getting Started	Definition of research question Possibly a prior constructs Neither theory nor hypotheses	 Research questions and prior constructs presented in Section 3.8 of the literature review. General theories discussed within the literature review and developed in Chapter 6.
Selecting Cases	Specific population Theoretical, not random, sampling	• Purposive sampling, rather than theoretical sampling is discussed in Section 4.4.1
Crafting Instruments and Protocols	Multiple data collection methods Qualitative and quantitative data combined Multiple investigators	 Instrumentation choice discussed in Sections 4.4.4 Findings strengthened through the triangulation of evidence. The use of secondary data to support the analysis is discussed in Section 4.4.8 Single investigator used for PhD thesis.
Entering the field	Overlap data collection and analysis, including field notes Flexible and opportunistic data collection methods	• Data analysis commenced before fieldwork was complete to allow for emerging themes from each case, as described in Section 4.5
Analyzing data	Within-case analysis Cross-case pattern search using diverging techniques	 Data analysis approach is discussed in Section 4.5 In-case analyses of each case are provided in Chapter 5, Cross-case analysis and outcomes of the analysis are presented in Chapter 6.
Shaping hypothesis	Iterative population of evidence for each construct Replication, not sampling, logic across cases Search evidence for "why" behind relationships	 Cross case analysis in Chapter 6 allows for pattern matching. Replication logic discussed in Section 4.3.2.
Enfolding literature	Comparison with conflicting literature Comparison with similar literature	• The discussion in Chapter 6 relates the findings from the empirical work to literature reviewed in Chapters 2 and 3.
Reaching closure	Theoretical saturation when possible	• Models contributing to theory presented in Chapter 6

Table 4.4: Process of theory building from case study research. Source: Eisenhardt(1989b)

Although, the original work of Glaser and Strauss (1967) has evolved, grounded theory has remained an iterative approach between data collection and data analysis (Strauss and Corbin, 1998). Within grounded theory, cases would be selected throughout the theory building process. However, in Eisenhardt's (1989b) model, the researcher is

directed to select cases that replicate or extend theory early in the process, before entering the field with the rationale being, in order to control extraneous variation and help to define limits for generalizing and findings. Another deviation from the grounded theory approach is that Eisenhardt (1989b) does not suggest a continuous cycle between data collection and analysis, instead the suggestion is to shape hypothesis and develop theory from the data previously collected. Although this study does not apply a grounded theory methodology, as justified in Section 4.2.4.3, Eisenhardt's (1989b) framework is considered valuable to guide the research design of the thesis. In drawing comparison between the models, the research within this study is closer aligned with Yin's (2014) model, but does draw on Eisenhardt (1989b) for guidance.



Figure 4.2: Overview of the research design

In considering the models presented in Figure 4.1 and Figure 4.2, an overview of the research design is illustrated in Figure 4.2. As shown, the research is divided into three stages of theoretical foundation; data collection and analysis; and contribution.

Building a theoretical foundation for the study involved a review of the extant literature on the two theoretical areas of strategic alignment, from within the strategic management literature, and temporary originations from within the literature on organizational theory, project management and construction management. Definition of the research questions followed a review of the literature on the two theoretical areas of strategic alignment and temporary originations. A conceptual model in Figure 3.10 was developed from themes emerging from the literature review and used as the basis for the empirical stage of the study.

The first phase of data collection and analysis involved selection of relevant casesstudies to address the research questions of the thesis, and design of the research methods for the collection of data. These processes are discussed in Section 4.3.1 and Section 4.4, respectively. The second phase involved, systematic, within-case and crosscase analysis of the case studies under investigation. The findings from each case study are discussed in Chapter 5 and synthesized in Chapter 6, where the research outcomes are presented.

The final stage of the research involves answering the research questions in light of the research findings and literature reviewed as part of the theoretical foundation, to achieve the research objectives. Two models making contributing to theory are presented in Chapter 6. The first model, presented in Figure 6.3, identifies how varied actors within the TMO measure project success. The second model, presented in Figure 6.5, is the definitive model of the thesis and explains the complexities of strategic alignment within the context of a TMO. Contributions to theory are discussed in Chapter 7, along with contribution to practice. Finally, implications and suggestions for further research are proposed.

4.3.2 Establishing case study type

The choice of case study will be guided by the objectives of the research (Hartley, 2004) or, and by, the overall study purpose (Baxter and Jack, 2008). Yin (2014) draws a distinction between, *'explanatory'*, *'exploratory'* and *'descriptive'* case studies. It is

proposed that the type of case study used will be dependant on three conditions; (1) the type of research question posed; (2) the extent of control a researcher has over actual behaviour events; and (3) the degree of focus on contemporary as opposed to entirely historical events. In contrast, Stake (1995) makes a distinction between *'Intrinsic'*, *'Instrumental'* and *'Collective'*. Baxter and Jack (2008) present a summary of varied case study types, adopted in Table 4.5.

Case Study Type	Definition
Explanatory	A case study whose purpose is to explain casual links in real-life interventions that are too complex for survey or experimental strategies. Explanatory case studies focus on "how" or "why" a sequence of events did or did not occur. The case study, therefore, considers the operational links that need to be traced over time, rather than mere frequencies (Yin, 2014).
Exploratory	A case study whose purpose is to explore those situations in which the situation being evaluated has no clear, single set of outcomes. This type of case is used to identify the research questions or procedures to be used in a subsequent research study, which may or may not be a case study (Yin, 2014). Exploratory case studies tend to be descriptive and focuses primary on "what" questions with the goal to develop a pertinent hypothesis.
Descriptive	A case study whose purpose is to describe a phenomenon in its real world context (Yin, 2014)
Intrinsic	A case study whose purpose is to gain a better understanding of a particular case. The purpose is not theory building but is pursued when the case itself is of primary interest, not secondary interest (Stake, 1994; Stake, 1995).
Instrumental	A case whose purpose is to facilitate understanding of particular phenomena. The case is of secondary interest and is examined mainly to provide insight into an issue or to redraw a generalisation (Stake, 1994; Stake, 1995).
Collective	Several cases within the same study (Stake, 1995). This explanation is consistent with Yin's (2014) description of multiple case studies, discussed in Section 4.3.3

 Table 4.5: Definitions and Examples of Case Study Types. Source: Adapted from Baxter and Jack (2008)

As discussed in Chapter 1, the primary objective of this study is to explore how varied organisations within a TMO align multiple strategic objectives through a single construction project. The purpose of the case studies are to gain a deeper understanding into the challenges of aligning multiple objectives within the context of a TMO and analysing strategic behaviour of TMO actors. They therefore fit within a number of case study types shown in Table 4.5, and not restricted to any single one.

4.3.3 Selecting between single and multiple case studies

Whereas, Eisenhardt's (1989b) process advocates the use of multiple case studies to build theory, Yin (2014) distinguishes between single and multiple case study design. Yin (2014) advises that a single case study may be an appropriate design when the case is either critical to the theoretical proposition; it represents an extreme case deviating from theoretical norms; it is a *'common case'* that captures circumstances and conditions of everyday situations; it is a revelatory case and provides the opportunity to observe and analyse a phenomenon previously inaccessible to social enquiry; or it is a longitudinal case, studying the same case at different points in time. Hartley (2004) further adds that single case studies may be the only feasible option when access difficulties are present, when there are limited resources or the rarity of the phenomena precludes wider study.

Dyer Jr and Wilkins (1991) also proposes that single-case research produces more and better theory than multiple-case research. However, this is disputed by Ozcan and Eisenhardt (2009), who argues that *"multiple cases are effective because they enable collection of comparative data, and so are likely to yield more accurate, generalizable theory than single cases"* (Ozcan and Eisenhardt, 2009: 249). Leonard-Barton (1990) also caution that a single case study is subject to limits in generalizability and several potential biases, such as misjudging the representativeness of a single event (Tversky and Kahneman, 1986), and, therefore, suggest that multiple cases augment external validity and help guard against observer biases.

The advantage of pursing a multiple case study design is that it allows for comparative analysis across a number of cases. Miles and Huberman (1994) suggest that by looking at a range of similar and contrasting cases adds confidence to the findings, as researchers can understand an individual case single case findings, by grounding it, specifying how and where, and if possible, why it behaves as it does. Eisenhardt (1991) also argues that comparison of cases clarify as to whether an emergent finding is only idiosyncratic to a single case or a consistent finding among multiple cases.

Yin (2014) proposes that the distinct advantage of pursing multiple case studies is that it allows for replication of design. Within Yin's (2014) framework, replication logic is emphasised. This involves selecting two or more cases and ensuring that the same methods are applied in each case so that findings can be compared. Yin (2014) makes

the argument that only with such replications would the original findings be considered robust. In considering replication logic to be analogous to that used in multiple experiments, Yin (2014) advises that each case in the study must be carefully selected so it either (a) predicts the same results (*a literal replication*) or (b) predicts contrasting results but for anticipated reasons (*a theoretical replication*). The use of replication logic also allows for analytical generalisation, in which a previously developed theory is used as a framework for comparison of a study's empirical results. As explained:

"If two more cases are shown to support the same theory, replication may be claimed. The empirical results maybe considered yet more potent if two or more cases support the same theory but do not support an equally plausible rival theory" (Yin, 2003: 38)

It is therefore concluded that multiple case studies provide more robust findings than single case studies. This is partially due to the opportunity for triangulation of evidence (Herriott and Firesone, 1983). In addition, Taylor *et al.* (2011) stresses the importance in considering that case-to-case differences in the construction industry are likely to vary more than in other industries, and therefore studying multiple case studies within this context is of particularly significance to generate insights that can be generalised.

4.3.4 Unit of analysis

Merriam (1998) proposes that the single most defining characteristic of case study research lies in delineating the object of study. Merriam (1998) further argues that if the phenomenon of study is not intrinsically bounded, it is not a case. A number of authors support this notion by viewing the case study as a '*bounded system*' (Smith, 1978), '*integrated system*' (Stake, 1995) or being limited by scope with clearly defined conceptual boundaries (Bromley, 1986).

However, Yin (2014) advises that identifying the units of analysis that form the boundaries of a case is not a straightforward process. Within the social sciences, a case could consist of single organisation, a single location, a person or a single event (Bryman and Bell, 2007), whereas, a unit of analysis "could be almost any activity, process, feature, or dimension of organisational behaviour" (McClintock et al., 1979: 3). A case study may also involve examination of more than one unit of analysis, as proposed by Fletcher and Plakoyiannaki (2010):

"...when a single case study examines only the global nature of an organization, a holistic design is used, but when sub-units are analyzed in a single setting an embedded single case study approach is used" (Fletcher and Plakoyiannaki, 2010: 839). To illustrate this, Yin (2014) presents a $2x^2$ matrix that distinguishes between holistic and embedded units when using both single and multiple case studies.



Figure 4.3: Basic Types of Design for Case Studies. Source: Adapted from Yin (2014)

In Yin's (2014) model, illustrated in Figure 4.3, holistic (single unit of analysis) is categorized as Type 1 case design, This is used when a single case study is examining only the global nature of the organisation or programme (Fletcher and Plakoyiannaki, 2010). Yin (2014) also recognises that a single case may involve analysis at more than one level. This occurs when, within a single case, attention is given to sub-units, such as groups within a single organisation. This type of design is categorized as Type 2, embedded (multiple units of design). Multiple, holistic case design is categorized as Type 3, and multiple case design, that involves the examination of embedded sub-units of analyse is categorised as Type 4.

Level	Units
Cases	Temporary multi-organisations
Units of analysis	Individual organisations participating in each TMO
Embedded unit of analysis	Individual actors within each TMO member organisation

Table 4.6: Cases, units of analysis and embedded units of analysis.

This study fits within Type 4 of Yin's (2014) matrix. The relationship between cases, units of analysis and embedded units of analysis are defined in Table 4.6. The research design adopts a multiple case study approach, where the cases forming the research are temporary multi-organisations in public sector construction projects. The units of

analysis in each case are the individual member organisations that form each TMO. The embedded unit of analysis are the individual actors representing each member organisations within each TMO. The boundaries of the each case include those actors within the client organisation and actors within the consultant and contracting organisations that are directly involved in the construction project. The boundary of each case study is illustrated in Chapter 5, when describing the structure of each TMO.

4.3.5 Validity and Reliability

Taylor *et al.* (2011) makes two valid criticisms of case study research deign. Firstly, they find that design tends to be limited to only one or a few cases. Hence, much of the criticism focuses on how investigators can make justifiable generalizations from a limited number of samples, or indeed a single case study (Flyvbjerg, 2006; Eisenhardt, 1989b). Talyor *et al's* (2011) second criticism concerns the general lack of rigour in case study design:

"... many case-study research projects are executed with insufficient precision, quantification, objectivity, or rigor in which investigators have not followed standard procedures or allowed a biased view to influence the direction of the findings." (Taylor et al., 2011: 303).

Hamel *et al.* (1993) are similarly critical at the lack of rigor in the collection, construction and analysis of empirical data in the social sciences. They propose a link between lack of rigour and research bias, which is introduced through the subjectivity of the researcher, as well as, the research participants (Hamel *et al.*, 1993). In order to address the criticism and concerns over standards in data collection and analysis, Yin (2014: 46) identifies four tests that are commonly used to establish quality of empirical social research. These are identified as follows:

- **Construct Validity**: Establishing correct operational measures for the concepts being studied.
- Internal Validity: Seeking to establish a causal relationship, whereby certain conditions are shown to lead to other conditions, distinguished from spurious relationships. However, within Yin's (2014: 46) framework, internal validity is only applicable for explanatory or causal case studies.
- External Validity: Defining the domain to which a study's findings can be generalised.
- **Reliability:** Demonstrating that the operations of a study, such as the data collection procedures, can be repeated with the same results.

Test	Case study tactic	Tactic applied	
Construct Validity	Use multiple sources of evidence	 Convergence of evidence from multiple interviews from different organisations and hierarchical levels, see Section 4.4.7 and multiple sources of secondary data, see Section 4.4 Triangulation of evidence to form within-case and cross case analysis discussed in Section 4.4.4 	
External Validity	Use replication logic	 Replication logic used in data collection, discussed in Section 4.4.1 Standardised guide used for semi-structured interviews, see Section 4.4.6 	
Reliability	Use case study protocol	 Case study protocol discussed in Section 4.4.2 Thematic matrices from template analysis discussed in Section 4.5.2 and attached in Appendix F. 	
	Develop case study data- base	 Develop Case study register. A sample in Appendix B Data source log for each case attached in Appendix C. 	

Table 4.7: Validation and reliability tactics applied within the research

Drawing on Yin (2014), the tests utilised in this study are summarised in Table 4.7, identifying the specific tactics applied to ensure validity and reliability within the research. Internal validity in isolation is not considered, as establishing a chain of events and causal relationships between variables fits within the positive paradigm.

4.4 Data Collection

This section describes the data collection process used within this thesis, in greater detail. This includes selection of cases for data analysis, within-case sampling, design of the data collection protocol and selection of data collection methods.

4.4.1 Number of case studies

Whereas, no specific guidance is given to the number of cases to be included within a study, Yin (2014) suggests that the number of replications adds certainly to the results and proposes that six to ten cases *"provide compelling support for an initial set of propositions"* (Yin, 2014: 57). Whereas, Eisenhardt (1989b) suggests that between four and ten cases studies would be sufficient for theory generalisation.

In contrast, Miles and Huberman (1994) discourage cases design with over fifteen cases, as this can result in large unwieldy volumes of data and consequent loss of detail distinctiveness and context. In considering the complexity of case studies under investigation, the number of interviews within each case, and the limitation of resources to collect data, it was considered that research questions could be satisfactory answered and objectives achieved through four in-depth case studies that satisfied the criteria listed in the following section.

4.4.2 Case study selection

In contrast to theoretical sampling as proposed in Eisenhardt's (1989b) framework for generation of theory, this study takes a purposive sample approach to select cases for the research. Within this non-probability sampling method, subjective assessment of the sample is appropriate (Remenyi *et al.*, 1998). Purposive sampling involves drawing on the judgement of the investigator to select cases that answer the research questions and meet the objectives of the study (Saunders *et al.*, 2009). This is appropriate only when a limited number of the population can serve as primary data sources, and when there is a requirement to select cases that are particularly informative (Neuman, 2006).

Following Yin's (2014) guidance for replication logic, the selected cases for this study were chosen to, either literally or theoretically, replicate other cases within the sample. Cases were purposely selected according to the following judgement criteria:

- The case must have involved a temporary multi-organisation within the construction sector, where members of the TMO are employed (either permanently or temporarily) by different member organisations. This is contrast to property developers that may employ varied construction skills within a single organisation.
- In order to investigate the strategic alignment of projects (Chapter 2), all cases must have involved a single client that has established a TMO to realise their strategic objective through a single construction project.
- In order to investigate temporary organisational theory (Chapter 3), within the context of each case, all construction projects must have had a fixed deadline by which time the project should have been completed.
- To ensure replication logic, it was it was necessary for comparative reasons that all cases were from the same industry sector. In considering the differences between the management of construction projects within the private and public sectors (Songer and Molenaar, 1996), it was judged that cases from within the public sector would

provide better results, due to the inherent formal structures, processes and transparency of procedures.

- To ensure international and regional consistency, all cases were based in UK, within a 50-mile radius of each other.
- It was judged that the construction projects forming the cases, needed to be a recently complete, preferably within 24 months from the start of data collection. This was because of the perceived difficulty in conducting historical construction-based case studies. As discussed in Chapter 3, the TMO inevitably disbands on completion on the project, and as reported by the Construction Industry Training Board (BMG, 2015), there is a high degree of mobility within the UK construction industry. As a consequence, it was considered that there would be difficulties in contacting actors that had left the construction firm represented within the TMO. Another reason for the use of recently completed case studies was to ensure that participants in the study would recall critical incidents.
- The willingness of client organisations to participate in the study and divulge the necessary data required for analysis to the support the snowballing sampling strategy, discussed in Section 4.4.6, and provides encouragement to other firms to participate in the research.

Cases studies selected for investigation are outlined in Table 4.8, including a brief description of the construction project and a summary justifying the selection of each. For reasons of confidentiality, cases are referred to a Case A, Case B, Case C and Case D respectively.
ient Reason for selection of case	 I Knowledge of client organisations structure and processes. Meets main selection criteria as discussed in section 4.4.1 Fixed deadline for completion before start of semester Willingness by client organisation to allow access to data 	 Build Accessibility due to contacts in the client organisation Meets main selection criteria as discussed in section 4.4.1 Fixed deadline for completion before the start of the preparations for the major sporting event. Willingness by client organisation to allow access to data 	 Build Accessibility due to contacts in the client organisation Meets main selection criteria as discussed in section 4.4.1 Fixed deadline aligned with relocation of staff. Willingness by client organisation to allow access to data 	 Meets main selection criteria as discussed in section 4.4.1 Fixed deadline for completion of 700th anniversary commemorations
Procurem	Traditiona	Design &	Design &	Traditiona
Duration	19 months	36 months	12 months	22 months
Budget	£12m	£26m	C£7m*	£9.2m
Client Organisation	Higher Education provider	Public sector authority	Environmental regulator and flood warning authority	Conservation charity for the protection and promotion of cultural heritage
Project Description	Construction of modern student accommodation to replace existing residences.	Second phase of a new 4-lane public carriageway and associated regeneration works	Refurbishments of new build office building for the relocation and accommodation of laboratory and office facilities.	New build of a tourist visitor centre and to replace existing and conservation of
Proiect Type	New build, student halls of residence	New road, civil engineering	Laboratory and office fit-out	New Build, visitor centre
Case	P	a	U	Q

Table 4.8: Summary of cases studies forming the research

95

4.4.3 Case study protocol

Both, Eisenhardt (1989b) and Yin (2014) highlight the need to develop a case study protocol that can be used as a guide for data collection across multiple cases studies. Miles and Huberman (1994) also suggest such a protocol should outline the rules and procedure that govern the research of the case. This is to ensure uniformity of data collection and analysis, thereby, increasing the reliability of the case study research (Yin, 2014).

As a single investigator conducted the research, the main purpose of the case study protocol was to provide a guide for the data collection and reporting procedures across all cases. This was required for replication logic and to support the reliability of the data. Drawing on Yin's (2014) guidelines for development of a case study protocol, the protocol for this study is summarised in Table 4.9.

Stage	Protocol
A. Overview of Case	• Establish background to case through available secondary data and
Study	initial meeting with contact from client organisation.
B. Data Collection	 Develop case study register of data sources, identifying organisation, role and contact details. A sample in Appendix B Send introductory e-mail to source explaining the purpose of the study and requesting a meeting. Arrange interview meeting with source. Confirm anonymity of case and participant Forward interview questions at least two days prior to meeting All interviews recorded permission of the participant Identify further sources data during meeting Assign catalogue number to all captured data. An explanation of cataloguing is explained in Appendix C1.
C. Data Collection Questions	• Interview questions were designed to address the research questions of the study and the research objectives. A copy of the interview guide is contained in Appendix A.
D. Case Study Analysis and Reporting	 Codes created for each data source recorded in the data source log for each case. Copies of the data source logs are contained in Appendix C. Each interview logged and transcribed using MS Word document format. Each interview coded using NVivo. Coding system used within this study is incorporated within the Initial Template contained in Appendix D. Summary of themes evolving from each case are summarised in the Final Template contained in Appendix E, and a thematic matrix for each case. Copies of each thematic matrix is contained in Appendix F.

Table 4.9: Summary of case study protocol

An overview of each case and a description of each TMO are provided within the introduction to each case study in Chapter 5. For the purpose of replication logic, the

findings of each case study are reported in a consistent structure, as described in Section 5.1.

4.4.4 Instrumentation Choice

Interviews, documentation and archival records were identified as the most appropriate instruments for collection of data. Semi-structured interviews allowed flexibility to investigate emerging factors and critical incidents that occurred during each construction project, whilst maintaining rigour and reliability in data collection. Documents and archival records were also collected as sources of data to support and augment evidence from the semi-structured interviews, as well as, providing additional information to support the findings from of each case study.

Saunders *et al.* (2009) advise when using a case study strategy it is necessary to triangulate multiple sources of data within a single case. Bryman and Bell (2007) also suggests that triangulation though multiple data sources, or methods, support the reliability and validation of the study. Denzin (1978) distinguishes between four forms of triangulation

- 1. Data triangulation, which entails gathering data through several sampling strategies, so that segments of data at different times and social situations, as well as, a variety of data sources, are gathered.
- 2. Investigator triangulation, which refers to the use of more than one researcher in the field to gather and interpret data.
- 3. Theoretical triangulation, which refers to the use of more than one theoretical position in interpreting data.
- 4. Methodological triangulation, which refers to the use of more than one method for gathering data.

Each form is considered for appropriateness for this research. The approaches to triangulation applied within PhD thesis are outlined in in Table 4.10

Levels of Triangulation	Elements of Triangulation	Description
Data sources	Semi-structured interviews Documentation Archival records	• Triangulation between different sources
Interviews	Multiple perspectives	• Triangulation between different perspectives across all projects within the study
	Perspectives between different organisations	• Triangulation between different organisations participating within each project and across projects
	Perspectives between different levels within organisations	• Triangulation between different the project level and the organisation levels within each project and across projects
Archival records	Recorded minutes of progression meetings	• Triangulation between perceptions of project events and recorded minutes within each project

Table 4.10: Triangulation methods pursued in data collection stage

Research diaries and an electronic case study register were kept throughout the study to record persons contacted, dates of correspondence, interview meetings and relevant notes. All data, whether interview or document based, was electronically captured, assigned a catalogue number and entered into a data source log for each case (see Appendix B), to support the reliability of the research.

4.4.5 Semi-structured interviews

Under an interpretive paradigm, interviews are concerned with exploring "*data on understandings, opinions, what people remember doing, attitudes, feelings and the like, that people have in common*" (Arskey and Knight, 1999: 144). Within the context of multiple case studies, interviews enables the researcher to discover multiple realities of participants (Stake, 1995), and develop an understanding of the respondents world (Easterby-Smith *et al.*, 1991).

Easterby-Smith *et al.* (1991) advise that interviews for qualitative data maybe either unstructured or semi-structured. However, it was considered that, within this study, unstructured interviews, where the researcher uses an *aide-mémoire* as a brief set of prompts to guide a range of questions and topics, would not efficiently support the replication logic of the methodology adopted for this study. By comparison, in semi-structured interviews, the interviewer uses a research guide containing a list of questions on specific topics to be covered during the interview (Bryman and Bell, 2007). Remenyi *et al.* (1998) also maintain that when the research is conducted across a number of sites,

as is the case within this study, it is particularly important that an appropriate research instrument be used to systematise the collection of evidence and enable comparisons to be made during the analysis phase. Otherwise, the research runs the risk of collecting a wealth of evidence that, collectively, is difficult to generalise from (Sharp *et al.*, 2002).

'*Face-to-face*' was the preferred method of conducting each interview. Being in the presence of the participant makes it easier for the investigator to seek clarification of answers (Dialsingh, 2008). In the majority of cases, interviews were conducted at the participant's organisation or where most convenient for the interviewee. Where it was not possible to conduct interviews in person, primarily through the restraint of distance, a telephone interview or video-conferencing through Skype was used. There was a preference for interview by via Skype over telephone interviews, as this enabled many of the advantages of face-to-face, using available technology. Follow-up interviews were conducted as necessary, in particular, where there was a requirement for clarification of critical incidents.

As highlighted in the case study protocol, Table 4.9, interview questions were sent, via e-mail, to participants at least 48 hours prior to the interview being conducted. This allowed participants to prepare initial answers and recall critical incidents. This was considered best practice, as most participants had relocated to other projects and in some cases, other organisations. Answers to questions were explored during the interview process. Also, in order to gain trust from the participant and encourage agreement for a meeting, it was important that sources were made aware that the interviewe would not be asked to divulge confidential or sensitive information, and that they were comfortable with answering the questions that would be posed to them during the interview process. This is disused further within the discussion on research ethics in Section 4.6.

4.4.6 Interview design

Standardised questions that formed the basis of discussion within each interview were developed and refined through pilot-tests with an independent construction professional from within the construction industry. The pilot participant was considered most appropriate for the study, as he worked for a client organisation, managed projects within the public and private sector and had participated in varied temporary multiorganisations at differing levels. The pilot was used to test the relevance of the questions, in terms of contribution to the research objectives. But also to test understanding of the questions by the interviewees. Following the initial pilot study, the questions were refined, re-tested and finalised for use within the research.

Each interview lasted between 45 and 60 minutes. Interviews consisted of fourteen standardised questions to ensure reliability was maintained within the study and data analysis is supported through replication logic. Depending on the role of the participant in the construction project, some questions were omitted and the wordings of some questions were refined during the interview. Also, following Saunders *et al's* (2009) guidance for conducting semi-structured interviews, the order of questions varied between interviews, depending on the flow of the conversion. Additional questions were also asked to explore participant responses to questions, or from critical incidents or observations highlighted by participants from previous interviews.

A copy of the interview guide is contained in Appendix A. The first set of questions sought to gain information about the participant. This was to allow categorisation of the participant for comparative purposes. An important part of the study was to draw comparison of perceptions between clients, consultants and contractors at different organisational levels. Introductory questions also asked about the role of the participant in the construction project, their employment status and when they joined the project. Thus, allowing for partial contribution to the construction project, as observed by Cherns and Bryant (1984).

The objective of the main interview questions was to gain individual perceptions of the underlying rationale behind the construction project, the strategic benefits of the project, and individual perceptions of project success. The remaining questions explored the perceptions and realities deriving from themes within the literature on temporary organisations. These included, perceptions of short deadlines within the construction industry, the procurement process, location factors and conflicting organisational objectives.

As discussed within the case study protocol in Table 4.9, all interviews were electronically recorded with the permission of the participant, and transcribed independently by a professional transcription agency. To ensure quality of the transcription was maintained, interviews were transcribed strict verbatim, including preliminary discussions, repeats, laughs and pauses. The researcher then listened to each recording and edited the transcription prior to coding. Whilst, being a lengthy and detailed process, this allowed for detailed understanding of the respondent's perception, whist supporting the reliability of the data.

4.4.7 Within-case sampling

To answer the research questions presented within the thesis, it was necessary to collect data from actors participating within each TMO, and relevant actors from within the client organisation and member organisations participating within the TMO. In order to identify and make initial contact with potential interview participants, the snowball sampling technique was used to collect the required data for the research. Following Bryman and Bell's (2007) guidance, initial contacts were made with relevant key personal within each client organisation, and used as a vehicle for introduction with key actors from the varied organisations participating in each TMO.

Within the traditional snowball sampling technique, the precise extent of the population is difficult to identify (Bryman and Bell, 2007). In this study, the population derives from the member organisations and actors in each TMO. Therefore, once the organisations participating within the TMO were identified, the extent of the population could be estimated from the number of representative actors in each TMO member organisation. This was considered important to plan the data collection stage of the study, but also to establish when theoretical saturation of the data would be achieved (Eisenhardt, 1989b).

Group	Levels	Description
Client	Project Sponsor Director level Senior Project Management level Senior Management level	Client representatives involved in the inception and implementation of the project and concerned with the strategic value of the project to the organisation. This could be from the long-term strategic perspective of the project to the organisation or from a particular departmental end-user perspective
Consultants	Director / Senior Manager Project Team member	Consultants represent the external organisations involved in the project. This includes, external project management consultants, architects, engineers, quantity surveyors, structural engineers and designers
Contractor	Director / Senior Contracts management Project Manager	Depending on the procurement process, the role of main contractor was to implement the project in accordance with guidelines of the client and consultants. Sub-contractors and utility suppliers were not included in the sample, as this would have extended the group to an unmanageable size.

Table 4.11: Participant groups for data collection

Determining relevant interview participants was based on the objectives of the research and the research questions of the thesis presented in Section 3.7. Multiple interviews were undertaken to gain perspectives across all organisations participating in the TMO and at differing hierarchical levels. For the purpose of comparative analysis, each actor was grouped as client, consultant or contractor. Within each group, the role of the actor within the construction project was recorded within a case study register and the relevant data source log. A description of each participant group is presented in Table 4.11.

Participant	Case A	Case B	Case C	Case D
Client	7	2	7	4
Consultant	4	1	4	3
Contractor	2	3	1	1
Total	13	6	12	8

Table 4.12: Number of interviews within each case data

39 participants were interviewed across the four cases studies. A code was created for each data source and recorded in the data source log for each case. Copies of the data source logs are contained in Appendix B. Table 4.12 shows the number of interviews and their positions within each case. Due to the different types of construction projects and procurement strategies, the number of participants varied across cases. The structure and relationship between participants is explored in Chapter 5.

4.4.8 Documents and archival records

Documentation and archival records were collected to provide the background to each case and to verify and supplement data used for triangulation. The nature of construction projects is that they produce a significant amount of documentation. Therefore, to avoid an overabundance of data, it was important to narrow and focus only on the documentation that would support the findings of the research. In addition, some organisations were cautious about releasing documentation that was, either potentially sensitive, or that could be seen as an advantage to competitors, such as records containing financial information.

The most useful archival documentation was the minutes of monthly progress meetings from each project. As is typical practice within construction projects, weekly meetings were held at the operations level to discuss specific technical issues, therefore outwith the scope of this study. Instead, monthly progress meeting were held at the management level to discuss the strategy and progress of the project, and were therefore judged to be more relevant to the research objectives.

Documentation Type	Case A	Case B	Case C	Case D
Minutes of monthly progress meeting	Х	Х		Х
Project Management Reports	Х	Х	Х	Х
Closure reports	Х		Х	
Journal / Magazine articles		Х		Х
Internet articles	Х	Х	Х	Х

Table 4.13: Secondary data types used in study

Other useful documentation included project management and closure reports. These documents provided rationale for the project and client organisational judgements on the success of the project. Finally, Internet articles also provided useful backgrounds to the client organisations commissioning the project and enabled greater insight into the strategic rationale behind the projects. The type of documentation used in each case is shown in Table 4.13. A code was created for each secondary data source and recorded in each data source log, contained in Appendix B.

4.5 Data Analysis

Yin (2014) proposes that analysis of case study evidence is one of the least developed aspects in case study research, as there does not appear to be a standardised procedure to follow. However, in general, data analysis in qualitative research consists of

preparing and organising the data, then reducing it into themes through a process of coding and condensing of codes, before representing the data in figures, tables and discussion (Creswell, 2007). These general procedures are summarised in Table 4.14 through three different perspectives. Madison (2005) presents a critical ethnography perspective, whereas Wolcott (1994) uses a more traditional approach to research taken from ethnography and case study analysis. Finally, Miles and Huberman (1994) adopt a systematic approach to data analysis, which is applicable to case study research.

Analytic Strategy	Madison (2005)	Huberman & Miles (1994)	Wolcott (1994)
Sketching ideas		Write margin notes in field notes	Highlight certain information in description
Taking notes		Write reflective passages in notes	
Summarize field notes		Draft a summary sheet on field notes	
Working with words		Make metaphors	
Identify codes	Do abstract coding or concrete coding	Write codes, memos	
Reduce codes to themes information	Identify salient themes or patterns	Note patterns and themes	Identify patterned regularities
Counting frequency of codes		Counting frequency of codes	
Relating categories		Factoring, noting relations among variables, building a logical chain of evidence	
Relating to categories to analytical to framework in literature			Contextualize in Frame work from literature
Creating points of view	For scenes, audiences and readers		
Displaying the data	Create a graph or picture of the framework	Make contrast comparisons	Display findings in tables, charts, and figures; compare cases; compare with standard

Table 4.14: Com	parison of	data models	of data a	analysis. ,	Source:	Creswell ((2007))
-----------------	------------	-------------	-----------	-------------	---------	------------	--------	---

As shown in Table 4.14, there are a number of consistencies between the three frameworks. All advocate the importance of identifying themes evolving from the research and displaying the data for comparison and analysis. However, drawing on

Miles and Huberman's (1994), in particular, three concurrent flows of activity; data reduction, data display, and conclusion drawing and verification, are applied within this study.

Data reduction refers to the process of "selecting, focusing, simplifying, abstracting, and transforming the data that appear in written up field notes and transcripts" (Miles and Huberman, 1994: 10). This was pursued through the development of abstract codes and concrete coding using template analysis, assisted by qualitative data analysis software, to identify the salient themes evolving from the data (Madison, 2005). Data display, is the second major flow of analysis activity in Miles and Huberman's (1994) definition. This was addressed through the creation of a thematic matrix and a casual display network for each case. This served to organise and compress information, making it amenable to further analysis and interpretation. The conclusions and findings drawn from the data set are reported in Chapters 5 and 6 of the thesis. Processes within each simultaneous stream are discussed in the following sections.

4.5.1 Data reduction - Computer assisted data analysis

The interview data, once transcribed, amounted to over 200,000 words, together with over 150 archival documents and reports forming the secondary data. The total capacity of data within the data management system was 1.04 GB, including audio recordings of each interview. In considering the volume of evidence collected to support the study, it as was necessary that a logical data management system was implemented and that a rigorous process of data reduction be developed.

Data reduction was supported through the use of computer assisted qualitative data analyse software (CAQDAS), as termed by Lee and Fielding (2011). The use of software designed for qualitative analysis has become widespread and significantly more advanced than early computerised coding and retrieving tools (Bazeley, 2013). Modern CAQDAS, programmes are particularly beneficial for data reduction when managing large amounts of texts, and have the ability to store multiple data types. This was was considered necessary in this study, where the data consisted of Word documents, portable document formats (PDF), and web-based sources. Creswell (2007: 165) summarises the advantages of CAQDAS programme as:

• Providing an organized storage file system that allows the investigator to quickly and easily locate material and store it in one place.

- Enable the investigator to locate material easily, whether this material is an idea, a statement, a phrase, or a word.
- Encourages the investigator to look closely at the data and think about the meaning of each sentence and idea.
- The concept-mapping feature of computer programs enables the investigator to visualize the relationship among codes and themes by drawing a visual model.
- A computer program allows the researcher to easily retrieve memos associated with codes, themes, or documents.

Despite the benefits of CAQDAS programmes, Bazeley (2013) expresses concern that that computers distance researchers from their data, and foster the dominance of a *'code and retrieve'* strategy for analysis. This was managed through a process of manually reviewing the data prior to digital coding, as discussed in Section 4.5.2. This ensured full understanding of themes emerging from each transcript, but also enabled a holistic overview of each case.

From the varied systems available, the NVivo CAQDAS programme was selected to extract evidence relevant to the research questions and identify salient the themes evolving from the data. The software supported multiple data types, provided structured organization of data and ideas emerging during the analysis, has the facility to query the data and find incidences with the text, and facilitates within and cross-analysis of multiple case studies. Furthermore, the software was made freely available within the Heriot-Watt University and enabled easy conversion of files between MAC and PC, which enabled the researcher to work from different locations.

To address concerns of time required to adequately learn to use CAQDAS programmes (Creswell, 2007; Saldana, 2013), the researcher undertook two training session in the use of NVivo, utilised the on-line support offered by the programme developers (QSR International), and followed the guidance offered within textbooks specifically written for qualitative data analysis with NVivo (Bazeley and Jackson, 2013).

4.5.2 Data Reduction – Template analysis

Prior to using the CAQDAS programme, a lean approach to coding was initially undertaken (Creswell, 2007), whereby preliminary coding generated a small list of categories that expanded as the research continued. As noted by Saldana (2013), coding

may need to go through a number of iterations before the coding set is adequate for the purpose of analysis. Therefore, drawing on the principles of template analysis (King, 1998), *a priori* codes were developed from the research questions stated in Chapter 3 and the interview guide in Appendix A.

Within the template approach, contextual data is thematically organized and analysed according to an initial template, developed from a set of codes *a priori* (King, 1998). The benefit of template analysis is that it also allows for codes to be added *a posteriori*, thus, enabling a final template to be modified and refined as the data is examined (King, 2004). This is in contrast to content analysis, where codes are predetermined and their distribution analysed statistically (Weber, 1985), and grounded theory where there is no *a priori* definition of codes (Glaser and Strauss, 1967). The approach of incrementally modifying the coding set throughout the data collection and analysis process, allowed the researcher to maintain a link between the literature, from which the research questions were developed, and themes evolving from the empirical data, without inhibiting the capture of new insights (Bazeley, 2013).

A priori codes within the initial template were organized hierarchically, with groups of similar codes clustered together to produce more general, higher-order codes. The higher order codes were also categorised under the themes emerging from the literature review. Following King's (1998) guidance, the main questions from the interview guide, which was developed from the research questions, served as the higher order codes, with subsidiary questions and probes as the lower order codes. To maintain clarity, initial codes evolving from the *a priori* themes were identified and categorised within a two level hierarchy. Although within the final template, this extended to three levels. The summary of initial template with *a priori* codes relative to the research questions is provided in Table 4.15. A copy of the Initial Template is contained in Appendix D.

Category	Code No	Code Description	Relative research question
Strategic objectives	1	Organisational strategic objectives	1
Client complexity	2	Levels of strategy	1
Alignment mechanisms	3	Governance mechanisms	2
	4	Procurement Strategies	2
Environmental influences	5	External Environment	3
	6	Client Leadership	3
TMO Strategy	7	TMO Behaviours	4
Project success	8	Project success criteria	5

 Table 4.15: Summary of initial coding set

The initial coding set in Table 4.15 was systematically applied to further data and modified as necessary in order to develop a final template for the purpose of guiding the analysis. Development of a single, final template enabled the researcher to identify both emerging themes and consistent themes between cases. This supported the within-case analysis in Chapter 5, and cross-case analysis in Chapter 6.

Development of the final template involved a manual review of transcripts from the semi-structured interviews and secondary data collected from each case and the support of computer based coding, as discussed in Section 4.5.1. Minutes from the monthly progress meetings were particularly beneficial in gaining a holistic overview of each case study. Using the initial template, data emerging from the transcripts and documentation collected over stages B and C of the case-study protocol, were thoroughly analysed and applied to appropriate a priori codes. Parallel coding of segments of texts was also applied whereby the same segment of text was classified within different codes at the same level (King, 1998). It was important that the initial coding set was not treated as prescriptive, and data was not forced to fit the initial coding set (Bazeley, 2013), rather a posteriori themes emerged from the early data sets. The iterative process enabled additional codes to be added as themes evolved during the analysis. Also, early and redundant codes were removed or redefined during the analysis process. It was only after a number of iterations and applying data from all four cases that the template was finalised. A copy of the Final Template is contained in Appendix E.

4.5.3 Data display and conclusions

The remaining components of data analysis involve presenting the interpreted data and displaying the analysed evidence, from which conclusions and verifications are drawn. Within Miles and Huberman's (1994) concurrent flow of activity, a display refers to "a

visual format that presents information systematically, so the user can draw valid conclusions and take needed action" (Miles and Huberman, 1994: 91). Whereas, the final template developed in Section 4.5.4 displays codes a priori and a posteriori, there remains a requirement to present the researchers interpretation of the data (King, 1998).

Coffey and Atkinson (1996) contend that, although, the use of codes and identification of themes are a critical part of the qualitative process, it is not an end in itself. As argued by Bazeley (2013), the description of thematic codes and categories identified in the data provides a useful starting point in developing a report of findings from the study. However, effective analysis requires using data to build a comprehensive, contextualised and integrated understanding of the findings, with an argument drawn from across the data that establishes the conclusions drawn.

Within this research, conclusions were drawn from within-case and cross-case analysis, following Eisenhardt's (1989b) process of theory building from case study research, as presented in Section 4.3.1. Reporting the findings of each case individually enables deeper understanding of the perspectives of individual participants, and helps ensure that the discussion of themes does not become abstract from participants accounts of their experience (King, 1998). Ayres *et al.* (2003) also argue that analyse of individual cases enables the researcher to understand those aspects of experience that occur not as individual *"units of meaning"*, but as part of the pattern formed by the *"confluence"* of meanings within individual accounts (Ayres et al., 2003: 873). Individual case analysis, therefore, enabled conclusions drawn from within the boundaries of a single case to be compared and with other cases in the study and synthesized to draw final conclusions (Miles and Huberman, 1994).

Detailed descriptions of each case study were informed through the process of template analyse, discussed in Section 4.5.2. Building on the initial template, a thematic matrix for each case study was developed in conjunction with the final template. The thematic matrices, contained in Appendix F, isolate the critical incidents and unique themes. These are further explored within the discussion of each case study, with conclusions drawn to answer each research question. A display of each TMO structure also provides a holistic overview of each case and the relationships between individual participants to the study. Furthermore, a casual diagram developed from the case study findings illustrates the influential relationship between themes to support the conclusion of each case study.

In order to address the aims and objectives of the research, the final stage of analysis included synthesis of themes and findings from each case study. Comparison of *a priori* and *a posteriori* codes emerging from data set allowed for themes emerging from the research to be explored in relation to the extant literature. At the same time, comparisons of the thematic matrices enabled parallels and divergences across the cases to be critically discussed. Using two, concurrent approaches to inform the aims and objectives, supported the verification of the findings, and led to more robust conclusions discussed in Chapters 6 and 7.

4.6 Research Ethics

For Berg (2001) scientists conducting social research have a greater ethical obligation to their study population than, perhaps, other discipline areas. The reason is that social science examines the social behaviours and personal perceptions of humans.

"From such excursions into private social lives, various policies, practices, and even laws may result. Thus, researchers must ensure the rights, privacy, and welfare of the people and communities that form the focus of their studies" (Berg, 2001: 39)

Miles and Huberman (1994: 288) also advise that as qualitative researchers, we must also consider the implication of our actions in relation to the people whose lives we are studying, to our colleagues, and to those who sponsor our work.

The research within this thesis was conducted in accordance with the research ethical guidelines of Heriot Watt University, which is monitored by the University Ethics Committee established to guide schools, monitor procedures and ensure ethical issues are being considered (HWU, 2010). Besides, the ethics policy of the university, varied scholars has also outlined basic principles of ethical research. Each area, relevant to this study is considered and discussed in the following sections.

4.6.1 Harm to participants

Research that has the potential to cause harm was the primary ethical concern within the a number of discourses on research ethics (Flick, 2009; Diener and Crandall, 1978).

These include physical harm, harm to participants development or self esteem, and stress or harm to career prospects or future employment (Bryman and Bell, 2007).

The collection of data did not present a risk of physical harm to either the researcher or participants. Psychological harm was avoided through the anonymity of participants and cases. This also mitigated any business risk to firms participating in the study. Interview questions were designed to be sufficiently probing, without being contentious or challenging the ethics and values of respondents. As stated in the case study protocol in Section 4.4.3, a copy of the questions were sent to the participant prior to the interview, in order to allow preparation of answers, but also, to ensure that the participant was comfortable with the questions being asked, and to provide the opportunity to request further clarification before the meeting.

4.6.2 Informed Consent

The principle of informed consent means that prospective research participants are given as much information as might be needed to make an informed choice about whether or not to participate in the study (Bryman and Bell, 2007), and that individuals fully understand the consequences of participation (Flick, 2009).

Using snowball sampling, as discussed in Section 4.4.6, prospective participants were sent an introductory e-mail by the researcher, respectfully requesting their participation in the study. The introductory e-mail clearly stated the purpose of the study, the reason for the interview, the length of time the interview was expected to take, and the name, title, institution and contact details of the researcher. The e-mail also assured participants that, in accordance with the ethical guidelines of the university, the case and all participants would remain anonymous. Further information pertaining to the study was supplied on requested.

Prior to the start of each interview, the researcher explained the research process and confirmed the ethical guidelines of the university. Permission was sought from the participant prior to recording, and the participant was assured that the recording and transcript would only be used for the purpose of the study.

4.1.1 Maintenance of privacy

Within this study, the maintenance of privacy relates to the anonymity of participants and firms contributing to the research. Being anonymous in a research context means that only the principle investigator can match the results of a study with the participant associated with those results (Flick, 2009)

Whereas, every endeavour was made to ensure anonymity of the participants, as previously discussed, there was a concern that it would difficult to guarantee anonymity of the cases. This is because of the high profile nature of the cases in this study and some of the characteristics that are unique to specific construction projects. Participants were made aware of this concern prior to the interviews taking place. Within Chapter 5, all direct references to each case have been disguised, as have the location of each construction project.

4.6.3 Confidentiality

To ensure anonymity of participants and maintain confidentiality, data was stored securely on a password-protected computer and backed-up on password-protected cloud storage. This included transcripts from interviews, digital recordings, the case-study register and reports received from participants. However, as the case studies involved public sector projects, a large majority of the secondary data was already in the public domain. Despite this, respect for confidentiality was maintained throughout the case study reporting process.

4.7 Conclusion to Chapter

This chapter presented a detailed description and rationale of the methods used in this study to realise the aims and of objectives of the research. Applying case study methodology within the intrepretivist paradigm allows for in-depth understanding of the multiple perceptions within a TMO. To date, perceptions of strategic alignment and project success within the context of a TMO has not been captured across multiple case studies.

The difficulty with case study design is the lack of consistent guiding principles within the literature (Meyer, 2001). This study has applied practices from both Eisenhardt's (1989b) grounded theory approach and Yin's (2014) post-positivist approach, to developed a robust design process. Despite this, limitations in the data collection and investigator triangulation is accepted, where it is recommended that multiple investigators collect and interpret the data (Eisenhardt, 1989b; Denzin, 1978). King (1998) also suggests that the initial template used in template analysis, be developed with two or more advisors, with knowledge of the research. However, as this thesis is for award of PhD, it was deemed appropriate that a single investigator conducted the research.

Ridged processes were implemented to ensure reliability. These include a case study protocol as guide for all interviews, data source logs for each case study, and the use of a final template and thematic matrices for individual cases to support analysis and conclusions. The evidence for each case study, collected from the methods discussed in this chapter are presented in Chapter 5 and synthesised in Chapter 6 to draw the final conclusions of the research.

CHAPTER 5 FINDINGS: WITHIN-CASE ANALYSIS

5.1 Introduction to chapter

This chapter is the first of two reporting the findings from the collection of data described in Chapter 4. Within this chapter, the empirical findings and analysis from each case study is reported, the following chapter syntheses the individual case study findings, in order to draw conclusions from the research. Evidence used to support the findings and analysis of individual cases are recorded in the data source logs, contained in Appendix C. The structure of the chapter is illustrated in Figure 5.1 and described in Section 5.1.1.



Figure 5.1: Within-case analyses chapter structure

As discussed in Section 4.5.2 of the methodology chapter, *a priori* themes were developed from the literature to create an initial template. Following a process of coding and analysis a final template, formed from all cases, and a thematic matrix from each case study was produced to report the findings and conclusions in this Chapter.

Case	Organisation Type	Project	Section
Α	Higher Education	New Student Residence	5.2
В	Local Authority	New 4 lane carriageway	5.3
С	Environmental Authority	New Office and Laboratory facility	5.5
D	Heritage & Conservation Agency	New tourist visitor centre	5.4

Table 5.1:	Case studies	informing	the	research
1 abic 5.1.	Cuse studies	, miorining	the	cocaren

The four case studies comprising the research are introduced in Table 4.8 of the previous chapter and summarised in Table 5.1, identifying the relevant section of each case study report.

Desc	cription of Section	Relative research question
•	A general background and holistic overview of the construction project informing the case study with a brief summary of key events	
• _	A description of the structure of the TMO formed by the sponsoring organisation to implement the construction project.	Q2
•	A discussion of the client strategic objectives being pursued through implementation of the construction project	Q1
•	A discussion of the varied TMO member's strategic objectives to be realised through participation in the project.	Q1
•	An examination of the mechanisms implemented to maintain alignment of client strategic objectives	Q2
•	A discussion of internal and external environmental factors that influenced the realisation of strategic objectives	Q3
• ,	An examination of project strategy developed by TMO actors to deliver the construction project	Q4
•	A discussion of the varied stakeholder perceptions of project success	Q5
• .	A summary of findings for each case and answers to questions raised in Chapter 3.	

 Table 5.2: Structure of each case study report

To support the replication logic of the research, as discussed in Section 4.4.3, the findings of each case study is presented using a consistent approach, as shown in Table 5.2. Each section is discussed to address the research question. A thematic matrix identifying the themes evolving from each individual case study, and the Final Template, is attached in Appendix E.

5.2 Case A – New Student Residence

Case study A involves the construction of a new student residence built on an existing suburban university campus, commissioned by a university within the higher education sector. A holistic overview of the construction project is provided in the following section, with discussions to answer the research questions, as shown in Table 5.2.

5.2.1 Background to the case

The new student residence formed part of an infrastructure programme to replace existing student accommodation was considered "...no longer fit for purpose and carrying an increasing risk of service failure as well as being materially below the standard that we [the University] would wish to be represented as..." (A-PMc-0210-report). The conclusion of inspections in 2009 found that the existing accommodation facility would not comply with the new guidelines for Licencing in Houses in Multiple Occupation (HMO) regulations (2010). An appraisal of the buildings, by the university Estates Department, subsequently led to the recommendation that refurbishment of the existing student residencies was not a commercially viable option to meet the new guidelines. Following submission of the Residences Business Case (A-PMc-0210-report), the university agreed to erect a modern residency block that comprised of 273 bedrooms, kitchens and shared social areas. A time-ordered display of the key events are summarised in Table 5.3 and discussed within the case study.

	2009	2010		2011		2012
•	July. PMC appointed	 February. Business case approved by PME March. Project approved by Court 	•	January. MC appointed	•	7 th September. Practical completion and handover
		October. Advertised for tender	•	February. Construction commenced on site	•	8 th September. Students take occupancy

Table 5.3: Case A: time-ordered display of key events

A project management consultant (PMC) was appointed in July 2009 to provide design and project management services for the new residencies. Following the process of design and application for statutory approvals, the project was advertised for tender on 11th October 2010. The main contractor (MC) for was formally appointed on 27th January 2011, with construction commencing one month later for a scheduled completion date of 29th June 2012 (A-CS-0710-schedule). However, due to a series of delays, the building was not practically complete until 7th September 2012. Student occupation of the new residences began on 8th September for a semester start date of 10th September 2012.



Figure 5.2: Case study A - TMO Structure

The structure of the TMO created to realise client strategic objectives is illustrated in Figure 5.2. As shown, the TMO consisted of three member organisations. These are identified as the Client Organisation (the university), the PMC as the consulting organisation and the main contractor as the MC organisation. Within this case study, sub-contractors are considered to be part of the main contracting organisation.

Contractual arrangements and governance process are discussed in Section 5.2.5 and 5.2.6 respectively.

5.2.2 Client organisation's strategic objectives

The strategic objectives to be achieved through implementation of the project as reported in the Residences Business Case (A-PMc-0210-report) to justify expenditure on the project were as follows:

- 1. To provide a long term solution to rapidly deteriorating student residences
- To address safety risks and concerns relating to non-compliance with UK Houses in Multiple Occupation (HMO) regulations
- 3. To provide more attractive campus accommodation for students to support the university strategy for growth.
- 4. To address marketing and sales issues faced by the conference office and increase the opportunities for growth to support the university's financial position.
- 5. To make a major contribution to implementing the university Estate's strategic plan in reducing the maintenance backlog

As explained by Director of Campus Services, the primary objective of the university was the provision of student accommodation that was considered to be of an acceptable standard:

"... I think, undoubtedly for the university is the fact that we make a promise to new and first year students that we will make sure they have accommodation. I think it's got the added benefit that it actually stacks up commercially, but the primary motivation for the university is, or certainly has not been because there's money to be made in this. They do it because they have made a very public statement that first year students will be provided with accommodation" (A-MD-031214-intvw)

Failure to provide appropriate accommodation for first year and international students would have had an impact on the reputation of the university, which is dependant on its duty of care for students. It was, therefore, critical that the residencies were complete as advertised, for students to take occupation at the start of the new academic year.

5.2.3 Business level strategies

The client strategic objectives, set the executive level of the university, were implemented at the business level through the Department of Student Services, the Department of Hospitality and Catering, and the Department of Estates. The interrelated strategies are outlined as follows.

5.2.3.1 Department of Student Services strategy

The Director of Student Services reported that the existing accommodation had deteriorated through age and consequently suffered from poor student feedback. "...the students didn't like living there and the quality was really poor (A-CJ-280314-intvw). Not only did this impact on the reputational risk of the university, but also on the university strategy for growth. "...unless we had high quality accommodation, or good quality accommodation then the students won't come" (A-CJ-280314-intvw). It was perceived that provision of student accommodation that was of a higher quality than standard university accommodation, would provide the university with "a competitive edge" (A-DW-170315b-intvw) and "enhance its reputation in the market" (A-MMc-291014-intvw). This was viewed as being of particular importance to the international market where universities competed on the offering of a "composite package of learning, welfare and accommodation" (A-DW-170315b-intvw).

5.2.3.2 Department of Hospitality Services strategy

It was also suggested that by increasing the quality of student accommodation the University could enhance their financial position by charging a "*premium rent*" for the rooms, thus, supporting the university's financial position (A-TD-240314-intvw). An initial strategic option to generate revenue from the residences was to offer conference facilities and accommodation to non-students. This would address the marketing and sales issues faced by the conference office, as highlighted in the Residences Business Case (A-PMc-0210-report). This was pursued through specification of larger rooms and auxiliary accommodation. However, the strategy for conference accommodation was later abandoned as the additional space required would reduce the capacity of rentable rooms and significantly increases the unit cost from "...£30,000 to, at least, £50,000 per bedroom" (A-MC-190314-intvw).

5.2.3.3 Department of Estates strategy

A key objective of the Department of Estate's strategic plan was to develop a sustainable and low maintenance facility. This was pursued through a process of *"life cycle costing"* that involved the use of materials and fittings that were perceived to be of an acceptable level of quality.

"... as opposed to a low cost budget delivered quickly that could be used for 10-12 years and then knock it down and start again... these buildings will be there for a long time" (A-BG-240314a-intvw)

Through a process of nominating suppliers for materials, fixtures and fittings, the Department of Estates could manage the supply chain with the objective of reducing the long-term maintenance costs of the new facility. It was considered that adherence to a strict *"Bible"* of nominated suppliers increased costs, as the option for competitive bidding among suppliers was removed (A-PW-071014-intvw).

5.2.4 TMO strategic objectives

A significant factor influencing the strategic objectives of TMO member organisations was the UK financial crises of 2008-2013, which witnessed a dramatic decline in the construction industry (CIC, 2009). During the crisis, student residencies emerged as one of the few growth markets. As a result of the increasingly competitive employment market, universities saw a sudden growth in student applications (UCAS, 2011), which led to an imbalance occurring within the student property market between the supply of accommodation and the demand for bed spaces (KnightFrank, 2011). Consequently, the Higher Education sector was perceived as an attractive market for consultants and contractors.

Therefore, besides the financial benefit of pursuing the residency project, both, the PMC and the MC identified potential market opportunities through membership of the TMO. Through alignment with the university strategy for delivery of a "*bespoke residence that was a higher quality than the normal market for an end user client or an owner-occupier*" (A-BG-240314a-intvw) the PMC could enhance their project portfolio of the PMC and expand business in the education sector.

Similarly, membership of the TMO enabled market entry into the student residence sector for the MC.

"...the education market was somewhere we were looking to develop an offering in and [The University] are by any measure a blue chip client in that market" (A-MMc-291014-intvw)

By achieving client satisfaction the MC would be able to apply for further work with the university, as part of the wider regeneration programme and apply for a favourable reference to support applications for other contracts within the higher education sector.

5.2.5 Procurement process

The services of the PMC were employed under the Office of Government and Commerce (OGC) Project Management & Full Design Team Services Framework RM457. This was considered a "*faster route to market*" than advertising through the Official Journal of European Union (OJEU) and the time required for the preceding prequalification process (A-BG-131014b-intvw). Under the conditions of the framework, the PMC would provide all design services, including civil and structural engineering, mechanical engineering (M&E), quantity surveying (QS), Construction Design and Management coordination (CDM) and architectural design. All consultants were existing employees of the PMC, apart from the architect whose services were contracted to the PMC from an independent architectural practice.

The decision was made to manage the contractual arrangements of the construction under the traditional procurement strategy. This was to guarantee "*control over the end product*" through nominated suppliers, in accordance with the Department of Estates strategic plan for reduction of building maintenance (A-MC-190314-intvw).

The decision of procurement strategy was made despite the recommendation of the PMC organisation to pursue a design and build procurement method, which was considered more suitable for buildings of modular construction, such student residencies, and would transfer the risk of cost over runs to a design and build contractor (A-GC-231214-intvw). All respondents later considered the decision to pursue a traditional procurement strategy to be the main failing of the project.

"It's the main reason why it got all cocked up without... it's 90% of the cause...it was the wrong choice of procurement route" (A-DW-170315b-intvw)

Another perceived failing in the choice of procurement was the inexperience of the client body to efficiently manage the design team responsible for managing the procurement process.

"So, you've got a group who are just not familiar with running projects of that scale and if you take responsibility for the design team you need to know what you're doing" (A-MD-031214-intvw)

As explained in Section 3.6.2, to ensure cost certainty for the client, design information needs to be complete at the time of going out for tender, otherwise the client runs the risk of additional costs. As a consequence of the economic decline in construction industry, contractors were forced to submit low and often unrealistic bids "*with stupid*

profit levels" (A-MD-031214-intvw) to secure contracts. With losses being recouped through claims for additional works.

"... the contractor's dependent on the design team to make sure that they make as many cock ups as possible so they can drive a bus through it and make money" (A-DW-170315b-intvw)

Referred to as "*suicidal pricing*" (A-MC-190314-intvw), the university accepted a tender application from the MC at 10% below budget. Incomplete design information and post-tender tender design changes left the university exposed to additional costs by the contractor, for variations from the original tender documents and charges for additional items.

5.2.6 Governance structure

The university established two new boards for the purpose of governance, as shown in Figure 5.2. A project board was created to provide direction and guidance for the TMO. Representation included the Director of Finance, the Director of Estates, Student Services, Hospitality Services and the Student Union. An oversight board was created to report directly to Court, as the highest governing body of the university, and included representation from the Finance Committee, the Audit and Risk Committee, the Estates Committee and representation from PME (A-PMc-0210-report).

The new governance structure was incorporated within the established governance framework of the university. Within the governance arrangements, design, budget and deadlines of the project required formal approval by the Court of the University before progressing. However, as Court only met four times per year, the key milestones within the project were aligned to the pre-arranged dates of Court meetings.

"The university governance structures imposed a timetable that didn't fit with the project...if we missed the Court deadline [for submission of paperwork], we wouldn't be able to start the next stage of the project. We rushed too much...we rushed the proposal and it wasn't complete and that proved to be the biggest mistake we made, because we hadn't fully costed the building and delivery of the facilities within them" (A-CJ-280314-intvw)

As a consequence of working towards the Court imposed deadlines, and seeking to meet the deadline of completion before the start of term, design information was not complete at the time of advertising for tender of main contractors. Concern was also expressed over the experience of newly established project board to effectively guide and direct the project to achieve the university strategic aspirations, as previously highlighted in Section 5.2.5.

"The main problem was that there was hardly anybody on the project board who had ever been involved in delivering new residences and so there was a lack of knowledge and a naivety about how we're going to build this really high specification building ... There was naivety, lack of expertise and a lack of honesty about delivering on time and we tried to deliver too fast, to a time scale that wasn't achievable" (A-CJ-280314-intvw)

A project manager (PM) for the Department of Estates was later appointed act as client representative and the main interface between the Project Board and TMO (A-HW-0311-report).

5.2.7 Client behaviour

The client complexity within the project board resulted in competing and often conflicting directive being issued to TMO actors. There was also evidence of disagreement between board members, which caused difficulties for members of the TMO.

"...in the early stages there were regular meetings where we would come and present to the university. That was a very tortuous drawn-out process and there was never an agreement between those bodies. So, how would you ever know if you were doing the right thing or the wrong thing because they would always be pulling in different directions? ... and they would have conversations, 'Don't listen to them. Listen to me'...That made it very difficult" (A-PW-071014-intvw)

There was frustration at the lack of leadership and direction from the university. The initial lack of a *"single decision-maker*" (A-PW-071014-intvw) and the uncertainty of who, in the university, had the authority to make the final decision had an impact on the progress of the project.

"That uncertainty ran through the entire university team. When we'd gone to speak to them ... that uncertainty about who is in charge percolated down through the entire team. It eventually came to us and at times you almost felt that you were caught in the middle. It was a bizarre situation for a while" (A-MMc-291014-intvw)

Although, the appointment of a client project management, to provide direction to the TMO, eased the difficulties caused by the client complexity, the leadership style of the appointed PM to recover delays on the project, did affect the motivation of the TMO.

"... the team broke down because of the pressure applied by me...I was simply putting them under pressure day to day, which is what a project manager is supposed to do because they have to deliver..." (A-DW-170315b-intvw)

As a consequence TMO actors also reported of "some very unreasonable behaviour on the part of the client" (A-PW-071014-intvw) and were of the opinion that the university was "very poor at relationships with all the team" (A-MMc-291014-intvw). As a consequence of the client behaviours, meetings between the client and TMO became 'fractious rather than conversational' with discussions between actors becoming 'significantly heated' (A-MMc-291014-intvw) for the duration of the project. As observed by the MC:

"... it just didn't promote team working. If anything, it promoted working in silos and looking after yourself...It's about a team working well and working well together, and the client has a lead role in which to claim that. And if the behaviours for the client are such that it actively discourages a team from working together, that's a recipe for disaster" (A-MMc-291014-intvw)

The evidence finds that the leadership behaviours led to a lack of trust in the client from members of the TMO, and also developed into a lack of trust in the PMC from the client, in which the professional integrity of the client was being questioned (A-GC-231214-intvw). Negative behaviours also had an impact on the internal relationships within the TMO, as discussed in Section 5.2.9.

5.2.8 Supply chain

The most threatening external influence to realisation of strategic objectives was the uncertainty in the MC's supply chain. The accepted tender price offered to the client was partially based on estimates received by potential sub-contractors for specific elements of the work. The MC was, therefore, dependant on the securing the services of specific sub-contractors at, either, the estimated price or lower, depending on the cost submitted (A-MC-190314-intvw). However, as observed by the PM:

"...the market had changed. Basically what happened was prices on materials, in particular, had levelled off during the recession period and just as they started to pick up again actual material costs went through the roof. Then labour costs went up as well so all in all it's a perfect storm for being a contractor" (A-DW-170315b-intvw)

A consequence of the turbulent market conditions was that the MC was unable to secure sub-contractors at the previous estimated cost. They, therefore, needed to return to the market in order to seek availability of alternative sub-contractors that would provide services at the lower costs. The time taken to secure alternative sub-contractors had an inevitable impact on the progress of the project.

A second difficulty in the supply chain was the availability of bricklayers for the traditional masonry and pre-cast concrete construction of the new residencies.

"They need the work in front of them to have continuity. What happened, then, when we then started hitting all these issues with design and all that, that affected continuity for the bricklayers" (A-MMc-291014-intvw)

As a result of adverse weather conditions and delays in design information, there were gaps in the construction of the masonry. Consequently, the MC was *"losing bricklayers from the job left, right and centre"* (A-MMc-291014-intvw), as they moved on to other projects, in order to ensure continuity of employment and income. Thus, causing further delays in in achieving the original deadline.

5.2.9 TMO strategy

The project strategy of the TMO was influenced by a number of internal and external environmental factors. This includes governance mechanisms, client complexity, client behaviours, and the varied strategic objectives being pursued through the project. The strategic behaviour of the TMO in delivering the project is explored under the themes of *time, team, task* and *transition*.

5.2.9.1 Time

All TMO actors recognised the need to complete the project before the start of the new term and fully understood the reputational risk to the university in not achieving the deadline.

"... the university had already marketed the accommodation and the damage that would have been done from the reputational point of view to the university. We were fully aware of that and we wanted to make sure we got the students in on time" (A-MMc-291014-intvw)

Actors were aware that if the residency was not complete in time for the students arriving, there would be a requirement to arrange alternative accommodation for students, and the new residences would "*basically, be empty for a year*" (A-MD-031214-intvw). Despite this, the residencies were delivered two months later than the original deadline and 15% over budget at completion.

In the latter stages of the project, the strategic focus of the TMO was to meet the deadline of completion prior to students arriving for the start of the new term. This was at the expense of other strategic priorities. "...come hell or high water, the students had

to be in!" (A-MMc-291014-intvw). However, as a result of increased efforts, the residences were only *'practically'* complete to allow occupancy of the students.

"Delivery to time was very difficult and very problematic and we delivered it five minutes before the students arrived. Five minutes! The paint was still wet on the walls when the students arrived with their parents, luggage and everything" (A-DL-270314-intvw)

Albeit, in the rush to make the residences accessible there were a number of quality issues and unfinished work that needed to be addressed after the students had taken occupation (A-GC-231214-intvw).

5.2.9.2 Team

Evidence shows that that the pressures to complete the student residencies in time for the student arriving influenced the behaviour of actors and increased tensions within the TMO.

"It obviously increased pressure. If you're trying to push people to complete by a date that they perceive as tight, there is obviously a knock-on impact on relationships and stress levels" (A-GC-231214-intvw)

As a result of the stress in meeting the deadline, together with the additional pressures imposed by the client, behaviours between actors became defensive, with reports of *"silos"* emerging between TMO members (A-MC-190314-intvw). There were also reports of communication breakdowns and a lack of integration from actors within the same PMC organisation.

"You know they would all travel down to the same meetings from the same office but they would travel in separate cars and all take different routes" (A-PW-071014-intvw)

Consequently, a culture of blame developed within the TMO, whereby the PMC was attributing slippages in the schedule to the difficulty the MC was having in securing sub-contractors. Conversely, the MC blamed the delays in construction on inaccurate and incomplete design information being issued by the PMC design team.

"...the contractor was trying to pick out items of errors in design ...the designers who are saying "that's not ours, it's your fault" ...both sides were defensive rather than proactive" (A-MC-190314-intvw)

As a result of pressure on actors to complete the project, the TMO became fragmented, rather than integrated, resulting in a poor working environment.

There was also evidence that the remote location of TMO actors caused difficulties. Whereas, most of the PMC were based in the same office, the structural engineer was based in a regional office, two hundred and fifty miles from the site

"They had that slightly dismissive attitude...I don't think they really appreciated that what we thought we'd been asked to deliver... so there was a bit of conflict there" (A-PW-071014-intvw).

As a result of the lack of integration between the structural engineer and the remaining members of the design team, the steelwork needed to be redesigned, causing further delays in the project and greater tensions between the TMO member organisations.

5.2.9.3 Task

The interpretation of the design requirements was also a cause tension for the architect in defending the alignment of the strategic directive to provide a high quality residency.

"We weren't asked to deliver a Premier Inn because we could have done that...for half the time and for half the cost, but that's not what we were asked to do, but I think that kind of got lost...so it became a bit of an issue" (A-PW-071014-intvw)

Whereas, the traditional masonry construction and pre-cast construction aligned with the Department of Estates maintenance strategy, in terms of robustness and low maintenance, it was a slower form of construction (A-BG-131014b-intvw) and dependent on the availability of bricklayers, as previously discussed (A-MMc-291014-intvw).

The PM was critical also of the design, in that was too "bespoke" and lacked "repetitive elements"

"...which means that you've got no economies of scale. You've got no opportunity to find slack in the programme with repetitive detail and repetitive forms of construction" (A-DW-170315b-intvw)

Despite the architect's perception of the client requirements to build a unique, high quality student residency, the evidence suggests that the remaining TMO actors were of the opinion that the design of facility was a factor in the difficulty of achieving the deadline.

5.2.9.4 Transition

As previously in Section 5.2.1, the new residences were originally scheduled for handover to the Department of Hospitality and Catering on 1st July 2011, in order to allow sufficient time for the necessary preparations required for student to take occupancy.

"When it became 8th September, with students arriving the following day, yes then we just thought "everything has gone to heck in a bucket, get on with it" (A-TD-240314-intvw)

Because of the late delivery of the rooms, there was inadequate time for inspection and efficient transition between handover and occupancy. These resulted in complaints from the new student residents that items had not been completed adequately or were unfinished (A-GC-231214-intvw). There were also reports that, in the rush to fix the outstanding items, additional damage was caused to finishes by sub-contractors, thus damaging relations further (A-PMc-101214-intvw).

5.2.10 Project success

The first measure of project success across the majority of participants was on delivery of the project within schedule and within the allocated budget. It was generally accepted that "*it wasn't exactly on time, but the objectives to have the students in place for the term start in 2012 was achieved. The September intake target was met*" (A-BG-240314a-intvw). It was also conceded "*the budget aspirations weren't achieved*" (A-BG-240314a-intvw).

From the PMC and MC perspective, an important measure of project success was client satisfaction (A-MC-190314-intvw). Through client satisfaction, the MC would be able to apply for further work with the university (A-MMc-291014-intvw), and enable the PMC to enhance their portfolio in the Higher Education Sector (A-BG-240314a-intvw).

Besides the efficiencies in project delivery, the main measure of project success, from the client perspective, was the end-user satisfaction. Through positive student feedback, the university would maintain its reputation. Despite the delays and the initial complaints, the new facility was well received and the feedback from the students was *"positive"* (A-GC-231214-intvw).

"Personally... I think it's a great success, a lot of people have got caught up in thinking it wasn't because of the financial impact but, longer term, people will *forget about that and think we have an amazing high quality building.* " (A-TD-240314-intvw)

The Director of Hospitality and Catering also reported that from a financial perspective "*It sold out quickly, which was nice to know that people would happily pay for quality. So, on that grounds, yes it was a success*" (A-TD-240314-intvw).

Therefore, despite being unsuccessful in achieving the short-term success criteria, the project was successful in achieving the long-term success criteria for the client. However, as a result of the delays and cost overruns, the MC and PMC were not invited to participate in future work with the university.




5.2.11 General findings from Case A

Report on the general findings of Case A are structured around the research questions discussed in Section 3.5. Findings are illustrated by a causal network diagram shown in Figure 5.3 that show the relationships between events and actions within the case study.

5.2.11.1 Organisational strategy

The policy level strategic objective of the client was to safeguard the reputation of the university and maintain its competitive position, which had been threatened by the deteriorating condition of existing residencies. Construction of high quality residences provided strategic opportunities at the business level to grow student numbers through the enhancement of the student experience; reduce the maintenance of the university estate through nomination of specified suppliers; and make contribution to the universities financial position through higher rents from the accommodation. There was evidence that client complexity has a negative effect on the project, with differing strategic objectives being communicated to the TMO. This was alleviated through the appointment of a project manager to act a single point of contact between the client and contracted parties.

As illustrated in Figure 5.3, through alignment with the client strategic objectives and participation in the PMC could gain entry into the Higher Education Sector, which was one of the few buoyant sectors at the height of the financial recession. Similarly, through delivery of a successful project the MC could seek further work with client as part of the universities infrastructure programme. However, as a result of the delays and disputes over claims, the relationship between the client and TMO organisations were poor. As a consequence, the strategic objectives of the PMC and MC were not achieved.

5.2.11.2 Mechanisms to maintain alignment

Findings suggest that formal mechanisms to ensure that the alignment of strategic objectives was maintained were not effective. The established governance structure within the university did not align with the schedule required to achieve the project objectives. Approvals of the tender application was required by the Court Committee of the university, which only met four times a year. In order to maintain the programme for the completion date, tender documents were, therefore, advertised with incomplete design information.

The traditional procurement strategy adopted by the university enabled the MC to submit an unrealistically low tender price and make claims for additional works that were not included in the tender documents. However, as a result of the financial crisis and incomplete design information, the MC was also unable to secure sub-contractors with the supply chain to carry out works at the previously estimated low cost. Findings also suggest that the structure of the TMO did not fit with procurement strategy. All consultants were employed by the PMC, and roles and authority were unclear.

5.2.11.3 TMO strategic behaviour

The delays in achieving the immovable deadline had a significant impact on the strategic behaviours of the TMO. This was also influenced by the client leadership behaviours towards realising the strategic objectives of the project. As a consequence of factors causing delays and additional pressures imposed by the client project management, undue stress and fragmentation within the TMO began to occur, with a loss of trust and a blame culture between TMO member organisations evolving. The evidence finds a lack of integration between TMO members, but also between actors within the same organisation.

The findings also suggest a strong task focus. This was particularly evident where team members were not co-located and with sub-contractors, who were only partially connected to the TMO. In general, the focus of actors was on the completion of the individual tasks that they would, individually, be measured on.

5.2.11.4 Project success

During the project and on completion, success of the project was measured on the shortterm project management objective of completing the facility, within the allocated budget and in time for the students arriving. Quality of the complete facility was also a success criterion, but notably, this was raised by only a small number of participants. The MC and PMC were aware that their performances were measured on the time, cost and quality these criteria. It was through, client satisfaction and the successful delivery of the project objectives that the strategic objectives of the contracting organisations would be achieved. At the client business level, success of the project was measured on end-user satisfaction. Through student satisfaction, the university would main its reputation, and support the strategy for growth through the provision of a more attractive campus. The Department of Hospitality and Catering also measured success on the ability to, comfortably, charge higher-level rental, thus contributing to the universities financial strategy. However, these measures of success could only be ascertained a period after completion of the project.

From a commercial perspective, the MC reported financial losses and significant management time dedicated to recouping costs that "*could have spent chasing other work and doing other work*" (A-DL-270314-intvw). As, neither, the MC nor PMC achieved their longer-term strategic objectives of attracting further business through participation in the project, the project was not considered successful for the TMO organisation.

5.3 Case B – New Four-lane Carriageway

Case study B involves the construction of the second phase of a four-lane carriageway, distributer road, routed through the east end of a major UK city, commissioned by the Local Authority of the City Council. A holistic overview of the design and construction phase is provided in the Section 5.3.1, with discussions to answer the research questions of the study.

5.3.1 Background to the case

The new carriageway formed part of a major regeneration strategy, which was originally developed in 1965 to connect new areas of the city that were being developed for the relocation of residents living in deprived neighborhoods. During that time, migration was needed to alleviate overcrowding and the deterioration of living conditions after the Second World War (Kirkpatrick, 1965). However, due to expenditure on the construction of motorway links to other towns in the country, and the typical five-year cycle of the serving local government, works on the new carriageway did not commence until 2010 (B-GO-0915-email).

The catalyst for investment of the new carriageway came in 2007 when the city won a bid to host a major, international athletics event (the Games) to be held in 2014 (B-LES-0911-report). As part of the application process, a commitment was made to invest \pounds 1.25 billion in preparation for the Games, which included the construction of the new carriageway. (B-GCG-0814-report). Due to the overall length of road, at 5.3km, it was planned that the complete carriageway would be constructed in three phases. The first phase, stretching 1.5km, was completed in April 2010. Construction of the second phase commenced in May 2010. This phase was the longest stretch of carriageway at 2.4km and included the incorporation of full-length cycle-ways, the widening of local footpaths and the construction of a relief tunnel for flood alleviation (B-AH-0713-doc). Construction of Phase 2 was complete on 26th April 2012, one week ahead of schedule. Approval for Phase 3 was granted in January 2016. A time-ordered display of key events for are summarised in Table 5.4.

2007		2009	2010	2012
•	City wins bid to host major sporting event in 2014	• March. Phase 1 Construction commenced on site	• January. Design approved by the City Council	• April 2012. Completion of Phase 2
		• April. Specimen design for Phase 2	 February. Design and Build Contractor appointed 	
		• May. Advertised for tender under OJEU procedure	• April. Completion of Phase 1	
		• November. Tender Return	May. Phase 2 Construction Commenced on site	

Table 5.4: Case B: Summary of key events

On completion of Phase 1 in April 2009, the Local Authority technical team produced a specimen design for Phase 2, which formed part of the tender documentation (B-GO-200814-intvw). Following the formal process of advertisement in the Official Journal of the European Union (OJEU), as set out in the Directives 2004/18/EC of the European Parliament on the coordination of procedures for the award of public works contracts, a joint main contractor (JMC) was appointed in February 2010 to provide design and construction of the second phase of new carriageway.

Figure 5.4 illustrates of the structure of the TMO created to realise the client organisation's strategic objectives. The TMO consisted of three member organisations. These are identified as the client organisation (Local Authority), the Joint Main Contractor (JMC) and Design Engineers (DE). Within the case the JMC was formed as a joint venture between two contractors. For the purpose of the case study, these are identified as Joint Venture Contractor 1 (JVC1) and Joint Venture Contractor 2 (JVC2). Within the conditions of the design and build contract the DE was employed by the JMC to supply design services for the project.



Figure 5.4: Case Study B - TMO Structure

5.3.2 Client organisation's strategic objectives

The primary strategic objective for construction of the new carriageway was to facilitate a major regeneration initiative in the east end area of the city. As stated in the Local Authority Business Plan (B-CGB-0608-report), the vision for regeneration was supported by three strategic objectives:

- 1. *'Sustainable place transformation'*, which focuses on the overall infrastructure and environment of the area. This involved improving the quality of the built and physical environment, and the provision of increased housing choices.
- 2. '*Increased economic activity*', which focused on attracting employers to locate to the area, thus increasing employment opportunities for local residents.
- 3. '*Develop community capacity*', which considers the long-term investment in into the community, through focus on education, health and increased community participation.

Construction of the new carriageway and associated works made contribution to all three strategic objectives. The strategy for sustainable transformation was supported through improvement of the physical infrastructure, which included construction of a relief tunnel for flood alleviation. Economic activity was stimulated through the attraction of businesses to the area and the subsequent increased opportunity for local employment.

"...so the purpose was to try and stimulate the local economy by encouraging local businesses to come in and set up and, therefore, the road would provide links to the adjacent motorway, to encourage businesses to set up on the route of the new scheme" (B-AH-200814-intvw)

A growth in local business and higher employment opportunities would attract further residents to migrate to the area. Thus, supporting the strategy for community capacity by stemming the loss of the local population and investment.

The short-term strategic objective was to provide access to the sporting facilities that were either being built or upgraded for the purpose of hosting the Games. The project incorporated the provision of link roads to varied arenas and the Athletes Village, in addition to the improvement of existing roads along the route.

"...without that, the arena would have been pretty much landlocked so that drove the timescales because, as I say, that needed to be done before the games" (B-AH-200814-intvw)

Due to the size and international nature of the athletic event, the date of the opening ceremony was determined within the original bid application to host the 2014 Games. To maintain the time-line for preparations for the Games, and maintain the international reputation of the city, it was essential that the carriageway was complete as scheduled.

5.3.3 TMO strategic objectives

Both JV contractors identified the strategic benefits of entering into a joint venture as a TMO member organisation. By combining the skill sets of each contractor they would, jointly, be in a stronger position to successfully tender for the project and deliver the project objectives. JVC1 perceived that by becoming the main contractor for construction of the carriageway would provide the organisation with the relevant experience to successfully tender for projects of a larger size than they would have normally been considered.

"So, it sort of helped to raise our game... we are now able to say that we've done a project of that sort of level. It helps being able to pre-qualify for other projects" (B-LC-111214-intvw)

Both JV contractors also recognised the significant impact the Games would have on the city. International media coverage and publicity would raise the city's profile and the profile of projects associated with the Games. Through participation in the project, TMO member organisations would be able to support marketing initiatives and raise brand awareness across the region (B-IW-220914-intvw).

All TMO member organisations shared the same strategic objective of pursing future contracts with the local authority. Besides being considered a "good employer" (B-LC-111214-intvw), securing more local contracts was a key strategic objective for JVC1. At the time of the study, the employees were working on jobs across the country, but this was incurring additional costs in terms of accommodation, travel and the transportation of owned plant fleet. The Finance Director also expressed concern over the work-life balance of employees.

"So, what we're finding is more and more of our staff are having to stay away on a weekly and regular basis and that becomes quite difficult for people with families, so if we can get more local work it's better for us" (B-LC-111214intvw)

Similarly, participation in the project aligned with the DE strategy of establishing a local presence. As an international organisation, the design engineer had offices throughout Europe, each with multiple disciplines. However, as the main government transport offices were based within the city and "*Transport* [UK] *being one of our crown jewel clients*" (B-SR-010914-intvw), the DE recognised the strategic benefits of developing their transportation design capability locally.

5.3.4 Procurement processes

All respondents were in agreement that the design and build procurement strategy was the most suitable approach to manage the project. Within the contractual arrangements, the DE further developed Technical Services preliminary designs developed for detail design and construction. The main benefit to the client organisation was that risk was transferred from the local authority to the JMC.

"... should the design not fit in terms of the land availability or should there be problems with the design, it's not back on us, the employer" (B-AH-200814-intvw)

The design and build approach also allowed for innovation and potential savings by the contractor (B-AH-200814-intvw). This was because the JMC had greater control of the design, and through a process of *"value engineering"*, the JMC would be able to provide additional environmental, cost and programme benefits (B-JW-230914-intvw). Not only was strategic alignment maintained through focus and innovative practices, but also under the conditions of the contract, any savings were shared between the client and the JMC.

5.3.5 Governance structure

Formal governance processes within the local authority were well established and an accepted part of the operating procedures of city council projects. The local authority established a project board for the purpose of governance. The project board maintained formal responsibility for direction of the project and TMO. Membership consisted of representation from stakeholder organisations involved in the regeneration initiative, and included the local authority, the local Water Authority, a Green Network Partnership created by central government to implement a programme of green space enhancements across cities, and an Urban Regeneration Company established from neighbouring local authorities to represent the community in the area with regards to regeneration, (B-GCC-0312-report).

Independent governance arrangements included inspection by an Operational Delivery Scrutiny Committee who has the responsibility to monitor the operational performance of all local authority services in relation to the policy objectives and performance targets (B-GCC-210915-doc).

5.3.6 Client behaviour

The local authority Technical Services Project Management team (TRPM) were based on site along with the JMC, DE and other sub-contractors. The local authority Resident Engineer was of the opinion that that by having a client presence during the works would ensure that the quality of workmanship was maintained (B-AH-200814-intvw). There were some underlying conflicts over ownership of the design between the JMC and the client. The DE did felt that the TRPM were "over protective" of their preliminary design.

"...there was a little bit of, "It's ours," and not recognising then that the detail was going to be done by someone else so in ... it's almost like, you know,

"Having done a preliminary design, my concept was this but you want to do something else therefore you're wrong"..." (B-SR-010914-intvw).

Under the contractual agreement, the DE was able to *"interpret"* the specification in the tender documents. In some parts, this differed as to what was originally intended by the client, although the final decision would fall with the JMC.

"That's where sometimes as a design and build team, the feeling you get from the Council is that they think they're running it as a standard contract where they're in control of making the decisions in quality, whereas that's not always the case. If you're design and build, you're taking that risk so it's then up to you to argue that case..." (B-IW-220914-intvw)

Despite this, TRPM perceived relations within the TMO to be, generally, positive. There were some discrepancies over design and cost, but problems did not escalate.

"...there will always be tensions in the industry between the contractor and a client, but nothing unusual," (B-AH-200814-intvw).

It was reported that when issues were encountered, TRPM were "...quite happy to come and talk and try and help find acceptable solution" (B-SR-010914-intvw), thereby, enabling a productive working environment to evolve.

5.3.7 Supply Chain

As with Case Study A, the biggest threat to the realisation of strategic objectives was the contractors supply chain, in particular, the utilities organisations providing water, electrical, gas and telecoms. Significantly, these external organisations do not operate within the boundaries of a single TMO, rather they operate across multiple projects. The JV contractors were frustrated that, as most utility organisations are single suppliers, they are able to work to unique timeframes and deadlines that did not align with deadlines for the project.

"...they have no contractual responsibility to install that cable when you want it and be out by the time you need them out. So, you know, two months later you're still waiting and you can't complete that section of road until they've come in and done their works and you're left kind of hanging about." (B-JW-230914intvw)

Due to delays in the completion of utilities works, other sub-contractors could not complete dependant tasks. This resulted in slippage of the project schedule and further frustration for all TMO actors.

5.3.8 TMO Strategy

The project strategy of the TMO was influenced by a number of environmental factors. These included the governance mechanisms, client behaviours, external factors and the varied strategic objectives being pursued through the project. The strategic behaviour of the TMO in delivering the project is explored under the themes of *time, team, task* and *transition*.

5.3.8.1 Time

Although the programme for completion was considered "*challenging for the size*" (B-AH-200814-intvw), all TMO actors recognised the importance of completing the carriageway by the set deadline. Delays in the delivery of the new road would have had implications on dependant projects, including the preparations for the Games

"If you look at the political commitments...being able to deliver to match commitments made by others is usually quite an important thing" (B-SR-010914-intvw)

The deadline was not considered to be a major difficulty for TMO members. Deadlines were considered normal within the construction industry and enabled *"focus"* on the project objectives.

"It's a normal dynamic in our industry. We're used to working to deadlines and pressure and all these sorts ..." (B-IW-220914-intvw)

The carriageway was opened one week before scheduled, as discussed in Section 5.3.1. This was despite a number of interruptions at the early stages of the project. Minutes from progress meeting reported a thirty-nine day delay at the start of the construction stage, as a result of late design information (B-LES-120910-report). The programme was delayed for a further four months due to the discovery of previously unknown underground services (B-LES-240211-report). This was exacerbated by the lack of historic services information and the slow reaction from utility companies to remove the obsolete services (B-JW-230914-intvw).

Emergent actions to manage the programme and realise the deadline included extended working hours and twenty-four hour site operations. The JMC also paid suppliers additional sums to expedite the delivery of concrete beams that could not be delivered in time to match the schedule.

"So, ultimately we said, "Well, it's going to cost us fifty grand in delays if we wait the three months so we'll pay you a couple of grand for your guys to work

next weekend and get your batching plant working and we can move things on"..."

Through the proactive behaviour of the JMC, the deadline was comfortably achieved and dependant projects, including the Games, were able to progress as scheduled.

5.3.8.2 Team

It was considered that the differentiation within the TMO was a contributing factor to the success of the project. As discussed in Section 5.3.3, the JMC was created as a joint venture of organisations with "*complementary sets of skills*" (B-JW-230914-intvw). JVC1 had specialist expertise in earthworks, road works and remediation, whereas, JVC2 had specialist expertise in structures and tunnelling. Although both contractors had different operating procedures, a learning environment evolved where both contractors gained experience through working as joint venture (B-JW-230914-intvw).

Relations within the TMO were positive, with actors reporting strong integration and support between team members. Early agreements and understandings were made between TMO actors that they would need to work to together to manage the complexities of the project.

"There was a harmony, there was positivity about everyone pushing in the right direction and I think that really contributed to an overall successful outcome for everyone" (B-AH-200814-intvw)

The positive internal environment was attributed to the co-location of TMO actors who were all based in the same site accommodation for the duration of the project (B-GO-200814-intvw). This not only enabled daily interaction with fellow TMO members, but also allowed for any arising issues to be dealt with quickly and efficiently, through rapid site inspection and appropriate solutions being implemented.

"...if people know each other on a much more personal level they're much more likely to work together on a team basis and resolve matters on a much more informal basis than necessarily going through formal channels and, you know, it allows relationships to develop" (B-JW-230914-intvw)

As a result of the strong integration, TMO actors reported of job satisfaction and positive social interaction.

"...we were a great team involved, going great, you know it really was an enjoyable period of time; busy, but enjoyable." (B-GO-200814-intvw).

There was also evidence of social networks within the TMO. Senior management of both JV contractors were Board Members of the Civil Engineering Contractors Association (CICA) and had worked together on previous projects, although not as a joint venture. Given the nature of the construction industry, actors, reported that they did expected to work together at some point in the future. Moreover, as TMO members operate in the *"same pool"* contractors and consultants often recommend or employ each other on other construction projects to ensure continuity and good working relations (B-JW-230914-intvw).

5.3.8.3 Task

Evidence suggests that the complexity of the task supported the integration of the TMO. The heritage of industrial activity, contamination and unknown ground conditions resulted in a high risk of unfavourable issues occurring on the site (B-SR-010914-intvw). It was therefore recognised at the early stages of the project that all TMO members would *"share the responsibilities and act as a team to try and help each other"* (B-AH-200814-intvw), in order to achieve the project objectives.

This involved integration with other projects contributing to the regeneration programme. Evidence suggest that there was strong interaction between the client, the DE and MC. Rather than separate teams, a matrix structure was adopted "...to make sure that things worked" (B-SR-010914-intvw). This also supported the implementation of innovative practices. In particular those applied to the tunnelling and flood alleviation, which was highly complex.

5.3.8.4 Transition

As illustrated in Table 5.3, the carriageway was completed as planned. This enabled associate works for the Games to commence, including access to venues. The Water Authority was also able to implement the flood alleviation system that was constructed as part of the project. Furthermore, through the completion of the carriageway, the Local Authority Highways were able to direct the volumes of traffic from the minor routes to "…a main arterial route, which was, ultimately, a lot more direct and built, suitable for purpose" (B-JW-230914-intvw).

In general, construction of the second phase of the carriageway successfully facilitated the initiatives to regenerate of the area, with no reported issues in the transition.

"[URC] are using it to promote redevelopment in the East End. So, post road opening, there has been a lot of good, I think, coming from it." (B-AH-200814-intvw)

Increased access has enabled the relocation of organisations and business to the east end of the city and stimulation of the local economy. New businesses have since located to the area and there is evidence of increased economic activity (B-AH-200814-intvw).

5.3.9 Project Success

Sources cited the delivery of the project being on time and within budget as the primary measurement of project success. Completing one week ahead of the deadline was viewed as a particular achievement, especially due to the complexity of tasks required for the completion of the project (B-AH-200814-intvw). Although, the deadline was the biggest driver (B-IW-220914-intvw), there was *"heavy penalties"* for delays (B-AH-200814-intvw). Furthermore, due to the high profile of the project, meeting the deadline was of political importance to the client and late delivery would have had an impact on dependent projects, in particular preparation for the Games.

Although client satisfaction was cited as a success measure, there was more emphasis on the wider stakeholder satisfaction and the impact the new carriageway had on the regeneration initiative. Positive feedback received from the local community towards the opening of the road, and the social impact it made to the regeneration area was an important success criterion across the TMO.

"The other thing that I really took a lot from was ...the comments that they made about how it opened up the area and gave it a new lease on life" (B-SR-010914-intvw)

The project was also awarded a national Saltire Civil Engineering Award for demonstrating value to the wider community, environmental improvement, and providing a catalyst for further economic development in the area (B-JV1-293015-web).

"I think that one of the major successes was getting the Saltire Commendation and the recognition that that brought with it" (B-SR-010914-intvw)

The prestige of the commendation supported the marketing initiatives all TMO actors, in terms of being considered for future contracts and developing local reputation.

Despite this, the JMC did not achieve the their financial objects through participation in the project. As a consequence of the 2008-2013 UK financial crises, and uncertain

market conditions, the JMC bid application was significantly lower than the Local Authorities estimate for the phase of work.

"I would say that when they had set their budget, the market was totally different, so I would say their budget level was high and they actually got the job for outstanding value for money" (B-IW-220914-intvw)

This did impact on the profit the JMC expected to receive from participation in the project, although, as a result of the design and build procurement strategy, the client did benefit from a lower than anticipated budget.





5.3.10 General findings from Case B

Report on the general findings of Case B are structured around the research questions discussed in Section 3.5. Findings are illustrated by a causal network diagram shown in Figure 5.5 that show the relationships between events and actions within the case study.

5.3.10.1 Organisational strategy

The policy level strategic objective of the client was to make contribution to the regeneration programme for the east end of the city. Construction of the road would increase access for new business and stimulate economic activity the area.

"It's facilitated regeneration, along with some other strategic investment that they've made ... In fact if you walk down there, compared to five years ago when I first got involved, it's a completely different place" (B-SR-010914-intvw)

As a result of the construction, regeneration targets were achieved. Businesses had established in the area and the construction of new housing saw people relocating to the regenerated area. Stakeholders also benefited from component parts of the project, such as the flood alleviation system, which has a significant impact on locals that had previously suffered from flooding of local waterways (B-JW-230914-intvw). Also, the local football team benefited from the road. By giving access to a new coach park, the police were able to disperse crowds after a game much more quickly (B-GO-200814-intvw). Access to the Games venues was also created, including paths, walkways and new transportation routes to support visitors to the city.

The success of the carriageway enabled the TMO strategies to be realised. The positive stakeholder satisfaction increased the reputation of the TMO members and led to the commendation of awards. Through participation, both organisations within the JMC and DE were able to increase their profile to attract future business. Although, profits of JMC was not as expected, they were *"reasonably happy"* with the final account agreed for the work done (B-LC-111214-intvw).

5.3.11 Mechanisms to maintain alignment

The established governance structure was the formal mechanism implemented to maintain alignment of strategic objectives. This was conducted at the higher level of the client's organisational hierarchy. Directives to the TMO were made through the TRPM as a single source of communication. This avoided issues of client complexity or multiple directives, as evident in Case Study A. It was considered that the co-location

of client with the TMO, enabled decisions to be made much faster and, with the removal of formal administrative channels, led to a more informal and productive environment.

The design and build procurement strategy enabled innovative practices and decisions to be made. There was evidence of the client seeking to maintain control of the design, which caused frustrations for the designers. As under the contractual conditions, design responsibility is transferred to the JMC.

5.3.11.1 TMO strategic behaviour

There was no evidence to suggest that the fixed deadline had an impact on the behaviour of the TMO or the alignment of strategic objectives. This was despite the complexity of the task and the international media attention the Games received. Deadlines were considered normal within the construction industry and managed accordingly.

Figure 5.5 shows how positive client behaviour had an influence on the behaviour of the TMO team. From the start, the TSPM set out to establish an environment of mutual understanding and shared responsibly. This led to efficient integration between TMO members and focus on the success of the project. Evidence suggests that the co-location of the TMO was a contributing factor in creating an effective environment for integration. With the client, designers and contractors all working in the same building, the TMO was able to provide efficient and rapid solutions to emerging project issues. Co-location also improved communication channels between actors, thus, enabling a greater awareness of the project activities amongst the TMO and the contribution of specific tasks towards the project objectives.

There was also evidence of strong social interactions. Participants reported job satisfaction and enjoyment, as a result of participation in the TMO. Reports find that as a result of good relations within the TMO, friendships continued to evolve and two years after completion of the project many of the team members were still interacting socially.

5.3.11.2 Project Success

During the project, success was on completion of the project by the deadline. Completing the project one week before the deadline early was cited as a significant achievement. However, due to the complexity, overcoming technical difficulties was also considered the key success driver (B-IW-220914-intvw). Success was hindered by the delays caused by utility suppliers, who were not members of the TMO. These organisations were highly task focused and concentrated only on their specific activities, within timeframes of their own organisational objectives.

On completion of the project, the main success criterion was stakeholder satisfaction and the positive feedback on the road received from members of the local residents. In particular, the contribution the carriageway made to the local community. This was acknowledged through the recommendation and receipt of awards for the positive impact on the wider community and environment. All TMO members, including the client, recognised the significance of this, for their reputation and future business success.

5.4 Case C – New Laboratory and Office Facility

Case study C involves the refurbishment and fit-out of a new-build commercial office unit for laboratory and office use, commissioned by a public sector Environmental Protection Agency (EPA). A holistic overview of the design and construction phase is provided in the Section 5.4.1, with discussions to answer the research questions of the study.

5.4.1 Background to the case

The new laboratory and office accommodation formed part of a three-year estates strategy "...to be smaller, more flexible, more responsive and more innovative" (C-EPA0309-report), which involved the relocation of staff from five existing laboratories and offices, dispersed across three geographical locations, into one central facility. Thereby, reducing running costs and increasing efficiencies.

Following submission of a business case to secure government funding for the rationalisation of the existing research and administration facilities, a 20-year lease was agreed to secure a strategically located building within a recently constructed business park. A TMO was formed to manage the required structural alterations of the new building, design and fit-out for laboratories and the offices, and installation of specialist equipment and furniture. Meanwhile, a temporary organisation was created, comprising of internal personnel, to manage the relocation of staff from existing offices and laboratories. A time-ordered display of key events is summarised in Table 5.5 and discussed within the case study.

	2009	2011	2012	2013
•	March. EPA Estates Strategy	• January. PMC led design team appointed to provide preliminary designs	• August. Advertised for tender under OJUE procedures	• July. Formal handover of facility to EPA
•	August. PMC appointed to support production of the business case.	• July. EPA secure lease 20 year lease on building	 November. Main Contractor appointed to provide design and build services. November. Construction commenced on site 	 Conclusion of migration and Occupation of facility by EPA

Table 5.5: Case C: Summary of key events

Following the formal process of advertising in the Official Journal of the European Union (OJEU), a project management consultant (PMC) led, design team was appointed to conduct a feasibility study and preliminary designs with the scientists key stakeholder departments for the layout of laboratories. Once approval for facility was agreed by Central Government in July 2011, a twenty-year lease was agreed on the new building.

The invitation for tender to execute the fit-out was advertised within OJEU on 11th August 2012. The main contractor (MC) was appointed on 2nd November 2012 to provide design and construction of the new facility along with specific members of the design team. Construction on site began on 26th November 2012 and was completed for occupation on 5th July 2013, as scheduled. Migration of existing personnel to the new facility commenced on 8th July.



Figure 5.6: Case study C - TMO Structure

The structure created to manage the fit-out of the new facility is illustrated in Figure 5.6. As shown, the TMO consisted of three member organisations. These are identified as the client organisation (EPA), the project management consultant organisation (PMC),

and the design and build, contracting organisation (MC). Within the case, the services of the architect, structural engineer and mechanical and electrical engineer (M&E) were formally contracted to the MC organisation. Similarly, construction design and management coordination (CDMC) and the assessor for the Building Research Establishment Environmental Assessment Methodology (BREEAM) were contracted to the PMC, who already had cost management (QS) capabilities within the organisation. Contractual and governance process are discussed in Sections 5.4.5 and 5.4.6.

5.4.2 Client strategic objectives

The estates strategy made contribution to the primary objective of the EPA 2009 strategic change programme to *"improve service delivery while reducing operating costs"* (C-EPA0913-report). This was realised through the following strategic objectives (C-EPA0309-report):

- 1. Cost efficiency through a reduction in operating costs as a result of funding shortfalls.
- 2. Improved working practices and processes through efficient deployment of resources and the development of a culture of inter-disciplinary problem solving.
- 3. Provide a good working environment that encourages and supports synergies between teams, functions and services.
- 4. Reduction in carbon emissions through the acquirement of new-build facilities with a BREEAM Excellent rating and an 'A' rated energy performance certificate

The strategic objective to achieve cost efficiencies was achieved through the rationalisation of the EPA estate and the sharing of equipment and facilities across existing departments.

"... it's not just about sharing services with laboratories; it's about sharing other accommodation with other people. If we begin to share other accommodation, we then maybe begin to share our admin, our meeting rooms, our IS infrastructure ... that's the longer term strategy" (C-LMc-100414-intvw)

Rationalisation also provided an opportunity to review current working practices through greater integration between the specialist disciplines that would be brought together within the single facility. By the provision of *"state-of-the-art"* facilities to support operations (C-NA040414-intvw), EPA would be in a position to upgrade current capabilities and realise further business opportunities, through the creation of synergies across functions (C-PH100414-intvw).

Besides being a funding requirement of Central Government, attainment of an 'Excellent' BREEAM rating and the reduction in carbon emissions also contributed to the cost efficiency strategy, through the reduction in energy costs and consumption, by operation of a sustainable building.

"...it considers a lot of elements and that benefits [EPA] in that they have a working environment that people enjoy and are comfortable in and there is staff retention and sickness levels should be lower ..."(C-FW190315-intvw)

Pursuit of an *'Excellent'* BREEAM rating also influenced the design and construction to provide a healthy working environment that catered for the wellbeing of the occupants (C-FW190315-intvw).

5.4.3 TMO strategic objectives

Both the architect and PMC had provided services for EPA on previous projects and recognised the benefit of retaining *"high profile"* clients for future work.

"...we pride ourselves on getting repeat business because, obviously, it demonstrates that we're providing a good service and that clients want to come back to us" (C-DJ121214-intvw)

Besides the prominence of EPA, the type and complexity of the facility also enhanced the architects project portfolio of "*developing environmental buildings*" (C-BR170315-intvw). By gaining further experience in laboratory projects, the PMC would be able to improve the organisation and individual curriculum vitae to attract further work within the science industry (C-DJ121214-intvw).

The MC also recognised the strategic benefit of being employed by EPA. Whereas, they had significant experience in the construction of commercial office developments, the unique services provided by the client organisation and the specialist laboratory fit-out would augment their experience and enable them to tender for similar specialist contracts (C-NS281114-intvw). Furthermore, as the main contractor for the business park, where the new facility was located, they benefited from being invited to tender for all fit-outs from future tenants considering relocation.

"We're not necessarily going to get everything our own way but we see that as an advertisement for being "okay, we built it" and we wouldn't like another contractor to come in and sort of, for wont of a better word, "muscle-in" on something that we want to keep within our grasp" (C-NS281114-intvw). By participating within the TMO, the MC would not only be in a better position to compete within the market sector, but would also protect its role as the first choice of contractor for fit-outs and refurbishments within the business park.

5.4.4 Procurement processes

Respondents were in general agreement that the design and build procurement strategy was the most suitable route for administration of the project. Through the contractual arrangements, EPA was able to maintain strategic alignment by transferring project risk and ensuring cost certainty (C-DJ121214-intvw). However, in order to guarantee that the new facility aligned with the intention for new working practices within the organisation, most of the design elements for the fit-out were developed prior to appointing the design and build contractor.

"...the case here is that they [EPA] wanted the design to be developed to a level that would be traditional contract but they want the contractor to take all the risks. So, they call it a design and build" (C-FW190315-intvw)

The design strategy was maintained by novating the contractual rights and obligations of the architect, structural engineer and mechanical and electrical engineer (M&E) to the MC at the construction stage. The benefit to the project was that there was consistency in the design team from initial consultation with the client through to construction.

"...so the process can be speeded up because you don't have to reacquaint new designers with the whole building" (C-BR170315-intvw)

Moreover, through the use of value engineering, the MC was also able to "give the client a better product or less costly solution" (C-NS281114-intvw), as they had greater influence over design and construction within the design and build procurement method.

5.4.5 Governance Structure

As shown in figure 5.6, EPA established three boards for the purpose of governance. Project Board 1, was responsible for directing the TMO and activities associated with the refurbishment. Membership consisted of the Head of Estates and Facilities (HoEF), facilities management, and representation from the varied scientific disciplines to be relocated to the new facility (C-RA-110414-intvw). Project Board 2 was responsible for the coordination of the staff relocation to the finished facility. Membership of Project Board 2 consisted of the Head of Business Support, Head of Business Development, the

Programme Office Manager, the Finance Director, and an external management consultant (C-LMc-100414-intvw). Both boards reported to an executive programme board consisting of the Chief Executive Officer and members of the Corporate Management Team (C-NA040414-intvw).

Despite having differing objectives, members of client organisation reported that having two project boards for governance caused a number of difficulties. The lack of communication between boards led to varied directives of the project requirements (C-MC100414-intvw) and differing priorities over what needed be done as part of the rationalisation programme (C-NA040414-intvw).

"...the two project boards ... it was difficult to know who was responsible for what ..." (C-LMc-100414-intvw)

The main difficulties were establishing clearly defined boundaries of responsibility between the two boards, and the separation of design and refurbishment of the new facility from the relocation of end users, and the development of new working practices.

"...there was a lot of overlap so some things were just falling down between the two groups. Some things involved duplication of effort because the two groups were doing them" (C-RA-110414-intvw)

The governance arrangement also caused frustration for TMO members, as directives were being instructed through both boards with no clear jurisdiction or single point of authority. Directives were often conflicting, inconsistent and brief (C-RA-110414-intvw). TMO members felt that they were being left with the responsibility of prioritising the requirements of each board and were required to self-govern, in order to ensure the design requirements of the project was maintained (C-BR170315-intvw). This was later resolved through the appointment of a new Head of Estates and Facilities (HoEF) who acted as the project manager and single point of contact between the boards and the TMO.

5.4.6 Client behaviour

There was evidence that the dual structure of the projects boards and the initial absence of a single point of responsibility from within the client organisation was a cause of initial frustration for the TMO.

"If you've got two heads and that person makes a decision, this person may disagree with it a month later or it may take them a month to actually change the decision of that person...". (C-BR170315-intvw)

The appointment of the HoEF prior to the commencement of construction made a significant impact to the project. "*That was where the success really started to happen*" (C-BR170315-intvw). Robust change control processes were implemented by the HoEF to enable a single point of contact between the TMO and the client organisation (C-RA-110414-intvw).

"...the [HoEF] attended pretty much every meeting; they were involved in discussions of any potential issues; they were involved in discussions of the solution and they were pretty quick at making decisions to aid progress. (C-DJ121214-intvw)

Although, the EPS personnel had difficulties in understanding the rationale behind the newly implemented processes, the *"rigour and structure"* in communications and the single point of control and leadership from within the client system, enabled the TMO to focus on the tasks, and efficiently progress with the project (C-BR170315-intvw).

5.4.7 Resistance to change

The greatest difficulty faced by the TMO was from the EPA personnel that, in general, were resistant to the relocation. There were reports of stress, and staff either leaving their jobs or seeking redeployment elsewhere in the organisation as a result of the consolidation plans (C-PH100414-intvw).

"Some people decided ahead of time that they would not be able to cope with driving on the motorway... and left the organisation because of the stress" (C-PH100414-intvw).

Relocated personnel were also resistant to the new way of working and the sharing of desk spaces to create a more flexible environment (C-KMc-110414-intvw). Despite a consultation process with the design team, varied departmental representatives reported frustration that their requests for particular requirements and layouts were not being provided as requested.

"...the feedback from the users was quite often that it was different from what was actually intended in terms of how the room was laid out and where the equipment was positioned" (C-NS281114-intvw).

Although, authorisation for the final layout were made by the project board, based on costs restrictions, capacity and the integration with other departmental needs (C-MC100414-intvw), the general perception of the end users was that the TMO lacked sufficient experience to understand the requirements of each department.

"If we'd done more up front for them to understand plans. If we'd also made sure, technically, that the external people had done this sort of stuff before...obviously, there would be things that people have never done before and that's not a problem in itself. I think it's about owning up to that..." (C-LMc-100414-intvw)

This was despite previous experience of working with EPA and previous science and laboratory experience.

5.4.8 TMO Behaviour

The project strategy of the TMO was influenced by a number of environmental factors. These included the governance mechanisms, client behaviours, resistance to change and the varied strategic objectives being pursued through the project. The strategic behaviour of the TMO in delivering the project is explored under the themes of *time*, *team*, *task* and *transition*.

5.4.8.1 Time

The imposed deadline for completion of the fit-out was driven by two factors. First, EPA was renting properties that were nearing the end of their lease and short-team leases were was not a viable option (C-RT161214-intvw). Secondly, in order to ensue continuity of the operations activities of the business, there was a fixed deadline for the migration of the staff, which was scheduled to commence in conjunction with completion of the fit-out programme (C-LMc-100414-intvw).

Evidence shows some difficulties in the early stages of the project that impacted on the schedule. Firstly, it was reported that the design team were appointed before the brief was finalised and therefore needed *"additional analysis"* for clarification of the requirements (C-BR170315-intvw). There was also a need for additional design and structural alterations within the new laboratory space, which extended the pre-contract stage of the project, consequently, delaying the invitation for tender being advertised

"...if you went at an earlier stage and then you were incurring issues like that during, while on site, that would obviously have had an impact on the programme but we had been through that process and we had a detailed robust design when we went out to tender" (C-DJ121214-intvw)

Despite this, the PMC sought to ensure that the design information was complete at the time of going out to tender, thereby avoiding delays at the post-contract stage and maintaining the programme throughout the construction period.

5.4.8.2 Team

Actors reported of collegiate and positive working relationships within the TMO. There were regular meetings and consistent communications to achieve integrated solutions to project difficulties "...*it was very much a team working solution*" (C-DJ121214-intvw)

A contributing factor to the successful integration of the TMO was that members of the design team had previously worked together on other EPA projects (C-RA-250914aintvw). Not only did this support communications between actors, but also enhanced understanding of the "*programme and commercial implications*" (C-DJ121214-intvw).

"...I think it also helped that there was an understanding, especially with the architects of what they would like to do and what they could do, but they understood that as well...and I've been there before when they get a very precious architect..." (C-RA-250914a-intvw)

Successful integration of the TMO contributed to completing the project by the deadline, and also towards a reduction in costs through the procurement processes. Contrary to end user perceptions, previous working relationships and experience on other EPA laboratories enhanced understanding of the requirements and understanding of the client systems (C-RT161214-intvw).

5.4.8.3 Task

As the building was originally designed for commercial office use, a number of alterations were needed in order to accommodate the construction of the new laboratories. These included risers through the building and structural reinforcement of the precast concrete floors, as previously mentioned in Section 5.4.8.1.

"...there was a lot of what you would call abnormal works...we had to demolish the precast; we had various holes throughout the building, four storeys high which were as long as this room and as wide as this room. So, there are huge health and safety issues there to deal with. (C-NS281114-intvw).

In addition, reinforcement was also required to minimise the "*vibrations*" in the concrete floor as this would have a potential impact on equipment readings, due to the sensitivity of the specialist instruments used by the EPA scientists.

"... fairly late in the day somebody said "I can feel this floor move"...whether they could or couldn't I'm not entirely sure but one of the people that happened to be there was one of the chief scientists who said "our instruments are really sensitive; they can be affected by, literally, somebody walking past when they're taking a measurement"... after that he started talking in a different language... I didn't understand it" (C-RT161214-intvw) The impact of the vibrations to the concrete floors had not been considered at the design or tender stage. This was attributed to the lack of input from EPA scientists, into the initial design and development of the laboratory layouts, and a *"misunderstanding of the level of specialist detail"* required to fulfil the requirements (C-RA-110414-intvw).

5.4.8.4 Transition

The refurbishment was complete in time for the migration of EPA staff to commence.

"We got the keys on 5th, we moved in on 8th. We ironed out some problems with the first lot of people that came in logistically around the move..." (C-LMc-100414-intvw)

The short gap between handover and occupation ensured minimum disruption and continuity of the services provided to EPA clients. However, the consequential stress of EPA personnel prior to, and following, the relocation resulted in a downturn in productivity.

"You know, it must have been like nine months when we went from being a high performance, motivated team to the lowest and then bringing back. And now, six months later, it's starting to get back to that sort of morale and motivation that we had before and we're starting to really motor in terms of delivery again. So you can add that up, 15 months is the impact of the move" (C-PH100414-intvw)

Despite the lack of integration between EPA scientists and the TMO, and the frustration of the end users in the perceived "*difficulty in terms of meeting our requirements*" (C-LMc-100414-intvw), the fit-out achieved the design objectives, in terms of creating an efficient working environment that allowed for greater integration and synergies between services, as proposed in the estates strategy (C-EPA0309-report).

5.4.9 Project Success

During the project, and after completion, success was measured on handover of the facility as scheduled, within budget, and to the specified level of quality. The HoEF reported that project was complete within the original budget and to the deadline, "*in terms of quality, there have been a number of issues but generally minor*" (C-RA-110414-intvw).

Client satisfaction was rated as a high success criterion by TMO member organisations. By achieving client satisfaction the consultant organisations would be in a better position to successfully apply for repeat business with the *"high profile client"* (C-DJ121214-intvw) and attract further contracts in the scientific research sector. The PMC reported that on completion of the facility, the client "was delighted with the end result and the building was functioning as it should have" (C-DJ121214-intvw).

A third measure of success for both the client and TMO member organisations was end user satisfaction. The perception of the EPA personnel presented mixed results. During construction and on completion, end users were frustrated that their individual requirements for working spaces had not been delivered, as expected.

"I think the fact that nobody went ballistic and the systems kept going was a good measure of success, because there are some people in here that could" (C-KMc-110414-intvw)

However, a post occupancy survey was reported as "*pretty good*" and that staff expectations had been met (C-RA-110414-intvw). Much of the frustration could be attributed to the resistance to change, as discussed in Section 5.4.7, which contributed to the initial negative perception of project success.





5.4.10 General Findings from Case C

Report on the general findings of Case C are structured around the research questions discussed in Section 3.5. Findings are illustrated by a causal network diagram shown in Figure 5.7 that shows the relationships between events and actions within the case study.

5.4.10.1 Organisational Strategy

The client policy objectives were made explicit within the EPA 2009 strategic change programme, to improve services while reducing operating costs. This was realised through the estates business level strategy for the rationalisation of EPA premises. By relocating staff from multiple laboratories and offices to a single facility, EPA could make substantial reductions in operating costs, whilst achieving targets for reduce carbon emissions, and creating synergies to enable new working practices to be developed.

However, findings identify conflicts between strategies at the business level. In the first instance there was resistance from personnel from within varied scientific departments, who were reluctant to move to the new relocation and, the consequential, changes in working patterns. Secondly, personnel perceived a lack of alignment between the specific requirements of each scientific department and the strategic objective of integration between specialisms. This resulted in the layouts of laboratories not being delivered, wholly as expected, and discontent expressed by end users.

The estates strategy for rationalisation was understood by the TMO. As illustrated in Figure 5.7, through alignment with the client business level objectives, the PMC and associated consultant organisations would be able to seek repeat business with the existing client, and enhance their market position in the scientific personnel. Whereas, the MC enhance their experience in specialist laboratory fit outs in order to expand their portfolio and attract future business in the sector.

5.4.10.2 Mechanisms to maintain alignment

The success of mechanisms to maintain strategic alignment were mixed. The design and build procurement strategy was successful. Design risk and the risk of delays, as a result of creating a design team, were mitigated through the novation of design team members to the contractor, after the contract was awarded. However, the efficiency of the two projects boards, created for the purpose of governance, suffered as a result of weak internal communications and inherent client complexity. Client behaviours also had an initial impact of the activities of the TMO. As discussed in Section 5.4.6, it was only when a single point of contact between the project boards and the TMO, was appointed, in the form of the HoEF, that the project began to progress productivity. Until this point, TMO members reported of frustration at the varied directives being given from the each board and the need to self-govern.

5.4.10.3 TMO strategic behaviour

There was no evidence to suggest that the fixed deadline had an impact on the behaviour of the TMO or the alignment of strategic objectives. TMO members did not report of any undue pressure as a result of the deadline, instead "*it probably focused the mind a bit*" (C-RT161214-intvw). Besides, additional works that had an impact one the schedule, the project was complete in time for the relocation of staff to commence.

There were also reports of good relations within the TMO, with success of the project being partially attributed to the positive working relations between TMO members (C-DJ121214-intvw). It was reported that prior working relations and previous employment with the client did enable greater understanding of the client strategic objectives and integration of TMO actors (C-BR170315-intvw). The greatest difficulty the TMO faced was the multiple directives being given by the two boards, and the lack of integration from EPA specialists in the design process. Findings suggest this could also be attribute to the composition of the project board. Difficulties were alleviated through the creation of a single point of contact between the boards and the TMO.

5.4.10.4 Project success

During the project and on completion, the general measure of success was on completion and handover of the new facility by the deadline, within budget and to the specified level of quality. As the construction project needed to align with the migration project, it was critical that construction was complete in time, in order to allow efficient relocation of staff and normal business activities to be maintained. It was reported that the deadline and budget targets were achieved, and the transition did progress as planned. TMO members reported of client satisfaction on completion of the project. Achieving client satisfaction was the important criterion in the pursuit of the individual TMO strategic objectives to attract repeat business and enhance market position in the scientific research and laboratory sector. Despite, initial dissatisfaction by EPA personnel and end-users, that request for facilities were not being delivered as expected, there was general satisfaction at the final result, after the transition, and when the building was in use.

5.5 Case D – New Tourist Visitor Centre

Case study D involves the design and construction of a new visitor centre for a battle site of historical significance, owned and managed by a chartable National Conservation Trust (NCT), as part of their estate. A holistic overview of the project is provided in the Section 5.5.1, followed by discussions addressing the research questions of the study.

5.5.1 Background to case

The new visitor centre was built to replace an existing heritage centre that was no longer *"fit for purpose...and did not give the right impression for the trust"* (D-DMc011214intvw). The existing 1960's building had received a number of minimal upgrades over the years. But, over time the exhibition and interpretation of the battle had become *"very poor and tired"* (D-JR241114-intvw) and in need of investment (D-TIC-041114intvw). Following a review of NCT's estate in 2010, funding was received from the Central Government to create a facility in which the core mission was to:

"To create a world class visitor attraction which reflects the battles...exceptional significance in British and European history which will be completed in time for the 700th year anniversary of the battle in 2014" (D-NCT0610-report)

A condition of the Government funding was that the project would be commissioned as a joint venture between NCT and a government executive agency responsible for a number of historic monuments (HEA). The HEA would also act as project managers for the scheme on behalf of NCT. The second condition was that the project would be complete in time for the 700th anniversary of the battle in 2014. Following a design competition, consultants, including interpretation design, architectural design and landscape design, were formally appointed. Appointments coincided with the official launch and public announcement of the project in October 2010.

The scheme design for the installation was complete in May 2011, followed by detail design in December 2011. An application was submitted for additional funding from the Heritage Lottery Fund in April 2011 and awarded in January 2012. Following the formal process of advertisement in the Official Journal of the European Union (OJEU) the Main Contractor (MC) was appointed in May 2012 with construction commencing on site in June of the same year. A time-ordered display of key events is summarised in Table 5.6.

2010	2011	2012	2013	2014
CG grant and	April. Application	January. HLF	October Handover	January.
HEA appointment	for HLF	grant application	of construction	Completion of
		approved		Interpretation fit-
				out
October 2010	May. Approval	Advertisement for	Fit-out of	1 st March. Opened
Appointment of	of Scheme Design	tender under	interpretation	to public
design team.		OJEU procedures		
October 2010	December.	May.	December.	$23^{rd} - 24^{th}$ June.
Project official	Approval of	Appointment of	Handover of	700 th year
launch	Detail Design	Main Contractor	facility to NCT	Anniversary
	-		-	celebrations
		June.		
		Construction		
		works commence		
		on site		

 Table 5.6:
 Case D: Summary of key events

The construction works were complete for handover, four months later than scheduled, at the end of October 2013. As a consequence, the fit-out of the interpretation was also delayed with completion in January 2014. NCT's visitor project team took occupancy of the building for commissioning and training in December 2013. Despite the delays, the visitor centre opened to the public on 1st March 2014, ahead of the 700th anniversary of the battle on 23rd June 2014.


Figure 5.8: Case D TMO Structure

Figure 5.8 illustrates the structure of the TMO created to realise the client organisation's strategic objectives. The TMO consisted of eight organisational members. These are identified as the client organisation, formed as a joint project management team between NCT and HEA, the main contractor (MC), and design consultants consisting of Interpretation Design (ID), the architect, landscape architect, mechanical and electrical engineer (M&E), structural engineer and quantity surveyor (QS). Within the case Computer Generated Imagery (CGI) were sub-contracted to the ID organisation as a condition of Government funding. Similarly, sub-contractors within the construction element of the project formed part of the MC organisation.

5.5.2 Client Strategic Objectives

As stated in the project definition statement, NCT's mission for the visitor centre was to "communicate the local, national and international significance of site to visitors and *to create an enjoyable and accessible learning experience for all*" (D-NCT0610report). This would be realised through the following strategic objectives:

- To improve the visual impact of the site to give it presence in the landscape and assist in attracting additional visitors.
- To conserve and enhance the site in its role in commemorating the battle
- To provide visitors with an experience which immerses them in the chaos, excitement and honour of battle
- To provide a facility which demonstrates NCT's values as a conservative charity
- To raise the profile of the NCT in its role to protect and promote the countries heritage
- To provide a financially sustainable resource with the flexibility to adapt to future change.

The project objective was to conserve the historical battle site and provide a suitable setting in which the story of the battle could be told in a modern and "*dignified manner that was fitting for the battle itself*" (D-DA220616-intvw).

"You've got a national war memorial, the site of an iconic battle ... that's the context that you're dealing with. An incredibly significant part of [the countries] heritage, which is set in trust and demands and justifies investment. It needs to be interpreted adequately and well... and there are huge opportunities in that" (D-TIC-041114-intvw)

The biggest challenge for NCT was that, although the iconic battle had been commemorated for hundreds of years, there were no physical artefacts in which to exhibit. The current exhibition consisted of replicated uniforms and displays that did not *"cater for a 21st Century the audience"* (D-CP271015-intvw). The precise location of the battle also remained unknown. The bore stone, on which the victory flag had been raised, had been chipped away over the years, with only a small fragment remaining. Furthermore, the existing monuments that had been erected over the years to commemorate the battle had begun to erode. As a consequence of the high competition in local tourism visitor numbers to the existing heritage centre were in decline (D-CW011214-intvw).

Construction of the new visitor centre contributed to the strategic objectives of NCT. Through Government and Lottery funding NCT was able to conserve the site and *"bring dignity back to a suite of monuments, making sense of them and improving the* *landscape setting for those monuments*" (D-DMc011214-intvw). It also provided an opportunity for an ambitious and innovative way of presenting the battle site that was considered *"appropriate for its historical significance"* (D-TIC-041114-intvw). This was achieved through the use of 3D technology to tell the story and immerse visitors in the battle itself.

"By creating a really innovative, new experience that's kind of cutting edge the Trust strengthened its position within the heritage sector and it strengthened and refreshed its own reputation amongst members and non-members as being a modern institution, being technologically savvy, knowledgeable about in terms of what we know about IT and heritage interpretation" (D-TIC-041114-intvw).

Through the new facility and the innovative approach to exhibiting, NCT was able to attract and significant number of additional visitors, which included educational visits from schools and overseas tourist groups. This not only provided financial sustainability for the site, but also made financial contribution for future projects. Besides raising the profile NCT, the use of 3D visualisation also achieved the ministerial strategic objectives to showcase the countries digital design technology (D-NCT0610-report). The project was complete in time to commemorate the 700th anniversary of the battle with a two-day celebration promoting national heritage and tourism.

5.5.3 TMO Strategic Objectives

Portfolio enhancement was the main strategic objective for consultants participating in the TMO. The opportunity to demonstrate innovation and creativity through the interpretation would enhance both, the IC and CGI organisations competitive position in a specialist market segment. By taking a unique approach to exhibition design and showcasing their talent through the visitor centre, the IC were of the opinion that they could redefine standards within the visitor centre and museum industry.

"... we created an exemplar within the industry which people are still looking at and they will be in 50 years to come...But, the real test, for us, will be going forward into the future. Will other large scale or medium scale visitor centres be more likely to view digital interpretation as a viable means?" (D-CW011214intvw)

The architect also recognised the strategic benefits of participating in the project.

Besides *"widening the experience"* of the firm within museum industry, positive public relations and the high profile of the project would benefit internal marketing efforts (D-DA220616-intvw). Both the Senior Associate and project manager for the CGI also

recognised the benefit for individual employees working on the project, in terms of enhancement of employee's curriculum vitae.

Similarly, portfolio enhancement was the long-term strategic objective for the MC. The contractor had previously performed works for HEA, and considered that construction of the visitors centre would complement their current and previous portfolio to attract future work.

"...it was the type of work that we had done in the past and we wanted to be able to demonstrate that we were still capable of doing that type of project in the future as well" (D-GMc2808150intvw)

The MC was also aware of the high profile attention the visitors' centre was attracting, and through successful participation in the TMO, the MC could expand their client base in the museum and tourism industry.

5.5.4 Procurement processes

The views of respondents to manage the contractual arrangements under the standard Government Conditions of Contract for general contracting (GC works), traditional procurement strategy, were mixed. Whereas, respondents understood that the GC works was the most common form of contract used for government projects, it was felt that it was only suitable for the build aspect of the visitor centre.

"...the problem is that the [GC works] contract is set up, brilliantly, for works like concrete bricks, foundations, steelworks, even shop fitting, things like that, but it doesn't contain any of the courses that are needed when you are dealing with less tangible deliverables like media, digital works, and it doesn't cover any of that intellectual property" (D-CW011214-intvw)

The interpretation aspect of the project was, therefore, managed under the Design and Build GC works conditions of contract. However, in contrast to normal practice, where a preliminary design may be developed prior to advertising for tender, and finalised by a design and build contractor, who has responsibility for the design and build, within digital media the design is usually developed in conjunction with a specialist contractor employed to undertake the build. As advised by the lead designer "...you don't complete a finished museum design for a tender, you can't" (D-CW011214-intvw), instead the IC submitted a "design response" and ideas for the interpretation under normal conditions of competitive tendering.

Under the conditions of Government funding, CGI were nominated as a sub-contractor to ID, to build and install the interpretation. The benefit of the contractual arrangement was that the client was able to draw on the creative expertise of IC to design the interpretation, and the technical expertise of CGI for development and installation. Thereby, realising the strategy for an innovative 3D immersive experience (D-JR241114-intvw).

5.5.5 Governance processes

The formal governance process for the new visitor centre involved the creation of a Strategic Board to represent the interests of Central Government, as the main funders, and the interest of NCT as the owners and operators. As shown in figure 5.8, membership included Chief Executives, Directors of Finance and project sponsors from both the Trust and HEA (D-HEA0613-report). The Strategic Board was responsible for reporting progress of the project to the Government Public Audit Committee and the Department of Culture and External Affairs. A representative from HLF was also in attendance to monitor spending of funding.

Although HEA and NCT had collaborated on previous projects, the visitor centre was the first project that had been managed as a joint venture between the two public sector organisations. It was perceived that the decision for the joint venture was political.

"Government ministers really want to push collaborative working across the heritage sector...we were able to sort of wave the flag and say this is the biggest collaboration ever between the two biggest heritage organisations in the country working happily together. Which, a lot of people were very sceptical about, actually, but it worked incredibly well" (D-JR241114-intvw)

The differences in organisational culture between the government agency and the charity did raise initial concerns regarding the collaboration and alignment of the varied stakeholders strategic objectives. In order to ensure the interest of both organisations were represented, a memorandum of agreement was drawn between the two heritage organisations, which also included the legal agreements for management of the project (D-JR241114-intvw).

Despite the Governments conditions, project management responsibilities were shared between the HEA Project Manager and the NCT Project Director, who jointly reported project, related issues to the Strategic Board. NCT also provided expertise to guide the design of the interpretation, which would be managed by a NCT's visitor project team when complete.

5.5.6 Client Behaviour

Despite the perceived challenges of the partnership arrangement between HEA and NCT, participants reported of strong leadership by the project managers of both HEA and NCT.

"There were always checks and balances. Nobody ever didn't know what was going on and there was proper time taken to ensure that things were not running over budget that things were running to time and if any dangers were flagged up then they were dealt with" (D-CP271015-intvw)

There were regular and inclusive meetings with all TMO actors, through which issues pertaining to the project would be openly discussed. This enabled information and decisions to be efficiently communicated to the varied levels within each TMO member organisation (D-DA220616-intvw).

At the time of construction the country was also preparing for a referendum on independence, for which the battle was interpreted, by the media, as a significant event for current government.

"Yes, everyone was on their toes. Very exposed as well, it wasn't like we were hidden away...it was a very transparent project. It was very public contract" (D-JR241114-intvw)

In contrast to the media perceptions, the HEA project manager attempted to "*depoliticise*" project activities (D-DA220616-intvw). It was felt that neither HEA nor NCT could be seen to be taking a political position within the referendum debate. Therefore, in order to avoid any impact on project progress, actions were put in place to prevent the media from contacting TMO members and causing any disruptions.

"There were political sensitivities... yes, but not directly. I think to be fair to [the PM] did protect us a wee bit from that, and God knows what they had to go through. There were meetings that they would have had to go to that I wouldn't ever thank you to go to. I like what I do. I couldn't do what they do. I'm not capable nor want to even if I was" (D-DA220616-intvw)

TMO actors were aware of the impact the project had on the current political environment and accepted the lack of flexibility in the programme. Therefore, where possible, support was given, in terms of "*extra money and the slight extension of the deadline*" (D-PC150415-intvw) in order to open in time for the 700th anniversary.

5.5.7 TMO Strategy

The project strategy of the TMO was influenced by a number of environmental factors. These included governance mechanisms, the procurement strategy, client leadership, the political environment, and the varied strategic objectives being pursued through the project. The strategic behaviour of the TMO in delivering the project is explored under the themes of *time, team, task* and *transition*.

5.5.7.1 Time

As discussed in Section 5.5.1, the driver for the published deadline was the 700^{th} anniversary of the battle, which was scheduled for commemoration by an official opening of the visitor centre on 23^{rd} and 24^{th} July 2014. Moreover, Central Government had stated in their manifesto that the project would be complete by the deadline (D-CG0511-report)

"It would have been a bomb going off if this project didn't complete on time... This project could not afford to fail under any circumstances!" (D-PC150415intvw)

Despite the pressure of achieving the immovable deadline, participants agreed that a fixed deadline was a necessary and positive attribute in the management of the project.

"Well, it's a double edged sword, isn't it? ... it focuses people. It did mean that decisions got made whereas I think if there was a slightly weaker deadline in terms of it being fixed there would have been more pressure" (D-CW011214-intvw)

As reported by the HEA project manager, "you can't argue with a deadline, you can't, everyone knows they've got to meet it" (D-JR241114-intvw). Through the implementation of a robust "change control and key issues reporting system" (D-DMc011214-intvw) the HEA project manager was able to manage excessive delays by keeping all TMO members informed and taking action when necessary. This included overlapping activities and making necessary amendments to the project schedule.

5.5.7.2 Team

The composition and differentiation of the TMO resulted in a number of non-standard collaborations. Despite the cultural differences between HEA and NCT, participants reported of a productive relationship between the two organisations.

"One is a charity and one is a Government funded organisation. So you would think that they wouldn't necessarily be natural bedfellows but on this particular project they worked pretty well together" (D-CP271015-intvw) However, there were notable tensions between other TMO member organisations. Whereas, IG were a "creative design company who are used to design and build and being in control" (D-DMc011214-intvw), ICG, as a nominated subcontractor, were part of an academic research unit and did not have the same level of commercial experience. "Differences in the way of working" were also identified by the NCT Projects Director (D-DMc011214-intvw), and intervention was sought to develop a stronger working relationship between actors.

The inherent sentience between disciplines was also attributed to the "*personality clashes*" between the MC and the architect (D-DA220616-intvw). As recognised by the HEA project manager in the early stages of the project, membership of creative organisations in the TMO would require additional management initiatives to integrate the varied actors.

"It's just when you work with creative people you're going to get these personality clashes. The creative ego is a strong one" (D-JR241114-intvw)

The Interpretation project manager for NCT reported that much of his role involved managing the relationship between actors (D-TIC-041114-intvw).

"I think I had to work hard to keep an even keel and hold things together, ... part of my job is stakeholder management and there was, in any good team, you've got people who are strong leaders and people who are ... who balance that and so there were inevitable clashes...So, part of my job was to sort of manage those to some extent for the good of the project as a whole" (D-TIC-041114-intvw)

Similarly, the HEA project manager reported that she was often required to *"tiptoe through the egos"* in order to motivate TMO actors to integrate.

"That was a lot of my job. "... Aye, you're brilliant...now get on with it. Get your finger out"..." (D-JR241114-intvw)

Despite the conflicts between actors, TMO organisations did understand each other's contribution to the project and reported that they would be working on projects together future.

5.5.7.3 Task

One of the greatest challenges for the project was the complexity of the interpretation. The 3D exhibition was unique, in that each element of the immersive experience needed to be computer designed and coded. Many of the activities involving this had not been done before, and a number of aspects, including the 3D modelling, were being designed and developed as the project progressed. This caused difficulties for the architect and the engineers who, under the conditions of contract, had designed the detailed elements of the building prior to the construction phase.

"The difficulty with the interpretation package for this, was that some of it was so innovative they couldn't actually tell us what they needed until we were on site...our processes aren't aligned" (D-DA220616-intvw)

The input of the academics advising on the accurateness of the interpretation also caused difficulties for the ID organisation. In the absence of existing artefacts, NCT wanted to ensure that the display was as accurate as possible. It was, therefore, decided that all content would be verified by an expert academic panel prior to CGI coding and production. Whereas, historical accuracy was of paramount importance to the interpretation, delays in academic agreement on specific attire or posture had an impact on the deadline, and the design and production of CGI activities

"They couldn't make their minds up...or even worse, and this happened on numerous occasions, they changed their mind after we'd actually produced something" (D-PC150415-intvw)

A second difficulty was the integration of tasks between TMO members. Due to the delays, the IG and MC were required to work simultaneously in the building. It was reported that in order to manage the cabling, IC technicians would typically cut holes in the newly built or freshly painted walls. The next day MC would return to fill the holes, inevitably creating dust near the newly installed stereoscopic projectors.

"...getting dust into one of those projectors is thousands and thousands of pounds" (D-DA220616-intvw)

Task focus was evident on the project. It was reported that nearing handover, actors within the CGI organisation were "*working through night*" in order to complete the installation (D-PC150415-intvw). Actors working on other areas that "*were not really that interested*" in the interpretation, focused on completion of the activities they were responsible for (D-DMc011214-intvw).

5.5.7.4 Transition

The new visitor centre was initially scheduled for formal handover in October 2013. Due to the series of delays, previously discussed, the NCT visitor project team were unable to take occupancy until January 2014. The delay left only six weeks for the transition period between handover and opening to the public. Due to the complexity of the interpretation and the new technologies involved, a reasonable transition period was required to allow for commission of the 3D interpretation. This was besides, the required staff training, testing and other activities involved in operationalizing a visitor centre, such as retail and catering.

"I think in terms of day to day running because we didn't know what it would actually be like when we were open to the public. We did a kind of soft opening, to an extent whereby we invited members of the public to come in and test it. We had a lot of schools come in and test it just to make sure the technology worked, the visitor experience worked and to see how people reacted as went into the building" (D-CP271015-intvw)

The new visitor centre had the 'soft opening' on 1st March 2014, prior 700th anniversary commemorations, with TMO organisations providing additional support as necessary. Furthermore, despite the tensions between creative actors during the project, the architect has reported that they have been working on another project with ID, and the MC has continued to work with HEA, NCT and the architect.

5.5.8 Project Success

The focus of the project during construction was opening the visitor centre in time for the 700th anniversary commemorations. Delivery on time, was the also the key criterion for Central Government under the conditions of sponsorship. While, the construction phase was completed later than schedule, the overall project was ready prior to the official opening and delivered within the constrained budget.

Although, completion in time for the official opening was of significant importance, only four sources cited the opening of the visitor centre on time and within budget as a key measure of project success. The primary measure of success was cited as the finished quality of the attraction, assessed after completion. The majority of actors perceived 'visitor feedback' as the being the most important criterion.

"I've always believed that the success of a project is really about how the public perceive it. I'm a great believer in it's all very well being right and accurate and being academic but it's only as good as the experience of individuals because this is all about heritage interpretation and museums, it's all about making things accessible" (D-TIC-041114-intvw)

Initial feedback from visitors was "a mixed bag, but overwhelmingly positive" (D-CP271015-intvw). The negative reaction was primarily due to the non-conventional and interactive way of exhibiting the story of the battle, in contrast to the traditional method of display and "learning by reading, passively" (D-TIC-041114-intvw). However,

social media sites such as Trip Advisor did show an improvement in public perception in months following the opening and engagement with schools was very successful.

"Within a couple of months we booked out for the whole school year for 2014 and 2015. That was ten schools a week, every week so we had a waiting list of schools. We couldn't cope with the demand" (D-CP271015-intvw)

Another measure of visitor satisfaction was the awards that the new attraction received. In the first year of opening, the visitor centre won the Stanford Award for school visits, and the Herald Digital Business Award for the learning website. In addition, NCT won a marketing award for the battle brand, and the project, as a whole, won a Civic Trust award for the contribution and benefit to the community. "*So, I think that basically it tells us we're doing something right*" (D-DA220616-intvw).

Client satisfaction was also perceived as a critical measure of project success. It was important to the MC and the architect that client relationships were positive on completion of the project.

"If a client shakes your hand at the end of the job and they are delighted with it then you know you've done the right thing" (D-DA220616-intvw)

Through client satisfaction and positive references, consultants and contractors would be able to tender for future work or seek repeat business with the same client. As summarised by the MC "...you're only as good as your last job" (D-GMc2808150-intvw).

NCT were evidently proud of the end result and creating a visitor centre "*that engages you emotionally*" in the battle. Indeed, the logo of the battle site was adopted as a brand to promote the work of the trust. The project manager for HEA also reported that Government ministers, who sponsored the project, were "*were certainly happy*" with the final project, as was the project manager herself (D-JR241114-intvw).





5.5.9 General Findings from Case D

Report on the general findings of Case D are structured around the research questions discussed in Section 3.5. Findings are illustrated by a causal network diagram shown in Figure 5.9 that illustrate the relationships between events and actions within the case study.

5.5.9.1 Organisational Strategy

Through the replacement of the existing heritage centre, NCT was able to realise the policy objectives set out in its mission to communicate the significance of the battle site on a local, national and international scale (D-NCT0610-report), and conserve the historic monuments. The high profile project and Government funding enabled conservation of existing monuments whilst enhancing NCT's reputation in its role to protect and promote national heritage. As a result of high visitor numbers the new visitor attraction became financially sustainable and made contribution to other heritage sites owned by the trust.

The strategic objective of Central Government was also achieved. The public attention that the visitor centre received and the positive feedback from visitors supported the government initiative for investment into the countries heritage, whilst, also securing a manifesto promise to be complete in time for the 700th Anniversary. Moreover, Central Government's strategic agenda to *"showcase digital design talent…"* (D-TIC-041114-intvw) was realised through the global publicity that the new facility attracted.

Despite the creative differences between TMO members during the project, there were no conflicts between the organisational strategies of TMO members. As illustrated in Figure 5.9, through alignment with the client strategic objectives, TMO members were able to demonstrate organisational capability and attract further work. Moreover, the creation of an original immersive visitor experience heightened the profile of designers, which supported marketing initiatives for future contracts in the museum and visitor centre industry.

5.5.9.2 Mechanisms to maintain alignment

The joint responsibility between HEA and NCT to govern the project was successful. Evidence suggests that much of the success was due to the project management processes implemented to manage the project. Besides the robust systems that were put in place, findings show that the effective leadership from the project managers was a contributory factor to the success if the project.

"...they were the people that we met with on a weekly basis, often daily, and they were the conduit through which decisions were discussed and made and escalated upwards, up the food chain, as it were, within each of their organisations" (D-DA220616-intvw)

The project managers engaged with all TMO actors and ensured focus remained on the project objectives and integration of activities.

The main external influence potentially impacting on the project was the political environment at the time. As a result of the referendum, the project did receive significant public attention that put additional pressure of the TMO to complete the project by the deadline. But also, as a result of increased publicity, ensure quality was to a high level. However, the HEA and NCT project management ensured that the TMO were safeguarded against media pressures and allowed to focus on the project objectives. Despite this, it was recognised that the publicity surrounding the project supported the TMO objectives of market enhancement and raising public profile.

5.5.9.3 TMO Strategic Behaviour

The evidence finds that the fixed deadline did add to the tensions within the TMO. The dependency on other tasks, such as the external ratification of academic expertise, caused delays and frustration for the TMO. Tensions were augmented by the high profile media attention that the visitor centre was receiving. As discussed in Section 5.5.7.2, evidence suggests that actors were focused on the completion of individual tasks, rather than integration of tasks to create the visitor experience. This appeared to be higher within TMO organisations that involved creative activities, such as the CGI, and architecture.

As a consequence of the TMO strategic objectives to demonstrate individual competencies, the time made available for transition activities was reduced. The evidence finds that the leadership role project managers were critical in ensuring that tasks were integrated and the deadline for completion was realised.

5.5.9.4 Project Success

During the project, success was measured on opening of the visitor centre by the 700th anniversary commemorations. This was a publicised deadline and also a principal condition of Central Government sponsorship. Although complete in time for the official opening, the schedule was hindered by the quality success criteria and external academics that were not full-time members of the TMO. Due to the limitation and sources of funding, completion within budget was also a key measure of project success.

On completion of the project, the primary success criteria were measured on the public perception and visitor feedback. The, generally positive feedback, was collated through reviews submitted to social media sites, and demonstrated an increase satisfaction through the year, following the opening. The medium-term success of the project was also endorsed through the multiple awards the visitor centre and NCT received, as listed in Section 5.5.8. All TMO members recognised the significance of the awards and public ratification, for the individual strategies of enhancing reputation to increase their market position.

5.5.10 Conclusion to Chapter

This chapter has reported on the empirical findings from the data collected for each case study. Themes from each case have been discussed and presented in thematic matrices, contained in Appendix F. The main themes evolving from the findings, in relation to each research question are discussed as follows:

Research Question 1: What are the varied organisational strategies being pursued within a temporary multi-organisation?

The findings confirm that the construction projects, from the four cases, are implemented to realise the organisational strategies of the client organisation. The consistent theme is the existence of a strategic driver behind the decision to build. In Case Study A, the driver was a change in housing regulations that determined the existing residences to be no longer fit for purpose. Similarly, in Case Study D the driver was the deterioration of historic monuments and the recognition that the existing exhibition and visitor centre did not meet current tourist expectations. In Case Study C, the strategic driver was the need to reduce operating costs and increase operational efficiency. Whereas in Case Study B, the strategic driver was the longer-term need for

regeneration of the east end of the city. Findings also show that organisational strategies at the policy level provided opportunities for strategies at the business level. These included strategies for growth, as in Case Studies A and D, and for cost reduction and efficiency, as in Case Studies A and C. In case study B, the carriageway provided a wider contribution to the community, in terms of transformation and increased economic activity.

The findings show how TMO organisations aligned their strategies with the client organisation. Across all four cases, TMO members pursued competitive strategies that would be realised through successful implementation of the construction project. The consistent themes were repeat employment, market entry into specialist industries, and enhancement of current portfolio. Findings suggest that TM strategies were driven by financial uncertainties at the time of tender.

Research Question 2: *What mechanisms are implemented to maintain the alignment of a construction project with the client strategic objectives?*

and

Research Question 3: *How effective are these mechanisms in maintaining the strategic alignment of a construction project?*

The main mechanisms implemented by clients were the formal contractual arrangements within procurement strategies. Apart from Case Study A, findings show that the choice of procurement route was appropriate and effective. Within Case Study A, there was ambiguity over roles and responsibilities. In all cases a project board was also created to provide guidance and direction to the TMO.

The emerging theme within this research question was the need for a single point of contact between the project board and the TMO. In Case Studies B and D, the project board communicated directives through a project manager. Within Case Studies A and C, the project board, initially, directed the TMO. The findings show that under such an arrangement, the project suffered from the client complexity of the client organisation, with mixed directives and frustration of the TMO.

Research Question 4: *How do varied TMO actors pursue strategic objectives within a single construction project?*

Findings show that the TMO the strategy developed to pursue the varied strategic objectives within a single construction project is influenced by governance processes, procurement, client leadership, external TMO members and the varied strategic objectives being pursued.

Evidence suggests that a fixed deadline caused tensions with each TMO, the levels of which varied across the cases. A consistent theme in meeting the deadline was the effectiveness of the client leadership and the processes implemented to realise the project objectives. All TMO members were task focused, and integration of tasks was created through the leadership within each TMO. However, an emerging theme across all cases was the impact of partial TMO membership by sub-contractors, utilities, and external consultants, which did not work towards the project deadlines.

A further finding was lack of integration between the TMO and client personnel responsible for the transition. In case studies A and D, inadequate time was allowed for efficient transition, yet effective transition was required to realise the strategic objectives.

Research Question 5: *How do the varied actors engaged within a TMO measure the success of a construction project?*

Across all cases, the main measure of project success during construction was delivery by the deadline. Adherence to budget was also cited by some respondents. On completion of the project, end-user satisfaction was the primary measure of success. By measuring the level of end-user satisfaction, client organisations could assess the extent to which strategic objectives had been achieved. In all cases, end-user satisfaction (students, residents, employees, visitors) was the main determinates of project success. All TMO actors recognised that in order to realise strategic objectives of repeat employment, market entry into specialist industries, and enhancement of current portfolio, it was necessary to achieve the success criterion of client satisfaction.

The following chapter of the thesis syntheses individual, case study findings in the presentation of within-case analysis and initial conclusions from the study.

CHAPTER 6 CROSS-CASE ANALYSIS AND DISCUSSION

6.1 Introduction to chapter

In the previous chapter, the findings from each case were presented and discussed. This chapter discusses the findings through the process of cross-case analysis, based on the *a priori* themes identified within literature review, in Chapters 2 and 3, and *a posteriori* themes emerging from the findings in Chapter 5. Themes are explored and examined in relation to the research objectives of the thesis. Figure 6.1 provides an overview of the process of cross-case analysis within the chapter.



Figure 6.1: Process of Cross Case Analysis

The chapter begins by presenting a comparative outline of the relevant characteristics of the four case studies. This is followed by cross-case analysis and discussion addressing each research objectives of the thesis. The aim of this thesis is to investigate how varied organisations within a temporary multi-organisation seek to align multiple strategic objectives through a single construction project, and realise project success. A model of strategic alignment within a TMO explaining this phenomenon is developed from the findings and presented and discussed in Section 6.8. The chapter concludes with a summary of the findings, before the final conclusion chapter of the thesis.

6.2 Case Study Characteristics

For the purpose of data reduction, a summary of the characteristics of each case is presented in Table 6.1. Similarities and differences relevant to the findings are identified for comparison.

	Case Study A	Case Study B	Case Study C	Case Study D
Public Sector Client Type	Higher Education	Local Authority	Environmental Authority	Heritage and Conservation Agency
Project Deliverable	New Student Residence	New 4-lane Carriageway	New Office and Laboratory facility	New Tourist Visitor centre
Project Governance	Projects Managed through Project Board.	Projects Managed through a single business unit	Projects Managed through Project Boards.	Projects Managed through external agency and client project team
Procurement Strategy	Traditional	Design & Build	Design & Build	Traditional with Design and Build element
TMO Composition	Composite design team and MC	Joint Venture D&B Contractor	D&B Contractor with novated design team	Individual design team and MC

Table 6.1: Summary of Case Study Characteristics

- *Client Type:* As discussed in Section 4.4.1 all clients were public sector organisations and, therefore, publicly funded. The student residency in Case Study A was funded through the universities own financial capabilities. The carriageway in Case Study B was funded through the local authority budget. In case studies C Central Government was the main source of direct funding. Similarly in Case Study D, where sponsorship was received from Central Government and a Heritage Lottery Fund.
- *Project Deliverable:* Apart from Case C, where the new office and laboratory facility was built for the use of personnel within the Environmental Authority, other construction projects were commissioned for external stakeholder use. In Case A, the end users were students who had a prior contractual agreement with the university to rent accommodation within the new residency. In Case D, the visitor centre was open to the public, schools and interest group. In Case B, the new 4-lane public carriageway provided access to both public and privately owned land.

- *Project Governance:* All client organisations had established hierarchies consisting of multiple business units and departments. Directives were top-down through the corporate level and disseminated across the organisations. In case studies A and C, projects were initially managed through project boards created to govern the project. During construction a single point of contact between the board and the TMO was appointed. In Case Study C the project was managed through a functional business unit within the client organisation, with directives to the TMO being communicated through the client project management. In Case Study D, the TMO was managed, as a joint venture, by an external government agency and the client project management capability,
- *Procurement Strategy:* Apart from Case A, all TMOs were created under the competitive tendering process, as discussed in Chapter 5. In Case Studies B and C, the construction was procured under a design and build strategy. In Case Studies A and D the construction was procured under the traditional procurement route. The variation was the IG in Case Study D, which was procured under the design and build procurement strategy.
- *TMO Composition:* The composition of the TMO varied across the four cases. Within Case A, the TMO consisted of the client project management, the MC, and the PMC, with the architect being seconded to the PMC from an external architectural firm. Case Study B consisted of only the client project management team and the JMC organisation, which comprised of joint venture contractors and the design engineer. Case C also consisted of three organisations, including the client organisation. In this case, members of the design team were novated from the PMC organisation to the MC on award of the contract. Case Study D had the largest TMO with each discipline being represented independently, apart from the ID organisation, which comprised of an interpretation consultant and an interpretation design team. The client project management from the client team.

The characteristics of each case are considered when conducting the cross-case analysis and forming conclusions of the research.

6.3 Hierarchy of Strategic Objectives

Objective 1 of the study seeks to explore how multiple organisations within a TMO align varied strategic objectives through a single construction project. Within the context of a single organisational boundary there is the expectation that the objectives being pursued, at each level of a strategic hierarchy, will align with the corporate strategy of the organisation Morris and Jamieson (2005). This is illustrated in Mintzberg's (1994) model of strategic planning, shown in Figure 2.1.

In contrast, within the construction industry, the TMO operates across multiple organisational boundaries, and therefore, seeks to align the varied levels of multiple organisational strategies, through a single construction project. This complexity is analysed through the theoretical lens of the strategic hierarchy, as discussed in Section 2.4. The strategic hierarchy of each case is presented in Table 6.2, with comparison of strategic objectives in Section 6.3.2. This is followed by a discussion of the theoretical implications of the findings in Section 6.3.3.

6.3.1 Hierarchy of Strategic Objectives within each case study

Drawing on the models of Archibald (1988) and Youker and Brown (1998) as discussed in Section 2.7.2, the hierarchy presented in Table 6.2 identifies three levels of organisational strategic objectives that were pursued within each case study. These include TMO strategic objectives, as well as, client objectives.

- Level 1: Policy objectives define the intended corporate strategy of the organisation, in terms of industries and markets in which it competes (Andrews, 1971). Client strategic objectives at this level, derive from the organisations mission, and explain the fundamental rationale for the construction project. The policy objectives of the TMO member organisations determine the long-term strategy and direction of the organisation.
- Level 2: Business objectives determine how the policy objectives, at the corporate level, are achieved through implementation of the construction project. Strategic objectives at this level focus on competition within a particular market segment (Beard and Dess, 1981). For, both client and TMO, it is through the business objectives, that the policy level objectives are realised.

• Level 3: Operational objectives describe the project deliverables. It is through the success of the implementation of the operational strategies, that the business level objectives and the policy level objectives of the client and TMO are realised.

The hierarchy of strategic objectives within a TMO, shown in Table 6.2, illustrates how business level objectives within the client organisation are pursued to achieve the policy objectives at the corporate level. The operational objectives of the construction project, established at level 3, is implemented to realise the business objectives of the client organisation.

As shown, in order to pursue the policy objectives of the consultant and contracting organisations, TMO member organisations align their business objectives with the operational objectives of the construction project. Strategies across all organisations are, therefore, realised through the implementation of the project, as demonstrated in the comparison of case studies within the study.

			Ī			Ì					
)	Case Study	Α		Case Study B		Case St	udy C	Ü	ase Study I	•
Client		Client			Client		Clie	int		Client	
Level 1:	Maintain (Competitive p	osition and	Regener	ate East End of	f the city	Improve services	whilst reducing	Conservation	of the histori	c monument
Policy Objective	reputa	tion of the uni	iversity				operatin	g costs			
Client	Grow	Contribute	Reduce	Increase	Support	Provide	Relocate staff to a	Create new	Enhance	Increase	Increase
Level 2:	international	to the	maintenance	economic	development	access to	central facility	working	public	visitor	financial
Business	student	universities	of	activity	of the	the Games		practices synergy	awareness	numbers	stability
Objective	numbers	financial	university	within the	community				of battle		
/		position	estate	East End		7		>	/ site		
Level 3:				Provide carri	ageway for gre-	ater access to	Provision of centra	al laboratory and	Provision of 1	new tourist v	isitor centre
Operational Objective	Provide high	ı quality stude	ent residences	the Eau implementa	st End of the C ition of flood re	ity and slief system	office f	acility	and inno	vative interp	retation
TMO /	Enhance expe	rience St	ecure repeat	Establish	1 Establ	lish reputation	Continue to work	Protect local	Enhance	De	monstrate
Level 2:	within the edu-	cation bus	siness with the	reputation in	local in le	ocal market	with existing	market	experience	orge	unisational
Business	sector		client	market			client		within the	ca	pabilities
Objective									museum indu.	stry	
		Seek	k opportunities	Secure repe	eat Wo.	rk on larger			Demonstrat	te	
		fo	or survival in	business with	h the	projects			organisation	lal	
		cn	urrent market	client					capability		
TMO	Seek opportui	nities Mari	ket entry in the	Business expa	insion Incre	sase business	Maintain	Market entry in	Maintain	V	1aintain
Level 1:	to maintain	and edu	ucation sector	in the UK m ⁶	arket op	portunities	competitive	the laboratory	competitiv	e co	mpetitive
Policy	increase						position	and science	position	4	osition
Objective	competitiv	ve						sector			
	position										
					V 22	Aake cost		Increase business	Enhance		
					effi	ciencies on		opportunities	competitiv		
					<u> </u>	contracts			position an	q	
	;			3			;		reputation		
	Consulta	nt	Contractor	Consultar	C	ontractor	Consultant	Contractor	Consultan	t CC	ntractor

Table 6.2: Hierarchy of strategic objectives within a TMO

6.3.2 Cross-case synthesis of strategic objectives

In Case Study A, the policy objective of the new residencies was to maintain the competitive position of the university. As discussed in Section 5.2, this had been threatened by the introduction of new HMO regulations that determined the existing residencies did not meet required standards, and would reflect negatively on the reputation of the university if new students were not offered the opportunity of on-campus accommodation. The competitive position of the university was pursued through the level 2 strategy to grow the intake of international students, and the offering of services for hospitality and conferences. Strategic objectives were achieved at level 3, through the deliverable of a high quality student residency. By offering student accommodation of a higher quality also enabled the Department of Hospitality Services to support the universities financial position through higher rental charges, and enabled the Department of Estates to reduce maintenance costs through the implementation of a sustainability strategy, in support of strategic objectives at level 2.

The level 1, policy objectives, of both, the PMC and MC was to survive in the current economic climate that was facing difficulty as a result of 2008-2013 financial crises. This was pursued through market entry into the higher education sector, in which the construction of student residencies was one of the few growth areas in the construction industry at the time (KnightFrank, 2011). Through alignment with the business objectives of university, in the delivery of a high quality student residency, the PMC would be able to expand their experience to attract further business. Whereas, the level 2 strategic objective of the MC, was to develop a working relationship and partnership with the university, in order to secure future contracts as part of a wider estates strategy.

As discussed in Section 5.3, the level 1 policy objective of the local authority, in Case Study B, was to support a major regeneration initiative of the east end area the city, that was developed fifty years prior. Regeneration was pursued through the level 2 strategic objectives of increasing economic activity of the regeneration area and supporting community development. The deliverable of the four-lane carriageway improved access to the regeneration area, which in turn, attracted new business. The carriageway also provided access to venues for hosting of the Games. This supported the level 1 policy objective to enhance the reputation of the city for tourism and commerce, thereby increasing economic activity.

The level 1 policy objectives of the JV contractor were aimed at increasing business opportunities and making cost efficiencies through reduced expenditure on projects. These were pursued at the business level through focus on larger construction contracts and on attracting more local business, thus saving on logistics. The level 1 policy objective of the European design-engineering firm was to expand its business within the UK. Successful delivery of the carriageway supported the level 2 strategic objective of seeking repeat business from the Local Authority.

In Case Study C, the level 1 policy objective was to improve services whilst reducing operating costs, as a result of reduced funding and the subsequent impact on the core operations of the business. The level 2 strategic objective was to relocate personnel from varied locations to a central facility, thus, making costs efficiencies through lower rental of facilities, reduced maintenance and unnecessary duplication of equipment and resources. The level 3, operational objective was implemented to deliver a new laboratory and office facility, whereby an environment would be created that enabled efficient sharing of equipment for new working practices and synergies to develop.

The PMC and architect, in Case Study C, had a previous working relationship with the EPA and were seeking to maintain this through future contracts. The policy objective was therefore to maintain their competitive position with the public sector client through the continuation of work. The level 1 policy of the design and build MC was to enable market entry into the specialist laboratory sector and to maintain their competitive position. This would be achieved at level 2, by protecting their position as the main contractor for refurbishment projects within the retail park, and securing the contract for construction of the new laboratory and offices with the park.

In case Study D, the battle site formed part of a portfolio of projects managed by the NCT. The level 1 policy objective of NCT was to conserve the historic monument. This was achieved through a level 2 strategic objective to raise public awareness of the battle site. Through increased visitor numbers and financial stability, NCT would be able to sustain the conservation of the historic monuments. The deliverable at level 3 not only included the new visitor centre, but also an innovative and unique interpretation that would attract wider and younger visitors. Through the uniqueness of the visitor experience and the media attention that project attracted, NCT were able increase visitor numbers and ensure continued conversation of the monuments at the battle site.

Similar to the other case studies, the policy 1 objective of the consultants and contractor was to maintain their competitive position. In the case of the ID organisation, the policy objective was to enhance their competitive position within the specialist 3D market. By pursing level 2 strategic objectives to demonstrate the unique design capabilities of the IC, they were able to improve their reputation and increase market position. Similarly, the high profile attention of the project enabled the architect and the MC to enhance their experience and demonstrate capabilities, thus, maintaining their competitive position in the museum and visitor centre market.

6.3.3 Alignment of strategic objectives

Existing studies investigating the alignment of strategic objectives with projects, consider only the levels of strategy within the boundary of a single organisation (Archibald, 1988; Kerzner, 2001; Morris and Jamieson, 2005). The hierarchy in Table 6.2 acknowledges the strategic intentions of all TMO member organisations, and demonstrates how the operational objectives supports the strategic objectives of both, the client organisation, and the consultant and contractor organisations participating in the project. Alignment is achieved when all TMO member organisations are able to realise individual strategic objectives through implementation of the construction project.

The findings demonstrate how clients develop and implement projects to realise unique strategies, of which the construction project is one component. As illustrated in Table 6.2, emergent opportunities, at the business level, also evolve as a result of the decision to invest in the construction project. This includes contribution to the financial strategy in Case Study A, development of new working practices in Case Study B, efficient transportation and flow to the Games in Case Study C, and, consistent with Mintzberg and Waters' (1985) and Mintzberg's (1978) proposition that organisational strategies emerge unintentionally during the event, a new branding strategy that emerged as result of the success of the visitor centre, in Case D.

In contrast to client organisations, where strategy is achieved through fit between internal elements, TMO consultant and contractor organisations align their business strategies with the external environment. Within the study, the policy and business objectives of the TMO member organisations were based on sustaining a competitive advantage within the market. At the time of the study, the 2008-2013 financial crises had a significant impact on the construction industry, resulting in construction organisations pursing survival strategies. Organisational strategies of TMO members were, therefore, formulated in line with the conditions of the competitive environment. These included diversification into specialist market groups and development of partnering arrangements with public sector clients.

In contrast to the assumptions made in the literature (Morris, 1982; Jones and Lichtenstein, 2008), the study did not find conflict between the strategic objectives within the TMO. Instead, analysis reveals that the intended, long-term, strategic objectives of all TMO organisational members were aligned through the operational objectives of the project. Consultant and contractor organisations were dependant on the successful delivery of the construction project for individual strategies to be realised. Despite this, TMO member organisations did not act as 'obedient servants' to the client organisation (Artto *et al.*, 2008). The short-term priorities of TMO members changed in accordance with the conditions of the external environment and the threats to individual, business level objectives. As a result, actors pursued courses of action that balanced the short-term objectives of their representative firm, with the operational objectives and delivery of the project objectives. The salient finding is that, at the start of the project, the strategic objectives of all TMO members were aligned. The difficulties arose through environmental changes, and actions taken by TMO members to maintain alignment.

6.4 Governance of temporary organisations

The second objective of the study explores the effectiveness of mechanisms implemented by the client organisation to maintain alignment of strategic objectives. Mechanisms are examined through the lens of project governance. This includes the analysis of procurement strategies and contractual relationships that determine the formal governance mechanisms, and the effectiveness of governance structures created by the client organisation to ensure that the strategic objectives of the project are realised.

6.4.1 Cross-case synthesis of procurement strategies

In Case Study A, the decision to adopt the traditional procurement strategy was perceived by the client as the main cause of failing of the project. Incomplete information at the tender stage enabled the MC to submit a low tender price and seek additional costs for items not included in the tender documentation (Curtis *et al.*, 1991). As a consequence, the project suffered from significant cost overruns and delays by the MC in securing sub-contractors at the reduced costs. Despite this, the traditional procurement process did support the strategic objective of the Department of Estates to reduce the maintenance of the estate. This was achieved through the nomination of specified materials and suppliers to be used in the construction.

Case Study D also used a tradition approach to procure the services of the MC for construction of the visitor centre. Similar to Case Study A, this also suffered delays as a result of securing sub-contractors and resources for particular work packages. Due to the technical nature of the interpretation and the design process, the CGI was implemented under the design and build procurement strategy. This empowered the IG organisation to be innovative and deliver an interpretation that aligned with the strategic objectives of NCT. Alignment was maintained through the approval process of academics employed to advise on the historic accuracy of the IG. However, delays in receiving academic approval did have an impact on the schedule and delivery of the operational objectives.

In Case Studies B and C, the design and build procurement strategy was adopted in order to reduce the risk of cost over runs for the public sector clients. In Case Study B, alignment of the design of the carriageway with the strategic objectives of the regeneration programme was maintained through the production of a preliminary design by the Department of Technical Services and on-site attendance by the client project manager. In Case Study C, the design of the layouts to facilitate new working practices and the requirements of each department using the facility were maintained through the novation of the design team from the PMC at pre-tender stage, to the design and build MC at the construction stage.

Analysis concurs with Rowlinson (1999) and Taylor *et al.* (1999) that no single approach to procurement is suitable for all contingencies. Although, it was found that in cases studies B and C, where a design and build procurement strategy was implemented, the projects completed either on time or earlier than planned. Under the design and build arrangements, it is the contractor that has greater control over the TMO activities and implementation of operational objectives. In contrast, within the traditional procurement strategy, the client has greater control over the final design. This is of

importance when the strategic objectives are dependent on the design of the new facilities, such as the maintenance strategy in Case Study A and innovative design in Case Study D.

The findings imply a need for project clients to have sufficient knowledge of the procurement processes when embarking on a construction process. This contradicts RICS's (2013) suggestion that the traditional route is suitable for inexperience clients. In comparing case studies, findings show a greater need for client management when the traditional procurement strategy is pursued than that of the design and build procurement strategy, where the directive of the TMO falls under responsibility of the design and build contractor. It is therefore argued that the traditional procurement strategic objectives, or the appointment of a capable project manager, as demonstrated in Case Study D.

6.4.2 Project Governance Structure

Findings support Cherns and Bryant's (1984) fifth proposition, listed in Table 3.2, that the actual performance of the TMO is determined more by the capabilities of its component organisations and their coordination than by the form of contract adopted in the procurement strategy. Focusing on coordination, the governance structure of each case is analysed in terms of the effectiveness in ensuring that the project meets the goals and expectations determined by the varied stakeholders (Turner, 2006), and maintaining alignment of the client organisation's strategic objectives (Müller, 2009).

In Case Study A, project governance was provided through a multi-stakeholder project board comprising of end user representation and varied stakeholder departments across the university. The project board, actually, had little formal authority over the project, and was dependant on the University executive and the Court of the University to approve appointments of contractors and make decisions pertaining to costs. The attempt by the project board to align the project schedule with the meetings of the Court was not successful. The lack of flexibility in the formal governance arrangement of the university resulted in the documents being issued for tender with incomplete design information. Conflicts between strategic objectives were also evident between project board members. The recommendation of the Department of Estates to pursue the traditional procurement strategy and enforce a strict specification of nominated materials and suppliers, supported the strategic objective to reduce the maintenance burden, but incurred additional costs, and contributed to the delays in completion of the residency. This had an impact on Department of Hospitality and Conferences who were seeking cost reduction and access to the facility to begin the preparations for students arriving. Also with the Department of Student Services who were seeking high level of student satisfaction through aesthetics, comfort and finishes within the new residencies.

Similarly, in Case Study C, a project board was established to provide guidance and direction to the TMO in the refurbishment of the laboratory and office facility, whilst a second project board was established to manage the relocation of personnel in the client organisations. Findings reveal that ambiguous authority and a lack of communication between the boards resulted in frustration for both, the TMO and end users, whose requests for specific layouts were not fully delivered as expected. As discussed in Section 5.4.5, the inefficiency of having two project boards was later recognised and subsequent action was taken to merge the two boards into one. A project manager was also appointed to take responsibility for the direction of the TMO, which was welcomed by TMO actors, as it enabled focus on project objectives without involvement in strategic objectives of the client organisation. This is consistent with point 8 of Cherns and Bryant's (1984) propositions, that TMO members have a preference for dealing with a single client representative within whom all politics of the client system can be contained.

In contrast, within Case D, the project board consisted of senior personal from NCT, HEA as the Government representative, and HLF as one of the funding organisations. The responsibility of the project board was to ensure conservation of the historic monuments in the interest of the public and the country. An HEA project manager was appointed to deliver the operational objectives of the project, as part of the Government funding agreement. The role of the project manager was to oversee the TMO in completion of the project by the 700th anniversary of the battle and report progress to the project board. TMO actors were directed by a single client entity consisting of an HEA project manager and a project manager from NCT. This guaranteed focus on

operational objectives and safeguarded the TMO from influence and involvement in the political and strategic issues surrounding the high profile conservation project.

Similarly, within Case Study B, the project board, consisting of representation of all interested stakeholders within the regeneration programme, governed the pursuit of strategic objectives. The operational objectives implemented by the TMO were managed through the capability of a distinct business unit within the Local Authority. Within this arrangement, TMO actors engaged only with a single client representative who reported to project board. This enabled on-site decisions to be made quickly and efficiently without ambiguity, and was preferred by TMO organisational members.

Consistent with the findings of Ruuska *et al.* (2011), across all cases, governance mechanisms within the TMO evolved during the project lifestyle. This was despite formal governance structures and the formation of project boards. In Case Studies A and C, where the project boards were not effective, the TMO developed a process of self-regulation, whereby project related decisions were made by TMO actors in order to ensure that the project progressed. This was a particular occurrence where the TMO received mixed and ambiguous directives. In Case Study B, self-regulation evolved as a result of the co-location of actors. Project decisions were made quickly, at a local level, which ensured that the project progressed accordingly. Ambiguity was also removed, with reports of strong integration between actors.

Also consistent with Ruuska *et al's* (2011) findings, the strategic objectives of key project actors appeared to take priority over other actors within the TMO. In particular when actor had a creative role within the project. In Case Study A, members of the project board focused attention on the architect to implement individual strategic objectives. Whereas, in Case Study D, key actors were defined as the interpretation consultants responsible for the design of the virtual exhibition and the architect, who were both reported as requiring additional attention by the client project management. It could therefore be concluded that, within a TMO, the priority of actors will vary in accordance with the strategic objectives of the project, but would appear to focus on the design capability.

6.4.3 Client Complexity

Ruuska *et al.* (2011) and Bekker (2014) also highlight the impact of stakeholder complexity on the governance of construction projects. Within the study the impact is greatest in Case Studies A and C where the multi-stakeholder project boards result in a pluralistic client body (Thomson, 2011). Under such conditions, competition emerges between organisational departments to manoeuvre the operational objectives of the project in favour of their own strategic requirements. This creates a situation where departments and SBU's are, not only, competing for resources and attention from the corporate level (Gupta and Govindarajan, 1986), but are also competing for prioritisation from the TMO to realise the strategic objectives of individual business level departments. Conflicts emerge between stakeholder departments, resulting in the TMO receiving mixed, and often opposing, directives by different client representatives. Within Case Study A, TMO actors were explicitly being told to favour the priorities of certain departments over others. In Case Study C ambiguity from within the client system resulted in frustration for TMO actors, with the risk of delays in development and progress of the project.

In contrast, within Case Studies B and D, a project manager was appointed to direct the TMO from the start of the design and construction phases, consistent with the structural arrangement in, both, Archibald's (1988) and Kerzner's (2001) hierarchy of objectives. In projects where a project board is created for governance, the project manager acts as the conduit between the multi-stakeholder project board and TMO actors. Thereby, mitigating the impact of pluralistic client complexity. Without the role of a single point of contact between and the project board and TMO actors, self-governance of the TMO, to ensure focus on the operational objectives of the project, becomes more evident.

It is relevant that in Case Studies A and C, project managers were later appointed by the client following the recognition that projects were not performing as planned. The findings show that formal governance structures are more effective when there exists a single point of contact between the client and the TMO. The authorisation of a project manager, to communicate the client's strategic objectives, ensures clarity of the project requirements, faster decision-making and supports TMO performance.

6.5 Leadership of Temporary Multi-Organisations

Related to project governance is the leadership of each TMO. The study finds a link between client leadership and the TMO strategy pursued to realise the operational objectives of the project. Findings show that client leadership influences the strategic behaviour of the TMO. The leadership styles of each client project manager, appointed to direct the TMO, are analysed through the characteristics of transformational and transactional leadership styles (Bass, 1990), as discussed in Section 3.5.

The case studies provide examples of, both, task-focused transactional leadership styles and person-focused transformational leadership styles, as presented in Avolio and Bass's (1991) full range leadership model. Consistent with Bryman *et al's* (1987b) proposition that, due to the limited duration of construction projects, leaders tend to be more task oriented than people focused, the findings of the study suggests that all client project managers were task-orientated and focused on completion of the task within the parameters of the project objectives. However, analysis also reveals the use of transformational leadership approaches to motivate TMO actors.

In case study B, TMO actors perceived the client project manager as highly supportive. The emphasis of the client project management, from the start of the project, was to establish a strong team-working environment that would allow focus on the project objectives. On site attendance enabled continuous support of the TMO and enabled issues to be addressed quickly and efficiently. This was recognised by all actors who reported of strong integration and a good team atmosphere amongst TMO actors.

Similarly, in Case Study C, the client project manager was also considered supportive in his attitude towards the TMO. He attended all meetings, provided immediate solutions to potential issues and demonstrated commitment in supporting the TMO in the achievement of the operational objectives of the projects. This included the implementation of a change control system to safeguard TMO actors against mixed directives from the project boards, thereby, enabling focus on the project tasks. As a consequence, TMO actors spoke highly of the client project manger and were explicit in the significant progress made on the project, following his appointment.

In Case Study D, the HEA project manager ensured mechanisms were in place to enable the TMO to remain focused on the task. To protect the TMO, the project manager deflected the, on going, political and media attentions surrounding the visitor centre, and ensured that the required support was given to TMO organisations to achieve the operational objectives. The vision of the project was clearly communicated by the client, and TMO actors were motivationally inspired to deliver a unique visitors centre that was attracting significant publicity. Whereas there were reports of *'personality clashes'* as a result of creative differences between actors, the acknowledgement of the creative capabilities of individuals, understanding of the individual needs of actors, and the enforcement of the project vision, ensured that TMO actors remained focused on the operational objectives of the project.

In contrast, the client project manager in Case Study A, exerted pressure on the TMO to achieve the programme deadline through coercive behaviour. This included abrasive interactions with TMO actors and openly criticising the capabilities of consultants and contractors during client meetings. Consistent with Cherns and Bryant's (1984) point in referring to the impact of conflicts and behaviours within the client organisation, the client behaviours from within the university, resulted in breakdowns in communication between the client and TMO organisations and, in general, demotivation of actors on the project.

Case Study A demonstrates the difficulties and consequences of adopting a transactional leadership style within the context of TMO. As observed by Tyssen *et al.* (2014), the challenge for leaders of temporary organisations is that they have little de facto authority. Within the case, the client project manager had no reward or punishment in which to transact with, in exchange for TMO performance. In response to the a symbolic transaction, in the form of performance in exchange for non-criticism, TMO actors deferred to the conditions of contract established within procurement strategy. Thus, negating the authority and, therefore, the leadership of the client project manager.

Findings also reveal challenges for transformational leadership styles. The limited duration of the construction project does not allow for transformational leadership interactions to develop, although, the co-location of actors in Case Study C enabled deeper social interactions to take place. However, there was evidence of transformational leadership through inspirational motivation, establishing a team spirit and expressing shared values (Bass, 1990), across the case studies. This was evident in case studies C and D, where actors understood and shared the values that the project had

on the wider stakeholder community, as a result of the client leadership. Also, within Case Studies B, C and D, actors spoke highly of the client project manager, whereas in Case Study A, reports were decisively negative.

6.6 TMO Strategy

Objective 3 of the study explores the strategy of the TMO in the pursuit of the varied strategic objectives to be realised from a single construction project. This is examined through the strategic behaviours and interaction of TMO actors in response to the procurement strategies, governance mechanisms, client complexity, client leadership, and other external factors. Strategic behaviour is analysed through the lens of Lundin and Soderholm's (1995) theory of temporary organisations, which generalises that actors within a temporary organisations will demonstrate particular behaviours, as discussed in Section 3.3. Consistent with the theory, the strategic behaviour of each TMO is analysed under the themes of *time, task, team* and *transition*. Cross case synthesis is made within each theme, and between the assumptions made within the literature on temporary organisations and findings from the analysis.

6.6.1 Time

As discussed in Section 3.3.1, the first perspective in the theme of *time*, views temporariness as being short in duration, due to the requirement for immediate goals (Palisi, 1970). This perspective is analysed from the potential impact of delays in the project schedule and the effect in the realisation of strategic objectives, as a consequence of project deadlines not being achieved.

The findings show that there was awareness among all TMO actors that completion of the project by the set deadlines was essential for the realisation of client policy objectives. Failure to meet deadlines would also impact on the MC in Case Study A and the DE in Case Study C, whose strategic objective was to seek repeat business with the client. In Case Study A, where the programme had slipped extensively, completion by the revised deadline of the new semester, took priority over cost and quality project objectives.

In each case, the schedule for completion was considered *'challenging'* by TMO member organisations. However, in recognising the strategic importance of meeting deadlines, all projects had time contingency built into the schedule, in order to mitigate

risk to the client strategy, including Case Study A. Apart from Case Study B, which finished one week early, all projects suffered delays in completion. In analysing the reason behind schedule slippage, it was found that many of the problems concerning delays originated from factors within the client organisation. Findings are consistent with Cherns and Bryant's (1984) ninth point that problems concerning design changes, delays and difficulties during the construction phase have their origins in the unresolved conflict within the client organisation.

In Case Study B, the project suffered schedule delays as a result of design changes and additional requirements from the end user. In Case Study D, the time taken for academic experts to advise on the historical accuracy of the interpretation caused delays in the progress and delivery of the 3D interpretation. In Case Study A, the attempt to align the project schedule with the formal university governance arrangements resulted in delays in completion of the design. Also, as a consequence of client complexity and conflict between the strategic objectives of representative departments on the project board, changes to the design were made late into the project schedule.

The study finds that the primary external factor for project delays was the dependency on organisations that only have partial engagement in the TMO. As a consequence of the economic climate at the time, the MC in Case Study A was unable to secure the availability of sub-contractors at the rate submitted to the client. The study finds that sub-contractors need continuity of work to ensure sustainability. However, this is dependent on the efficiency of the project management and other trades to complete dependant works before the sub-contractor can start (Eccles, 1981). As such, timeframes and availability of specific sub-subcontractors may not align with the operational objectives of the project, especially, when changes in the project schedule occur. Similarly, in Case Study B, construction was delayed as a result of the time taken for utility providers to install services. As found in the study, utility suppliers do not operate within the boundaries of a single TMO, rather they operate across multiple TMOs at any given point in time, and therefore only work to the required conditions of contract.

The second perspective within the theme of *time* views temporariness as the relatively short duration of the participation between actors involved in a project (Lanzara, 1983). It is proposed that as a consequence of the limited life span of a temporary organisation,
actors do not anticipate future interaction with each other beyond the imminent deadline of the project (Bennis, 1965; Goodman and Goodman, 1976). As such, concern will be on the effective completion of tasks, rather than developing long-term efficiency of project processes (Saunders and Ahuja, 2006) and the development of long-term relationships with other TMO actors (Bakker *et al.*, 2013).

Across the case studies, TMO member organisations were employed under specified conditions of contract to provide particular services or resources over a defined period, at certain stages of a project. Consequently, the primarily concern of TMO actors was the successful accomplishment of the tasks they were contracted to deliver, by the set deadline. Within Case D, focus on individual tasks resulted in tensions between varied disciplines within the TMO, with the need for the project manager to adopt stakeholder management practices. Within Case Study A, there were conflicts between the architect and the engineer over the design strategy of the residences and also between the PMC and MC organisations. Despite this, actors reported that tensions and conflict were considered normal behaviours within the construction industry, especially when there was pressure on time to jointly deliver a specific output.

The findings of the study supports the proposition that TMO actors primary concern will be on completion of project tasks over the development of interpersonal relations (Bakker, 2010). This was not as a consequence of the limited duration of the TMO or the anticipated future interactions with TMO members (Bennis, 1965; Goodman and Goodman, 1976), but as a result of contractual obligations, delegation of responsibility and the actors necessity to realise the operational objectives of the project in order to support the strategic objectives of their own organisation.

6.6.2 Task

The second theme in the research of temporary organisations considers the complexity of *tasks* executed by the TMO. As discussed in Section 3.3.2, the concern within the literature is the uniqueness of tasks performed by the temporary organisation and the difficulties of repetition (Goodman and Goodman, 1976). However, within the construction industry, optimization in the planning of repetitive activities can lead to efficiency in the scheduling and costs of a project (Hyari *et al.*, 2009). Indeed, infrastructure projects and housing programmes are characterized by the repetition of activities, as the elements of construction are repeated from one unit or location to

another (El-Rayes and Moselhi, 1998). In Case Study A, a reason for cost over runs and delays was cited by the client project manager as a lack of modularisation within the design and construction of new student residency, which would normally be expected.

In contrast to the propositions, within the temporary organisation literature, that unique tasks creates uncertainty in the project (Gann and Salter, 2000; Prencipe and Tell, 2001), and inhibits performance (Lundin and Soderholm, 1995), analysis reveals that the competitive strategy of TMO organisations was the ability to demonstrate capability in performing unique tasks. In Case Study B the complexity of the design and construction of the flood relief tunnel supported the JV contractors strategic aspirations to attract larger contracts. The unique interpretation in Case Study D would provide a competitive advantage for the CGI consultant within the museums and visitor centre market segment. Similarly, in Case Study A, the participation in the residences project enabled differentiation into the higher education sector for the PMC. Whilst, the construction of unique laboratories, in Case Study C, enabled entry into the specialist science sector for the Design and Build MC to realise their strategic objectives.

6.6.3 Team

The main concern within the theme of team, was the lack of time available for confidence building activities to support the integration and development of trust between the diverse actors within a temporary organisation (Meyerson et al., 1996). As discussed in Section 3.3.3, the concept of 'swift trust' follows from the notion that actors have not worked together previously, and act as if trust was inherent from commencement of the project (Saunders and Ahuja, 2006). However, participants across all case studies confirmed that, due to the high level of mobility of labour in the construction industry and the relatively small network of construction specialists, construction professionals were frequently engaged within the same TMO, and fully expected to work together in the future. In Case Studies B and C, consultants and contractors reported that they had worked on previous contracts for EPA. Also, in Case Study D, the architect had conducted previous work for HEA. Furthermore, NCT and HEA had been involved in other projects together. As a result of the high mobility, social networks of actors, and membership of professional organisations within the construction industry, structural embeddedness (Nahapiet and Ghoshal, 1998b) was evidence between component TMO organisations.

The notion of swift trust is therefore questioned within the context of TMOs. Consistent with Jones and Lichtenstein's (2008) proposition, the study finds that, due to the temporal nature of construction projects and the high mobility in the construction industry, it is common for actors to have prior working relations with other TMO actors. Analysis also supports the proposition that prior working relations between actors within a TMO can enhance project performance (Rowlinson, 1988; Shoesmith, 1996). In case studies B and C where TMO actors had worked together previously, participants reported of a high degree of integration between TMO members.

A common theme across the contracting and consultant organisations within the study was also the desire to establish partnering arrangements. In Case Studies A and C, both, the MC and PMC advocated the need for partnering arrangements to maintain the strategic objectives for repeat business and future contracts within the market sector, consistent with Latham's (1994) report. Through the development of partnering networks, the strategic objectives of repeat business with the same client would be achieved, thereby, supporting the policy objectives of the TMO organisation.

6.6.4 Transition

As discussed in Section 3.3.4, Lundin and Soderholm's (1995) framework proposes two meanings for the term of *'transition'*. The first refers to the distinctive change to the permanent organisation as a result of the project. The second meaning focuses on the perceptions of causal relationships, and how to proceed to the *"end state"* (Lundin and Söderholm, 2013). This study investigates transition through the lens of the second meaning. Specifically, it considers the transition processes that occur between the permanent and temporary organisation (Burström and Jacobsson, 2012).

In Case Study A, delays in the project schedule did not allow adequate time for an efficient transition processes between the project and the permanent organisation to take place. The new residency was still in construction as students began to arrive. There were also reports of initial concerns about quality as a consequence of the rush to complete the facility. A result of the late delivery of the project, there was additional pressure on the Department of Hospitality and Catering, and Student Services, to ensure the accommodation was ready to receive the students. Findings suggest that the absence of time for transition was a contributory factor in the poor attitudes towards contractor and consultant organisations, post completion.

Similarly, delays in construction and installation of the interpretation had an impact on the transition process in Case Study D. As a result of the schedule slippage, personnel responsible for the management of the visitor centre had less time to decommission the system and operationalize the facility. However, as a contingency for possible delays, a 'soft opening' was planned two months before the official 700th anniversary commemorations to allow for '*operational fine-tuning*', prior to the official opening, consistent with Burström and Jaccobson's (2012: 414) findings.

In contrast, within Case Studies, B and C, the transition processes were implemented in parallel to the construction of the carriageway and facility. In Case Study B, key stakeholder members of the project board were able to plan related projects within the regeneration programme, in accordance with the progress of the carriageway. Within Case Study C, the project board, specifically established to manage the migration of EPA personnel to the new facility, implemented the transition process. Despite difficulties of having two governing two boards directing the TMO as discussed in Section 5.4.5, a dedicated board for the transition ensured a reasonably smooth relocation of personnel from other locations.

The findings of the study provide an alternative to Jacobsson *et al's* (2013) perspective of transitory units within permanent organisations, which questioned as to whether the temporary organisation can really exist detached from the permanent organisation. Jacobsson *et al.* (2013) propose that within the intra-organisational perspective, the temporary organisation is formed from personal from within the permanent organisation that remain involved in the project through to the transition. However, within the context of TMOs, there is little overlap between the permanent and temporary organisations, with the permanent organisation providing only guidance and directives to the TMO, as demonstrated across the case studies.



Figure 6.2: Transformational process from temporary to permanent organisation

Developed from on the findings of each case, Figure 6.2 illustrates the transformation between the permanent and the TMO across the cases studies. Consistent with Jacobsson *et al.* (2013), a subset of the permanent organisation (i) is formed as a temporary organisation to develop the strategic requirements of the project. This is developed either within a single business unit, as in case studies B and D, or involves internal stakeholders from across the organisation, as in case studies A and C. Once the decision to build is confirmed, the permanent organisation will begin to form the TMO. In Case Study A, the university employed the expertise of the PMC to advise on and manage the procurement process. Within case study C, the architect was employed to support the application for funding. In Case Study D, the HEA project manager led the TMO as a provision of Government funding for the new visitor facility. This is shown at point (ii), in Figure 6.2, as the first overlap between the permanent and temporary organisations.

The TMO is formed when responsibility for implementation of the operational objectives of the project are officially transferred from the client organisation to the temporary multi-organisation, shown at point (iii). The overlapping organisational boundaries between permanent and TMO, at point (ii), constitutes the client project management responsible for providing leadership and directive to the TMO. It could also involve active members of the project board. But, as demonstrated in Case Studies A and D, the integration between project board members and the TMO tends to be weak.

The TMO has greater responsibility from the point of agreement of the contract and work starting on site. Within the case studies, members of the permanent organisation involved in the initiation of the projects remained within the permanent organisation, with the client project manager remaining within the TMO. Inevitably, membership of the TMO will cease to exist, as tasks are complete and actors return to their own permanent organisation. It is at this point (iv) that the second overlap occurs and the transition process begins. This involves operationalization of the facility for occupation or use by the permanent organisation.

It can therefore be proposed that the strategic role of the TMO is to provide a platform for delivering the strategic benefits as a result of implementation of the project. This occurs at point (iii). The realization of the client's policy and business objectives is normally dependant on the efficiency of the parent organisations transition processes, at point (v). However, analysis reveals that effective transition processes support the realisation of short-term strategic objectives, but are not critical to the long-term strategic aspirations of the client organisation, as shown in Case Study A. Once the project is complete, it is remains the responsibility of the permanent organisation to realise the expected benefits from investment in the construction project.

6.7 Project Success

The final objective of this thesis is to explore the linkages between project success and the intended strategic objectives of TMO member organisations and actors. Firstly, the varied perceptions of project success within a construction project are explored through a model of project success within a TMO, presented in Figure 6.3, developed from the findings of each case study. The linkages between project success and the contribution to organisational strategic objectives are then analysed through the lens of Shenhar *et al's* (2001) success dimensions framework, presented in Figure 6.4.

6.7.1 Success criteria within a TMO

As previously determined, the realisation of the policy, strategic and operational objectives identified in Table 6.2 are dependent on the successful delivery of the construction project by the TMO. However, as established in Section 2.6, within a typical project, there exist varied perceptions of success due to the differing interests of stakeholders involved in a project (Pinto and Slevin, 1988; Baccarini, 1999; Cooke-Davies, 2007). As such, success criteria will be more varied in a TMO where there exists greater complexity of stakeholders. Consequently, there is a greater possibility that perceptions of success, between stakeholder groups, will be conflicting.



Figure 6.3: Model of project success within a TMO

Analysis of the case studies reveals five possible tensions where differences in project success criteria may occur within a typical TMO. These are illustrated in Figure 6.3. Each tension is examined in respect to the influences on the behaviour of TMO actors in the pursuit of the operational objectives of each construction project. Within the model, alignment is achieved when there is a degree of fit between the varied project success criteria and the strategic objectives of each organisation within the TMO.

6.7.1.1 Tension 1: Between project objectives

The first measure of project success, across all case studies, was assessed on the efficiency of the TMO in achieving the time, cost and quality project objectives, identified in Tension 1 of Figure 6.3. Consistent with Baccarini's (1999) framework, at the operational level, the perception of success is determined, primarily, by TMO performance. Tensions occur, when there is a need to prioritise between meeting delivery of the project by the deadline, budget or expected level quality. It is proposed that in such circumstances, the TMO pursues a completion strategy based on the relative

importance of the client's strategic objectives. Analysis reveals that completion of the project will take priority as the first measure of success, in particular when handover is of significant importance and there is a schedule dependency on completion by the end users of the facility.

In Case Study A, the TMO prioritised completion of the new residency over budget and quality objectives. Actors were aware of the reputational damage to the university should there be a failure to meet the commitment made to students to offer accommodation in the new facility. Furthermore, failure to meet the revised deadline would have further implications on the, already, tense relationship between the client and the TMO. The strategy of the TMO was, therefore, to handover the residency to the university before the students arrived and complete the works during the transition period. This allowed the residency to be operational in time for students arriving, but incurred quality defects, cost-over runs and hindered the transition processes of the client organisation.

Similarly in Case Study C, completion of the office and laboratory facilities in accordance with the schedule also took precedence. Migration from the five existing laboratories and offices were planned and scheduled in accordance with the relocation project. Any delays in completion of the new facility would have had an effect on productivity of ECA and impact on business continuity. The new office and laboratory were complete by the deadline, although there were some minor quality defects that were addressed, post handover.

Completion of the visitor centre by the 700th anniversary of the battle was also the key focus of the TMO in Case Study D. The Trust had publicly announced an official opening to commemorate the anniversary of the battle, the year prior to completion. Moreover, the Government, as the main sponsor, had included the opening of the visitor centre by the significant anniversary, in their public manifesto. Due to the high public profile of the project, failure to meet schedule commitments would have caused reputational damage to NCT, the Government, and HEA as project managers. However, in the absence of artefacts from the battle site, the success of the visitor centre was also dependant on the quality of the visitor experience. Potential, impact on the quality objectives as a result of the deadline was, therefore, mitigated through the overlapping

of project activities and the reduction of the time dedicated for transition and operationalization of the visitor centre.

In contrast, although completion by the deadline was the main priority in Case Study B, there were no trade-offs required between schedule, budget and quality, as the project completed one week early. This allowed completion of other related projects within the regeneration programme that were dependent on access from the new carriageway. These included the new facilities being constructed for the high profile Games that were dependant on completion of the new carriageway by a set deadline in order to commence preparations and provide access to venues.

6.7.1.2 Tension 2: Between completion of tasks

Tension 2, identified within Figure 6.3, occurs between actors in the completion of the project tasks. Analysis confirms that, as a result of the task focus inherent in TMOs, actors will operate in *silos* and concentrate, primarily, on the successful implementation of the individual tasks they are contractually responsible for. As a consequence, there is general lack of integration between actors from different TMO member organisations, as discussed in Section 6.5.2. Tensions evolve when there is a dependency on the completion of specific tasks to complete particular parts of the project.

In Case Study A, it was found that the lack of integration between the structural engineer and architect led to numerous reiterations of plans and subsequent delays in design information being issued to the MC. As a result of the pressure on TMO member organisations to complete the project, a blame culture developed between the client, the PMC and the MC, which caused further tensions between actors. There was also evidence of *silo* behaviour between TMO organisations in Case Study D. This was particularly strong in tasks involving creative disciplines, as previously discussed. Analysis reveals that the main tensions between actors involved the installation of the CGI, where designers focused only on the success of the interpretation, independent of other tasks comprising the project. This included the main contractor responsibilities and construction of the building. However, in contrast to Case Study A, mechanisms were put in place through effective leadership, to ensure focus remained on delivery of the operational objectives.

Analysis finds that there is a greater prospect of achieving project objectives when tasks are efficiently integrated. This was evident in Case Studies B and C. Findings suggests that increased social interaction and prior collaboration does lead to improved integration of tasks. In Case Study C, task integration was attributed to the on-site, co-location of TMO actors. Consistent with the findings of Kavanagh and Kelly (2002), increased face-to-face communication through co-location enabled rapid response to project issues and greater inclusion among actors through informal discussions. Also, in Case Study B, where TMO actors had previously worked together on EPA projects, actors had already developed the necessary trust and understanding prevalent in permanent organisations (Meyerson *et al.*, 1996).

6.7.1.3 Tension 3: Between TMO member organisations

Tension 3, considers the success criteria of each TMO member organisation, which is measured on achieving the anticipated strategic objectives through participation in the project. Whereas, analysis finds alignment between the long-term strategic aspirations of each TMO member organisation, shown in Table 6.2, tensions occur in the pursuit of individual, operational objectives, that each TMO organisation expects to be achieved as a result of investment of resources in the project.

Consistent with Curtis *et al's* (1991) vicious circle theory, the main source of tension in Case Study A was between the unversity who were seeking to maximize value for money through the procurement process and the MC who was seeking to maximize profit. Findings reveal that the financial strategy of the MC was to submit a low tender application and look for alternative means to recoup lost income, mainly through claims for additional works (Ioannou and Leu, 1993). The JVC in Case Study B, and consultants in Case Study C also reported that they did not meet organisational financial objectives expeted through participation in project.

In Case Study D, tensions evolved between the MC and other consultants over the lack of resources made available to the project. In the attempt to make cost savings through minimising the resources allocated to the project, the construction of the building fell behind schedule. Not only did this have an impact on the progress of dependant activities, but also had an impact on the realisation of project objectives and the perception of the efficiency of the TMO in achieving them. Analysis reveals that, despite comprising of separate organisations, the successful performance of the TMO is assessed as a single entity by the client.

6.7.1.4 Tension 4: Between organisations and TMO actors

The fourth tension occurs between TMO member organisations and their representative actors participating in the project. The review of literature on project success in Chapter 2 finds no distinction made between the success criteria of specific stakeholders. However, analysis reveals that a TMO creates a context where actors often need to balance the short-term operational objectives of their employer organisation with the operational objectives of the construction project. This occurs, in particular, when there is an impact on the time of resources committed by a TMO member organisation to complete project tasks.

In Case Study A, the extended time spent on the construction of the residences increased the operating costs for both the PMC and the MC. Furthermore, as a result of the delay, resources committed to completion of the project were unable to transfer to other projects. Lengthy negotiations with the client over claims, also incurred additional management costs for the MC. Similarly, in Case Study D, the architect and ID were required to spend more time on the visitor centre than originally planned. This created a risk to the progress of other projects the organisations were participating in.

Moreover, in Case Studies A and C, the client project manager prioritised the project objectives over the objectives of the stakeholder departments. In both cases, the client project managers prevented the project boards from directing the TMO and, instead, focused on delivery of the project by the deadline. Across all cases, findings suggest that TMO actors, including the client project manager, will focus on the project objectives over the strategic objectives of organisations within the TMO.

6.7.1.5 Tension 5: Between stakeholders within the client organisation

As shown in Figure 6.3, Tension 5 occurs within the client organisation, itself. The concept of project success within this context becomes particularly complex due to pluralistic client complexity (Thomson, 2011) and the varied objectives being pursued by different client stakeholders. The study finds general agreement across all case studies that the primary measure of success was the efficiency of the TMO in delivering schedule, budget and quality objectives. All projects were considered a success by the

client organisation, apart from Case Study A, which failed to meet schedule and budget objectives. This was despite the residency being, practically, complete in time for students arriving.

Whereas there was general agreement on TMO efficiency as a success measure, analysis finds that tensions occur between departments competing to realise strategic objectives through implementation of the project. At the business level, success is measured on the realisation of individual strategic objectives. Within the case studies, this was assessed on end user satisfaction. In Case Study A, Student Services reported high level of student satisfaction, despite issues with quality, and schedule and cost over runs that had an impact on the operations of the Department of Hospitality and the Department of Estates. Similarly, in Case Study C, staff feedback on the new facility was reported as being positive. This was despite varied departments reporting that layouts of laboratories and facilities were not delivered as requested. In Case Study D, visitor feedback was also very positive, which supported the strategic objective to increase visitor numbers. However, in Case Study A, where the project did not meet time and budget objectives, the focus of success remained on the performance of the TMO, despite positive end user feedback.

The final measure of success from within the client organisation was on the contribution the project made to achieve the corporate strategy as described in the Level 1, policy objectives in Section 6.3.1. Across all cases this was the longer-term measure of project success, although indications of achievement were realised soon after completion of the project. In Case Study A, the reputation of the university was enhanced through the offering of a higher quality student residency. In Case Study C, EPA was able to reduce operating costs through the migration of staff to a single location. In Case Study B, the project received a national award for its contribution to regeneration of the city. The visitor centre in Case Study D also received varied awards and increased public awareness, thus, securing the conservation of the historic monuments at the battle site.

6.7.2 Dimensions of project success

Analysis of each case study confirms the subjective nature of project success (Pinto and Slevin, 1988; Freeman and Beale, 1992). Within a TMO, the success of a construction project is measured at micro and macro levels, consistent with Lim and Mohamed's (1999) proposition for projects, in general. The findings reveal that the dominant

measure of success remains on the efficiency of the TMO to deliver the project within the time, cost and quality project objectives. However, the thesis also finds that, in addition, project success is measured on the contribution the project makes to the strategic objectives of the client and participating organisations within the TMO.

Consistent with Shenhar *et al's* (2001) success dimensions framework, the study confirms that perceptions of success are time framed. Analysis reveals that short-term assessment of project success made during project execution and immediately on completion focus on the successful delivery of the operational objective. Long-term assessment of the project focuses on the strategic value the project adds to the organisation, as a result of investment. Figure 6.4 builds on Shenhar *et al's* (2001) success dimension framework, shown in Figure 2.2, to illustrate the relationship between time and strategic value, in the context of a TMO.



Figure 6.4: Strategic value of project success over time

As illustrated in Figure 6.4, during execution, success of a construct project is measured on the efficiency of individual TMO member organisations to execute individual project tasks within, the micro level, time, cost and quality objectives. In isolation, project tasks have little strategic value to the organisation and contribute only to the operational objectives of the project. As established in Section 6.7.1.2, it is the successful integration of project tasks that leads to the successful execution of the project. Consistent with Shenhar *et al's* framework (2001), during and immediately after project execution, success of the project is measured on the TMO efficiency in delivering the operational objectives of the project within the specified timeframe, budget and level of quality. It is through successful execution that the strategic objectives expected from the project can be realised. The first objectives to be realised through successful completion of each project is the operational impact on the client and TMO member organisations.

For client organisations the operational impact was the immediate organisational transition as a result of the new facility. Within Case Study A, the university were able to accommodate new students. Within Case Study C, EPA was able to close existing buildings and relocate staff within a single location. Within Case Study D, NCT were able to publicly display an innovative interpretation that reflected the historic importance of a battle site. Finally, in Case Study D, Local Highways were able to direct traffic to the new carriageway, thus, alleviating heavy traffic in other areas. Furthermore, the risk of local flooding was reduced, as result of the installation of the flood prevention system.

In contrast, the operational impact for TMO member organisations was the profit, or loss, as result of participation in the construction project, and lessons or capabilities developed as a result of participation in TMO. In Case Study B, operational benefits were developed through joint ventures, and within Case Study D, operational benefits were developed through sharing of capabilities in the design and installation of the interpretation.

Project success in the short-to-medium-term is measured on achieving the strategic objectives at the business level, as identified in Table 6.2. Success at this level is measured over a period of time after implementation of the project. In Case Study A, a measure of success for the client organisation included a tangible reduction in the maintenance of the university estate, and also a contribution to the financial position as a result of income generated from the high quality student residences. Similarly, in Case Study D, strategic success was measured on the reduction in operating costs and increased efficiency savings as a result of the relocation. In Case Study D, medium term success was measured on the increased visitor numbers and, the consequential, increased financial income. Whereas, in Case Study B, strategic success was measured

on the ease of traffic flow, but also the use of the new carriageway and visitors to the East end of the City. In contrast, for TMO member organisations, strategic successes are measured on the number of new contracts or enquiries the member organisations attracted as a result of successful delivery of the construction project.

Within the model in Figure 6.4, long-term project success is measured on the contribution the project makes to organisations long-term strategic aspirations. This includes the policy objectives of client organisation, as a result of investment in the project, and also the competitive market position of TMO member organisations as a result of contribution to the construction project. In Case Study A, long-term success for the university could be measured on the growth in student numbers as a result of increased reputation for the provision of high quality residencies. In Case Study C a long-term measure of project success for EPA would be the capabilities and new ways of working as a result of the relocation. In Case Study D, long term success is measured on increased visitor numbers across NCT's tourist attractions, and the consequential financial stability, enabling the conservation of other historic monuments. Whereas, in Case Study B, the long-term success of the project for the city is measured on the increased economic activity within the regeneration area, as a result of new businesses and housing relocating to it.

6.8 Model of strategic alignment within a TMO

As discussed in Section 1.5, an aim of the research was to develop a theoretical model of strategic alignment within the context of a TMO, to explain how varied organisations within a TMO seek to align multiple strategic objectives through a single construction project. The review of the extant literature in Chapter 2 finds that current models explaining the strategic alignment of projects with organisational strategy have been developed within the context of a single organisational boundary. Within these, strategic objectives established at the corporate level, cascade through the organisational hierarchy to be implemented as projects to support the corporate level strategy (Archibald, 1988; Mintzberg, 1994; Morris and Jamieson, 2004). The temporary organisation created to implement the project therefore operates within the boundaries of the permanent organisation.

In contrast, as discussed in Section 3.6, a TMO operates across multiple organisational boundaries. Whereas construction projects are implemented to support the strategic

objectives of the sponsoring organisation, as discussed in Section 6.3.1, alignment between the varied strategic objectives of organisations participating within the TMO become highly complex. This complexity is explained through the model of strategic alignment within the context of a TMO, presented in Figure 6.5.

6.8.1 Model development

The conceptual model, presented in Figure 3.10, was developed from the initial themes evolving from the extant literature review. Themes emerging from the literature review were listed in the Initial Template, contained in Appendix D, and investigated through the four case studies outlined in Chapter 4. Following development of the Final Template, contained in Appendix E, the model in Figure 6.5 was developed and validated through the findings of case studies in Chapter 5 and cross-case analysis within this chapter. In contrast to existing models of strategic alignment, the model of strategic alignment within the context of a TMO recognises that construction projects provide a context through which the strategic aspirations of multiple organisation are realised.

6.8.2 Discussion of model

The conceptual model in Figure 3.10 illustrates how the strategy of the TMO, that emerges to pursue operational objectives and realise the project success criteria of the client, is influenced by formal governance mechanisms, TMO behaviours, and internal and external environmental factors. The definitive model in Figure 6.5 acknowledges that the strategy of the TMO is also influenced by the strategic objectives of the component TMO member organisations.

Drawing on the findings from the research, each factor potentially influencing the strategy of the TMO is discussed in detail, with the implications for the realisation of project success and an explanation of how multiple organisations within a TMO seek to align their varied strategic objectives through a single construction project. The discussion of the model begins with identification of the strategic drivers that determine the client project success criteria. This is followed by identification of the influential factors that determine the TMO strategy and consequential project success during the design and construction phases of the construction project.



Figure 6.5: Model of strategic alignment within a TMO

6.8.2.1 Strategic drivers

The findings of the case studies, discussed in Chapter 5, confirm that construction projects are implemented to support a client's strategic objectives. Within this thesis, all case studies involved public sector client organisations, and were therefore determined by strategic drivers originating from the external, political environment. In Case Study A, the strategic driver was the change in HMO regulations that meant that the existing

residencies no longer met acceptable housing standards. In Case Study C, the strategic driver was the financial restrictions that forced the public sector organisation to seek operating and cost efficiencies. Whereas, in Case Studies B and D, the projects were driven by political initiatives, that saw the release of public funding to deliver the the project benefits.

However, despite the attention and effort dedicated to a construction project, the case studies show that the new facility, or carriageway, is only one component of the organisations strategy. The strategic benefits from investment into a construction project are dependent on other related projects and tasks, and efficient transition processes implemented as the TMO disbands.

6.8.2.2 Project success criteria

As shown in Figure 6.5, the strategic objectives of the client organisation are translated into project success criteria for implementation by the TMO. Findings from the research show that TMO members are aware of their client's strategic objectives and will plan the project accordingly. However, as discussed in Section 2.6, the success of a project is both subjective and time dependant. In the first instance, the pluralistic complexity of the client body often results in competing perceptions of success criteria as a consequence of conflicting strategic objectives at the business level. Consistent with Thomson's (2011) and Newcombe's (2003) studies, this was evident in Case Studies A and C where competition evolved between stakeholders to influence the TMO and manoeuvre the operational objectives of the project in the favour of a particular SBU. Secondly, as previously established, the client's strategic objectives can only be realised on completion of the project. Therefore, at the strategic level, the primary focus of project success is ensuring that the scope and requirements of the construction project align with the client's strategic objectives.

6.8.2.3 Procurement strategies

Consistent with practice, the model in Figure 6.5 shows that construction clients will implement procurement strategies and establish formal governance mechanism to direct the TMO in the realisation of the project success criteria. The model proposes that the TMO strategy that emerges to achieve the success criteria and realise the client's strategic objectives, is influenced through the choice of procurement strategy and the formal governance processes created by the client organisation to direct the TMO.

220

The study finds a relationship between the choice of procurement strategy and the level of control that the client sought to maintain over the design. In Case Study A, the traditional procurement strategy was selected to ensure that materials and suppliers, used in the construction of the residences, aligned with the estates strategy for low maintenance. In contrast, within Case Study D, responsibility for the design of the interpretation was transferred to the interpretation consultant, who had the required specialist knowledge to achieve the strategic objective of the visitor centre. Whereas, in Case Study B, there was evidence of tensions between the client and design engineer over responsibility of the design that had been procured under a design and build strategy.

6.8.2.4 Governance mechanisms

The model in Figure 6.5 proposes that procurement strategies require formal governance mechanisms to be managed effectively. This involves the formation of a project board created to ensure that the TMO operate within the conditions of contract, and the strategic interests of the client stakeholders are maintained throughout the execution of the project.

However, as found in the case studies, the effectiveness of the project board is dependent on the efficiency in communicating directives to the TMO. In Case Studies A and C, the project board provided conflicting and mixed directives to the TMO, which had an negative impact on its performance, consistent with point 14 of Chern's and Bryant's (1984) framework. The model supports Ruuska *et al's* (2011) findings that governance mechanisms emerge during the project life cycle, and draws a distinction between the formal governance processes implemented by the client organisation and the informal governance mechanisms that evolve through a process of self-regulation. Analysis of the case studies shows that self-regulation of the TMO will be more prominent when the impact of client complexity on the TMO performance is at its highest. This was demonstrated in Case Studies A and C. The findings show that the TMO will alter and pursue its strategy accordingly.

6.8.2.5 Environmental influences

As shown in Figure 6.5, the strategy of the TMO is also influenced by internal and external factors. Within the model, internal influences include client behaviours driven

by the leadership from within client organisation. Analysis reveals that the consequences of pluralistic client complexity are mitigated through effective leadership by a client project manager who acts as a single point of contact between the client organisation and the TMO. Within the context of TMOs, effective leadership ensures that the strategic objectives of the client organisation are communicated to TMO actors without ambiguity. Moreover, in overlapping the organisational boundaries between the client organisation and the TMO, the client project manager has a greater understanding of the strategic objectives of the project and the factors that determine the strategy of the TMO.

In Case Study A, the behaviours of the client had a negative impact on the performance of the TMO. In contrast, in Case Studies B and D, client behaviours supported and enhanced TMO performance. Other client behaviours identified within the Final Template, contained in Appendix E, include communication of directives to the TMO and creating environments of trust. Within the temporary organisation literature, research on trust appears to focus on relationships between TMO actors, and does not consider the necessary trust between the client organisation and TMO member organisations.

The external influences on the TMO strategy are those factors outwith the control of the TMO. These include the environmental factors that impact on the construction industry such as social, political and economic issues. But also, the availability of sub-contractors within the contractors supply chain and utility suppliers that are not part of the TMO but work to unique contractual conditions and schedules.

6.8.2.6 TMO Strategic objectives

Unlike existing models of strategic alignment of projects, the model of strategic alignment within the context of a TMO acknowledges the strategic objectives of TMO member organisations being pursued through a single construction project. As demonstrated within the study, TMO member organisations participate in construction projects to realise short-term and long-term strategic objectives. Consultant and contractor, corporate level strategies may include diversification into other industries, or enhancing reputation in order to develop and sustain a competitive position, as demonstrated in Case Studies A, C and D. Business level strategies may include the development of partnership and securing future work with an existing client, as pursued

in Case Studies A, B and C. However, at the operational level, project success is measured on the short-term strategic objectives and the realisation of the commercial objectives of the TMO member organisation, which includes profit in exchange for services. The strategic priority, and hence the TMO strategy, is dependent on the time available to complete the project, and the strategic significance of the project to the TMO and the client organisation.

6.8.2.7 TMO Behaviours

Within the final model TMO strategy is driven by the strategic behaviours of TMO actors and member originations. The strategy to deliver the project is therefore dynamic and evolves over the duration of the project, rather than being developed at a fixed point in time. Whereas, TMO actors are aware of strategic rationale of the project, and the strategic intentions of their representative organisations, the performance of the TMO will, inevitably, be measured on the efficiency in delivering the construction project within the project success criteria of time, cost and quality. It is in the pursuit of these criteria, along with the environmental influences, that the final strategy of the TMO is determined.

Drawing on the theory of the temporary organisations (Lundin and Soderholm, 1995), the model in Figure 6.5 acknowledges that TMO behaviour will be subject to the integration of project tasks. As established within Case Studies A and D, TMO actors focus on the element of project activities they are contractually responsible for, as success for individual TMO member organisations is measured on the performance of individual project activities. However, as illustrated in Figure 6.4, it is the integration of the project tasks that contribute to the successful performance of the TMO and the realisation of the operational objectives of the project.

6.8.2.8 Project success

As established within this thesis, the concept of success is diverse and subjective. The model of strategic alignment within a TMO shows that the success of a construction project is assessed at varied levels, both, within client organisation, and TMO member organisations. Across the case studies, the primary measures of project success are on achieving the operational objectives of the project within the boundaries of time, cost and quality. As previously discussed, it is against these criteria that the TMO is judged. The study finds that completion by a specified date is of particular importance,

especially when the delivery of the construction facility is of strategic significance to the client organisation. Due to the tangible nature of a construction project, failure to meet the deadline will impact on reputation of the client and the TMO.

The next level of project success is measured on achieving stakeholder satisfaction. For the client, this includes the end-user satisfaction of the complete construction and the internal and external stakeholders involved in the project. For the TMO, the success of the project is measured on the satisfaction of the client. It is only through client satisfaction that the longer-term strategic objectives of repeat contracts, partnering and strategic positioning, can be realised.

The model of strategic alignment shows that all organisations involved within a TMO, are dependent on successful implementation of the operational objectives of a construction project for organisational strategies to be realised. Efficiency in delivery of the project objectivises enables TMO member organisations to realise strategic objectives through participation in the TMO. Inventively, project success of the client organisation is measured on the contribution the construction project makes to the long-term strategic objectives articulated within the organisations vision. However, as shown in Figure 6.5, project success for the client is dependent on the implementation of efficient transition processes.

6.8.3 Benefits of the model

The intention of the model of strategic alignment within a TMO is to identify those factors that influence the strategic behaviour of the TMO in execution of the operational objectives of a construction project. It is proposed that awareness and greater understanding of the varied strategic objectives being pursued through a single construction project will support the decision-making process and performance of the TMO through the project life cycle of the project. The model recognises the varied strategic objectives being pursued through a TMO and the varied measures of project success. However, as established within the study, it is only through successful execution of the operational objectives that strategies within the TMO can be realised. The model supports this by demonstrating the influences on the strategic behaviour of the TMO, which in turn, impacts on the operational objectives, and consequently, on strategic objectives of the client and TMO member organisations.

6.9 Summary and conclusion to chapter

Analysis of the findings of the case studies, detailed in Chapter 5, reveals that the varied strategies within a TMO are aligned, rather than competing, as current literature suggests (Jones and Lichtenstein, 2008). Client organisations implement construction projects to provide a facility, or structure, through which the long-term strategic aspirations of an organisation are realised. Construction projects also enable emergent opportunities for the client at the business level. The hierarchy of strategic objectives in Table 6.2 shows how TMO member organisations develop and align their individual business objectives through the client operational objectives. It can therefore be determined that the realisation of strategic objectives, across all organisations, is dependent on the successful delivery of the construction project.

The research finds that the greatest challenge for the alignment of strategic objectives originates from within the client organisation, consistent with Cherns and Bryant's (1984) proposition. Whereas, vertical alignment within the client organisation has been established, conflict arises between the business level objectives of each SBU. This is particularly evident in public sector organisations where there exists a level of pluralistic client complexity (Thomson, 2011; Newcombe, 2003). The study finds, that in such cases, SBUs and departments will seek to the position the project in favour of a particular business strategy.

The study finds that when pluralistic client complexity is present, alignment is maintained through the leadership of a client project manager who acts a single point of contact between the client and the TMO. However, despite formal and informal governance process, the TMO will begin to self-regulate. The extent of self-regulation is dependent on the efficiency of the governance processes and the behaviours of the client as a result of the pluralistic complexity within organisation. Significantly, when client leadership is transactional, TMO actors will defer to the behaviours determined within the conditions of contract.

Whereas, the study demonstrates that the varied strategic objectives of TMO members are aligned through a single construction project, the research finds different perceptions of project success among actors. As discussed in the model of strategic alignment within the context of a TMO, presented in Figure 6.5, organisational strategic objectives can only be realised on completion of the construction project and are dependent on other initiatives and efficient transition processes for achievement of strategic benefits. Actors within the client and TMO therefore focus attention on project success criteria that enable strategic objectives to be realised. As illustrated in Figure 6.3, the perceptions of project success differ between actors, organisations and strategic levels within a TMO. However, the TMO strategy that emerges to deliver the project will, inevitably, focus on the short-term objectives, and prioritise delivery by a predetermined deadline over quality and budget, as the most important success criteria. As shown in Figure 6.5, it is the client's governance processes, internal and external environmental factors, the strategic behaviours of TMO actors, and strategic objectives of TMO member organisations that influence the strategy of the TMO. It is therefore concluded that the project is successful when there is alignment between the strategic intentions of all organisations within the TMO, and the operational objectives of the project.

CHAPTER 7 CONCLUSION

7.1 Introduction to chapter

The previous chapters to this thesis provided the theoretical foundations, data collection and analysis, and discussion of the theoretical models developed from the findings. This chapter presents the contribution and final conclusions of the research. The chapter begins with a review of the aims and objectives of the study. This is followed by a discussion of the contribution to theory and the contribution to management practice provided by the research. In reflecting upon the limitations of the study, recommendations for future research are proposed. Finally, the concluding remarks of the thesis are offered.

7.2 Review of aims and objectives

The aim of this thesis was to investigate how varied organisations within a temporary multi-organisation seek to align multiple strategic objectives through a single construction project, and realise project success. This is explained through a model of strategic alignment within the context of a TMO, presented in Figure 6.5. The definitive model of the study identifies the complex interactions and influences that determine the strategy of a TMO in pursuit of client operational objectives and project success. To address the aims of the thesis, the following research objectives were achieved.

7.2.1 Objective 1: Alignment of strategic objectives

The first objective of the thesis was to explore how the multiple organisations within a TMO align varied strategic objectives through a single construction project. This objective was satisfied through a review of the theoretical perspectives explaining the domain of fit, and the development of a hierarchy of objectives within a TMO, from the empirical findings of the study.

The review of the strategic management literature, presented in Chapter 2, concluded that that strategy occurs at different levels within an organisation. The dominant discourse proposes that each level be constrained by the upper level within a strategic hierarchy (Hofer and Schendel, 1978). Strategic alignment is achieved by setting objectives at the corporate level and cascading down the strategic hierarchy to be implemented as projects (Archibald, 1988; Youker and Brown, 1998; Kerzner, 2004).

However, the literature also recognises that each level of the hierarchy will be constrained by distinctive, and often competing, internal and external environmental forces (Haniff and Fernie, 2008; Morris and Jamieson, 2005). As such, objectives at each strategic level will be developed, and change, in response to the environmental conditions in which it competes. The filtering of strategic objectives from the corporate level is, therefore, subject to a complexity of interactions, processes, and varying objectives that, incrementally, influence the project.

As demonstrated in the review of the literature, models explaining the hierarchy of strategic objectives predominantly focus on alignment of projects within a single organisational boundary. Within the intra-organisational context, the general assumption is made that the temporary organisation, created to implement the project, will act as an *'obedient servant'* to a single parent organisation (Artto *et al.*, 2008), and the strategic objectives formulated at the corporate and policy levels of a hierarchy are fixed for the temporary organisation to implement as directed.

In taking an inter-organisational perspective to strategic alignment, the research conducted an empirical investigation into four public sector construction projects whereby the temporary organisation, created to implement the project, consisted of multiple organisations. The assumption within the literature is that varied organisations that constitute a TMO will have disparate and competing strategic objectives (Morris, 1982; Jones and Lichtenstein, 2008). However, the findings of the research suggest that the strategies of all TMO organisations are aligned and are, at least partially, dependent on each other.

Drawing on Archibald's (1988) and Youker and Brown's (1998) strategic hierarchies, discussed in Section 2.7.2, the hierarchy of strategic objectives within the context of a TMO, presented in Table 6.2, identifies the varied strategic objectives of the client, consultant and contractor organisations to be realised through the construction projects within the study. The model demonstrates how the project deliverable at the operational level of the hierarchy supports the varied strategic objectives of the client organisation, and through successful execution, enables component TMO member organisations to realise strategic aspirations through participation in the project.

As shown in the hierarchy, the long-term strategic intentions, to be realised through investment in the construction project, are determined at the policy level of the client organisation. Policy objectives are supported at the business level through competitive strategies and strategic opportunities that emerge from the decision to build. Significantly, the analysis finds that it is at the business level, of the client's organisational hierarchy, where conflicts between strategic objectives are likely to occur. The findings of the study support the notion of pluralistic client complexity (Newcombe, 2003; Thomson, 2011). As discussed in Section 6.4.3 this creates a condition whereby SBU's, compete for attention and resources from the corporate level of the client organisation (Gupta and Govindarajan, 1986) whilst seeking to manoeuvre the operational level of the project in favour of their own, unique, strategic requirements.

The hierarchy demonstrates how TMO member organisations align unique strategic objectives with the operational objectives of the project. The findings of case studies confirm that consultant and contractor organisations select the projects they intent to participate in for reasons other than commercial. Across the case studies, involvement in the TMO supported the long-term strategic positioning of consultant and contractor organisations. This included diversification and market entry into specialist industries, expansion in the current market, and maintaining and enhancing current competitive advantages. Business level objectives were pursued through demonstrating organisational capability, establishing reputation in the local market, and securing repeat contracts with the same public sector client. However, as determined within the research, the realisation of strategic objectives is dependent on the successful execution of the project.

It is therefore concluded that organisations participating within a TMO, align their varied strategic objectives through the successful deliverable of a construction project, as demonstrated within this thesis. It is through the project deliverable that the clients business objectives, and subsequent, policy objectives can be realised. Component TMO member organisations are able to pursue their long-term strategic intentions through the successful realisation of the client's operational objectives.

7.2.2 Objective 2: Effectiveness of governance mechanisms

The second objective of the theses was to examine the effectiveness of mechanisms implemented to maintain alignment of strategic objectives. This objective was satisfied though a review of theoretical formal and informal governance processes, and an investigation of governance practices across the four case studies, forming the empirical stage of the research.

The review the literature in Chapter 3, finds that formal governance mechanisms focus on procurement strategies and conditions of contract, implemented by the client organisation, to ensure that the TMO maintains alignment with the operational objectives of the project. Section 3.6.2 provides an overview of the common contractual relationships that determine the structural configuration of the TMO. Within these, the findings from case studies enabled a comparison between construction projects that had adopted the traditional, general contracting, procurement strategy, with TMOs that were employed under a design and build contract route. The cross-case analysis in Chapter 6, suggests that the efficiency of the procurement strategy, in terms of maintaining strategic alignment, was dependant on the coordination of the TMO and the formal governance structures established by the client organisation.

Across all case studies, a formal project board was created for the purpose of governance. The research found that when a project board was responsible for direct coordination of the TMO, governance mechanisms were less effective. This was partially due to the pluralistic client membership of the project boards that resulted in slow decision-making and mixed, and often competing, directives being given to the TMO. Consistent with Ruuska *et al's* (2011) findings, within their study of major construction projects, this research also found evidence of self-governance. The findings of the case studies suggest, that in the absence of efficient client leadership, a TMO will begin to self-govern and focus on completion of the project within the conditions of contract, established within the procurement strategy.

An important finding of the research is that informal governance processes and client leadership, which evolves throughout the duration of the project, has the greatest impact on maintaining strategic alignment and the realisation of operational objectives. Crosscase synthesis in Chapter 6, firmly establishes that a single point of contact between the client and the TMO was the most effective structural configuration and led to greater efficiency within the project in coordination of the TMO. In case studies B and D, where a client project manager was appointed at the start of the project, decisions were made quickly and TMO actors were able to focus on the project objectives, without being involved in the internal politics and issues relating to client complexity. These findings are consistent with Cherns and Bryant's (1984) points five and eight of their propositions, listed in Table 3.2.

7.2.3 Objective 3: TMO Strategy

The third objective of the research was to explore the strategy of the TMO in the pursuit of the varied organisational strategies inherent within a TMO. This objective was satisfied through an overview of the theoretical perspectives of project strategy in Chapter 2, and a detailed critique of the extant literature on temporary organisational forms in Chapter 3. As with the strategic management literature, the review found that research, primarily, focuses on the characteristics of a temporary organisation within the single organisational paradigm. To address this gap, themes evolving from the literature were empirically explored within an inter-organisational context. Internal and external environmental factors influencing the strategy of the TMO were identified from the findings and analysed for inclusion within the model of strategic alignment within a TMO, presented in Figure 6.5.

Analysis from the research found that the greatest influences on the TMO strategy originate from the client organisation. These included coordination of the procurement strategies and the formal governance arrangements, and leadership of the TMO, as previously discussed. In addition, the study found that the client leadership style has a direct impact on the strategic behaviours of the TMO. Whilst it was established that leaders within the project environment were task focused, the findings show that project leadership, which understands and supports the strategic objectives of individual TMO organisations, will be more effective.

External influences were categorised as those factors outwith the control of the client or TMO. These were identified as the current environmental issues that impact on the construction industry, and behaviours within the contractor supply chain. Within these, the economic financial crises of 2008-2013 had a substantial impact on the construction industry. This included risk and uncertainty for consultants and contractors, but also the

availability of specialist sub-contractors and trades, which was critical in the completion of projects and realisation of operational strategy.

The strategic behaviours of TMO actors were analysed from the four concepts of *time*, *task, team* and *transition*, within Lundin and Soderholm's (1995) theory of temporary organisations. An important observation from the research was that the concepts are not static and behaviours evolve over the duration of the project, in reaction to other influences. Evidence shows that towards the final stages of the project, the strategy of the TMO is on completion of the project by the established deadline. This is at the expense of other success criteria, including cost, quality, and the business objectives of the TMO member organisations. The primary concern of TMO actors is, therefore, the completion of the individual tasks they are contracted to achieve, and individual actors will pursue their strategy accordingly. In contrast to literature on temporary organisations, development of social relations and task integration are of secondary concern. Finally, in terms of transition processes, this becomes the responsibility of the client organisation, as illustrated in Figure 6.2.

7.2.4 Objective 4: Perceptions of project success

The final objective of the thesis was to explore the linkages between the varied perceptions of project success and the intended strategic objectives of TMO member organisations. A review of the literature on project success, presented in Chapter 2, concluded that the concept of project success is multifarious and subjective. Contemporary scholars have sought to address these ambiguities by drawing a distinction between success in the efficiency of project delivery by the temporary organisation, and business success as a result of the project outcome (Baccarini, 1999; Cooke-Davies, 2002; Shenhar *et al.*, 2001).

Drawing on these distinctions, the model of project success within a TMO, shown in Figure 6.3, was developed from the varied perceptions of project success gathered from all actors across the case studies. From the analysis of the findings, the model identifies five possible tensions where differences in perceptions of project success criteria may occur within a typical construction project. These are identified at the project level, where success is measured on the efficiency of the TMO in delivering the project within time, cost and quality objectives; between actors within the TMO on successful completion of individual tasks; between organisations within the TMO in pursuit of

individual business objectives; between TMO member organisations and their representative actors in the balancing of project objectives with business objectives; and within the client organisation as result of client complexity.

The linkages between the varied perceptions of project success, identified in Figure 6.3, and the contribution to the multiple strategic objectives are analysed through a model of strategic value of project success over time, presented in Figure 6.4. This model demonstrates the time-framed characteristic of project success. During and immediately after execution of a construction project, the primary measure of success is measured on the combined efficiency in delivery of project tasks within the time, cost and quality objectives. It is only through client satisfaction that the operational objectives have been achieved, that TMO member organisations are able to effectively pursue their individual strategic objectives. However, it was also observed, within the research, that the realisation of the operational objectives are dependent on effective transformation processes from the within the client organisation. As shown in Figure 6.4, the strategic objectives, will only be realised sometime after completion, once the TMO have disbanded. It is for this reason that the primary measure of project success, remains on the time, cost and quality objectives.

7.3 Theoretical Contribution

The study set out to explore how varied organisations within a temporary-multi organisation attempt to align individual strategic objectives through a single construction project. The aim was based on the initial assumption that, when compared to projects within a single organisational boundary, varied TMO member organisations will pursue diverse strategic objectives and exhibit differing perspectives of project success.

As established, current literature investigating temporary organisations forms focus on the intra-organisational characteristics of temporary systems created by a single parent organisation, whereas, research into TMOs focus on formal governance mechanisms and procurement strategies. Consequently, there exists a gap in empirical studies investigating the strategic behaviours of TMOs and, to date, no other research has conducted a study exploring the strategic alignment of multiple strategies within a single construction project. The specific theoretical contributions of the thesis are identified and discussed in the following sections.

7.3.1 Contribution to theories of strategic fit

As discussed in Chapter 2, previous models of strategic fit focus on the formation and implementation of strategy within a single organisation. This is illustrated in Mintzberg's (1994) planning hierarchy, reproduced in Figure 2.1. The assumption is made that objectives determined by senior management are implemented accordingly. Within the project environment, models of strategic alignment propose that the temporary organisation, created to implement the project, will coordinate activities to realise the strategic objectives of a single parent organisation (Archibald, 1988; Kerzner, 2004; Morris and Jamieson, 2005).

In contrast, the model of strategic alignment within the context of a TMO, shown in Figure 6.5, reveals that the actions of the TMO are subject to the strategic objectives of multiple organisations converging on a single project. In explaining this through the model, the complexities of implementing strategy through a construction project are exposed. Consistent with current perspectives of strategic fit, Figure 6.5 proposes that strategy is formulated at the corporate level, but takes into consideration the implications of client complexity in defining the client's project success criteria. In comparison to existing hierarchies, the model of strategic alignment within the context of a TMO identifies the varied external and internal factors that influence the emergent strategy of the TMO in actions taken to achieve success of the project.

Whereas, integrated perspectives of strategic fit seek to establish alignment between components within an organisational system and the external environment (Nadler and Tushman, 1980; Carmeli *et al.*, 2010), this research identifies the components of the project organisation that need to be aligned for project success to be realised, and the internal and external factors that create challenges for the successful alignment of strategic objectives within a TMO.

7.3.2 Contribution to perception of project success

The second important contribution to theory is enabling a greater understanding of project success, as an outcome of the thesis. Whereas it is generally accepted that project stakeholders may have differing perceptions of project success, studies have yet

to match success criteria to the varied stakeholder types. Figure 6.3 presents a unique model of project success that identifies the varied success criteria within a TMO. This includes consideration of perceptions of all stakeholders involved in a construction project, comprising of the client, TMO member organisations and TMO actors. The model also identifies possible situations where tensions between the pursuits of conflicting success criteria may occur, a topic that has not been considered within previous studies.

The model, not only, contributes to a greater understanding of project success, but also contributes to the greater understanding of stakeholder motivations within a project context. Moreover, in considering the diverse stakeholders within a typical project, the model can be adapted for any project type, and is not constrained for use only within a TMO.

7.3.3 Contribution to the theory of temporary organisations

The third contribution to theory enhances understanding of temporary organisations to include TMOs. Lundin and Sodeholm's (1995) theory of temporary organisations makes no distinction between temporary organisational types. Rather, the assumption is made that all temporary organisations are characterised by the same general conditions (Jacobsson *et al.*, 2015). However, as demonstrated within the study, the idiosyncratic nature of the construction industry results in unique differences between a temporary organisation operating within a single organisational boundary and a TMO, where the boundaries of multiple organisations overlap.

Cherns and Bryant (1984) have previously defined the characteristics of a TMO through a pilot study proposing twenty points for consideration. This thesis empirically supports a number of Cherns and Bryant's (1984) points, but also draws a comparison between the characteristics of temporary organisations from the review of the literature, and the characteristics of TMOs emerging from the empirical findings of the study.

	Characteristics	Characteristics
	of Temporary Organisations	of Temporary Multi-Organisations
Time	 Limited time to develop communities of practice and shared task relevance (Lindkvist, 2005) Attention on immediate present (Bakker <i>et al.</i>, 2013) Stress due as a result of limited time (Morley and Silver, 1977) Pressure to deliver outcomes (Turner and Muller, 2003) Limited time to foster group relations (Bryman <i>et al.</i>, 1987a) 	 Participation within a TMO over different points in time. Contracted to perform specified tasks to a predefined deadline, within a predefined budget Deadlines considered normal in the industry.
Task	 Task complexity (Meyerson <i>et al.</i>, 1996) Difficulty of task (Hanisch and Wald, 2011; Liu, 1999) Non-repetitive tasks (Goodman and Goodman, 1976) Uniqueness of task results in unfamiliarity of how to execute (Goodman and Goodman, 1976; Lundin and Soderholm, 1995) Limited time to develop task familiarity (Morley and Silver, 1977) 	 Focus on project complexity, including managerial processes and systems Uniqueness relatively low, preference for repetitive tasks TMO comprising of specialist skills
Team	 Do not anticipate working together in the future (Bennis, 1965; Goodman and Goodman, 1976) Have never worked together in the past Limited time to develop trust – swift trust (Meyerson <i>et al.</i>, 1996; Saunders and Ahuja, 2006) 	 High mobility and prior working relationships. Preference for partnering arrangements and repeat collaboration Structural embeddedness through social networks
Transition	• Temporary organisation part of the permanent organisation (Jacobsson <i>et al.</i> , 2013)	• TMO detached from permanent organisation

 Table 7.1: Distinctions between temporary organisations and TMOs

Table 7.1 identifies noteworthy differences between a temporary organisation and a temporary multi-organisation. These should be considered when conducting research into TMOs. The salient differentiation between the types is that TMO actors are employed under specific contractual conditions, determined within the procurement strategy. As such, accomplishment of tasks will take priority over social relations with other actors. However, as established across all cases within the study, the construction industry is also characterised by the high level of mobility among actors. Therefore, prior working relationships are likely, and, indeed actively encouraged through formal partnering arrangements and strong social networks through professional bodies. These

latter points are considered important for future studies on social behaviour and studies of social embeddedness involving TMO actors.

7.3.4 Contribution to project leadership

The fourth contribution to theory is within the study of project leadership. As previously discussed in Section 3.4, there is a lack of empirical studies investigating leadership styles within the context of projects. The definitive model of the thesis, in Figure 6.5, recognises the impact of client leadership on the strategy of the TMO.

The empirical research raises unique questions for studies of transactional and transformational leadership within a project environment. Firstly, as proposed by Tyssen *et al.* (2014), client project managers have little *de facto* authority for transactional leadership to be effective. Conversely, there is insufficient time within a construction project, where membership is at different points in the project life cycle, for the benefits of transformational leadership to evolve. As previously established within the research, actors and leaders within a TMO will focus on completion of tasks over development of relationships, and therefore, integration of tasks becomes difficult for the transformational leader. When transactional leadership is ineffective, TMO actors will revert to the conditions of contract, as demonstrated in Case Study A.

7.4 Contribution to practice

The thesis also makes a number of contributions to management practice. The hierarchy in Table 6.2 demonstrates how consultant and contractor organisations align their strategic objectives through the client's operational objectives for a construction project. This contradicts common perceptions within the construction industry that strategies of TMO member organisations will be conflicting. Instead, the research reveals that all component organisations within the TMO are dependent on the successful delivery of the construction project for strategies to be realised. The research did find conflicts between the TMO and the client, but the source of these derived from the procurement strategy, as a consequence of the competitive bidding process. Even within these circumstances, the TMO member organisation will seek to achieve the client's operational objectives, unless the strategy of the consultant, or contractor, is only shortterm. Secondly, The model of strategic alignment within a TMO, presented in Figure 6.5, identifies forces that influence the strategy of the TMO. Understanding of these influences will be particularly beneficial for project management practice. The model shows that TMO behaviour is not localised, but driven by the internal behaviours of the client organisation and the emergent influences within the external environment. Factors for consideration include, the strategic importance of the construction project that determines the deadline, the complexity of the project itself, and the structure of the TMO, in terms of task integration. The research also finds that strategy of the TMO is influenced by the long and short-term strategic objectives of the consultant and client organisations, and understanding of these would better support the management of projects.

The model in Figure 6.5 also makes explicit the impact of client leadership and behaviours on the strategy of the TMO. A critical lesson, from the findings of the study, for client organisations, is the need for efficient coordination of the TMO. Within public sector organisations, it is common to establish project boards as a governance mechanism. Within varied methodologies this is seen as best practice (APM, 2004). However, as demonstrated across the case studies informing this thesis, there need for a single point of contact between the project board and the TMO. The research shows that when a pluralistic client project board is responsible for instructing the TMO, directives received are often conflicting and ambiguous, which can result in frustration for the TMO and delays in the project.

7.5 Limitation

The research makes contribution to both theory and practice, as discussed in the preceding section. Despite these, the limitations of the research are acknowledged. Methodological limitations are identified in Section 4.7 and include the limitations in data collection and triangulation, where it is recommended by Eisenhardt (1989b), Denzin (1978) and King (1998) that multiple investigators collect, interpret and analyse data. However, this was deemed as inappropriate for PhD research where a single investigator is expected to conduct the research for award. Rigour was taken in the data collection and analysis thorough the case study protocol, summarised in Table 4.9 and the replication logic within the data collection.
The findings of the research were also limited by the available data collected. As discussed in Section 4.4.1, the four cases studies involved public sector projects. To ensure international and regional consistency, the construction projects were located within a 50-mile radius of each other. Perspectives were therefore taken from participants within a specific region of the UK, although it is accepted that cases from other parts of the UK might have presented additional findings, due to regional, economic and political differences. Furthermore, findings may be different to those that might emerge from commercial projects within the private sector.

Whereas, the investigation attempted to interview all relevant stakeholders within each case study, this was not possible. Due to the high mobility within the construction industry, some actors had since changed organisation and were not contactable. Others did not respond to the invitation for interview, or were not available for follow-up interview. Despite this, triangulation of data was employed and all interviews were transcribed as verbatim, for reliability.

Finally, the construction projects within each case study were also planned and executed at the height of 2008-2013 UK financial crises. As discussed within the thesis, the economic environment, at the time, did influence the strategic decisions and drivers of the clients, contractors and consultants within the study. There was also an impact on the availability of sub-contractors that was perceived as an external influence to the TMO strategy. Due to the limitation of time and resources to conduct the research for the thesis, the boundaries of the case study did not include perceptions of sub-contractors or utility supplies. Although, it is acknowledged that additional data from these external organisations may have provided interesting insight.

7.6 Further research

Taking into consideration the limitations of the research, identified in Section 7.5, areas for further research are identified.

The main focus of further research from the study would be to test the components of the theoretical model in Figure 6.5, to support its validation. The model of strategic alignment within the context of a TMO was developed from the qualitative findings from four case studies forming the research. Quantitative studies investigating each of the dimensions that influence the strategy would further elaborate on the themes evolving from the research. Important areas of study include the impact of client behaviours and leadership on project strategy and project success. It would also be interesting to draw comparison of external environmental factors outwith the economic uncertainty that influenced strategies within this thesis. Within this area, future research could consider the strategic motivations of sub-contractors and utility providers. Both of which have received little attention in the literature.

The research also recognised that client complexity has a major effect on the alignment of strategic objectives. Very few studies have investigated the role of the client in achieving project success within a pluralistic client body. Studies investigating client influence have tended to perceive the client as a single entity. However, the growth in joint ventures and industry collaborations, would suggest that the client is becoming even more multifaceted. The model in Figure 6.5 could, therefore, be applied to settings where the project client consists of multiple organisations. This would increase the complexity of organisations seeking to align multiple strategic objectives through a single construction project.

The implications of the research are not only limited to the construction industry. The thesis has successfully explored the implications of multiple organisations being involved in a single project. Further research could investigate this phenomenon across other industries. This could include product development projects involving multiple firms, or service projects involving multiple agencies. Within this environment, issues extend to the leadership of the pluralistic client body, as well as, the leadership of the TMO.

7.7 Final conclusions

Jacobsson *et al's* (2015) follow-up paper to Lundin and Soderholm's (1995) theory of temporary organisations suggests a need to address the ambiguities and confusion regarding projects and temporary organisations. This thesis makes significant contribution by providing an additional perspective of how projects can be understood. Previous studies have predominantly focused on the temporary organisation formed within the boundaries of single organisation. Within these studies, models propose that strategy is set at the highest level of a strategic hierarchy and cascades down to be implemented through projects.

However, the review of the literature establishes a lack of empirical studies exploring the concept of alignment of projects with organisational strategy. Moreover, the literature does not adequately address the complexities of how temporary organisations realise an organisations strategic objectives. This study has set out to explore how varied organisations within a temporary-multi organisation attempt to align individual strategic objectives through a single construction project. In achieving this aim, the study exposes a number of ambiguities and difficulties organisations face when seeking to realise strategic objectives through a TMO.

These include the strategy of the TMO, strategic priorities, and actions taken to deliver a construction project. The study shows that these are dynamic and subject to internal and external influences. An important contribution is the recognition of clients influence on the strategic behaviours of the TMO and consequential success of project. At an organisational level, this includes formal governance structures and conditions of contract, but also includes project leadership and client behaviours towards the TMO. Moreover, the study clarifies some of the ambiguities pertaining to the concept of project success. Through the devolvement of models from the findings of the empirical research, the thesis identifies the varied and competing perceptions of project success, and how they relate to organisational strategy.

The thesis has challenged a number of common perceptions regarding projects and temporary organisations. Strategies between organisations comprising a TMO do not conflict, as the literature suggests. Instead, the research demonstrates alignment between strategic objectives. However, there are differences of opinion as to perceptions of project success and, therefore, how success of a project should be pursued. But, as demonstrated within the study, success of a project is time-dependant and perceptions of success change over time.

Therefore, this thesis has achieved its aims and objectives. Through the three stages of research design, shown in Figure 4.2, the thesis developed a theoretical foundation through an extant literature review on the strategic context of projects and temporary organisational forms. Themes emerging from the literature guided the empirical research conducted across four case studies of construction projects within the public sector. Models developed from the findings and analysis of the study makes

contribution to theory and practice of strategic alignment, effectiveness of governance mechanisms, TMO strategy, and perceptions of project success.

- AALTO, T. (2000) 'Strategies and Methods for Project Portfolio Management', Tu-22.451 Seminar in Project Management Helsinki Helsinki University of Technology.
- ABEDNEGO, M. P. & OGUNLANA, S. O. (2006) 'Good Project Governance for Proper Risk Allocation in Public–Private Partnerships in Indonesia', *International Journal of Project Management*, 24, 622-34.
- AGA, D. A., NOORDERHAVEN, N. & VALLEJO, B. (2016) 'Transformational Leadership and Project Success: The Mediating Role of Team-Building', *International Journal of Project Management*, 34, 806-18.
- AHOLA, T., RUUSKA, I., ARTTO, K. & KUJALA, J. (2014) 'What Is Project Governance and What Are Its Origins?', *International Journal of Project Management*, 32, 1321-32.
- AKINTOYE, A. (1994) 'Design and Build: A Survey of Construction Contractors' Views', *Construction Management & Economics*, 12, 155.
- ALVESSON, M., KÄRREMAN, D., STURDY, A. & HANDLEY, K. (2009)
 'Unpacking the Client(S): Constructions, Positions and Client–Consultant Dynamics', *Scandinavian Journal of Management*, 25, 253-63.
- ANDREWS, K. (1971) *The Concept of Corporate Strategy*, Homewood, Illinois, Dow Jones Irwin.
- ANSOFF, H. I. (1965) Corporate Strategy: An Analytic Approach to Business Policy for Growth and Expansion, New York, McGraw-Hill
- APM (1995) APM Body of Knowledge, Buckinghamshire, Association for Project Management.
- APM (2004) Directing Change: A Guide to the Governance of Project Management, High Wycombe, UK, Association for Project Management.
- ARCHER, N. P. & GHASEMZADEH, F. (1999) 'An Integrated Framework for Project Portfolio Selection', *International Journal of Project Management*, 17, 207-16.
- ARCHIBALD, R. D. (1987) 'Key Milestones in the Early Pert/CPM/PDM Days', *Project Management Journal*, 18, 29-31.
- ARCHIBALD, R. D. (1988) 'Projects: Vehicles for Strategic Growth ', *Project* Management Journal, 19, 31-33.
- ARNAUD, G. (1998) 'The Obscure Object of Demand in Consultancy: A Psychoanalytic Perspective', *Journal of Managerial Psychology*, 13, 469-84.
- ARSKEY, H. & KNIGHT, P. (1999) Interviewing for Social Scientists London, Sage.
- ARTTO, K. (2013) 'A Chunk View of Research into Temporary Organizations', International Journal of Managing Projects in Business, 6, 595-603.
- ARTTO, K. & KUJALA, J. (2008) 'Project Business as a Research Field', *International Journal of Managing Projects in Business*, 1, 469-97.
- ARTTO, K. A., MARTINSUO, M., DIETRICH, P. & KUJALA, J. (2008) 'Project Strategy: Strategy Types and Their Contents in Innovation Projects', *International Journal of Managing Projects in Business* 1, 49-70.
- AVOLIO, B. J. & BASS, B. M. (1991) *The Full Range Leadership Development Programs: Basic and Advanced Manuals*, Binghampton, NY, Bass, Avolio & Associates.
- AVOLIO, B. J. & BASS, B. M. (2002) *Developing Potential across a Full Range of Leadership* [™]: *Cases on Transactional and Transformational Leadership*, Mahwah, NJ, Lawrence Erlbaum Associates.
- AVOLIO, B. J., WALDMAN, D. A. & YAMMARINO, F. J. (1991) 'Leading in the 1990s: The Four I'S of Transformational Leadership', *Journal of European Industrial Training*, 15, 9-18.

- AVOTS, I. (1969) 'Why Does Project Management Fail?', *California Management Review*, 12, 77-82.
- AYRES, L., KAVANAUGH, K. & KNAFL, K. A. (2003) 'Within-Case and across-Case Approaches to Qualitative Data Analysis', *Qualitative Health Research*, 13, 871-83.
- BACCARINI, D. (1996) 'The Concept of Project Complexity: A Review', *International Journal of Project Management*, 14, 201-04.
- BACCARINI, D. (1999) 'The Logical Framework Method for Defining Project Success', *Project Management Journal*, 30, 25.
- BAIN, J. S. (1950) 'Workable Competition in Oligopoly: Theoretical Considerations and Some Empirical Evidence', *American Economic Review*., Vol.40, 35-47.
- BAKER, W. E. & FAULKNER, R. R. (1991) 'Role as Reseource in the Hollywood Film Industry', *American Journal of Sociology*, 96, 279-309.
- BAKHSHI, J., IRELAND, V. & GOROD, A. (2016) 'Clarifying the Project Complexity Construct: Past, Present and Future', *International Journal of Project Management*, 34, 1199-213.
- BAKKER, R. M. (2010) 'Taking Stock of Temporary Organizational Forms: A Systematic Review and Research Agenda R.M. Bakker Temporary Organizational Forms', *International Journal of Management Reviews*, 12, 466-86.
- BAKKER, R. M., BOROŞ, S., KENIS, P. & OERLEMANS, L., A. G. (2013) 'It's Only Temporary: Time Frame and the Dynamics of Creative Project Teams', *British Journal of Management*, 24, 383-97.
- BANA E COSTA, C. A., NUNES DA SILVA, F. & VANSNICK, J.-C. (2001) 'Conflict Dissolution in the Public Sector: A Case-Study', *European Journal of Operational Research*, 130, 388-401.
- BASS, B. M. (1985) *Leadership and Performance Beyond Expectations*, New York, Free Press.
- BASS, B. M. (1990) 'From Transactional to Transformational Leadership: Learning to Share the Vision', *Organizational Dynamics*, 18, 19-31.
- BASS, B. M. (1999) 'Two Decades of Research and Development in Transformational Leadership', *European Journal of Work and Organizational Psychology*, 8, 9-32.
- BASS, B. M. & AVOLIO, B. J. (1994) *Improving Organizational Effectiveness Through transformational Leadership*, Thousand Oaks, California Sage Publications, Incorporated.
- BAXTER, P. & JACK, S. (2008) 'Qualitative Case Study Methodology: Study Design and Implementation for Novice Researcher', *The Qualitative Report*, 13, 544-59.
- BAZELEY, P. (2013) *Qualitative Data Analysis: Practical Strategies*, London, Sage Publications Ltd.
- BAZELEY, P. & JACKSON, K. (2013) *Qualitative Data Analysis with Nvivo*, London, Sage Publications Ltd.
- BEARD, D. W. & DESS, G. G. (1981) 'Corporate-Level Strategy, Business-Level Strategy, and Firm Performance', *Academy of Management Journal*, 24, 663-88.
- BECHKY, B. A. (2006) 'Gaffers, Gofers, and Grips: Role-Based Coordination in Temporary Organizations', *Organization Science*, 17, 3-21.
- BEKKER, M. & STEYN, H. (2009) 'Defining 'Project Governance' for Large Capital Projects', 2009, 20.
- BEKKER, M. C. (2014) 'Project Governance: "Schools of Thought", South African Journal of Economic and Management Sciences, 7, 22-32.
- BELASSI, W. & TUKEL, O. I. (1996) 'A New Framework for Determining Critical Success/Failure Factors in Projects', *International Journal of Project Management*, 14, 141-51.

- BENNIS, W. G. (1965) *Beyond Bureaucracy: Will Organization Men Fit the New Organization*, Philadelphia, PA, Trans-Action.
- BENNIS, W. G. & SLATER, P. E. (1968) *The Temporary Society*, New York, NY, Harper & Row.
- BERG, V. (2001) *Qualitative Research Methods for the Social Sciences*, Needham Heights, MA, Allyn and Bacon.
- BERTELSEN, S. & EMMITT, S. (2005) 'The Client as a Complex System', *13th* Annual Conference in the International Group for Lean Construction. Sydney, Australia
- BLISMAS, N. (2001) 'Multi-Project Environment of Construction Clients', Loughbororough, University of Loughborough
- BLISMAS, N., SHER, W., THORPE, A. & BALDWIN, A. (2004) 'A Typology for Clients' Multi-Project Environments', *Construction Management and Economics*, 22, 375-71.
- BLUMER, H. (1962) 'Society as Symbolic Interaction', In Rose, A. M. (ed) *Human Behviour and Social Processes.* Boston, MA, Houghton.
- BMG, R. (2015) 'Workforce Mobility and Skills in the Uk Construction Sector 2015', UK, Construction Industry Training Board (CITB).
- BOSCH-SIJTSEMA, P. M. & POSTMA, T. J. B. M. (2010) 'Governance Factors Enabling Knowledge Transfer in Interorganisational Development Projects', *Technology Analysis & Strategic Management*, 22, 593-608.
- BOULDING, K. E. (1956) 'General Systems Theory the Skeleton of Science', Management Science, 2, 197-208.
- BOURGEOIS, L. J., III (1980) 'Strategy and Environment: A Conceptual Integration', *The Academy of Management Review*, 5, 25-39.
- BOURNE, L. & WALKER, D. H. T. (2006) 'Visualizing Stakeholder Influence--Two Australian Examples', *Project Management Journal*, 37, 5-21.
- BRADY, T. & DAVIES, A. (2004) 'Building Project Capabilities: From Exploratory to Exploitative Learning', *Organization Studies*, 25, 1601-21.
- BRESNEN, M., GOUSSEVSKAIA, A. & SWAN, J. (2005) 'Managing Projects as Complex Social Settings', *Building Research & Information*. Routledge.
- BROMLEY, D. B. (1986) *The Case-Study Method in Psychology and Related-Disciplines*, Chichester, John Wiley & Sons.
- BRYANT, D. T., FOSTER, P. M., SPINK, P. K. & LUCKMAN, J. (1978)
 Multiorganizational Relationships on Large Building Sites and Their Influence on Morale and Effectiveness, London, Tavistock Institute of Human Relations.
- BRYMAN, A. & BELL, E. (2007) *Business Research Methods*, New York, Oxford University Press.
- BRYMAN, A., BRESNEN, M., BEARDSWORTH, A. D., FORD, J. & KEIL, E. T. (1987a) 'The Concept of Temporary the System: The Case of the Construction Project', *Research in the sociology of organizations: A research annual*, 5, 253-83.
- BRYMAN, A., BRESNEN, M., FORD, J., BEARDSWORTH, A. & KEIL, T. (1987b)
 'Leader Orientation and Organizational Transience: An Investigation Using Fiedler's LPC Scale', *Journal of Occupational Psychology*, 60, 13-19.
- BSI (1996) 'Project Management. Guide to Project Management', *BS 6079.* British Standards Institute.
- BUDAYAN, C., DIKMEN, I. & BIRGONUL, M. T. (2015) 'Alignment of Project Management with Business Strategy in Construction: Evidence from the Turkish Contractors', *Journal of Civil Engineering and Management*, 21, 94-106.
- BURGESS, R. G. (1982) Field Research: A Sourcebook and Field Manual (Contemporary Social Research), Abingdon, Oxon, Routedge.

BURNS, J. M. (1978) Leadership, New York, Harper and Row.

- BURNS, T. & STALKER, G. M. (1961) *The Management of Innovation*, London, Tabistock.
- BURRELL, G. & MORGAN, G. (1979) Socialogical Paradigms and Organisational Analysis: Elements of the Socialogy of Corporate Life, Hants, Ashgate Publishing Limited.
- BURSTRÖM, T. & JACOBSSON, M. (2012) 'Transition Processes in an Interorganizational Platform Project', *International Journal of Managing Projects in Business*, 5, 400.
- CARMELI, A., GELBARD, R. & GEFEN, D. (2010) 'The Importance of Innovation Leadership in Cultivating Strategic Fit and Enhancing Firm Performance', *The Leadership Quarterly*, 21, 339-49.
- CHAFFEE, E. E. (1985) 'Three Models of Strategy', *The Academy of Management Review*, 10, 89-98.
- CHANDLER, A. D. (1962) Strategy and Structure: Chapters in the History of the American Industrial Enterprise Cambridge, MA, MIT Press.
- CHERNS, A. B. & BRYANT, D. T. (1984) 'Studying the Client's Role in Construction Management', *Construction Management & Economics*, 2, 177-84.
- CHORN, N. H. (1991) 'The "Alignment" Theory: Creating Strategic Fit', *Management Decision*, 29, null.
- CIC (2009) 'The Impact of the Recession on Construction Professional Services', London, Construction Industry Council.
- CLARK, P. (1985) 'A Review of Theories of Time and Structure for Oganizational Sociology ', *Research in the sociology of organizations*, 4, 35-79.
- CLELAND, D. I. (1998) 'Project Stakeholder Management', In Cleland, D. I. & King, W. R. (eds) *Project Management Handbook*. New York, John Wiley & Sons.
- CLELAND, D. I. (2007) 'Strategic Management: The Project Linkages', In Morris, P. W. G. & Pinto, J. K. (eds) *The Wiley Guide to Project, Program & Portfolio Management*. New Jersey, John Wiley & Sons.
- CLELAND, D. I. & IRELAND, L. R. (2006) *Project Management: Strategic Design* and Implementation New York, McGraw-Hill, Inc.
- COFFEY, A. & ATKINSON, P. (1996) *Making Sense of Qualitative Data.*, Thousand Oaks, CA, Sage Publications Ltd.
- COGHLAN, D. & BRANNICK, T. (2005) *Doing Action Research in Your Own Organization*, London, Sage Ltd.
- COHEN, L., MANION, L. & MORRISON, K. (2007) *Research Methods in Education*, Abingdon, Oxon, Routledege.
- COLLINS, A. & BACCARINI, D. (2004) 'Project Success: A Survey', *Journal of Construction Research*, 5, 211-31.
- COLLIS, J. & HUSSEY, R. (2009) Business Research: A Practical Guide for Undergarduate and Postgraduate Students, Basingstoke, Hampshire, Palgrave Macmillan.
- CONNAUGHTON, J. & WELLER, S. (2013) 'Improving Collaboration in Construction: An Opportunity for Action Research', In Smith, S. D. & Ahiaha-Dagbui, D. D. (eds) 29th Annual ARCOM Conference. Reading, UK, Association of Researchers in Construction Management.
- COOKE-DAVIES, T. (2002) 'The "Real" Success Factors on Projects', International Journal of Project Management, 20, 185-90.
- COOKE-DAVIES, T. J. (2007) 'Project Success', In Moran, P. & Pinto, J. K. (eds) *The Wiley Guide to Project, Program & Portfolio Management*. New Jersey, John Wiley & Sons Inc.

- COOKE-DAVIES, T. J., CRAWFORD, L. H. & LECHLER, T. G. (2009) 'Project Management Systems: Moving Project Management from an Operational to a Strategic Discipline', *Project Management Journal*, 40, 110-23.
- COOPER, G. R., EDGETT, S. J. & KLEINSCHMIDT, E. J. (1999) 'Portfolio Management in New Product Development: Lessons from the Leaders, Phase II', In Dye, L. D. & Pennypacker, J. S. (eds) *Project Portfolio Management: Selecting and Prioritizing Projects for Competitive Advantage*. Havertown, Center for Business Practices.
- CRAWFORD, L. (2005) 'Senior Management Perceptions of Project Management Competence', *International Journal of Project Management*, 23, 7-16.
- CRESWELL, J. W. (2007) *Qualitative Inquiry and Research Design: Choosing among Five Approaches,* Thousand Oaks, CA, Sage Publications Ltd.
- CROTTY, M. (1998) *The Foundations of Social Research: Meaning and Perspective in the Research Process*, St Leonards, Australia, Sage Publications Limited.
- CURTIS, B., WARD, S. & CHAPMAN, C. (1991) *Roles, Responsibilities and Risks in Management Contracting,* London, Construction Industry Research & Information Association (CIRIA).
- CYERT, R. M. & MARCH, J. G. (1963) *A Behavioral Theory of the Firm* Englewood Cliffs, N.J., Prentice-Hall
- DAVIES, A., GANN, D. & DOUGLAS, T. (2009) 'Innovation in Megaprojects: Systems Integration at London Heathrow Terminal 5', *California Management Review*, 51, 101-25.
- DAVIES, A. & MACKENZIE, I. (2014) 'Project Complexity and Systems Integration: Constructing the London 2012 Olympics and Paralympics Games', *International Journal of Project Management*, 32, 773-90.
- DAVIES, P. R. & WALKER, D. H. T. (2008) 'Trust, Commitment and Mutual Goals in Australian Construction Industry Project Alliances', In Walker, D. H. T. & Rowlinson, S. (eds) *Procurement Systems: A Cross-Industry Project Management Perspective*. Abingdon, Oxon, Taylor & Francis.
- DAVIS, K. (2014) 'Different Stakeholder Groups and Their Perceptions of Project Success', *International Journal of Project Management*, 32, 189-201.
- DE BLOIS, M. & LIZARRALDE, G. (2010) 'A System of Classification of Temporary Multi-Organizations in the Building Sector', *CIB World Building Congress*. Salford.
- DE WIT, A. (1988) 'Measurement of Project Success', International Journal of Project Management, 6, 164-70.
- DEFILLIPPI, R. J. & ARTHUR, M. B. (1998) 'Paradox in Project- Based Enterprise: The Case of Film Making', 40, 125-39.
- DENZIN, N. K. (1978) Research Act: A Theoretical Introduction to Sociological Methods, McGraw-Hill.
- DIALSINGH, I. (2008) 'Face-to-Face Interviewing', In Lavrakas, P. J. (ed) *Encyclopedia of Survey Research Methods.* Thousand Oaks, CA, SAGE Publications, Inc.
- DIENER, E. & CRANDALL, R. (1978) *Ethics in Social and Behavioural Research,* Chicago, University of Chicago Press.
- DOD/NASA (1962) *DoD/NASA Pert/ Cost Guide*, Washington, DC, US Govermenet Printing Office.
- DVIR, D., LIPOVETSKY, S., SHENHAR, A. & TISHLER, A. (1998) 'In Search of Project Classification: A Non-Universal Approach to Project Success Factors', *Research Policy*, 27, 915-35.
- DYE, L. D. & PENNYPACKER, J. S. (2002) 'Project Portfolio Management and Managing Multiple Projects: Two Sides of the Same Coin?', In Pennypacker, J.

S. & Dye, L. D. (eds) *Managing Multiple Projects*. New York, NY, Marcel Dekker.

- DYER JR, W. G. & WILKINS, A. L. (1991) 'Better Stories, Not Better Constructs, to Generate Better Theory: A Rejoinder to Eisenhardt', *Academy of Management Review*. Academy of Management.
- EASTERBY-SMITH, M., THORPE, R. & LOWE, A. (1991) Management Research: An Introduction, London, Sage.
- ECCLES, R. G. (1981) 'The Quasifirm in the Construction Industry', *Journal of Economic Behavior & Organization*, 2, 335-57.
- EGAN, J. (1998) 'Rethinking Construction: Report of the Construction Task Force.', London, HSMO.
- EISENHARDT, K. M. (1989a) 'Agency Theory: An Assessment and Review', Academy of Management Review, 14, 57-74.
- EISENHARDT, K. M. (1989b) 'Building Theories from Case Study Research', *The Academy of Management Review*, 14, 532-50.
- EISENHARDT, K. M. (1991) 'Better Stories and Better Constructs: The Case for Rigor and Comparative Logic', *Academy of Management Review*, 16, 620-27.
- EL-RAYES, K. & MOSELHI, O. (1998) 'Resource-Driven Scheduling of Repetitive Activities', *Construction Management & Economics*, 16, 433-46.
- ELORANTA, K., KUJALA, J. & ARTTO, K. (2006) 'Subcontractors' Inter-Organizational Relationships as Risk Sources in Project Networks', Seventh International Conference of the International Research Network on Organising by Projects, IRNOP VII. Xi'an.
- EVARISTO, R. & VAN FENEMA, P. C. (1999) 'A Typology of Project Management: Emergence and Evolution of New Forms', *International Journal of Project Management*, 17, 275-81.
- FIELDER, F. E. (1967) *A Theory of Leadership Effectiveness*, New York, McGraw-Hill.
- FISH, G. (1961) 'Funtional Teamwork', Harvard Business Review.
- FITZGERALD, B. & HOWCROFT, D. (1998) 'Competeing Dichotomies in IS Research and Possible Strategies for Resolution', 19th International Conference on Information Systems ICIS. Helsinki, Finland.
- FLETCHER, M. & PLAKOYIANNAKI, E. (2010) 'Sampling', In Mills, A. J., Durepos, G. & Wlebe, E. (eds) *Encyclopedia of Case Study Research. SAGE Publications, Inc.* Thousand Oaks, CA, SAGE Publications, Inc.
- FLICK, U. (2009) *An Introduction to Qualitative Research*, London, SAGE Publications Ltd.
- FLORICEL, S. & MILLER, R. (2001) 'Strategizing for Anticipated Risks and Turbulence in Large-Scale Engineering Projects', *International Journal of Project Management*, 19, 445-55.
- FLYVBJERG, B. (2006) 'Five Misunderstandings About Case-Study Research', *Qualitative Inquiry*, 12, 219-45.
- FONDAHL, J. W. (1987) 'Precedence Diagramming Methods: Origins and Early Development ', *Project Management Journal*, 18, 33-36.
- FREEMAN, M. & BEALE, P. (1992) 'Measuring Project Success', Project Management Journal, 23, 8-17.
- FRY, L. W. & SMITH, D. A. (1987) 'Congruence, Contingency, and Theory Building.', Academy of Management Review, 12, 117-32.
- GADDIS, P. O. (1959) 'The Project Manager', Harvard Business Review, 37, 88-97.
- GALBRAITH, J. R. (1971) 'Matrix Organisation Designs: How to Combine Functional and Project Forms', *Business Horizons*, 14, 29-40.

- GALBRAITH, J. R., NATHANSON, D. A. & KAZANJIAN, R. K. (1986) Strategy Implementation: Structure, Systems and Process, St. Paul, MN, West Publishing.
- GANN, D. M. & SALTER, A. J. (2000) 'Innovation in Project-Based, Service-Enhanced Firms: The Construction of Complex Products and Systems', *Research Policy*, 29, 955-72.
- GAREIS, R. (1989) ''Management by Projects': The Management Approach for the Future', *International Journal of Project Management*, 7, 243-49.
- GAREIS, R. (1991) 'Management by Projects: The Management Strategy of the 'New' Project-Oriented Company', *International Journal of Project Management*, 9.
- GINSBERG, A. & VENKATRAMAN, N. (1985) 'Contingency Perspectives of Organizational Strategy: A Critical Review of the Empirical Research', *Academy* of Management Review, 10, 421-34.
- GLASER, B. & STRAUSS, A. (1967) *The Discovery of Grunded Theory: Strategies of Qualitative Research*, London, Wiedenfeld & Nicholson.
- GLUCH, P. & RAISANEN, C. (2009) 'Interactional Perspective on Environmental Communication in Construction Projects', *Building Research & Information*, 37, 164-75.
- GOODMAN, R. A. & GOODMAN, L. P. (1976) 'Some Management Issues in Temporary Systems: A Study of Professional Development and Manpower--the Theater Case', *Administrative Science Quarterly*, 21, 494-501.
- GRABHER, G. (2004) 'Temporary Architectures of Learning: Knowledge Governance in Project Ecologies', *Organization Studies*, 25, 1491-514.
- GRANOVETTER, M. (1992) 'Problems of Explanation in Economic Sociology ', In Nohria, N. & Eccles, R. G. (eds) *Networks and Organizations: Structure, Form and Action.* Boston, Harvard Business School Press.
- GRANT, R. M. (2005) *Contemporary Strategy Analysis*, Malden, MA, Blackwell Publishing.
- GRANT, R. M. (2010) 'Corporate Strategy: Managing Scope and Strategy Content', In Pettigrew, A., Tomas, H. & Whittington, R. (eds) *Handbook of Strategy & Management*. London, Sage.
- GRUN, O. (2004) Taming Giant Projects: Management of Multi-Organization Enterprises (Organization and Management Innovation), Heidelberg, Germany, Springer-Verlag.
- GRUNDY, T. (1998) 'Strategy Implementation and Project Management', *International Journal of Project Management*, 16, 43-50.
- GUANGQUAN, X., ZHIYONG, F., HUABEI, W. & DEXIN, Z. (2007) 'Swift Trust in a Virtual Temporary System: A Model Based on the Dempster-Shafer Theory of Belief Functions', *International Journal of Electronic Commerce*, 12, 93-126.
- GUBA, E. G. & LINCOLN, Y. S. (1998) 'The Landscape of Qualitative Research: Theories & Issues', In Denzin, N. K. & Lincoln, Y. S. (eds) *Handbook of Qualitative Research*. Thousand Oaks, California, Sage Publications.
- GUPTA, A. K. & GOVINDARAJAN, V. (1986) 'Resource Sharing among SBUs: Strategic Antecedents and Administrative Implications', *Academy of Management Journal*, 29, 695-714.
- HAMBRICK, D. C. (1983) 'Some Tests of the Effectiveness and Functional Attributes of Miles and Snow's Strategic Types', *The Academy of Management Journal*, 26, 5-26.
- HAMBRICK, D. C., MACMILLAN, I. C. & DAY, D. L. (1982) 'Strategic Attributes and Performance in the Bcg Matrix: A Pims-Based Analysis of Industrial Product Businesses', *Academy of Management Journal*, 25, 510-31.

- HAMEL, J., DUFOUR, S. & FORTIN, D. (1993) *Case Study Methods,* Newbury Park, California, Sage Publications.
- HANIFF, A. P. & FERNIE, S. (2008) 'Projects: Where Strategies Collide', In Carter, K., Ogunlana, S. & Kaka, A. (eds) *Transformation through Construction. Joint* 2008 CIB W065/W055. Dubia CIB.
- HANISCH, B. & WALD, A. (2011) 'A Project Management Research Framework Integrating Multiple Theoretical Perspectives and Influencing Factors', *Project Management Journal*, 42, 4-22.
- HARTLEY, J. (2004) 'Case Study Research', In Cassell, C. & Symon, G. (eds) *Essentail Guide to Qualitative Methods in Organizational Research*. London, Sage Publications.
- HARTMAN, F. (2000) Don't Park Your Brain Outside: A Practical Guide to Improving Shareholder Value with Smart Management, Project Management Institute.
- HATTEN, K. J. & SCHENDEL, D. E. (1977) 'Heterogeneity within an Industry: Firm Conduct in the Us Brewing Industry, 1952-1971', *Journal of Industrial Economics*, 26, 97-113.
- HAUC, A. & KOVAC, J. (2000) 'Project Management in Strategy Implementation: Experiences in Slovenia', *International Journal of Project Management*, 18, 61-67.
- HELLGREN, B. & STJERNBERG, T. (1995) 'Project Management and Temporary Organozationsdesign and Implementation in Major Investments - a Project Network Approach', *Scandinavian Journal of Management*, 11, 377-94.
- HENDERSON, J. C. & VENKATRAMAN, N. (1993) 'Strategic Alignment: Leveraging Information Technology for Transforming Organizations', *IBM Systems Journal*, 32, 472-85.
- HERON, J. (1996) *Co-Operative Inquiry: Research into the Human Condition*, London, Sage Publishing.
- HERON, J. & REASON, P. (1997) 'A Participatory Inquiry Paradigm', *Qualitative Inquiry*, 3, 274-94.
- HERRIOTT, R. R. & FIRESONE, W. A. (1983) 'Multisite Qualitative Policy Research: Optimizing Description and Generalizability', *Educational Researcher*, 12, 14-19.
- HOBDAY, M. (2000) 'The Project-Based Organisation: An Ideal Form for Managing Complex Products and Systems?', *Research Policy*, 29, 871-93.
- HODGSON, D. & CICMIL, S. (2006) 'Are Projects Real? The PMBok and the Legitimation of Project Management Knowledge ', In Hodgson, D. & Cicmil, S. (eds) *Making Projects Critical*. New York Palgrave MacMillan.
- HOFER, C. (1973) 'Some Preliminary Research on Patterns of Strategic Behaviour', In Proceedings, A. O. M. (ed).
- HOFER, C. & SCHENDEL, D. (1978) *Strategy Formulation: Analytical Concepts* West Publishing Company.
- HOLDEN, M. T. & LYNCH, P. (2004) ' Choosing the Appropriate Methodology: Understanding Research Philosophy', *Marketing Review*, 4, 397-409.
- HREBINIAK, L. G. & JOYCE, W. F. (1985) *Implementing Stratgy*, New York, Collins McMillan.
- HWU (2010) 'Research Ethics Policy and Procedures', Heriot Watt University.
- HYARI, K. H., EL-RAYES, K. & EL-MASHALEH, M. (2009) 'Automated Trade-Off between Time and Cost in Planning Repetitive Construction Projects', *Construction Management & Economics*, 27, 749-61.
- IBERT, O. (2004) 'Projects and Firms as Discordant Complements: Organisational Learning in the Munich Software Ecology', *Research Policy*, 33, 1529-46.

- IKA, L. A. (2009) 'Project Success as a Topic in Project Management Journals', Project Management Journal, 40, 6-19.
- IOANNOU, P. & LEU, S. (1993) 'Average Bid Method: Competitive Bidding Strategy', Journal of Construction Engineering and Management, 119, 131-47.
- JACOBSSON, M., BURSTRÖM, T. & WILSON, T. L. (2013) 'The Role of Transition in Temporary Organizations: Linking the Temporary to the Permanent', *International Journal of Managing Projects in Business*, 6, 576-86.
- JACOBSSON, M., LUNDIN, R. A. & SÖDERHOLM, A. (2015) 'Researching Projects and Theorizing Families of Temporary Organizations', *Project Management Journal*, 46, 9-18.
- JANOWICZ-PANJAITAN, M., BAKKER, R. M. & KENIS, P. (2009) 'Research on Temporary Organizations: The State of the Art and Distinct Approaches Towards "Temporariness"', In Kenis, P., Janowicz-Panjaitan, M. & Bart, C. (eds) *Temporary Organizations: Prevalence, Logic and Effectiveness*. Cheltenham, UK, Edward Elgar.
- JARVENPAA, S. L., KNOLL, K. & LEIDNER, D. E. (1998) 'Is Anybody out There?', Journal of Management Information Systems, 14, 29-64.
- JARVENPAA, S. L. & LEIDNER, D. E. (1999) 'Communication and Trust in Global Virtual Teams', *Organization Science*, 10, 791-815.
- JENSEN, M. C. & MECKLING, W. H. (1976) 'Theory of the Firm: Managerial Behavior, Agency Costs and Ownership Structure', *Journal of Financial Economics*, 4, 305-60.
- JONES, C. & LICHTENSTEIN, B. B. (2008) 'Temporary Inter-Organizational Projects: How Temporal and Social Embeddedness Enhance Coordination and Manage Uncertainty', In Cropper, S., Huxham, C., Ebers, M. & Ring, P. S. (eds) *The Oxford Handbook of Inter-Organizational Relations*. Oxford University Press.
- JOSHI, M. P., KATHURIA, R. & PORTH, S. J. (2003) 'Alignment of Strategic Priorities and Performance: An Integration of Operations and Strategic Management Perspectives', *Journal of Operations Management*, 21, 353-69.
- JUGDEV, K. & MULLER, R. (2005) 'A Retrospective Look at Our Evolving Understanding of Project Success', *Project Management Journal*, 36, 19-31.
- KANGIS, P. & LEE-KELLEY, L. (2000) 'Project Leadership in Clinical Research Organisations', *International Journal of Project Management*, 18, 393-401.
- KATZ, D. & KHAN, R. L. (1966) *The Social Psychology of Organizatons*, New York, Wiley.
- KAVANAGH, D. & KELLY, S. (2002) 'Sensemaking, Safety, and Situated Communities in (Con)Temporary Networks', *Journal of Business Research*, 55, 583-94.
- KEITH, P. M. (1978) 'Individual and Organizational Correlates of a Temporary System', *The Journal of Applied Behavioral Science*, 14, 195-203.
- KELLY, J., MORLEDGE, R. & WILKINSON, S. (2002) *Best Value in Construction*, Oxford, Blackwell Publishing.
- KELLY, J. R. & LOVING, T. J. (2004) 'Time Pressure and Group Performance: Exploring Underlying Processes in the Attentional Focus Model', *Journal of Experimental Social Psychology*, 40, 185-98.
- KERZNER, H. (2001) Strategic Planning for Project Management Using a Project Management Maturity Model, New York. NY, John Wiley and Sons.
- KERZNER, H. (2003) Project Management: A Systems Approach to Planning, Scheduling, and Controlling, Hoboken, New Jersey, John Wiley & Sons Inc.
- KERZNER, H. (2004) Advanced Project Management: Best Practices on Implementation, Hoboken, New Jersey, John Wiley & Son.

- KING, N. (1998) 'Template Analysis', In Symon, G. & Cassell, C. (eds) *Qualitative Methods and Analysis in Organizational Research: A Practical Guide.* Thousand Oaks, CA, Sage Publications Ltd.
- KING, N. (2004) 'Using Tempates in the Thematic Analysis of Text', In Cassell, C. & Symon, G. (eds) *Essential Guide to Qualitative Methods in Organizational Research*. London, Sage Publications Ltd.
- KIRKPATRICK, S. W. (1965) 'A Highway Plan for Glasgow,' Glasgow, Glasgow City Council.
- KNIGHT, K. (1976) 'Matrix Organisation: A Review', *Journal of Management Studies*, 17, 111-30.
- KNIGHTFRANK (2011) 'Student Property 2011: The Case for Investment in Light of Changes to Higher Eduction '.
- LAM, E. W. M., CHAN, A. P. C. & CHAN, D. W. M. (2004) 'Benchmarking Design-Build Procurement Systems in Construction', *Benchmarking: An International Journal*, 11, 287-302.
- LAMPEL, J. & JHA, P. P. (2004) 'Models of Project Orientation in Multiproject Organizations ', In Morris, P. W. G. & Pinto, J. K. (eds) *The Wiley Guide to Managing Projects*. London, UK, John Wiley & Sons Inc.
- LANZARA, G. F. (1983) 'Emphemeral Organizations in Extreme Environments: Emergence, Strategy, Extinction', *Journal of Management Studies*, 20, 71-95.
- LARSON, E. W. & GRAY, C. F. (2014) 'Project Management: The Managerial Process.', 6th ed. New York, NY, McGraw Hill Education.
- LATHAM, M. (1994) 'Constructing the Team: Joint Review of Procurement and Contractual Arrangements in the United Kingdom Construction Industry', London, HMSO Publications.
- LAU, E. & ROWLINSON, S. (2011) 'The Implications of Trust in Relationships in Managing Construction Projects', *International Journal of Managing Projects in Business*, 4, 633-59.
- LAWRENCE, P. & LORCH, J. (1967) *Organizations and Environment,* Cambridge, MA, Harvard University Press.
- LEE, R. M. & FIELDING, N. G. (2011) 'Tools for Qualitative Data Analysis', In Hardy, M. & Bryman, A. (eds) *The Handbook of Data Analysis*. London, Sage Publications Ltd.
- LENZ, R. T. (1980) 'Environment, Strategy, Organization Structure and Performance: Patterns in One Industry', *Strategic Management Journal*, 1, 209-26.
- LEONARD-BARTON, D. (1990) 'A Dual Methodology for Case Studies: Synergistic Use of a Longitudinal Single Site with Replicated Multiple Sites', *Organization Science*, 1, 248-66.
- LEWIN, K. (1946) 'Action Research and Minority Problems', *Journal of Social Issues*, 2, 34-46.
- LIM, C. S. & MOHAMED, M. Z. (1999) 'Criteria of Project Success: An Exploratory Re-Examination', *International Journal of Project Management*, 17, 243-48.
- LINDEROTH, H. (2002) 'Bridging the Gap between Temporality and Permanency.', In Sahlin-Andersson K & Söderholm, A. (eds) *Beyond Project Management: New Perspectives on the Temporary-Permanent Dilemma*. Malmo, Oslo Copenhagen Business School Press.
- LINDKVIST, L. (2005) 'Knowledge Communities and Knowledge Collectivities: A Typology of Knowledge Work in Groups', *Journal of Management Studies*, 42, 1189-210.
- LIU, A. M. (1999) 'A Research Model of Project Complexity and Goal Commitment Effects on Project Outcome', *Engineering Construction & Architectural Management (Wiley-Blackwell)*, 6, 105-11.

- LIZARRALDE, G., BLOIS, M. D. & LATUNOVA, I. (2011) 'Structuring of Temporary Multi-Organizations: Contingency Theory in the Building Sector', *Project Management Journal*, 42, 19-36.
- LIZARRALDE, G., TOMIYOSHI, S., BOURGAULT, M., MALO, J. & CARDOSI, G. (2013) 'Understanding Differences in Construction Project Governance between Developed and Developing Countries', *Construction Management & Economics*, 31, 711-30.
- LOCH, C. (2000) 'Tailoring Product Development to Strategy: Case of a European Technology Manufacturer', *European Management Journal*, 18, 246-58.
- LORANGE, P. & VANCIL, R. F. (1976) 'How to Design a Strategic Planning System', *Harvard Business Review*, 54, 75-81.
- LUNDIN, R. A. (1995) 'Editorial: Temporary Organizations and Project Management', Scandinavian Journal of Management, 11, 315-18.
- LUNDIN, R. A. & SODERHOLM, A. (1995) 'A Theory of the Temporary Organization', *Scandinavian Journal of Management*, 11, 437-55.
- LUNDIN, R. A. & SÖDERHOLM, A. (2013) 'Temporary Organizations and End States: A Theory Is a Child of Its Time and in Need of Reconsideration and Reconstruction', *International Journal of Managing Projects in Business*, 6, 587-94.
- LYCETT, M., RASSAU, A. & DANSON, J. (2004) 'Programme Management: A Crtitical Review', *International Journal of Project Management*, 22, 289-99.
- LYNCH, R. (2006) Corporate Strategy. Harlow, UK, Pearson Education Ltd.
- MADISON, D. S. (2005) *Critical Ethnography: Methods, Ethics, and Performance.*, Thousand Oaks, CA, Sage publications.
- MAYLOR, H. (2001) 'Beyond the Gantt Chart: Project Management Moving On', *European Management Journal*, 19, 92-100.
- MAZUR, A., PISARSKI, A., CHANG, A. & ASHKANASY, N. M. (2014) 'Rating Defence Major Project Success: The Role of Personal Attributes and Stakeholder Relationships', *International Journal of Project Management*, 32, 944-57.
- MCCLINTOCK, C. C., BRANNON, D. & MAYNARD-MOODY, S. (1979) 'Applying the Logic of Sample Surveys to Qualitative Case Studies: The Case Cluster Method', *Administrative Science Quarterly*, 24, 612-29.
- MCGRATH, J. E. (1991) 'Time, Interaction, and Performance (Tip): A Theory of Groups', *Small Group Research*, 22, 147-74.
- MCGREGOR, D. (1966) Leadership and Motivation, Cambridge, MA, MIT Press.
- MENACERE, K. (2016) 'When More Does Not Mean Better: Selectig a Research Methodology and Methods', In Ahmed, V., Opoku, A. & Aziz, Z. (eds) *Research Methodology in the Built Environment: A Selection of Case Studies*. Abingdon, Oxon, Routledge.
- MEREDITH, J. E. & CHINOWSKY, P. S. (2000) 'Strategic Management in Construction'.
- MERRIAM, S. B. (1998) *Qualitative Research and Case Study Applications in Education*, San Francisco, California, Jossey-Bass
- MESKENDAHL, S. (2010) 'The Influence of Business Strategy on Project Portfolio Management and Its Success: A Conceptual Framework', *International Journal* of Project Management, 28, 807-17.
- MEYER, C. B. (2001) 'A Case in Case Study Methodology', Field Methods, 13, 329-52.
- MEYERSON, D., WEICK, K. E. & KRAMER, R. M. (1996) 'Swift Trust and Temporary Groups', In Kramer, R. M. & Tyler, T. (eds) Swift Trust and Temporary Groups. Trust in Organizations: Frontiers of Theory and Research. Thousand Oaks, CA, SAGE Publications, Inc.

- MILES, M., B. (1964) 'On Temporary Systems', In Miles, M. B. (ed) *Innovation in Education*. New York, John Wiley, .
- MILES, M. B. (1977) 'On the Origin of the Concept of the "Temporary System"', *Administrative Science Quarterly*, 22, 134-35.
- MILES, M. B. & HUBERMAN, A. M. (1994) *Qualitative Data Analysis: An Expanded Sourcebook,* Thousand Oaks, California, Sage Publications.
- MILES, R. E. & SNOW, C. C. (1978) *Organisational Strategy, Structure and Process,* Tokyo, Japan, McGraw Hill Kogakusha Ltd.
- MILLER, R., LESSARD, D. R., MICHAUD, P. & FLORICEL, S. (2014) *The Strategic Management of Large Engineering Projects: Shaping Institutions, Risks, and Governance,* Cambridge, Massachusetts, MIT Press.
- MILOSEVIC, D. Z. & SRIVANNABOON, S. (2006) 'A Theoretical Framework for Aligning Project Management with Business Strategy', *Project Management Journal*, 37, 98-110.
- MINTZBERG, H. (1978) 'Patterns in Strategy Formation', *Management Science*, 24, 934-48.
- MINTZBERG, H. (1987) 'The Strategy Concept I: Five Ps for Strategy', *California Management Review*, 30, 11-24.
- MINTZBERG, H. (1994) *The Rise and Fall of Strategic Planning*, New York, Free Press.
- MINTZBERG, H., AHLSTRAND, B. & LAMPEL, J. (1998) *Strategy Safari*, Harlow UK, Prentice Hall.
- MINTZBERG, H. & WATERS, J. A. (1985) 'Of Strategies, Deliberate and Emergent', *Strategic Management Journal*, 6, 257-72.
- MODIG, N. (2007) 'A Continuum of Organizations Formed to Carry out Projects: Temporary and Stationary Organization Forms', *International Journal of Project Management*, 25, 807-14.
- MOHAMMED, S. & NADKARNI, S. (2011) 'Temporal Diversity and Team Performance: The Moderating Role of Team Temporal Leadership', Academy of Management Journal, 54, 489-508.
- MORGAN, G. & SMIRCICH, L. (1980) 'The Case for Qualitative Research', Academy of Management Review, 5, 491-500.
- MORLEDGE, R. (2002) 'Procurement Strategies', In Kelly, J., Morledge, R. & Wilkinson, S. (eds) *Best Value in Construction*. Oxford, Blackwell Publishing.
- MORLEY, E. & SILVER, A. (1977) 'A Film Director's Approach to Managing Creativity', *Harvard Business Review*, 55, 59-70.
- MORRIS, P. W. G. (1982) 'Project Organizations: Structures for Managing Change', In Kelley, A. J. (ed) *New Dimensions in Project Management*. Lexington
- MORRIS, P. W. G. (1997) The Management of Projects, London, Thomas Telford.
- MORRIS, P. W. G. & JAMIESON, A. (2004) *Translating Corporate Strategy into Project Strategy: Realizing Corporate Strategy through Project Management* Pennsylvania, Project Management Institute Inc.
- MORRIS, P. W. G. & JAMIESON, A. (2005) 'Moving from Corporate Strategy to Project Strategy ', *Project Management Journal*, 36, 5-18.
- MÜLLER, R. (2009) *Project Governance*, Farnham, England, Gower Publishing Limited.
- MÜLLER, R. & TURNER, J. R. (2007) 'Matching the Project Manager's Leadership Style to Project Type', *International Journal of Project Management*, 25, 21-32.
- MURDOCH, J. & HUGHES, W. (1997) Construction Contracts: Law and Management London, E&F Spon.

 MURRAY, M., LANGFORD, D., HARDCASTLE, C. & TOOKEY, J. (1999)
 'Organisational Design', In Rowlinson, S. & Mcdermott, P. (eds) *Procurement Systems: A Guide to Best Practice in Construction*. London, E & FN Spon.

NADLER, D. A. & TUSHMAN, M. L. (1980) 'A Congruence Model for Diagnosing Organizational Behavior ', In Miles, R. H. (ed) *Resourcebook in Macro Organizational Behaviour*. California, Goodyear Publishing Company.

NAHAPIET, J. & GHOSHAL, S. (1998a) 'Social Capital, Intellectual Capital, and the Organizational Advantage', *Academy of Management Review*, 23, 242-66.

NAHAPIET, J. & GHOSHAL, S. (1998b) 'Social Capital, Intellectual Capital, and the Organizational Advantage', *Academy of Management Review*, 23, 242-66.

- NEUMAN, W. L. (2006) Social Research Methods: Qualitative and Quantitative Approaches, Boston, MA, Pearson.
- NEWCOMBE, R. (1996) 'Empowering the Construction Project Team', *International Journal of Project Management*, 14, 75-80.

NEWCOMBE, R. (2003) 'From Client to Project Stakeholders: A Stakeholder Mapping Approach', *Construction Management and Economics*, 21, 841-48.

- NORBURN, D. & MILLER, P. (1981) 'Strategy and Executive Reward: The Mis-Match in the Strategic Process', *Journal of General Management*, 6, 17-27.
- OECD (2004) *Oecd Principles of Corporate Governance*, Parice, France, Organisation for Economic Co-operation and Development.
- OGC (2003) 'Procurement and Contract Strategies ', *Achieving Excellence in Construction Procurement Guide.* London, Office of Government Commere.
- OISEN, R. P. (1971) 'Can Project Management Be Defined?', *Project Management Quarterly*, 21, 12-14.
- OLANDER, S. (2007) 'Stakeholder Impact Analysis in Construction Project Management', *Construction Management & Economics*, 25, 277-87.
- OYEGOKE, A., S, DICKINSON, M., KHALFAN, M., M. A., MCDERMOTT, P. & ROWLINSON, S. (2009) 'Construction Project Procurement Routes: An in-Depth Critique', *International Journal of Managing Projects in Business*, 2, 338-54.
- OYEGOKE, A. S. (2001) 'Uk and Us Construction Management Contracting Procedures and Practices: A Comparative Study', *Engineering Construction & Architectural Management (Wiley-Blackwell)*, 8, 403-17.
- OZCAN, P. & EISENHARDT, K. M. (2009) 'Origin of Alliance Portfolios: Entrepreneurs, Network Strategies, and Firm Performance', *Academy of Management Journal*, 52, 246-79.
- PACKENDORFF, J. (1995) 'Inquiring into the Temporary Organization: New Directions for Project Management Research', *Scandinavian Journal of Management*, 11, 319-33.
- PALISI, J., BARTOLOMEO (1970) 'Some Suggestions About the Transitory-Permanence Dimension of Organizations', *British Journal of Sociology*, 21, 200-06.
- PARTINGTON, D. (1996) 'The Project Management of Organizational Change', International Journal of Project Management, 14, 13-21.
- PATANAKUL, P. & SHENHAR, A. J. (2012) 'What Project Strategy Really Is: The Fundamental Building Block in Strategic Project Management', *Project Management Journal*, 43, 4-20.
- PEARCE, J. A. & ROBINSON, R. B. (2003) Strategic Management: Formulation, Implementation, and Control, New York. NY, McGraw-Hill Irwin.
- PELLEGRINELLI, S. & BOWMAN, C. (1994) 'Implementing Strategy through Projects', *Long Range Planning*, 27, 125-32.

- PELLEGRINELLI, S., PARTINGTON, D., HEMINGWAY, C., MOHDZAIN, Z. & SHAH, M. (2007) 'The Importance of Context in Programme Management: An Empirical Review of Programme Practices', *International Journal of Project Management*, 25, 41-55.
- PEMSEL, S. & MÜLLER, R. (2012) 'The Governance of Knowledge in Project-Based Organizations', *International Journal of Project Management*, 30, 865-76.
- PERROW, C. (1967) 'A Framework for the Comparative Analysis of Organizations', *American Sociological Review*, 32, 194-208.
- PERRY, C., RIEGE, A. & BROWN, L. (1999) 'Realism's Role among Scientific Paradigms in Marketing Research', *Irish Marketing Review*, 12, 16-23.
- PINTO, J. K. (2013) *Project Management: Achieving a Competitive Advantage*, Harlow, Essex, Pearson Education Limited.
- PINTO, J. K. & MANTEL, S. J., JR. (1990) 'The Causes of Project Failure', Engineering Management, IEEE Transactions on, 37, 269-76.
- PINTO, J. K. & SLEVIN, D. P. (1988) 'Project Success: Definitions and Measurement Techniques', *Journal of Project Management*, 19, 67-72.
- PMI (1987) 'Project Management Body of Knowledge of the Project Management Institute (PMI)', In Commitee, P. M. I. S. (ed) Upper Derby, IL, Project Management Institute.
- PMI (2013) 'A Guide to the Project Management Body of Knowledge: Pmbok Guide', 5th ed. Pennsylvania Project Management Institute.
- POLLACK, J. (2007) 'The Changing Paradigms of Project Management', *International Journal of Project Management*, 25, 266-74.
- PORTER, M. E. (1980) Competitive Strategy, New York, US, The Free Press.
- PORTER, M. E. (1985) Competitive Advantage: Creating and Sustaining Superior Performance, New York, Free Press.
- POWELL, A., PICCOLI, G. & IVES, B. (2004) 'Virtual Teams: A Review of Current Literature and Directions for Future Research', *SIGMIS Database*, 35, 6-36.
- PRAHALAD, C. K. & HAMEL, G. (1990) 'The Core Competence of the Corporation', *Harvard Business Review*, 68, 79-91.
- PRASAD, P. (2005) Crafting Qualitative Research: Working in the Postpositive Traditions, New York, M. E. Sharp Inc.
- PRENCIPE, A. & TELL, F. (2001) 'Inter-Project Learning: Processes and Outcomes of Knowledge Codification in Project-Based Firms', *Research Policy*, 30, 1373-94.
- PRYKE, S. D. (2004) 'Analysing Construction Project Coalitions: Exploring the Application of Social Network Analysis', *Construction Management & Economics*, 22, 787-97.
- REMENYI, D., WILLIAMNS, B., MONEY, A. & SWARTZ, E. (1998) Doing Research in Business Management: An Introduction to Process and Method, London, Sage Publications Lts.
- RICS (2013) 'Developing a Construction Procurement Strategy and Selecting an Appropriate Route', *RICS guidance note*. 1st ed. London, Royal Institute of Chartered Surveyors.
- RITSON, G., JOHANSEN, E. & OSBORNE, A. (2012) 'Successful Programs Wanted: Exploring the Impact of Alignment', *Project Management Journal*, 43, 21-36.
- ROBERT JR, L. P., DENNIS, A. R. & HUNG, Y.-T. C. (2009) 'Individual Swift Trust and Knowledge-Based Trust in Face-to-Face and Virtual Team Members', *Journal of Management Information Systems*, 26, 241-79.
- ROBSON, C. (2002) Real World Research Oxford, Blackwell.
- ROWLINSON, S. (1999) 'A Definition of Procurement Systems', In Rowlinson, S. & Mcdermott, P. (eds) *Procurement Systems: A Guide to Best Practice in Construction.* London, E & F Spon.

- ROWLINSON, S. M. (1988) 'An Analysis of Factors Affecting Project Perforamance in Industrial Building (with Particular Reference to Design Build Contracts)', *Department of Building Technology*. Hong Kong, Brunel University.
- RUMELT, R. P. (1974) *Strategy, Structure, and Economic Performance* Cambridge, Massachusetts, Harvard University Press.
- RUMELT, R. P. (1982) 'Diversification Strategy and Profitability', *Strategic Management Journal*, 3, 359-69.
- RUUSKA, I., AHOLA, T., ARTTO, K., LOCATELLI, G. & MANCINI, M. (2011) 'A New Governance Approach for Multi-Firm Projects: Lessons from Olkiluoto 3 and Flamanville 3 Nuclear Power Plant Projects', *International Journal of Project Management*, 29, 647-60.
- SALDANA, J. (2013) *The Coding Manual for Qualitative Researchers*, London, Sage Publications Ltd.
- SANDERSON, J. (2012) 'Risk, Uncertainty and Governance in Megaprojects: A Critical Discussion of Alternative Explanations', *International Journal of Project Management*, 30, 432-43.
- SAUNDERS, C., S. & AHUJA, M., K. (2006) 'Are All Distributed Teams the Same? Differentiating between Temporary and Ongoing Distributed Teams', *Small Group Research*, 27, 662-700.
- SAUNDERS, M., LEWIS, P. & THORNHILL, A. (2009) *Reasearch Methods for Business Students*, Harlow, Essex, Pearon Education Limited.
- SCHWARTZ, H. & DAVIS, S. M. (1981) 'Matching Corporate Culture and Business Strategy', *Organizational Dynamics*, 10, 30-48.
- SHARAD, D. (1986) 'Management by Projects: In Ideological Breakthrough', *Project* Management Journal, 17, 61-63.
- SHARP, K., PETERS, J. A. & HOWARD, J. (2002) *The Management of a Student Research Project*, Aldershot, Gower Publishing Ltd.
- SHENHAR, A. J. (2004) 'Strategic Project Leadership: Toward a Strategic Approach to Project Management', *R&D Management*, 34, 569-78.
- SHENHAR, A. J. & DVIR, D. (1996) 'Toward a Typological Theory of Project Management', *Research Policy*, 25, 607-32.
- SHENHAR, A. J., DVIR, D., GUTH, W., LETHLER, T., MILOSEVIC, D.,
 PATAKANAKUL, P., POLI, M. & STEFANOVIC, J. (2007) 'Project Strategy: The Missing Link', In Shenhar, A. J., Milosevic, D., Dvir, D. & Thamhain, H. (eds) *Linking Project Management to Business Stratgy*. Pennsylvania, Projecy Management Institute, Inc.
- SHENHAR, A. J., DVIR, D., LEVY, O. & MALTZ, A. C. (2001) 'Project Success: A Multidimensional Strategic Concept', *Long Range Planning*, 34, 699-725.
- SHIRAZI, B., LANGFORD, D. A. & ROWLINSON, S. M. (1996) 'Organizational Structures in the Construction Industry', *Construction Management & Economics*, 14, 199-212.
- SHIRLEY, R. C. (1982) 'Limiting the Scope of Strategy: A Decision Based Approach', *The Academy of Management Review*, 7, 262-68.
- SHOESMITH, D. R. (1996) 'A Study of the Management and Procurement of Building Services Work', *Construction Management & Economics*, 14, 93-101.
- SIDWELL, A. C. (1990) 'Project Management: Dynamics and Performance', *Construction Management & Economics*, 8, 159.
- SIEMENS, N. (1971) 'A Simple Cpm Time-Cost Tradeoff Algorithm', *Management Science*, 17, B354-B63.
- SKULMOSKI, G. J. & HARTMAN, F. (2000) 'The Project Achilles Heel: Misalignment', *Cost Engineering*, 42, 33-37.

- SLACK, N., CHAMBERS, S., JOHNSTON, R. & BETTS, A. (2006) Operations and Process Management: Principles and Practice for Strategic Management, Harlow, UK, Pearson Education Limited.
- SMITH, L. M. (1978) 'An Evoloving Logic of Participant Observation, Educational Ethnography and Other Case Studies.', In Shuman, K. (ed) *Review of Research in Education*. Itasca, IL, Peacock.
- SNOW, C. C. & HAMBRICK, D. C. (1980) 'Measuring Organizational Strategies: Some Theoretical and Methodological Problems', *The Academy of Management Review*, 5, 527-38.
- SODERLUND, J. (2004a) 'Building Theories of Project Management: Past Research, Questions for the Future', *International Journal of Project Management*, 22, 183-91.
- SODERLUND, J. (2004b) 'On the Broadening Scope of the Research on Projects: A Review and a Model for Analysis', *International Journal of Project Management*, 22, 655-67.
- SÖDERLUND, J. (2002) 'Conceptualizing Project Management: From Optimistic Optimization to Critical Questioning', *EURAM Conference*. Stockholm.
- SÖDERLUND, J. (2011) 'Pluralism in Project Management: Navigating the Crossroads of Specialization and Fragmentation', *International Journal of Management Reviews*, 13, 153-76.
- SONGER, A. D. & MOLENAAR, K. R. (1996) 'Selecting Design-Build: Public and Private Sector Owner Attitudes', *Journal of Management in Engineering*, 12, 47-53.
- SRIVANNABOON, S. (2006) 'Linking Project Management with Buisness Strategy', *Project Management Journal*, 37, 88-96.
- STAKE, R. E. (1994) 'Case Studies', In Denzin, N. K. & Lincoln, Y. S. (eds) *Handbook* of *Qualitative Research*. 3rd ed. London, Sage Publications.
- STAKE, R. E. (1995) *The Art of Case Study Research*, Thousand Oaks, California, Sage Publications.
- STERMAN, J. D. (2000) Business Dynamics: Systems Thinking and Modelling for a Complex World, New York, NY, McGraw-Hill Inc.
- STRAUSS, A. & CORBIN, J. M. (1998) *Basics of Qualatative Research: Grounded Theory Procedures and Techniques*, Newbury Park, California, Sage Publications.
- STRINGER, J. (1967) 'Operational Research for "Multi-Organizations"', OR, 18, 105-20.
- SUTRISNA, M. & BARRETT, P. (2007) 'Applying Rich Picture Diagrams to Model Case Studies of Construction Projects', *Engineering, Construction and Architectural Management*, 14, 164-79.
- SYMON, G. & CASSELL, C. (2012) *Qualitative Organizational Research: Core Methods and Current Challanges*, London, Sage Publications Ltd.
- TAN, Y., SHEN, L., M.ASCE & LANGSTON, C. (2012) 'Competition Environment, Strategy, and Performance in the Hong Kong Construction Industry'.
- TAYLOR, C. (1971) ' Interpretation and the Sciences of Man', *The Review of Mataphysics*, 25, 3-51.
- TAYLOR, J., DOSSICK, C. & GARVIN, M. (2011) 'Meeting the Burden of Proof with Case-Study Research', *Journal of Construction Engineering and Management*, 137, 303-11.
- TAYLOR, R. G., NORVAL, H. M., HINDLE, P. D., RWELAMILA, P. D. & MCDERMOTT, P. (1999) 'From Conventionally Orientated to Developmentally Orientated Procurement Systems: Experiences from South Afrcan', In

Rowlinson, S. & Mcdermott, P. (eds) *Procurement Systems: A Guide to Best Practice in Construction*. London, E & F Spon.

- TEECE, D. J., PISANO, G. & SHUEN, A. (1997) 'Dynamic Capabilities and Strategic Management', *Strategic Management Journal*, 18, 509-33.
- THOMAS, J. (2006) 'Problematising Project Management ', In Hodgson, D. & Cicmil, S. (eds) *Making Projects Critical*. New York, Palgrave MacMillan.
- THOMSON, D. (2011) 'A Pilot Study of Client Complexity, Emergent Requirements and Stakeholder Perceptions of Project Success', *Construction Management & Economics*, 29, 69-82.
- TOOR, S.-U.-R. & OGUNLANA, S. O. (2010) 'Beyond the 'Iron Triangle': Stakeholder Perception of Key Performance Indicators (KPIs) for Large-Scale Public Sector Development Projects', *International Journal of Project Management*, 28, 228-36.
- TURNER, J. R. (1999) The Handbook of Project-Based Management: Improving the Process for Achieving Strategic Objectives, Maidenhead, UK, McGraw-Hill.
- TURNER, J. R. (2006) 'Towards a Theory of Project Management: The Nature of the Project Governance and Project Management', *International Journal of Project Management*, 24, 93-95.
- TURNER, J. R. & MÜLLER, R. (2005) 'The Project Manager's Leadership Style as a Success Factor on Projects: A Literature Review', *Project Management Journal*, 36, 49-61.
- TURNER, J. R. & SIMISTER, S. J. (2001) 'Project Contract Management and a Theory of Organization', *International Journal of Project Management*, 19, 457-64.
- TURNER, R. J. & MULLER, R. (2003) 'On the Nature of the Project as a Temporary Organization', *International Journal of Project Management*, 21, 1-8.
- TVERSKY, A. & KAHNEMAN, D. (1986) 'Rational Choice and the Framing of Decisions', *The Journal of Business*, 59, Part 2, 5251-78.
- TYSSEN, A. K., WALD, A. & SPIETH, P. (2014) 'The Challenge of Transactional and Transformational Leadership in Projects', *International Journal of Project Management*, 32, 365-75.
- UCAS (2011) 'Ucas: End of Cycle Report 2010/2011', CHELTENHAM, UCAS.
- UNGER, B. N., KOCK, A., GEMÜNDEN, H. G. & JONAS, D. (2012) 'Enforcing Strategic Fit of Project Portfolios by Project Termination: An Empirical Study on Senior Management Involvement', *International Journal of Project Management*, 30, 675-85.
- VAN DE VEN, A. H. (1979) 'Organizations and Environments', *Administrative Science Quarterly*, 24, 320-26.
- VAN DER MERWE, A. P. (2002) 'Project Management and Business Development: Integrating Strategy, Structure, Processes and Projects', *International Journal of Project Management*, 20, 401-11.
- VENKATRAMAN, N. (1989) 'The Concept of Fit in Strategy Research: Toward Verbal and Statistical Correspondence', *Academy of Management Review*, 14, 423-44.
- VENKATRAMAN, N. & CAMILLUS, J. C. (1984) 'Exploring the Concept of "Fit" in Strategic Management', *Academey of Management Review*, 9, 513-25.
- VON BERTALANFFY, L. (1968) General System Theory: Foundations, Development, Applications, New York, George Braziller.
- WALKER, A. (2007) *Project Management in Construction* Oxford, Blackwell Publishing.
- WALKER, D. & HAMPSON, K. (2008) *Procurement Strategies: A Rerlationship Based Approach*, Oxford, UK, Blackwell Science Ltd.

- WALKER, D. H. T. & JOHANNES, D. S. (2003) 'Construction Industry Joint Venture Behaviour in Hong Kong: Designed for Collaborative Results?', *International Journal of Project Management*, 21, 39-49.
- WATERMAN JR., R. H. (1982) 'The Seven Elements of Strategic Fit', *The Journal of Business Strategy*, 2, 69-73.
- WATERMAN JR., R. H., PETERS, T. J. & PHILLIPS, J. R. (1980) 'Structure Is Not Organization', *Business Horizons*, 23, 14.
- WEBER, M. (1947) *The Theory of Social and Economic Organization*, New York, Free Press.
- WEBER, R. P. (1985) *Basic Content Analysis*, Beverley Hills, CA, Sage Publications Ltd.
- WESTERVELD, E. (2003) 'The Project Excellence Model®: Linking Success Criteria and Critical Success Factors', *International Journal of Project Management*, 21, 411-18.
- WHITLEY, R. (2006) 'Project-Based Firms: New Organizational Form or Variations on a Theme?', *Industrial and Corporate Change*, 15, 77-99.
- WILLIAMS, M. & MAY, T. (1996) *Introduction to the Phiosphy of Social Research*, London, UK, UCL Press Ltd.
- WILLIAMSON, O. E. (1975) Markets and Hierarchies: Analysis and Antitrust Implications, New York, Free Press.
- WILLIAMSON, O. E. (1985) The Economic Institutions of Capitalism: Firms, Markets, Rational Contracting, New York, Free Press.
- WINCH, G. (1988) *The Construction Firm and the Construction Project,* Warwick, Institute for Management and Research Development, University of Warwick.
- WINCH, G. (2007) 'Managing Project Stakeholders', In Morris, P. W. G. & Pinto, J. K. (eds) *The Wiley Guide to Project, Program and Potfolio Management*. Hobeken, New Jersy, John Wiley & Son.
- WINCH, G. M. (2001) 'Governing the Project Process: A Conceptual Framework', *Construction Management & Economics*, 19, 799-808.
- WINCH, G. M. (2008) 'The Governance of Project Coalitions: Towards a Research Agenda', *Commercial Management of Projects*. Blackwell Publishing Ltd.
- WINTER, M., ANDERSEN, E. S., ELVIN, R. & LEVENE, R. (2006a) 'Focusing on Business Projects as an Area for Future Research: An Exploratory Discussion of Four Different Perspectives', *International Journal of Project Management*, 24, 699-709.
- WINTER, M., SMITH, C., MORRIS, P. & CICMIL, S. (2006b) 'Directions for Future Research in Project Management: The Main Findings of a UK Government-Funded Research Network', *Project Management Journal*, 24, 638-49.
- WOLCOTT, H. F. (1994) *Transforming Qualitative Data: Description, Analysis, and Interpretation.*, Thousand Oaks, CA, Sage Publications.
- YANG, L.-R. (2012) 'Implementation of Project Strategy to Improve New Product Development Performance', *International Journal of Project Management*, 30, 760-70.
- YIN, R. K. (2003) *Case Study Research: Design and Methods*, Thousand Oaks, California, Sage Publications.
- YIN, R. K. (2014) *Case Study Research: Design and Methods*, California, Sage Publications Inc.
- YOUKER, R. & BROWN, J. (1998) 'Defining the Hierarchy of Project Objectives', *IPMA*. Slovenia.
- YOUNG, R. & GRANT, J. (2015) 'Is Strategy Implemented by Projects? Disturbing Evidence in the State of Nsw', *International Journal of Project Management*, 33, 15-28.

- YOUNG, R., YOUNG, M., JORDAN, E. & O'CONNOR, P. (2012) 'Is Strategy Being Implemented through Projects? Contrary Evidence from a Leader in New Public Management', *International Journal of Project Management*, 30, 887-900.
- ZAJAC, E. J., KRAATZ, M. S. & BRESSER, R. K. F. (2000) 'Modeling the Dynamics of Strategic Fit: A Normative Approach to Strategic Change', *Strategic Management Journal*, 21, 429-53.

APPENDIX A – INTERVIEW GUIDE

Cas	e		1	Date	
Con	tact Type		1	Interviewer	
Sect	ion A				
1	Name				
2	Company			Company Type	
3	Employment	Permanent / Ter	nporary		
4	Position / role	e			
5a	At what stage	did you join the			
5b	project? At stage did ye	ou finish on the			
	project?	• • • •			
Sect	ion B – Strateg	gic Alignment			
6	What was the	purpose of the proj	iect?		
7	What was the	long-term benefit t	o (client) in pursuing the	project?	
0	¥¥77 1				
8	What were the project?	e strategic benefits	to your (organisation/de	partment/section) ir	i taking part in the
9	In your opinio	on how was success	of the project measured	?	
10	How did vou r	personally measure	project success?		
	Jan Jan Jan P		r J		
11	In your opinio	on were there any is	ssues that affected projec	t success from bein	g fully realised.
12	How did the since the second s	hort duration / imp	osed deadline of the proj	iect impact on the p	erformance of the
	<i>project:</i>	a. Completion of	tasks (complexity)		
	l What was don	b. Behaviour of the	he team (trust)		
	what was don	e in mese suudion	3		
13	How did the s	tructure of the tean	n (project organisation) i	mpact on the perfor	rmance / success of
	the project?	- D		- 1 5	5
	c What	a. Procurement p t was the reaction i	rocess n these situations?		
14	In your opinio project?	on were there any c	onflicting objectives betw	veen organisations	involved within the
	projeci:				

APPENDIX B – CASE STUDY REGISTER

Interviewee	Organisation	Role	E-mail	Contact Number	Interview Date	Transcribed	Coded

APPENDIX C – DATA SOURCE LOGS

- C.1 Case Study A: New student residence
- C.2 Case Study B: New 4-lane carriageway
- C.3 Case Study C: Office and laboratory facility
- C.4 Case Study D: New tourist visitor centre

Reference code

Case - Interviewee initials - date data collected - data type

Reference No.	Source description	Data Type	Data Capture	Source	Organisation	Role	Date Collected
A-PMc-101214-intvw	Notes from interview	intvw	Unrecorded interview	PMc	Client	Finance Director	10/12/14
A-MD-031214-intvw	Transcript of interview	intvw	Audio recorded interview	MD	Client	Director of Campus Services	03/12/14
A-DW-270314a-intvw	Transcript of interview	intvw	Audio recorded interview	DW	Client	Project Manager	27/03/14
A-DW-170315b-intvw	Transcript of follow-up interview with DW	intvw	Audio recorded interview	DW)	17/03/15
A-GC-231214-intvw	Transcript of interview	intvw	Audio recorded interview	GC	Client	Project Manager	23/12/14
A-AJ-270314-intvw	Transcript of interview	intvw	Audio recorded interview	AJ	Client	Project Manager	27/03/14
A-BG-240314a-intvw	Transcript of interview	intvw	Audio recorded interview	BG	Consultant	Operations Director	24/03/14
A-BG-131014b-intvw	Transcript of follow-up interview with BG	intvw	Audio recorded interview	BG			13/10/14
A-DL-270314-intvw	Transcript of interview	intvw	Audio recorded interview	DL	Contractor	Project Manager	27/03/14
A-CJ-280314-intvw	Transcript of interview	intvw	Audio recorded interview	CI	Client	Director of Student Support	28/03/14
A-TD-240314-intvw	Transcript of interview	intvw	Audio recorded interview	TD	Client	Director of Hospitality	24/03/14
A-PW-071014-intvw	Transcript of interview	intvw	Audio recorded interview	PW	Consultant	Architect	07/10/14
A-MMc-291014-intvw	Transcript of interview	intvw	Audio recorded telephone interview	MMc	Contractor	Area Manager	29/10/14
A-MC-190314-intvw	Transcript of interview	intvw	Audio recorded interview	MC	Consultant	Quantity Survey	19/03/14
A-IG-190314-intvw	Transcript of interview	intvw	Audio recorded interview	IG	Consultant	CDM	19/0314
A-PMc-0210-report	PME Business Case	report	Copy of report	PMc	Client		27/05/15
A-CBA-0714-report	Report on Cost &	report	Copy of report	CBA	Consultant		27/05/15
	Programme over-runs						
A-CS-0710-schedule	Development programme	programme	Copy of programme	CS	Consultant		27/05/15
A-HW-0311-report	Project Execution Plan	report	Copy of report	DW	Client		27/03/14
A-HW-0314-figure	University Committee	diagram	University Website	NWH	Client		24/06/15
	Structure						

C.1 Case Study A - Data Source Log

265

B-G0-200814-intw B-MT-200814-intw Transcript of interviewintwAudio recorded interviewGOClient ClientB-AH-200814-intw B-WY-230914-intw Transcript of interviewTranscript of interviewintwAudio recorded interviewAHClientB-LC-111214-intw B-WY-230914-intw Transcript of interviewTranscript of interviewintwAudio recorded interviewAHClientB-LC-111214-intw B-WY-230914-intw Transcript of interviewTranscript of interviewintwAudio recorded interviewAHClientB-LC-111214-intw B-WY-220914-intw B-WY-230914-intw B-GC-0814-reportTranscript of interviewintwAudio recorded interviewAHClientB-AH-0713-doc B-GC-0814-reportPresentation slidesdocument documentCopy of sportLCContractorB-GCC-0814-reportNummary Document B-GCC-0814-reportdocument Copy of reportCopy of reportLESClientB-GCC-0814-reportBusiness planReportCopy of reportLESClientB-GCC-0811-reportProgress meeting No.1ReportCopy of reportLESClientB-LES-120910-reportProgress meeting No.5ReportCopy of reportLESClientB-LES-240211-reportProgress meeting No.5ReportCopy of reportLESClientB-LES-240211-reportProgress meeting No.5ReportCopy of reportCHCHB-LES-240211-reportProgress meeting No.5ReportCopy of reportLESCHB	Reference No.	Source description	Data Type	Data Capture	Source	Organisation	Role	Date Collected
B-AH-200814-intvw Transcript of interview intvw Audio recorded interview AH Client B-W-230914-intvw Transcript of interview intvw Audio recorded interview JW Contractor B-W-230914-intvw Transcript of interview intvw Audio recorded interview JW Contractor B-W-230914-intvw Transcript of interview intvw Audio recorded interview JW Contractor B-W-230914-intvw Transcript of interview intvw Audio recorded interview JW Contractor B-M-0713-doc Presentation slides document Copy of slides AH Client B-AH-0713-doc Presentation slides document Copy of report LES Contractor B-GCC-0814-report Summary Document document Copy of report LES Client B-LES-10910-report Business plan Report Copy of report LES Client B-LES-120910-report Business plan Report Copy of report LES Client B-LES-120910-report Business meeting No.1 Report Copy of report LES CHW	B-GO-200814-intvw	Transcript of interview	intvw	Audio recorded interview	GO	Client	Section Engineer	20/08/14
B-JW-230914-intvw Transcript of interview intvw Audio recorded interview JW Contractor B-LC-111214-intvw Transcript of interview intvw Audio recorded interview JW Contractor B-LC-111214-intvw Transcript of interview intvw Audio recorded interview JW Contractor B-W-220914-intvw Transcript of interview intvw Audio recorded interview JW Contractor B-M-0713-doc Presentation slides document Copy of slides AH Cilent B-GCC-0814-report Summary Document document Copy of report LES Consultant B-GCC-0814-report Business plan Report Copy of report LES Cilent B-GCC-0814-report Business plan Report Copy of report LES Cilent B-GCC-0814-report Business plan Report Copy of report LES Cilent B-GCC-0814-report Progress meeting No.1 Report Copy of report LES Cilent B-GCC-0810-report Progress meeting No.5 Report Copy of report LES LES	B-AH-200814-intvw	Transcript of interview	intvw	Audio recorded interview	AH	Client	Resident Engineer	20/08/14
B-LC-111214-intvw Transcript of interview intvw Audio recorded interview LC Contractor B-W-220914-intvw Transcript of interview intvw Audio recorded interview LC Contractor B-MU-0210914-intvw Transcript of interview intvw Audio recorded interview LC Contractor B-AH-0713-doc Presentation slides document Copy of slides AH Client B-GCC-0814-report Summary Document document Copy of report GCC Consultant B-GCC-0814-report Buneary Document document Copy of report GCC Consultant B-GCC-0814-report Buneary Document document Copy of report LES Client B-GCC-0814-report Buneary Document Copy of report LES Client Client B-GCC-0811-report Bunual Report Copy of report LES Client Client B-LES-120910-report Bress meeting No.1 Report Copy of report LES Client B-LES-240211-report Progress meeting No.5 Report Copy of report LES LES	B-JW-230914-intvw	Transcript of interview	intvw	Audio recorded interview	JW	Contractor	Sub-Agent	23/09/14
B-IW-220914-intvwTranscript of interviewintvwAudio recorded interviewIWContractorB-SR-010914-intvwTranscript of interviewintvwAudio recorded interviewIWConsultantB-SR-010914-intvwTranscript of interviewintvwAudio recorded interviewSRConsultantB-SR-010914-intvwTranscript of interviewintvwAudio recorded interviewSRConsultantB-GCC-0814-reportSummary DocumentdocumentCopy of slidesAHClientB-GSC-0814-reportBusiness plandocumentCopy of reportLESClientB-LES-0911-reportProgress meeting No.1ReportCopy of reportLESCHWB-LES-120910-reportProgress meeting No.1ReportCopy of reportLESClientB-LES-240211-reportProgress meeting No.5ReportCopy of reportLESLESB-LES-240211-reportProgress meeting No.5ReportCopy of reportLESB-LES-240211-reportProgress meeting No.5ReportCopy of reportLESB-LES-240211-reportProgress meeting No.5ReportCopy of reportLESB-LES-240211-reportProgress meeting No.5ReportCopy of reportLESB-GCC-293015-webLocal Authority Web site 1Web pageGCCGCCB-GCC-293015-webContractor Website 2Web pageGCCGCCB-GCC-29312-reportImplementation ReportReportCopy of reportLESB-GCC-0312-report<	B-LC-111214-intvw	Transcript of interview	intvw	Audio recorded interview	LC	Contractor	Finance Director	11/12/14
B-SR-010914-intwTranscript of interviewintwAudio recorded interviewSRConsultantB-AH-0713-docPresentation slidesdocumentCopy of slidesAHClientB-GCC-0814-reportSummary DocumentdocumentCopy of reportGCCCB-LES-0911-reportSummary DocumentdocumentCopy of reportGCCLESB-LES-120910-reportBusiness planReportCopy of reportLESCHWB-LES-120910-reportProgress meeting No.1ReportCopy of reportLESCHWB-LES-120910-reportProgress meeting No.5ReportCopy of reportLESCHWB-LES-240211-reportProgress meeting No.5ReportCopy of reportLESCIentB-LES-240211-reportProgress meeting No.5ReportCopy of reportLESLESB-LES-240211-reportProgress meeting No.5ReportCopy of reportLESB-LES-240211-reportProgress meeting No.5ReportCopy of reportLESB-GCC-293015-webLocal Authority Web site 1Web pageWeb pageGCCB-GCC-293015-webContractor Website 2WebWeb pageJV1B-GCC-293015-webContractor Website 2WebWeb pageJV1B-GCC-293015-webContractor Website 2WebWeb pageJV1B-GCC-293015-webContractor Website 2WebWeb pageJV1B-GCC-30312-reportImplementation ReportReportCopy of reportGCC<	B-IW-220914-intvw	Transcript of interview	intvw	Audio recorded interview	IW	Contractor	Project Manager	22/09/14
B-AH-0713-docPresentation slidesdocumentCopy of slidesAHClientB-GCC-0814-reportSummary DocumentdocumentCopy of reportGCCB-LES-0911-reportSummary DocumentdocumentCopy of reportGCCB-LES-10910-reportBusiness planReportCopy of reportLESB-LES-120910-reportBusiness planReportCopy of reportLESB-LES-120910-reportB-LES-120910-reportLESClientB-LES-120910-reportB-LES-240211-reportReportCopy of reportLESB-LES-23015-webProgress meeting No.5ReportCopy of reportLESB-GC-233015-webLocal Authority Web site 1Web pageWeb pageGCCB-GC-0312-reportImplementation ReportReportCopy of reportGCCB-GC-0312-reportImplementation ReportReportCopy of reportGCCB-GC-0312-reportImplementation ReportReportCopy of reportGCCB-GC-0312-reportImplementation ReportReportCopy of reportGCCB-GC-0312-reportImplementation ReportReportCopy of reportGCCB-GCC-0312-reportImplementation ReportReportCopy of reportGCCB-GCC-0312-reportImplementation ReportReportCopy of reportGCCB-GCC-0312-reportImplementation ReportReportCopy of reportGCCB-GCC-0312-reportImplementation ReportReportCopy of reportGCC	B-SR-010914-intvw	Transcript of interview	intvw	Audio recorded interview	SR	Consultant	Operations Manager	01/09/14
B-GCC-0814-report Summary Document document Copy of report GCC B-LES-0911-report Annual Report document Copy of report LES B-LES-10910-report Business plan Report Copy of report LES B-CGB-068-report Business plan Report Copy of report LES B-LES-120910-report Progress meeting No.1 Report Copy of report LES B-LES-120910-report Progress meeting No.1 Report Copy of report LES B-LES-120910-report Progress meeting No.5 Report Copy of report LES B-LES-240211-report Progress meeting No.5 Report Copy of report LES B-LES-240211-report Progress meeting No.5 Report Copy of report LES B-LES-240211-report Progress meeting No.5 Report Copy of report LES B-GCC-293015-web Local Authority Web site 1 Web Web page JV1 B-GCC-293015-web Contractor Website 2 Web Web page JV1 B-GCC-293015-web Contractor Website 2 Web Web page JV1	B-AH-0713-doc	Presentation slides	document	Copy of slides	АН	Client		
B-LES-0911-report Annual Report document Copy of report LES Client B-CGB-0608-report Business plan Report Copy of report LES CHW B-LES-120910-report Progress meeting No.1 Report Copy of report LES B-LES-120910-report Progress meeting No.1 Report Copy of report LES B-LES-120910-report Progress meeting No.5 Report Copy of report LES B-LES-240211-report Progress meeting No.5 Report Copy of report LES B-LES-240211-report Progress meeting No.5 Report Copy of report LES B-GCC-293015-web Local Authority Web site 1 Web page GCC GCC B-GCC-293015-web Contractor Website 2 Web page GCC GCC B-GCC-293015-web Contractor Website 2 Web page JV1 GCC B-GCC-293015-web Contractor Website 2 Web Web page JV1 B-GCC-293015-web Contractor Website 2 Web Web page JV1 B-GCC-03015-web E-mail correspondence e-mail Correspondence <td>B-GCC-0814-report</td> <td>Summary Document</td> <td>document</td> <td>Copy of report</td> <td>GCC</td> <td></td> <td></td> <td>14/08/14</td>	B-GCC-0814-report	Summary Document	document	Copy of report	GCC			14/08/14
B-CGB-0608-reportBusiness planReportCopy of reportCHWB-LES-120910-reportProgress meeting No.1ReportCopy of reportLESminutesminutesCopy of reportLESB-LES-240211-reportProgress meeting No.5ReportCopy of reportLESB-LES-240211-reportProgress meeting No.5ReportCopy of reportLESB-GCC-293015-webLocal Authority Web site 1WebWeb pageGCCB-GCC-293015-webContractor Website 2WebWeb pageGCCB-GCC-293015-webContractor Website 2WebWeb pageGCCB-GCC-293015-webContractor Website 2WebWeb pageGCCB-GCC-203015-webE-mail correspondencee-mailCorrespondenceGCCB-GCC-0312-reportImplementation ReportReportCopy of reportGCCB-GCC-03012-reportPublic Status ReportReportCopy of reportGCCB-GCC-0809-renortPublic Status ReportReportCopy of reportGCC	B-LES-0911-report	Annual Report	document	Copy of report	LES	Client		03/09/15
B-LES-120910-report Progress meeting No.1 Report Copy of report LES minutes minutes Report Copy of report LES B-LES-240211-report Progress meeting No.5 Report Copy of report LES B-GCC-293015-web Local Authority Web site 1 Web Web page GCC B-GCC-293015-web Contractor Website 2 Web Web page GCC B-JV1-293015-web Contractor Website 2 Web Web page GCC B-GCC-29315-web Contractor Website 2 Web Web page GCC B-GCC-0312-report Implementation Report Report Copy of report GCC B-GCC-0809-renort Public Status Renort Report Copy of report GCC Client	B-CGB-0608-report	Business plan	Report	Copy of report	CHW			04/09/15
B-LES-240211-report Progress meeting No.5 Report Copy of report LES B-GCC-293015-web Local Authority Web site 1 Web Web page GCC B-JV1-293015-web Contractor Website 2 Web Web page GCC B-GCC-293015-web Contractor Website 2 Web Web page GCC B-GCC-203015-web Contractor Website 2 Web Web page GCC B-GCC-0312-report Implementation Report Report Copy of report GCC B-GCC-0809-remort Public Status Report Report Copy of report GCC	B-LES-120910-report	Progress meeting No.1	Report	Copy of report	LES			22/08/14
B-LES-240211-report Progress meeting No.5 Report Copy of report LES minutes minutes minutes Copy of report LES B-GCC-293015-web Local Authority Web site 1 Web Web page GCC B-JV1-293015-web Contractor Website 2 Web Web page GCC B-GCC-0915-email E-mail correspondence e-mail Correspondence GCC B-GCC-0312-report Implementation Report Report Copy of report GCC Client		minutes						
B-GCC-293015-webLocal Authority Web site 1WebWeb pageGCCB-JV1-293015-webContractor Website 2WebWeb pageJV1B-GO-0915-emailE-mail correspondencee-mailCorrespondenceGOB-GCC-0312-reportImplementation ReportReportCopy of reportGCCC0004-renortPublic Status RenortRenortRenortConv of renort	B-LES-240211-report	Progress meeting No.5 minutes	Report	Copy of report	LES			22/08/14
B-JV1-293015-web Contractor Website 2 Web Web page JV1 B-GO-0915-email E-mail correspondence e-mail Correspondence GO Client B-GCC-0312-report Implementation Report Report Copy of report GCC Client	B-GCC-293015-web	Local Authority Web site 1	Web	Web page	GCC			29/30/15
B-GO-0915-email E-mail correspondence e-mail Correspondence GO Client B-GCC-0312-report Implementation Report Report Copy of report GCC Client B-GCC-0809-remort Public Status Remort Remort Conv. of remort GCC	B-JV1-293015-web	Contractor Website 2	Web	Web page	JV1			29/30/15
B-GCC-0312-report Implementation Report Report Copy of report GCC Client B-GCC-0809-report Dublic Status Report Report Conv. of report GCC	B-GO-0915-email	E-mail correspondence	e-mail	Correspondence	GO	Client	Section Engineer	02/09/15
B-GCC-0800-remort Dublic Status Remort Remort Conv. of remort GCC	B-GCC-0312-report	Implementation Report	Report	Copy of report	GCC	Client		03/09/15
$\mathbf{r} \sim \mathbf{r} \sim $	B-GCC-0809-report	Public Status Report	Report	Copy of report	GCC			09/09/15
B-GCC-210915-doc Council Web site document web page GCC Client	B-GCC-210915-doc	Council Web site	document	web page	GCC	Client		21/09/15

C.2 Case Study B - Data Source Log

266

Source Log
- Data
Study C
Case
C.3

Reference No.	Source description	Data Type	Data Capture	Source	Organisation	Role	Date Collected
C-LMc-100414-intvw	Transcript of Interview	intvw	Audio recorded interview	LMc	Client	Project Manager	10/04/14
C-RA-110414-intvw	Transcript of Interview	intvw	Audio recorded interview	RA	Client	Head of Procurement, Estates and Facilities	11/04/14
C-RA-250914a-intvw	Transcript of follow-up interview with RA	intvw	Audio recorded interview	RA	Client		25/09/14
C-NS281114-intvw	Transcript of Interview	intvw	Audio recorded interview	NS	Contractor	Construction Director	28/11/14
C-RT161214-intvw	Transcript of Interview	intvw	Audio recorded interview	RT	Consultant	Senior Project Manager	16/12/14
C-DJ121214-intvw	Transcript of Interview	intvw	Audio recorded interview	DJ	Consultant	Associate Director	16/12/14
C-FW190315-intvw	Transcript of Interview	intvw	Audio recorded interview	FW	Consultant	BREEAM Assessor	19/03/15
C-BR170315-intvw	Transcript of Interview	intvw	Audio recorded interview	BR	Consultant	Architect	17/03/15
C-PH100414-intvw	Transcript of Interview	intvw	Audio recorded interview	Hd	Client	Programme Office	10/04/14
						Manager/Project	
						Assurance	
C-NA040414-intvw	Transcript of Interview	intvw	Audio recorded interview	NA	Client	Head of Business Support	05/04/14
C-MC100414-intvw	Transcript of Interview	intvw	Audio recorded interview	MC	Client	Senior Ecologist	10/04/14
C-KMc-110414-intvw	Transcript of Interview	intvw	Audio recorded interview	KMc	Client	Space Planner	11/04/14
C-EPA0309-report	Estates Strategy	Report	Copy of report	EPA	Client		03/09
C-EPA0913-report	Project Closedown report	Report	Copy of report	EPA	Client		09/13
C-RT231015-email	E-mail correspondence	e-mail	Copy of e-mail	RT	Consultant	Senior Project Manager	23/10/15
C-EPA01209-report	Step Change Report	Report	Copy of report	EPA	Client		12/09

\mathbf{Log}
Source
: - Data
\circ
Study
Case
C.4

Reference No.	Source description	Data Type	Data Capture	Source	Organisation	Role	Date Collected
D-TIC-041114-intvw	Transcript of Interview	intvw	Audio recorded interview	TIC	Client	Interpretation Project	04/11/14
D-DMc011214-intvw	Transcript of Interview	intvw	Audio recorded interview	DMc	Client	Project Director	01/12/14
D-JR241114-intvw	Transcript of Interview	intvw	Audio recorded interview	JR	Client	Project Manager	24/11/14
D-CP271015-intvw	Transcript of Interview	intvw	Audio recorded interview	CP	Client	Learning Manager	27/10/15
D-DA220616-intvw	Transcript of Interview	intvw	Audio recorded interview	DA	Consultant	Senior Associate. Architect	22/06/15
D-CW011214-intvw	Transcript of Interview	intvw	Audio recorded interview	CW	Consultant	MD Interpretation Design	01/12/14
D-GMc2808150-intvw	Transcript of Interview	intvw	Audio recorded interview	GMc	Contractor	Project Surveyor	28/08/15
D-PC150415-intvw	Transcript of Interview	intvw	Audio recorded interview	PC	Consultant	Project Manager	15/04/15
D-NTC0610-report	Project Definition and Brief	Report	Copy of Report	NCT	Client		06/10
D-HEA0613-report	Project Control Document	Report	Copy of Report	HEA	Client		06/13
D-CG0511-report	Government Manifesto	Report	Copy of Report	CG	Client		05/11

Category	Rela	tive Research Questions	Code No	A priori theme
Strategic Objectives	Q1	What are the varied organisational strategies being pursued within a temporary multi-organisation?	1	Organisational Strategies Objectives
			1.1	Client Strategic Objectives
			1.2	Consultant Strategic
				Objectives
			1.3	Contractor Strategic Objectives
Client Complexity			2	Strategic levels
complexity			2.1	Policy level
			2.3	Business level
			2.5	Operational level
Strategic alignment mechanisms	Q2	What mechanisms are implemented to maintain the alignment of a construction project with the client strategic objectives?	3	Procurement Strategies
			3.1	Contractual arrangements
			3.2	TMO Structure
			4	Governance mechanisms
			4.1	Governance Processes
			4.2	Client Structure
			4.3	Communication channels
Environmental Influences	Q3	How effective are these mechanisms in maintaining the strategic alignment of a construction project?	5	Client Behaviours
		1 5	5.1	Client Structure
			6.2	Client Leadership
			6	External Environment
			6.1	Supply Chain
			6.2	Construction Industry
TMO Strategy	Q4	How do varied TMO actors pursue strategic objectives within a single construction project?	7	TMO Behaviours
		1 5	7.1	TMO Time
			7.2	TMO Task
			7.3	TMO Team
			7.4	TMO Transition
Project success	Q5	How do the varied actors engaged within a TMO measure the success of a construction project?	8	Project Success criteria
			8.1	Cost
			8.2	Time
			8.3	Quality
			8.4	Stakeholder Satisfaction
			8.5	Business Success

APPENDIX D – INITIAL TEMPLATE

Category	Rese	earch Questions	Code No	A priori theme
Strategic Objectives	Q1	What are the varied organisational strategies being pursued within a temporary multi-organisation?	1	Organisational Strategies Objectives
		I J G	1.1	Client Strategic Objectives
			1.1.1	Strategic Drivers
			1.1.2	Strategic Benefits
			1.2	Consultant Strategic Objectives
			1.3	Contractor Strategic Objectives
Client Complexity			2	Strategic levels
r			2.1	Policy level
			2.3	Business level
			2.5	Operational level
Strategic alignment mechanisms	Q2	What mechanisms are implemented to maintain the alignment of a construction project with the client strategic objectives?	3	Procurement Strategies
		0	3.1	Contractual arrangements
			3.1.1	Contractor claims
			3.1.2	Contractor Autonomy
			3.1.3	Client control
			3.2	TMO Structure
			4	Governance mechanisms
			4.1	Governance Processes
			4.1.1	Project Boards
			4.1.2	Scrutiny monitoring
			4.2	Client Structure
			4.3	Communication channels
Environmental Influences	Q3	How effective are these mechanisms in maintaining the strategic alignment of a construction project?	5	Client Behaviours
			5.1	Client Structure
			5.2	Client Leadership
			5.3	Client Experience
			5.4	Trust
			5.5	Communication
			6	External Environment
			6.1	Supply Chain
			6.2	Construction Industry
TMO Strategy	Q4	How do varied TMO actors pursue strategic objectives within a single construction project?	7	TMO Behaviours
		~ ~	7.1	TMO Time
			7.2	TMO Task
			7.3	TMO Team
			7.3.1	Integration
			7.3.2	Co-location
			7.3.3	Social networks
			7.4	TMO Transition

APPENDIX E – FINAL TEMPLATE

Project success	Q5	How do the varied actors engaged within a TMO measure the success of a construction project?	8	Project Success criteria
		1 5	8.1	Cost
			8.2	Time
			8.3	Quality
			8.4	Stakeholder Satisfaction
			8.4.1	End User Satisfaction
			8.4.2	Client Satisfaction
			8.5	Business Success
			8.6	Team Dynamics
			8.7	Health and Safety

APPENDIX F – THEMATIC MATRICES

- F.1 Case Study A: New student residence
- F.2 Case Study B: New 4-lane carriageway
- F.3 Case Study C: Office and laboratory facility
- F.4 Case Study D: New tourist visitor centre

Category	Node No	Node	Description
Strategic Objectives	1	Organisational Strategic Objectives	
	1.1	Client Strategic Objectives	Client strategy to build high quality residences would command a higher rental income
	1.1.1	Strategic Drivers	Existing residencies unable to meet HMO regulations
	1.1.2	Reputation	Promise made to first year students to first year students to provide on
			campus accommodation
	1.1.3	Student experience	Contribution to growth strategy
	1.1.4	Revenue generation	Increase revenue through higher quality accommodation
	1.2	Consultant Strategic Objectives	Market entry strategy
			Working the university on previous projects
	1.3	Contractor Strategic Objectives	Market entry strategy - Working the university on previous projects
Client Complexity	2	Strategic levels	
	2.1	Policy level	Maintaining University reputation through provision of accommodation
			Enhancing growth strategy through provision of higher quality accommodation
	2.2	Business level	Estates - Specification of selected materials for reduced maintenance costs
			and life cycle costing Student Services - Attract students through higher quality of
			accommodation
	с С	Ommericanol Invest	Hospitality - Generate income through conference facilities
	C.7	Operational rever	
Strategic alignment mechanisms	3	Procurement Strategies	
	3.1	Contractual arrangements	Traditional procurement to maintain control of construction
	3.1.1	Claims by contactor	Contractor claims and as result of Incomplete design information
	3.2	TMO structure	Consultants under single PMC
	4	Governance mechanisms	
	4.1	Governance Processes	Oversight and Project Board unable to reach agreement within the client
			system

F.1 Case Study A – Thematic Matrix

umental i

Category

Strategic Objectives
Client Complexity
•
Strategic alignment processes
1
Environmental Influences

F.2 Case Study B – Thematic Matrix

	6.3	Market Conditions	Low tender bid as a result of competitive market. Favours client
TMO Strategy	, , , , , , , , , , , , , , , , , , ,	TMO Behaviour	- - - - - - - - - - - - - - - - - - -
	7.1	TMO Time	Challenging programme. Deadlines normal in the industry. Stress caused by utility or sanisations outside of the TMO
	7.2	TMO Task	Highly complex task with additional items, including flood alleviation.
	7.3	TMO Team	Creating synergy through joint venture. Sharing of responsibilities. Skill
	7.3.1	Co-location	Co-location of team members - improved social interaction
	7.3.2	Social networks	Prior relations and social networks
	7.4	TMO Transition	
Perceptions of project success	8	Project Success criteria	
	8.1	Budget & Schedule	Time and cost, primary measure of success. Delivered one week before the
	8.3	Quality	Briefly mentioned by the contractor
	8.4	Safety & maintenance	Briefly mentioned as a critical success factor
	8.5	End User Satisfaction	Satisfaction of the local community and key stakeholders
	8.6	Client Satisfaction	Briefly mentioned by the contractor
	8.7	Team Dynamics	How well the team got on with each other

Category	Code No	Node	Description
Strategic Objectives	1 1.1.1 1.1.2 1.1.3 1.1.3 1.1.4 1.2.1 1.3	Organisational Strategic Objectives Client Strategic Objectives Rationalisation Environment Funding Lease Consultant Strategic Objectives Actor strategic objective Contractor Strategic Objectives	Savings on operating cost, creating a flexible working environment. Creating a flexible and transparent working environment Government funding for BREEAM and efficiency initiatives Favourable rental conditions Enhance portfolio for future and continuous work CV enhancement Enhance portfolio and market entry into Laboratory fit- out
Client Complexity	2 2.3 2.3 2.4	Strategic levels Policy level Business level Operational level	Upgrading capability and working more efficiently Rationalise estate and deliver more flexible working patterns Provision of central laboratory and office facility
Strategic alignment Mechanisms	ω ω ω	Procurement Strategies Contractual arrangements TMO Structure Governance Mechanisms Governance Processes Communication channels Stakeholder Engagement	Design and Build - Transfer of risk and cost certainty. Novation of design team post contract Creation of project board to oversee and direct TMO Through Board End user consultation
Environmental Influences	5 5.1 5.3 6 6.1	Client Behaviour Client Experience Leadership Resistance to change External Environment Market Conditions	Clients knowledge and experience in construction projects Single point of contact between boards and TMO End-user resistance Lower tender price

F.3 Case Study C – Thematic Matrix

TMO Strategy	7	TMO Behaviour	
	7.1	TMO Time	Focus on completion for relocation project to commence
	7.2	TMO Task	Communication of end user requirements
	7.3	TMO Team	Efficient teamwork and integration
	7.4	TMO Transition	Completion to enable continuity of service
Drajaat enoraes	ø	Draiaat Survass aritaria	
	0	I I Office Duccess CITUTIA	
	8.1	Time	Time, cost and quality
	8.2	Cost	
	8.3	Quality	Quality of design and fit for purpose
	8.4	End User Satisfaction	Staff survey feedback
	8.5	Client Satisfaction	Successful population, client relationships
	8.6	Team Dynamics	Positive team working relationships

Category	Node No	Node	Description
Strategic Objectives	1 1.1	Organisational Strategic Objectives Client Strategic Objectives	Conservation of historic monuments
	1.1.1	Strategic Drivers	Current exhibition dated and "not fit for purpose"
	1.1.2	National reputation	Government sponsorship
	1.2	Consultant Strategic Objectives	Market enhancement, market dominance
	1.2.1	Project profile	Show-case talent
	1.2.2	Actor strategic objective	Enhance CV
	1.3	Contractor Strategic Objectives	Portfolio enhancement
Client Complexity	6	Strategic levels	
2	2.1	Policy Jevel	Conserve the battle site and historic monuments
	2.2	Business level	Enhance public awareness of the site, increase financial stability
	2.2.1	Industry competition	Increase visitor numbers
	2.4	Operational level	Provision of new tourist visitor centre and innovative interpretation
Strategic alignment mechanisms	e	Procurement Strategies	
1	3.1	Contractual arrangements	GC Works traditional contract and GC Works Design and Build
	3.2	TMO Structure	Individual consultant organisations
	3.2.1	Joint ventures	Client Joint venture
	4	Governance Mechanisms	
	4.1	Governance Processes	Creation of project board to oversee and direct TMO
	4.2	Collaborative working	Joint project management
	4.2	Communication channels	Through project manager
Environmental Influences	Ś	Client Behaviour	
	5.1	Leadership	Stakeholder/ team management, strong leadership
	y	Estanal Environment	
		Dolitical environment	Political implications of project
	6.2	Media attention	TMO protected by PM leadership

F.4 Case Study D – Thematic Matrix

TMO Strategy	7	TMO Behaviour	
ł	7.1	TMO Time	Focus on completion for opening
	7.2	TMO Task	Task focus of TMO actors
	7.3	TMO Team	Different working patterns. Task focus causing tension between actors
	7.4	TMO Transition	Time required for transition to meet short-term success criteria
Perceptions of project success	×	Project Success criteria	
	8.1	Cost	Delivery by budget
	8.2	Time	Measured by multiple stakeholders
	8.3	Quality	Visitor perception of interpretation
	8.4	End User Satisfaction	Visitor feedback
	8.5	Client Satisfaction	Stakeholder and sponsor satisfaction